

ANALYZING THE ENVIRONMENTAL IMPACT OF INNOVATIVE BIO-BASED PRODUCTS FROM MULTIFUNCTIONAL SYSTEMS

CHRISTIAN MORETTI



ANALYZING THE ENVIRONMENTAL IMPACT OF INNOVATIVE BIO-BASED PRODUCTS FROM MULTIFUNCTIONAL SYSTEMS



CHRISTIAN MORETTI



Analyzing the environmental impact of innovative bio-based products from multifunctional systems

Christian Moretti

ISBN:	978-94-6458-222-2
Cover design & lay-out:	Publiss www.publis
Print:	Ridderprint www.ri

ss.nl dderprint.nl

© Copyright © 2021, Christian Moretti

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, by photocopying, recording, or otherwise, without the prior written permission of the author.

Analyzing the environmental impact of innovative bio-based products from multifunctional systems

Analyse van de milieu-impact van innovatieve biobased producten uit multifunctionele systemen

(met een samenvatting in het Nederlands)

Proefschrift

ter verkrijging van de graad van doctor aan de Universiteit Utrecht op gezag van de rector magnificus, prof.dr. H.R.B.M. Kummeling, ingevolge het besluit van het college voor promoties in het openbaar te verdedigen op

vrijdag 10 juni 2022 des middags te 2.15 uur

door

Christian Moretti

geboren op 31 december 1992 te Varese, Italië

Promotor:

Prof. dr. H.M. Junginger

Copromotoren:

Dr. L. Shen Dr. B. Corona To Elisa,

CONTENTS

1.	Int	roduc	tion	11
	1.1	Bioed	conomy and its role in the EU green transition	12
	1.2	The i by-pr	nterest in locally sourced agro-industrial and food-processing roducts	14
	1.3	LCA a prod	as a key tool to measure the impacts of innovative bio-based ucts	15
	1.4	Chall	enges of modeling bio-based products in LCA	16
		1.4.1	The LCA multifunctionality issue and the ISO hierarchy interpretation issue	17
		1.4.2	The multifunctionality uncertainty in LCAs of bio-based by- products	19
	1.5	Rese	arch objectives and structure of the thesis	22
2.	Re [.] les	view o sons l	of Life Cycle Assessments of lignin and derived products: earned	27
	2.1	Intro	duction	29
	2.2	Mate	rial and methods	32
		2.2.1	Selected studies	32
		2.2.2	Aims and structure of the review	33
	2.3	Resu	lts	34
		2.3.1	Product systems	34
		2.3.2	Analysis of methodological choices	35
		2.3.3	Environmental impact	47
	2.4	Conc	lusions	57
3.	Re ^r	viewiı vironr	ng ISO compliant multifunctionality practices in nental Life Cycle modeling	63
	3.1	Intro	duction	65
	3.2	Meth	odology	67
	3.3	The c	critical review combined with text mining	70
		3.3.1	Debate on the interpretation of ISO's system expansion	70
		3.3.2	Selection of the ISO allocation criterion	75
	3.4	The b	pibliometric review based on main path analysis	79
		3.4.1	Bilateral beginning (1994-1998)	79
		3.4.2	The ISO 14041 influence (1999-2003)	81

	3.4.3	Consequential LCA influence (2004-2008)	85
	3.4.4	. ISO 14044:2006 application (from 2009)	86
	3.5 Discu	ussion and conclusions	88
	3.A Appe	endix	91
4.	Kraft lig attributi	nin as a bio-based ingredient for Dutch asphalts: an onal LCA	97
	4.1 Intro	duction	99
	4.2 Mate	rials and Methods	100
	4.2.1.	Goal and scope definition	100
	4.2.2	Life cycle inventory	103
	4.3. Resu	lts	113
	4.3.1	Kraft lignins	113
	4.3.2	SMAs	117
	4.3.3	ACs	119
	4.3.4	ZOABs	121
	4.4 Discu	ussion	121
	4.4.1	Multifunctionality	121
	4.4.2	Change of functional unit (to 1 m²) and product system	125
	4.5. Conc	clusions	127
	4.A Appe	endix	130
5.	Combini and pow	ing biomass gasification and Solid Oxid Fuel Cell for heat /er generation: an early-stage Life Cycle Assessment	139
	5.1 Intro	duction	141
	5.2 Mate	rial and methods	143
	5.2.1	Goal and scope definition	143
	5.2.2	Life cycle inventory	146
	5.3 Resu	lts	153
	5.3.1	Environmental impact of the HBP technology	153
	5.3.2	Benchmarking with competing technologies	157
	5.3.3	Alternative methods for solving multifunctionality	162
	5.3.4	Sensitivity analysis on potentially sensitive parameters	166
	5.4 Cond	lusions	168
	5.A Appe	endix	170

6.	Attribut Dutch p	tional and consequential LCAs of a novel bio-jet fuel from ootato by-products	173
	6.1 Intro	oduction	175
	6.2. Mat	erials and Methods	178
	6.2.1	. Goal and scope definition	178
	6.2.2	2. Life cycle inventory	182
	6.3. Res	ults	191
	6.3.1	ALCA: identification of the environmental hotspots	192
	6.3.2	CLCA: potential environmental impacts due to changes in demand	194
	6.3.3	6 Comparison with petrochemical kerosene	197
	6.3.4	 ALCA: Environmental performance of the other bio-based products 	199
	6.4. Disc	ussion	204
	6.4.1	Uncertainties in the attributional model	204
	6.4.2	2 Uncertainties in the consequential model	205
	6.4.3	3 Indirect land use change	208
	6.4.4	 Advantages and disadvantages of attributional or consequential LCAs 	209
	6.5. Con	clusions	213
	6.A App	endix	214
7.	Environ used co	mental Life Cycle Assessment of polypropylene made from oking oil	219
	7.1 Intro	oduction	221
	7.2 Mat	erial and methods	223
	7.2.1	Goal and scope definition	223
	7.2.2	Life cycle inventory modeling	227
	7.3 Res	ults	233
	7.3.1	Impact assessment and interpretation	233
	7.4 Disc	ussion	236
	7.4.1	Data uncertainty	236
	7.4.2	2 Model uncertainty: multifunctionality	239
	7.4.3	Environmental benchmarking of UCO-based PP	244
	7.5 Con	clusions	246
	7.A App inte	endix: Detailed breakdown per life cycle stage and rpretation	248
	7.В Арр	endix: Multifunctionality sensitivity analysis	250
	7.С Арр	endix: Petrochemical polypropylene?	251

8.	Sur	mmar	y, conclusions and recommendations	253
	8.1	Resea	arch question 1	256
		8.1.1	Lignin use-review	257
		8.1.2	ISO guidelines -review	259
		8.1.3	List of recommendations	263
	8.2	Resea	arch question 2	264
		8.2.1	Lignin-based asphalts	264
		8.2.2	Bioenergy from wood chips	265
		8.2.3	Bio-jet fuel from potato by-products	266
		8.2.4	Polypropylene from used cooking oil	267
		8.2.5	Summary of lessons learned and recommendations of RQ2	268
	8.3	Resea	arch question 3	271
		8.3.1	Lignin-based asphalts	271
		8.3.2	Bioenergy from wood chips	272
		8.3.3	Bio-jet fuel from potato by-products	273
		8.3.4	Polypropylene from used cooking oil	274
		8.3.5	Summary of lessons learned and recommendations of RQ3	275
	8.4	Overa	arching limitations of this thesis and future research	277
9.	Eng	glish s	summary	281
10.	Ne	derlar	ndse samenvatting	287
	Ref	ferenc	es	293
	Acl	knowl	edgments	321
	Cu	rriculu	ım Vitae	322



CHAPTER

Introduction

1.1 BIOECONOMY AND ITS ROLE IN THE EU GREEN TRANSITION

Bio-based products have emerged as a possible alternative to reduce the consumption of fossil resources in worldwide society. An economy relying on biobased resources is often referred to as "bioeconomy" or "bio-based economy". Using the definition of the European Commission, "bioeconomy comprises those parts of the economy that use renewable biological resources from land and sea – such as crops, forests, fish, animals and micro-organisms – to produce food, materials and energy" [1]. Hence, in simple words, bioeconomy is an economy relying on biomass conversion into a broad range of products, from more established ones such as food and feed to innovative products such as fuels for aviation and plastic materials.

The transition from the current fossil-based economy to a bio-based economy is seen as a new paradigm for a future sustainable economy, potentially generating environmental, societal and economic benefits [2–4]. Besides reducing dependence on fossil fuels, a bioeconomy is expected to reduce climate change impacts compared to the current fossil-based economy [5,6]. Such climate change benefits can be obtained by making use of the short carbon cycle of biomass and the possibility of storing biogenic carbon [4,7,8]. These environmental benefits could be accompanied by socio-economic benefits such as creating new job opportunities for sustainable economic growth [4,9].

The potential of bioeconomy to play a leading role in the sustainable transition towards climate neutrality has been widely recognized [10]. More than 50 countries have already created policy strategies related to bioeconomy [11,12]. In particular, the promotion of bio-based commodities is one of the main action plans of the European Green Deal launched by the European Commission [13]. The European Green deal looks at bioeconomy as a part of multiple policies related to renewable energy and chemicals and materials' sustainability and circularity, and as a way to address the Paris Agreement's climate objectives and meet the Sustainable Development Goals (SDGs) set by the United Nations [2,14,15].

Bioeconomy's present economic volume is currently difficult to estimate due to a lack of statistics on emerging bio-based products and the fact that a broad range of industries is partially bio-based [16]. However, the European Commission-Joint Research Center has made a first attempt to estimate the value-added generated by bioeconomy based on 2017 data. In the EU-27 in 2017, the bio-based part of the

economy generated €614 billion in value-added [17]. This value-added represents 4.7% of the 2017 EU-27 GDP [17]. The EU-27 bioeconomy employed 17.5 million full-time jobs in the same year, i.e. 8.9% of the EU-27 workforce [17]. Nowadays, the bioeconomy sectors employing most of the workforce are still the traditional sectors i.e. agriculture and food industries (see Table 1 for a detailed breakdown per sector).

Nonetheless, in the last decade, novel bio-based products have emerged in sectors traditionally dominated by fossil (or mineral) resources. Over the last decade, the bio-based chemicals and plastics industry experienced a 45% total added value growth accompanied by a 10% increase in the number of jobs created [18]. Electricity from biomass doubled its added value and almost tripled the number of jobs [18]. Conversely, the labor force in traditional bioeconomy sectors i.e. agriculture, and aquaculture, textiles and furniture/wood products declined significantly (see Table 1).

Sector	Value added 2017 (billion €)	Total growth of value added 2008-2017 (%)	Employment 2017 (million jobs)	Employment growth 2009-2017 (%)
Agriculture	189	37%	9.3	-16%
Forestry	25	50%	0.5	7%
Fishing and aquaculture	7	24%	0.2	-3%
Food and beverages	215	23%	4.4	6%
Natural textiles	21	16%	0.7	-21%
Furniture and other wood products	47	18%	1.4	-8.5%
Paper	42	25%	0.6	2%
Bio-based chemicals and plastics	60	45%	0.4	10%
Liquid biofuels	3	35%	0.02	-26%
Bioelectricity	4	105%	0.02	185%

Table 1. EU-27 bioeconomy value-added and employment per sector (2017 data based on [17] and [18])

The growing trend of innovative bio-based commodities is expected to continue under the push from the European Green Deal. In particular, the EU bioeconomy is expected to rise from the current 4.7% of GDP to 7.4% by 2050 to limit the global temperature rise to 2 °C above pre-industrial levels [19]. Besides the expected growth in the production of food products, most of the bioeconomy growth will be

driven by innovative bio-based commodities i.e. bio-based chemicals, plastics and fuels [19]. Between 2020 and 2030, all these innovative commodities are expected to grow between 25% and 35% to reach the 2 °C target in 2050 [19]. In particular, the EU Bio-based Industry Consortium already targets at least 25% growth by 2030 [20]. Given this significant growth in the short term, there is an urgency to understand the climate and environmental impacts of innovative bio-based products.

1.2 THE INTEREST IN LOCALLY SOURCED AGRO-INDUSTRIAL AND FOOD-PROCESSING BY-PRODUCTS

Beyond the shared acknowledgement of the bioeconomy role in the EU green transition, there are various points of view regarding how a future biobased economy should look, within which boundaries it should operate and its relationships to the rest of the economy [4,21]. In fact, the promotion of biobased products in Europe has led to significant imports of bio-based products from outside the EU [22]. This fact may generate conflicts of interest in biogenic resources' international trade since sustainable biomass is a scarce resource and its availability in the EU is not sufficient to meet EU ambitions [2,21,23]. For example, EU ethanol imports doubled in 2019 compared to 2018, with imports from the US increasing by 233% [24]. Similarly, 43% of the bio-based solvents consumed in the EU are imported [25].

In particular, among the most discussed trade-offs generated by bio-based production, there are the competition for crops that might threaten food security [26] and the impacts on biodiversity of large-scale crop production [27]. Given the concerns about food security and land competition, locally sourced bio-based "residues" have attracted increasing attention to extend the amount of bio-based products produced sustainably in the EU [28,29].

This thesis investigates several innovative bio-based products from this type of feedstock that have been either recently commercialized or are close to reaching the market. The explored feedstocks are food processing residues i.e. potato by-products and used cooking oil, forestry residues i.e. wood chips and a by-product from paper/pulp industries i.e. lignin.

Besides avoiding land competition with food, these bio-feedstocks can be employed in biorefineries and bioenergy technologies that rely on local availability, reducing fossil fuel dependency and biomass imports [30]. Moreover, this type of residual by-products are usually cheaper than crops from dedicated cultivation [30] and their production usually has a lower climate change impact than dedicated crops [31–34].

1.3 LCA AS A KEY TOOL TO MEASURE THE IMPACTS OF INNOVATIVE BIO-BASED PRODUCTS

The last two decades have seen the development of numerous bio-based alternatives in sectors that have been (and currently still are) dominated by the fossil industry. Among these products, we can find bio-based plastics (in some cases also biodegradable ones), bioenergy, bio-based fuels, bio-based chemicals and other unconventional bio-based products (e.g. asphalts). These products are named "bio-based" since they are wholly or partly derived from materials of biological origin. Nevertheless, the name "bio-based" does not directly imply being environmentally friendly and their potential environmental impacts need to be minimized using a proper monitoring tool.

Life Cycle Assessment (LCA) is one of the main tools used by worldwide policymakers, scientists, and companies to monitor the environmental impact of bio-based products [9,21,35–37]. LCA is an internationally standardized method to assess the environmental impacts of products or services in all their life cycle, from raw materials extraction to waste management [38]. The results of LCAs have already been used to build environmental sustainability criteria in several policy instruments and mechanisms covering several bioeconomy sectors e.g. biofuels [3,35,39–41].

So far, policy instruments typically only cover a few impact categories (mostly climate and/or energy demand). However, trade-offs typically exist between bio-based and petrochemical products [7,42]. When the scientific literature considered other categories, the LCAs of bio-based products often showed higher eutrophication, acidification and water depletion impacts than their petrochemical counterparts due to biomass production [43,44]. So, it is necessary to apply LCA to provide a comprehensive overview of environmental impacts to understand if these tradeoffs could also be observed for the emerging products made from biobased "residues" streams.

1.4 CHALLENGES OF MODELING BIO-BASED PRODUCTS IN LCA

The user of results from LCAs of bio-based products should be aware that several methodological uncertainties exist despite the existence of ISO LCA methodology standards. In particular, there are three major modeling uncertainties behind the LCAs of bio-based products: multifunctional processes, land-use changes (LUCs) and biogenic carbon [45–49]. Of these three methodological complexities, this thesis focuses on modeling multifunctional processes. This issue is considered one of the most "controversial" in the LCA scientific literature and with low convergence in LCA guides [50–55].

Since most innovative bio-based products are produced from multi-output systems [56,57], their LCAs regularly run into the "multifunctionality issue". The multifunctionality issue affects all LCA environmental impact categories, whereas LUCs and biogenic carbon predominantly affect the climate change impact of bio-based products. In addition, LUC impacts are generally more significant for products made from dedicated crops than bio-based by-products [58]. For these reasons, a major focus was given to the multifunctionality issue in this thesis.

Most methodological decisions made in LCA strictly depend on the goal and scope of the LCA. The goal and scope of the LCA have a direct link with the selection of the modeling approach and life cycle stages. For example, the modeling approach influences multifunctionality practices and the inclusion/exclusion of land-use changes in the system boundaries [59–61]. The literature distinguishes between two "major types" of modeling approaches i.e. attributional and consequential LCAs [59,62].

Attributional LCAs aim to understand which environmental impacts should be attributed to an economic activity [51]. Consequential LCAs aim to understand the direct and indirect changes in worldwide environmental impacts caused by changing or adding to some degree an economic activity [51]. Attributional LCAs usually apply an allocation share (e.g. based on the mass of co-products) to partition the environmental impacts of multifunctional processes and rarely include (indirect) LUCs [45,63,64]. In consequential LCAs, the so-called substitution method is often used to avoid allocation and both direct and indirect LUCs are generally included in the life cycle inventory [45,63,64].

The selection of the life cycle stages has a major effect on the modeling of biogenic carbon. The LCA modeling of biogenic carbon is under debate mainly for two aspects i.e. the credits from temporary or permanent biogenic carbon storage and the timing of carbon emissions [65]. The practices in the LCA literature regarding these two modeling aspects mainly vary depending on the inclusion or not of the end-of-life in the accounted life cycle stages and the type of product i.e. durable application or not (e.g. a fuel) [65].

1.4.1 THE LCA MULTIFUNCTIONALITY ISSUE AND THE ISO HIERARCHY INTERPRETATION ISSUE

The multifunctionality problem occurs every time a process provides more than one function. Since the LCA method calculates environmental impacts per functional unit, it is necessary to find a way to allocate the environmental impacts of a process between these functions or avoid such allocation [38]. Multifunctionality is usually present in bioeconomy systems since biorefineries, bioelectricity systems, paper industries and food industries produce more than one single economically valuable product.

ISO 14044:2006 recommends a three-level hierarchy that practitioners should follow in dealing with this problem:

- 1. The allocation should be avoided either by dividing the process into subprocesses or by system expansion.
- 2. Performing allocation based on the "physical relationships" between the coproducts.
- 3. Using allocation methods reflecting "other relationships" that are present between the products (for example, their economic value).

Unfortunately, selecting the best allocation rule based on the abovementioned hierarchy is difficult. As pointed out by Wilfart et al. [66] among others, one of the reasons is that "following this decision hierarchy encounters several unresolved issues that ultimately result in a variety of strategies, some of which are supported only by dubious interpretations". Consequently, the literature and sectorial/national ISO-compliant guidelines provide different (and sometimes contradictory) rules for applying such a hierarchy in practice [66–69].

There are two major debates on the interpretation of the ISO multifunctionality hierarchy. As explained in section 1.4, selecting the modeling approach (attributional or consequential) is crucial for selecting the multifunctionality method [61,69–71]. The first debate regards the use of the substitution approach as a system expansion method in attributional LCAs (ALCAs). This is not perceived as rationale by many LCA experts [51,52,68,71–73] since 1) ALCA modeling should not rely on perturbation logic or counterfactual notions, such as avoidance [51,59] and 2) the sum of the impacts accounted by ALCAs should sum up to the worldwide impacts, and this would not be valid anymore if substitution were applied [71,74]. However, some guides and authors consider system expansion as a synonym for substitution [75]. On this basis, substitution should also be used in attributional modeling instead of allocation since ISO prefers system expansion (e.g. [76–79]). This concept should be tracked back to the conceptual equivalency between substitution and system expansion illustrated by [80,81].

The second debate regards the meaning of "physical relationships". Some guides and practitioners interpret the term "physical relationships" as "physical causal" relationships i.e. reflecting quantitatively how a change in the product mix changes the inputs, outputs and impacts of the system (causality concept) [41,82]. However, some guides and practitioners interpret allocation based on "physical relationships" as an allocation based on a physical parameter [83–85].

The rationales of the current interpretations of the ISO 14044 hierarchy have been summarized by Pelletier et al. [86]. However, it is not fully understood how the different implementation practices of the ISO hierarchy developed and how the discrepancies were generated, leading to current inconsistencies in various ISOcompliant LCA guides.

To picture the relevance of this interpretation problem, we can take a widely investigated case regarding a well-established product as an example, i.e. meat products. For the same meat production system, up to 7 different allocation methods can be applied following different LCA ISO-compliant guidelines [66], leading to a significantly different environmental impact for meat products [51,66,87]. As a result, since milk is a co-product of meat production, interpreting and comparing the results of LCAs of dairy products is challenging [87–92]. Economic allocation between milk and beef favors beef production, while substitution drastically reduces the environmental impact of milk [93]. Similar issues were pointed out by

Agostini et al. [94] in their review of the 100 most cited articles on LCA of bioenergy and by Brankatschk and Finkbeiner [69] for agricultural systems.

While the multifunctionality uncertainty on the climate change impact of established bio-based products has been broadly investigated, the effect of this issue on innovative bio-based products is still unknown. In fact, many bio-based products have been recently commercialized for the first time or are close to commercialization [10,95]. Moreover, multifunctionality practices in LCAs of bio-based products largely vary depending on the specific bioeconomy sector and the leading ISO-compliant guide followed by practitioners of that sector. While inconsistencies have been remarked already by several authors [52], a quantitative analysis of such inconsistent practices is still missing.

The relevance of the problem becomes even more important considering the following three facts:

- the LCA literature has broadly acknowledged that using a different allocation method to assess the same bio-based product can change the results significantly, leading to conflicting recommendations to the study's commissioner and/or policymakers [50,96].
- previous literature showed that comparing two products fulfilling the same function with multifunctionality practices consistent on both sides is important [69,97]. However, the same process might provide products with multiple applications for various sectors and actors involved. Consequently, part of the environmental burdens are probably "either unaccounted or doubly accounted for" [69], leading to limited comparability of the results.
- 3. the environmental performance of bio-based products is often benchmarked with petrochemical products, which are as well affected by the same multifunctionality issue at the refinery level. So, comparing the environmental impact of bio-based with petrochemical products has a double uncertainty related to the multifunctionality issue on both sides.

1.4.2 THE MULTIFUNCTIONALITY UNCERTAINTY IN LCAS OF BIO-BASED BY-PRODUCTS

As mentioned in section 1.2, among the main reasons for interest in bio-based products produced from bio-based "residues", there is the expected lower climate

change impact than that of bio-based products from dedicated crops [31–34]. However, their lower climate change impact also originates from the allocation method applied to the bio-based by-products used as feedstock. A cut-off allocation does not assign upstream production burdens to the by-product, i.e. it is considered a "waste" in ISO LCA terms. Economic allocation assigns a much lower upstream impact to the by-product due to its lower market price than the main product.

A recent meta-analysis [65] of 30 LCA studies of plastic materials from bio-based residue or side streams highlighted that the most common multifunctionality practices are using a cut-off approach, i.e. no upstream production burdens are allocated to the feedstock (9 out of 30) and economic allocation (7 out of 30). Several studies did not clearly specify the allocation method adopted (5 out of 30). A few LCAs used mass allocation (3 out of 30). Other studies used system expansion to account for the consequences of redirecting the by-product feedstock from their current use (5 out of 30). One study compared economic allocation and system expansion [98].

Mass allocation assigns the same environmental impact per unit of mass to byproducts and main products (e.g. 1 kg of rice and 1 kg of rice straw), leading to a much higher environmental impact for by-products compared to the impacts resulting from cut-off and economic allocations [65]. Accounting for the environmental consequences of diverting the by-product feedstock from their alternative fate is in line with consequential modeling and leads to a higher impact than cut-off allocation and sometimes also of economic allocation depending on the current use of the by-products [48,98,99]. However, the selection of approaches to deal with multifunctionality in the LCA literature is often inconsistent with the study's goal and consequent modeling choice. As highlighted by a recent review of the 100 most cited LCAs of bioenergy products, the choice of the modeling approach based on the goal of the study, consequent selection of unit processes to be included in the system boundaries and multifunctionality approaches are made inconsistently in the majority of these 100 LCAs leading to misleading or inconsistent results in 67 of them [94].

Based on the experience gained by recent literature on established bio-based products from low-economic/physical significance streams, the environmental impact of this type of product is expected to be more affected than products from dedicated crops by the LCA multifunctionality uncertainty [55,57]. Given the lack of a shared interpretation of the ISO multifunctionality hierarchy and consequent inconsistency in multifunctionality practices, significantly different LCA results for products from low physical/economic magnitude flows like by-products are expected [55,57].

The reason for the high variability of LCA results for such products is straightforward for allocation with the following example. Suppose a pulp mill extracts lignin from the black liquor instead of using it as a fuel. Lignin becomes a co-product. The main product i.e. pulp, represents almost 90% of the mass of the two co-products (pulp and lignin) and a bit more than 90% of the economic revenues generated [100]. Suppose a slightly different assumption/data (e.g. on the price or moisture content), the allocation share of pulp could probably change to e.g. 85% and that of lignin (i.e. the by-product) to 15%. Therefore, the environmental burden allocated to the lignin by-product would increase by 50%, while the impact of the main product, i.e. pulp, would decrease only by 6%. Hence, the environmental impact of bio-based by-products is highly sensitive to the input data e.g. moisture content or prices used for determining allocation factors. As a result, the carbon footprint of one kilogram of lignin from the same pulp mill can vary between a negative impact ("substitution-based allocation") and 4 kg of CO₂eg/kg, by just varying the allocation method [55]. Consequently, the savings of climate change impact allowed by lignin-derived adipic acid can range between -90% (savings) and + 100% (an increase of impact) in the literature [67].

Therefore, it is important to increase consistency in selecting the multifunctionality practices to avoid generating arbitrary or extreme results with consequent erroneous interpretation and misleading recommendations to the study's commissioner and consequent decision-making [57,67]. Given the expected growth of innovative products from bio-based by-products/residues reaching the market, it is necessary to quantify the multifunctionality uncertainty behind the reported environmental impacts. However, the multifunctionality uncertainty for bio-based by-products has been investigated only incidentally so far when dealing with the climate change impact a specific product and rarely taking a holistic perspective on multiple products and at the same time on multiple impact categories [57].

1.5 RESEARCH OBJECTIVES AND STRUCTURE OF THE THESIS

This thesis aims to analyze the life cycle environmental performance of several innovative bio-based products from agro-industrial and food-processing by-products compared to their petrochemical counterparts. Since multifunctionality modeling causes high uncertainties for bio-based by-products, a major focus of this thesis is LCA multifunctionality. Accordingly, multifunctionality practices and uncertainties were deeply discussed for all investigated products for the entire spectrum of indicators considered useful for decision-making.

The ISO 14044:2006 hierarchy is the key selection tool for multifunctionality practices, but various ISO-compliant LCA guides have interpreted it differently. These different interpretations have led to inconsistent multifunctionality practices in LCAs of bio-based products. However, how these interpretations originated is partly unclear. Hence, the first research question of this thesis is:

1. What multifunctionality practices are adopted in LCAs of bio-based products and how can the consistency of the life cycle inventory model be improved?

A first review looks at multifunctionality practices and their effect on results' comparability for one of the feedstocks of interest i.e. lignin. A second review quantified the practices in implementing ISO 14044 multifunctionality recommendations in all LCAs of multifunctional processes in the literature (94% resulted in being somehow related to bio-based products). The same review investigates the debates on the interpretations of the ISO multifunctionality hierarchy and their historical origins.

Afterwards, this thesis aims to fill the lack of environmental LCAs of innovations from bio-based by-products that have been either only recently commercialized or are close to commercialization.

Hence, this thesis aims to answer the following research question:

2. What is the environmental impact of innovative bio-based products made from by-product/waste streams compared to their fossil counterparts?

Among products from bio-based "residues", the following products were analyzed: *polypropylene from used cooking oil* (Chapter 4), *bioenergy from wood chips* (Chapter 5), *bio-jet fuel from potato peels* (Chapters 6) and *lignin-based asphalts* (Chapters 7). An LCA was conducted based on primary data for their innovative conversion technologies for all these products. Moreover, LCAs of *other bio-based products from lignin* (Chapter 2) from the literature were also reviewed.

These emerging products belong to different sectors within bioeconomy. Therefore, they were considered suitable to provide an overview of the environmental impacts of innovative bio-based products in a holistic way. Compared to LCAs of other products from bio-based by-products in the literature, the spectrum of the environmental impacts was broadened to other categories than just climate change, such as acidification, eutrophication and human toxicity. In this way, it is possible to understand if the typical environmental trade-offs of bio-based products from bio-based by-products from bio-based by-products.

Based on the insights gained from their LCAs, e.g. on their environmental hotspots, this thesis aims to provide a list of recommendations to produce them sustainably. Potentially, these recommendations can be extended to similar innovative bio-based products to guide investment and research towards environmental sustainability.

As abovementioned, the environmental impacts of products from loweconomic/physical significance streams are usually significantly affected by the multifunctionality issue. So, this thesis aims to answer a third research question:

3. How are the LCA results of such products influenced by the approach adopted to deal with the multifunctionality issue?

The answer to this research question is provided by broadly discussing the effect of the allocation approaches among the ones allowed by ISO 14044:2006. To show the magnitude of multifunctionality uncertainty on the environmental impacts of the investigated products, performing various sensitivity analyses is necessary. This exercise is essential to detect the impact categories for which the comparison with petrochemical products is hampered by such uncertainty. Based on the insights from the case studies and the reviews, this thesis aims to provide several recommendations for good multifunctionality practices for bioeconomy products. Table 2 provides an overview of the topics of each chapter and their respective research objectives.

Chapter	Торіс	Res	Research question		
		1	2	3	
2	Review of Life Cycle Assessments of lignin and derived products: lessons learned	Х			
3	Reviewing ISO compliant multifunctionality practices in environmental Life Cycle modeling	Х			
4	Kraft lignin as a bio-based ingredient for Dutch asphalts: an attributional LCA		Х	Х	
5	Combining biomass gasification and Solid Oxid Fuel Cell for heat and power generation: an early- stage Life Cycle Assessment		Х	Х	
6	Attributional and consequential LCAs of a novel bio- jet fuel from Dutch potato by-products		Х	Х	
7	Environmental Life Cycle Assessment of polypropylene made from used cooking oil		Х	Х	

Table 2. Overview of chapters and corresponding research objective



2

CHAPTER

Review of Life Cycle Assessments of lignin and derived products: lessons learned

Published as:

C. Moretti, B. Corona, R. Hoefnagels, I. Vural-Gürsel, R. Gosselink, M. Junginger. Review of Life Cycle Assessments of lignin and derived products: lessons learned.

> Sci. Total Environ, 770 (2021), 10.1016/j. scitotenv.2020.144656

ABSTRACT

In the last decade, the use of lignin as a bio-based alternative for fossil-based products has attracted significant attention, and the first LCAs of lignin and derived products have been conducted. Assessing side-stream products like lignin and potential benefits compared to their fossil counterparts presents complex methodological issues. This article provides a critical review of forty-two peerreviewed LCAs regarding lignin and derived products. Methodological issues and their influence on the LCA results include the choice of the modeling approach and system boundaries, functional unit definition, impact categories considered, type of data used, handling multifunctionality and biogenic carbon modeling. The review focused on climate change impacts, as this is also the main impact category considered in most studies. Other impact categories in the comparison between lignin-based products and counterparts was also discussed with examples from the studies. Based on ten lessons learned, recommendations were provided for LCA practitioners to increase future consistency of environmental claims made about lignin and lignin-based products. The finding suggest that the environmental performance of lignin-based products is significantly affected by both 1) LCA methodological problems such as allocation practices and biogenic carbon modeling and 2) technical aspects such as the percentage of lignin in the composition of products and the selection of the fuel to replace lignin in internal energy uses. Beyond this, the reviewed LCAs showed that often lignin-based products offer better environmental performances than fossil-based products, especially for climate change.

2.1 INTRODUCTION

Next to cellulose and hemicellulose, lignin is the second most abundant natural biopolymer on Earth and accounts for approximately 30% of the organic carbon in the biosphere [101]. In nature, lignin is an aromatic-composed binder that provides stiffness and strength to the stems of plants [102]. From a bioeconomy perspective, lignin is currently mainly used to produce bioenergy (electricity and heat) but has recently received attention as a renewable raw material for the production of chemicals and materials to replace petrochemical resources and sometimes provide also technical improvements. For example, plastic polymers can take advantage of the complexity of the lignin molecule to avoid the transformation steps to convert simple molecules into complex ones [103]. Other examples of interesting applications where lignin can be used to replace conventional materials are displacing ureaformaldehyde in adhesives [104], bitumen in asphalts [105], polyacrylonitrile in carbon fibers [55], polyol in polyisocyanurate foams [103] and liquid fuels [106]. Yet, lignin is currently largely unexploited for these purposes [107]. Moreover, lignin can be used in other industrial applications that can benefit from the good surface activity of lignin [108] such as adsorbents for CO₂ capture [109,110] and catalysts [111,112].

Lignin is mainly produced as a side stream of either the pulp and paper industry or from lignocellulosic biorefineries [102,107]. In the pulping industry, lignin can be extracted from black liquor which is a by-product of the wood pulping process of pulp mills [103]. In biorefineries, lignin is obtained as a non-fermentable side stream separated during biomass pre-treatment [56]. With the expected development of lignocellulosic biorefineries more lignin is expected to become available. In both cases, lignin is currently mostly used internally to deliver energy needs [102] but it can also be marketed [103]. Moreover, in both pulp mills and lignocellulosic biorefineries, the lignin extracted often exceeds the internal energy demand and can be sold externally [102,113].

For pulp mills, extracting the lignin from the black liquor can be economically advantageous to have an extra source of revenue and diversify the products. Moreover, in most pulp mills, the recovery boiler works at maximum capacity since the upgrade of such a boiler is economically prohibitive [100,114]. By extracting lignin, part of the solids from black liquor are taken away and the recovery boiler can be de-bottlenecked. This de-bottlenecking can increase the production of pulp and soap generating additional revenues [55,114].

The final application of technical lignin is largely influenced by the chemical and physical characteristics of the lignin [115]. Beyond the feedstock used and the distinction between pulp mills and (lignocellulosic) biorefineries, the chemical structure of lignin is often influenced by the lignin production process and extraction techniques [116]. Figure 1 summarizes the different types of lignin according to their extraction process and main suitable applications and indicative market price ranges for each type of lignin. Market prices depend amongst others on purity and potential application.



Figure 1. Example of possible applications and indicative market price ranges of various types of lignin [117–119].

The Kraft process is the dominant process in the pulping industry [120,121]. Other conventional pulping processes include the sulfite process and the soda process. With the Kraft process, lignin is obtained from hardwoods and softwoods using sodium hydroxide and sodium sulfide mixed in hot water [122]. This mix is named white liquor and the residue of this process is the black liquor, from which lignin can be isolated. Among extraction techniques, acid precipitation through CO₂ and/or

sulphuric acid (also commercially known as the Lignoboost process) are the most common [122]. Alternatively, the organosoly process (solvent pulping) is a promising option that enables the extraction of relatively pure lignin but is only used at a small scale [122]. Sulfite pulping allows isolation of lignosulfonates from spent sulfite liquor. In (liqnocellulosic) biorefineries, the most common techniques for liqnin separation from lignocellulosic biomass are steam explosion, acid pretreatment, enzymatic hydrolysis and alkaline hydrolysis pretreatment [107,123]. Moreover, the abundant presence of aromatics in lignin makes it attractive for chemicals and fuels, and different depolymerization routes to aromatics (BTX and phenolic compounds) exist [124]. Among novel routes for the production of bio-based aromatics from lignin, there are pyrolysis technologies, direct hydrodeoxygenation, and hydrothermal upgrading [124]. These routes aim to achieve good separation of the lignin from the cellulose and hemicellulose without changing it chemically or physically. This allows to utilize fully lignin's macromolecular structure in materials such as asphalt binders, adhesives, carbon fibers, resins and polymer composites. This has led to the lignin-first biorefinery concept that considers strategies to prevent structural degradation of lignin during biomass fractionation [125]. Furthermore, vanillin can be produced from the oxidation of lignosulfonates that find use in foods and fragrances [126].

Lignin has the potential to substitute fossil fuels in both energy and non-energy use sectors to improve energy supply security and to contribute to climate change mitigation. For this reason, important development efforts are made by bioeconomy firms to make such a replacement possible. However, it is necessary to consider unambiguous sustainability criteria to assess if these alternative products allow actual environmental benefits compared to their fossil counterparts. In the bioeconomy, the tool that is often used to perform such a comparison between conventional products and bio-based alternatives is life cycle assessment (LCA) [35]. LCA is a standardized tool to model the entire life cycle of a product or system from resource extraction to final waste management [38,127]. In the last decade, many peer-reviewed LCAs have been conducted to assess the environmental impact of lignin and the potential environmental benefits that lignin-based products can offer.

However, assessing the environmental impacts of bio-based products with LCA can be challenging since multiple life cycle modeling choices have to be defined

by the practitioners [42,128]. In particular, the assessment of lignin and lignin-based products is among the most challenging case studies in bioeconomy. The origins of these challenges can be found in both *LCA methodological uncertainties* (e.g. handling co-products) that affect products from residual streams/bio-based by-products like lignin, and *data uncertainties* related to the low level of maturity of the production processes for lignin products for which often only lab-scale measurements are available. For these reasons, the environmental impact of lignin and lignin-based products is affected by high variability in the various LCAs reported in the literature [67]. The carbon footprint of one kilogram of Kraft lignin can vary between -23 kg CO₂eq to 4 kg of CO₂eq depending on the selected allocation method [55] while the savings of GHG emissions allowed by lignin-derived adipic acid can range between -90% (savings) and +100% (an increase of impact) depending on the data used and methodological choices applied [67].

This article is a critical review of peer-reviewed LCA studies of lignin and ligninbased products from the scientific literature. Given the methodological challenges in assessing lignin and lignin-derived products, the aim of this review is to obtain insights from the main findings of these studies and to evaluate qualitatively and quantitatively the methodological choices made in these LCAs and their consistency and robustness. Moreover, based on the results of these LCAs, potential environmental benefits of lignin-based products compared to the petrochemical products that they can replace are discussed. The insights from this review can be an important added value for LCA practitioners in the bioeconomy sector.

2.2 MATERIAL AND METHODS

2.2.1 Selected studies

The LCA studies on lignin and lignin-based products were retrieved from the Scopus database (www.scopus.com) on July 8th 2020. In particular, the search¹ was based on two main keywords i.e. "life cycle assessment" or its acronym "LCA" and "lignin" looking at their presence in titles and abstracts. Only studies published in English and documents published in scientific journals were considered. As a result of these parameters, 62 peer-reviewed articles were retrieved. After further

¹ The search string was: TITLE-ABS (("Life Cycle Assessment" OR Ica) AND (lignin)) AND (LIMIT TO (LANGUAGE, "English")) AND (LIMIT-TO (SRCTYPE, "j")). It is possible that some biofuel studies that include biochemical processes making assumptions about lignin were not considered if they did not mention lignin in the abstract.

screening, 48 articles concerning LCA of either lignin or lignin products were identified by excluding, for example, studies where the acronym LCA was not used as an acronym of life cycle assessment (but for "lignin-based activated carbon"). In particular, we focused on the studies published in the last decade which represents 41 studies out of 48. Of these 41 studies, about 85% were published in the last five years, which highlights the increasing interest in the topic. Moreover, a recent study published in August 2020 [129] and not yet present in Scopus at the time of the search was also considered which resulted in the end in a total of 42 LCA studies to be assessed.

2.2.2 Aims and structure of the review

After the LCAs of lignin and lignin-based products were selected, the analysis was conducted in four main steps. The structure of the following sections resembles these four steps and the analyses conducted in each step (see Figure 2).



Figure 2. Steps of the review

The first step of the review was focused on understanding the content of the articles. In particular, we mainly answered these two questions: 1) what was the goal of the LCA studies? and 2) what lignin production system was investigated?. The first step aimed at providing recommendations to increase future consistency and was targeted at LCA practitioners only. To achieve this objective, we reviewed how the LCA methodology was applied in the lignin case studies taking ISO standards and major EU LCA guidelines as methodological reference documents. This allowed us to identify the state of art in assessing the environmental impact of lignin products. This part of the review can be found in section 2.3.2 and sub-sections. The second step of the review aimed at comparing quantitatively the environmental impacts of lignin and lignin-based products reported in these LCAs. The result of this analysis (described in section 2.3.3) was used to provide an overview of the impacts of lignin and the environmental performances of various lignin valorisation options compared to fossil counterparts. This second step was targeted as well to LCA practitioners, but insights could be also interesting for policymakers and lignin producers. A major focus was out on climate change since it was the most considered impact in the selected studies (see section 2.3.2.4), it is among priorities in policy agendas and the impact assessment methods are (almost) standardized allowing a (direct) comparison of the results. The other environmental impacts of lignin were also considered in section 2.3.3.3 (and following). In the last step of the review, the main findings were summarised and recommendations for future research provided.

2.3 RESULTS

2.3.1 Product systems

The 42 environmental LCA studies considered in this review (see supplementary materials for studies' categorization at the online link of the published article) can be divided into the following categories based on their *main* product system investigated:

- Assessing Kraft lignin (3 studies fall in this category [55,100,103]);
- Assessing organosolv lignin (1 study [129]);
- Assessing a biorefinery delignification process using natural malic acid (1 study [130]);
- Assessing lignin-based applications from various lignins (15 studies);
- Assessing major biorefinery products such as ethanol or lactic acid (21 studies);
- Performing a meta-analysis of life cycle energy and GHG emissions of biobased chemicals (among them, some produced from lignin (1 study [67])

In the LCAs of Kraft lignin of Bernier et al. (2013) and Culbertson et al. (2016), the focus was on evaluating the environmental implications of introducing lignin

extraction in Kraft mills. A similar study [119] was also performed for biorefineries with and without marketed lignin.

Concerning the LCAs that look at the products using lignin, their aim was often twofold: identifying the environmental hotspots in the production processes and evaluating possible environmental advantages in comparison with petrochemical products. Among the investigated lignin-based products, there were *adhesives* especially for wood fiberboards and laminates [115,131,132], *phenol* and *propylene* [133], *transportation fuels* [106], *asphalt* [134], *nanoparticles* [135], *polyurethane foams* [136], *fertilizers* [8], *vanillin* [137], *adipic acid* [138,139], *catechols* [140] and *carbon fibers reinforced polymers* [141].

Concerning the studies investigating major biorefinery products (e.g. ethanol), some LCAs focused on assessing products of biorefineries where lignin is not a product since it is fully used for internal needs. For example, Vera et al. (2020) assessed a biorefinery producing ethanol and lactic acid which was equipped with a combined heat and power plant (CHP) where lignin was combusted for internal uses of the biorefinery without any surplus of heat and electricity. In other LCAs of biorefineries e.g. [142], the heat and electricity from the process were fully externally sourced from fossil fuels and all the lignin produced by the biorefineries, lignin was only an intermediate product which was further processed to obtain biofuels and/or chemicals [133,143].

2.3.2 Analysis of methodological choices

The goal of the study has strong implications in all the choices that the practitioner has to make to conduct the LCA. In particular, it strongly affects the definition of the scope of the LCA. In fact, based on the goal, the following parts of the scope are strictly defined: the unit processes included in the system boundaries, the modeling approach to be used (and the type of data to be used), the functional unit (FU) and the methods to deal with co-products. All these aspects are crucial to interpret the results of an LCA and understand what can be concluded and what not from the LCA results. Accordingly, the goal of the selected studies and the modeling choices made by the LCA practitioners were noted in the following sections where relevant.
2.3.2.1 Modeling approaches and system boundaries

The appropriate modeling approach (attributional or consequential) is directly linked with the goal of the study. An attributional approach should be selected if the goal is to assess the environmental hotspots of a process or the determination of the environmental impact of a single bio-based product to compare with fossil products. A consequential approach should be used instead to assess a change in a specific system and the overall consequence of this change in the system and in the world outside. For example, in view of the worldwide environmental impact, is the current use of the black liquor in Kraft mills for internal energy better than isolating lignin from it to be marketed?

Despite the fact that ISO 14044:2006 does not distinguish between attributional and consequential LCAs, for many practitioners [61,144] and handbooks (e.g. ILCD handbook [145]), it is important to select the modeling approach based on the goal of the study. In a 2020 study [146], using a text mining process, it was shown that 75% of the LCAs assessing multifunctional case studies did not clearly mention the modeling approach. When applying the same text mining method to the reviewed studies on lignin and lignin-based products, it was also found that 78% of the studies did not use the keywords "attributional" or "consequential" to specify the approach followed. Of the remaining studies, 8 LCAs were defined by the practitioners as attributional while only one article [138] defined the approach followed as consequential. The selection of the modeling approach also affects other decisions that the practitioners have to take to conduct an LCA. First of all, attributional studies require average data while in consequential studies marginal data are used. Second, depending on if the study is consequential or attributional, the system boundaries and the unit processes included within the system boundaries change (see a simplified example for dealing with electricity surplus in Figure 3). Third, depending on the goal, the type of system expansion method that can be applied is different: enlargement (only expansion of the boundaries) or substitution (expansion followed by substitution). While enlargement can be used in both attributional LCAs (ALCAs) and consequential LCAs (CLCAs), the use of substitution as a system expansion method is inconsistent with attributional modeling [51]. Further details about the use of system expansion to deal with multifunctionality can be found in section 2.3.2.3.



Figure 3. A simplified example of differences between system boundaries in attributional (black) and consequential LCAs to deal with a surplus of electricity generated using lignin burned internally. In red, system boundaries and unit processes that would be included within the boundaries in a consequential analysis.

Concerning consequential LCAs, this type of study uses economic market modeling to forecast what will happen as a consequence of the assessed change. For example, market modeling is needed to understand what mix of technologies will be replaced by a by-product that is produced because of the assessed change in the system (e.g. lignin extracted from a pulp mill and no longer used for internal combustion). To avoid malpractices, it is necessary to understand what are the main market drivers of the system (what is the purpose of the system) and what products satisfy such market drivers. For example, the Kraft black liquor currently exceeds the demand for non-fuel uses [103]. So, the demand for paper (pulp) is the market driver of the Kraft mills and not the market applications of lignin [103]. Another aspect that is important to consider in a consequential LCA of Kraft mills is the energy source used to replace black liquor. Since black liquor is mainly combusted for internal energy needs, a likely scenario is that fossil fuels will be used to produce the part of steam that cannot be produced anymore from black liquor [103,114]. On the other hand, depending on regulations policy schemes in place, biomass is also an option especially if there is the availability of low-quality biomass in the vicinity (e.g. bark). Similarly, also in biorefineries, different alternatives are possible as replacement of lignin for internal energy purposes. Among them, the

most probable options are the use of natural gas or biofuels such as wood chips or biogas [106]. While the use of natural gas is the most cost-effective (and better water footprint), the use of lignocellulosic biofuels is generally the best solution if the main goal is to minimize GHG emissions [106].

Concerning the life cycle stages considered, most of the studies (87%) were cradle to gate studies i.e. the use and end of life of the products delivered by the system were not included in the assessment. As exceptions, the following cases were found:

- Well to wheel studies [106,147,148]. In well to wheel studies, the combustion of the transportation fuels produced is considered (end of life=use phase).
- Investigation of a specific lignin extraction process (using natural malic acid), which was performed by Yiin et al. (2018). The boundaries of the systems were gate to gate: from the harvested oil palm empty fruit bunch to the extracted lignin.
- A full cradle-to-grave study for polyurethanes produced using lignin-derived polyols [136].
- A full cradle-to-grave study for adhesives used in fiberboard production [132]. In this study, the same end of life (landfill) was assumed for bio-based and fossil-based products and the dataset for inert waste processed in a landfill was retrieved from ecoinvent [149]. Hence, the different compositions were not taken into account.

2.3.2.2 Functional unit

In LCA, the functional unit is the "quantified performance of a product system for use as a reference unit" [38] and depends on the final function of the products delivered by the product system. Lignin can be used for products that have very different functionalities. How each product can fulfill a specific function has to be accounted for in the functional unit. For example, one of the main functions of polyurethane foams is to provide thermal resistance. Hence, in a comparative LCA of polyurethane foams, the differences in thermal resistances have to be accounted for by the functional unit. A good functional unit could e.g. be the amount of foam needed to achieve a specific thermal resistance [136]. Only if the physical properties and mechanical characteristics that are important for the final applications are comparable, a simplified functional unit based on a mass or a surface is a possible

option. For example, Hildebrandt et al., after checking that the tensile modulus and strength were comparable, defined the functional unit as 1m² of a laminate board [131]. However, most studies, e.g. [134], using simplified functional units did not make a similar check.

The functional units selected in these studies could be cataloged as follows:

- Simplified mass FU based on the output. For example, 1 tonne of pulp [100] or 1 g of vanillin [137]. This type of FU was used by about 50% of the LCAs;
- Simplified energy FU based on the output. For example, 1 MJ of ethanol [150] or 1 MJ of jet fuel [148]. This type of FU was used by about 15% of the LCAs;
- Simplified volume FU based on the output. This FU was used in two studies: 1 liter of ethanol [151] and 1 m³ of finished medium density fiberboard [104];
- Simplified area FU based on the output. This FU (1 m²) was used in two studies [131,132] assessing adhesives for wood fiberboards;
- Input based FU. This type of FU was used in the study authored by González-García et al. [152] where the functional unit was the input of the biorefinery (100 kg dried Pinus pinaster chips);
- Entire biorefinery. This FU was used by two studies [153,154];
- *Multiple FUs (one per each main co-product).* This type of FU was used by several studies [56,133,155];
- *Distance*. The FU of 1 km, which is typical for well-to-wheel assessments and was used by one study [147];
- *Ultimate final application.* This functional unit was used by one study [141] i.e. an automotive part under consideration.

In particular, an *input-based functional unit* or the assessment of the *entire biorefinery* allows to avoid the allocation between the co-products (among them, lignin). This approach is also one of the enlargement methods to solve the multifunctionality problem (for details, see section 2.3.2.3). In this way, the modeling uncertainty generated by the multifunctionality problem is avoided. This approach is applicable if the goal of the study is the identification of the environmental hotspots of a process or an entire biorefinery.

However, this approach is not applicable if the goal requires the determination of the impact of a single co-product. When this is the case (for example to compare it with its fossil counterpart), multifunctionality uncertainty cannot be avoided (except the few cases where subdivision solves the multifunctionality problem, see section 2.3.2.5). Under these circumstances, it is good practice to define multiple functional units to increase transparency and show what is the impact of all coproducts after the allocation is applied. Only in this way, the reader of the LCA can directly understand the effect of the allocation method on the environmental performance of each co-product. For example, Modahl et al. (2015) defined the FUs of their LCA of a Norwegian biorefinery as 1 tonne of product for cellulose, lignin and vanillin and 1 m³ for ethanol and showed the results per each FU.

2.3.2.3 Multifunctionality

Lignin is always a product of multifunctional systems, i.e. systems delivering multiple products. To perform an LCA of this type of system, the selection of a criterion to apportion the impact on each product is necessary. This selection is one of the main sources of uncertainty of LCAs [50,156]. In particular, ISO 14044:2006 provides a three-level hierarchy to deal with this problem. The first level of the hierarchy recommends avoiding allocation, either by dividing the process into sub-processes which are no more multifunctional or by system expansion e.g. by re-defining the boundaries of the system in a way that the system enveloped by the new boundaries is no more multifunctional. Applying subdivision in mills and biorefineries rarely solve the multifunctionality problem. System expansion can be applied in two ways: enlargement (system expansion alone) or (system expansion followed by) substitution. A summary of possible enlargement methods can be found in Moretti et al. [146]. Enlargement is not a solution if the goal of the study requires the determination of the impact of a single output of the system and not of the whole system. For example, in the study by Shinde et al. [154], the impact of the entire biorefinery is considered avoiding the allocation to each of the three coproducts ellagic acid (EA), lignin, and pectin. Substitution is the main option used in consequential studies. Considering Figure 3, once the unit process representing the avoided marginal production is included in the system boundaries, substitution allows to subtract the impact of this unit to the one of the entire system inside the boundaries. In biorefineries and mills, lignin is often a by-product (and not the main product) contributing to less than 50% of the revenues of the system. Accordingly, in these cases, substitution can be an option [57,145,146]. Nevertheless, substitution is sometimes used as a system expansion method in attributional studies leading to either erroneous results or misleading interpretations which emerge especially when multiple impact categories are assessed [51,57]. For example, Akmalina and Pawitra performed an LCA of ethylene from empty fruit bunch (residue of palm oil processing) and compared the obtained impact with the one of fossil ethylene [142]. The interpretation of the results was that bio-based ethylene was much better than fossil ethylene from a climate change perspective (about half impact). Akmalina and Pawitra (2020) solved the multifunctionality issue by substituting the lignin produced with electricity and chemicals. This credit reduced the impact of bio-based ethylene by 83.9% [142], leading to a climate change impact of 1.15 kg of kg CO₂ eq per kg of ethylene. However, the palm oil extraction unit alone had a contribution of 7.17 kg CO₂ eq/kg ethylene. When applying allocation (partitioning) instead of substitution, a completely different conclusion would have been obtained.

The second level of the hierarchy recommends allocation methods reflecting the way "in which the inputs and outputs are changed by quantitative changes in the products" [38], which in the literature has been often referred to as "physical causal relationships" allocation. To lignin, this allocation can be applied "by varying the quantity of lignin precipitated and then observing direct variations in the environmental loads" [103]. In general, the changes modeled using this type of allocation can be either marginal, incremental or average (listed in order of magnitude) [157]. A physical causality allocation based on average changes was used by Bernier et al. (2013) who eliminated a functional output completely. Different from other allocation methods (e.g. energy or economic value), physical causality allocation does not apportion the impact of the system with a static share for all impact categories. This implies that if extracting lignin does not have consequences on the ratio wood chips/pulp, the land occupation impact caused by the wood chips used in the wood pulping process is not allocated to lignin.

The third and last level recommends allocation methods based on parameters such as mass, energy or economic value selected based on their ability to reflect other causal relationships. A comprehensive study on lignin allocation was conducted by Hermansson et al. (2020), who applied 12 types of methods to deal with the multifunctionality of a Kraft mill. Among the methods applied there were system expansion followed by substitution, allocations based on mass, energy, exergy, economic values, marginal allocation, substitution-based allocations and mixed allocations (e.g. mass plus energy). Based on the sample of allocation methods selected, Hermansson et al. (2020) concluded that the impact of Kraft-lignin and derived products could be significantly affected by the allocation choices. The results were highly influenced by the following allocation parameters: (1) the choice of the main product/function (driver of the system), (2) the price of lignin and (3) the choice of displaced outputs. With respect to the first parameter, Hermansson et al. obtained the highest variation of results because some of the allocation scenarios considered lignin instead of pulp as the main product. However, this is very unlikely, since lignin represents about 3-5% of the overall revenues of Kraft mill [100]. So, apportioning all the impact of the Kraft mill to lignin (and no impact to the pulp) or substituting pulp looks unreasonable. When considering that pulp is the main driver of the system, the variation of the impact of lignin calculated by Hermansson et al. becomes much narrower (see section 2.3.3.1). Concerning economic allocation, the price assumed for lignin should reflect the specific lignin under investigation. In fact, the price of lignin is highly variable dependent on the source and quality of the lignin (see Figure 1). However, specific quality-level lignin "has relatively stable prices through the years and seasons" [117]. So, the scenario applied by Hermansson et al. (2020) assuming a tenfold increase in price for lignin in the future was also not considered in the ranges of climate change impacts identified in section 2.3.3.

In some studies, the type of allocation used was not clear (e.g. in [134]) while in most of the LCAs, a sensitivity analysis on the allocation method was performed. As an example, Culbertson et al. (2016) analyzed the impacts of producing pulp applying system expansion by substitution to the co-products in the baseline calculations. In particular, the two co-products (i.e. surplus electricity and lignin) were substituted with grid electricity and phenolic resin [100]. In their sensitivity analysis, mass and economic allocations were used in combination with substitution (keeping the credit for the surplus of electricity).

A summary of the adopted multifunctionality practices in the selected LCAs is shown in Figure 4. Although mass allocation was the most adopted method to deal with multifunctionality, Figure 4 shows that a wide variety of methods were applied between the reviewed studies. The fact that various methods were used is not a problem per se. However, it becomes a problem if the different practices derive from a different interpretation of ISO 14044:2006 recommendations, which has not been uniform in the LCAs of bioeconomy systems in the literature [146]. This problem emerges clearly from the case of lignin. For example, substitution was often used as both a system expansion method or as a basis for the application of an allocation. However, as Montazeri et al. (2016) observed in their meta-analysis, substitution "can produce distorted LCA results for biofuel systems in which coproducts constitute a significant fraction of total economic value, energy flow, or mass flow". For this reason, a check on physical/economical significance should be performed before applying substitution [145]. Applying substitution without this check, Hermansson et al., (2020) obtained negative climate change impacts for Kraft lignin. Since practices are not harmonized and to avoid the abovementioned problem, Montazeri et al. (2016) suggested that "to avoid such pitfalls, it is recommended that LCA practitioners, sustainability scientists, and the chemicals industry collaborate to form a consensus on a standardized LCA approach to account for coproduct flows for bio-based chemicals".



Figure 4. Summary of the adopted multifunctionality practices in the selected 42 LCAs.

2.3.2.4 Impact categories assessed

The selection of the impact categories is part of the scope of the LCA. The impact categories considered are important, especially to understand the claims in the interpretation of the results such as "product A is more sustainable than product B". What does more sustainable mean? As it is possible to observe from Figure 5, climate change impact was investigated in all the LCAs. The main reason is that

climate change is the main driver for the development of bio-based products given the short time carbon cycle of the biomass used. Fossil depletion was also often investigated (55% of studies). This type of impact category has lower uncertainty than others and is linked with the results of climate change impacts. Hence, once data are collected to assess climate change, all data needed for assessing fossil depletion are available. Eutrophication and acidification were also assessed in more than 50% of the studies. These two impact categories are important for biomass products since agricultural production (and the emissions resulting from the application of fertilizers) can accelerate the decrease of the pH of the soil over time. Among the least assessed impact categories are land use and water depletion, which were assessed only in 13% and 10% of the LCAs respectively. These figures resemble the numbers presented by [158] in their review of 222 LCAs of solid waste management systems. In their study, they showed that land use and water depletion were assessed in less than 15% of the LCAs. As Laurent et al. (2014) observed, the reason behind the lack of consideration of these two impacts can be found in the absence of consensus in their impact assessment methods. Doubtless. these two impacts are important for biomass systems and should be assessed. New methods are emerging for their assessment as for example the LANCA method for land use [159] and AWARE method for water depletion [160]. Despite this lack of consensus, at least an estimation of the hectares of land needed per functional unit and a water balance should be performed in an LCA of products derived from biomass.

2.3.2.5 Type of data used

Concerning inventory data used, in 55% of the studies, primary data were partially available. In most of these LCAs, these data were generated at the laboratory scale and then system modeling was conducted for their approximation on a large scale. In 28% of the studies, all data were generated through specific modeling software (without validation with lab experiments). In 18% of studies, all data were retrieved from the literature or LCA databases. Among the main literature sources for data of Kraft lignin, Culbertson et al. (2016), Benali et al. (2016) and Bernier et al. (2013) were the main sources used. For example, Culbertson et al. (2016) was used as a data source for Kraft lignin by [55] and [136].



Figure 5. Impact categories covered in the evaluated sample of LCAs.

Since lignin-based products are recently emerging, there is a problem with data availability and data quality. In particular, some lignin-based products have a technological readiness level below 5. Early-stage LCAs (e.g. Koch et al. (2020) for lignin nanoparticles) were conducted to support the development of the technology identifying environmental hotspots and possible modifications for environmental improvements. Early-stage assessments are characterized by problems related to lack of high-quality data and results are more affected by uncertainties than LCAs based on data collected from actual operating plants [162,163].

2.3.2.6 Biogenic carbon accounting procedure

The selected system boundaries and the timeframe influence how the biogenic carbon of the lignin is considered in the LCA.

In cradle-to-grave studies, one option is to include the biogenic carbon of lignin deriving from biomass as stored in lignin (with credit) and in the future product (e.g. plastic application) derived from lignin. If during the lifetime of the product, the biogenic carbon from lignin embedded in the product does not degrade, the biogenic carbon is entirely sequestered in the product. However, if the product is, for example, incinerated within 100 years (global warming is often assessed over 100 years) after the production phase, a cradle to grave LCA would have to account for the CO_2 emissions from lignin. The way that biogenic carbon intake is accounted for in lignin studies can highly affect the results in climate change. Bernier et al. (2013) estimated in 0.6 kg of CO_2 eq the cradle-to-gate impact of 1 kg of Kraft lignin. This value already includes a credit based on the biogenic carbon content of lignin (2.3 kg of CO_2 eq per kg of Kraft lignin) [103]. The subtraction of such a credit in a cradle-to-gate study implies that the biogenic carbon remains stored for more than 100 years.

If the carbon content of the lignin-based product is released in less than 100 years, another option is to assign a characterization factor equal to zero for biogenic emissions over the entire life cycle. This is also an option in cradle-to-gate studies and is often referred to as the "carbon neutrality" assumption. This assumption was for example made by Shuai et al. (2016) and Hermansson et al. (2020). In most of LCA guidelines and policy recommendations, a zero discount rate is applied to biogenic emissions. This means that the time difference between the moment when the biomass absorbed the carbon and the moment when the carbon dioxide is released is not accounted for; it is as if they both happen at the same time. This approach is followed by European commission guidelines [85,145,165], European directives for renewable energies and alternative fuels [9,37] and the US Environmental protection agency [166]. Alternatively, the release of biogenic emissions and carbon storage can be discounted in time as proposed by UK PAS 2050 [167].

As an alternative to the carbon neutrality assumption, Culbertson et al. (2016) accounted for the biogenic intake with a characterization factor of -1. This elementary flow representing the biogenic intake was accounted for in the inventory of biomass (softwood). As a result, this flow was then allocated to all products with the allocation method applied to apportion the impact (and the biogenic credit) to the co-products. This method is consistent with the EU PEF guide and PEFCR guidance which recommends that the "allocation rules used for all other elementary flows shall also apply to model the biogenic carbon flows" [85,165]. However, this can lead to carbon accounting inconsistencies when the allocation rules applied do not reflect the actual biogenic content of the product. In such cases, should the biogenic carbon content (as it would happen in reality), or to the allocated biogenic carbon content (as accounted for in the model)?. Using the method applied by Culbertson, a good practice is to separate the inventory and

characterization results for climate change into two categories: fossil and biogenic (as done by [131] and suggested by recent EU LCA guidelines [165]). Only in this way, it would be possible to use cradle-to-gate results (and inventory) as input to other cradle-to-grave studies. For this reason, the EU PEF guide and PEFCR guidelines recommend that "the biogenic carbon content at factory gate (physical content and allocated content) shall always be reported as additional technical information" [85,165].

Moreover, since most LCAs were conducted from the cradle to the gate, the possible biodegradation of the carbon embedded in the lignin during the use phase and end-of-life of the products was not modeled in these studies.

2.3.3 ENVIRONMENTAL IMPACT

2.3.3.1 Climate change impacts of lignin

As mentioned in section 2.2.2, this review pays increased attention to results for climate change than to other impact categories, since climate change was assessed in all the studies and offers higher comparability among studies. In particular, Figure 6 shows the climate change impact of Kraft lignin as reported in the LCA studies assessing lignin from Kraft mills [55,100,103] and two LCAs on lignin-based products for which it was possible to retrieve/back-calculate the values obtained from their inventory data.

From Figure 6, it is possible to notice that the cradle-to-gate impact of 1 kg of dry Kraft lignin varies between 0.1 and 2.7 kg CO₂eq. But Figure 6 is not self-explanatory and needs to be handled carefully. From Figure 6, it appears as if Arias et al. (2020) estimated a much higher impact than the other 4 LCAs and that the impact calculated by Bernier et al. (2013) is perfectly in line with the upper values from Hermansson et al.(2020) and Culbertson et al. (2016) while the result of Tokede et al. (2020) is just a bit less than their lower values. On the other hand, these studies should be compared with consistent modelings for biogenic emissions. However, it is unclear how the biogenic carbon was accounted for in Tokede et al. and Arias et al. The other three studies used unharmonized accountings. Hermansson et al. used the carbon neutrality assumption, Bernier et al. subtracted the biogenic carbon content of lignin as a carbon dioxide credit and Culbertson et al. accounted for the biogenic intake from biomass with a characterization factor of -1, which was afterward allocated. Although different, the methods applied by Hermansson et al.

and Culbertson et al. provide consistent cradle-to-gate results (as shown by Figure 6 and considering that Culbertson et al. was also the main data source used by Hermansson et al.). Conversely, although the value reported by Bernier et al. looks numerically aligned with these two, the biogenic accounting is not accounted in a similar way. The value reported by Bernier et al. becomes consistent once the biogenic carbon intake (2.3 kg CO₂eq) is added, becoming about 2.9 kg CO₂eq and therefore much closer (and higher) to the value reported by Arias et al. (2020). This means that the kraft mill modelled by Bernier et al. is much more impacting on climate change than the one modelled by Culbertson et al. (2016). The key reason is the (allocated) consumption of natural gas per kg of lignin which is one order of magnitude higher in Bernier et al.



Cradle-to-gate climate change impact (kg CO2eq /kg of dry lignin)

Figure 6. Cradle-to-gate climate change impact of Kraft lignin as reported in the reviewed LCAs. Ranges represent mainly the testing of different allocation methods in the baseline calculations conducted in the LCA. With respect to the allocation methods tested by Hermansson et al. (2020), once the assumptions regarding the main product and current steadiness of lignin price were revised, as mentioned in section 2.3.2.3, the impact of 1 kg of Kraft lignin was in the range between 0.2 and 0.6 kg CO₃eq per kg dry lignin.

Regarding other types of lignin, in the LCA conducted by [115], the climate change impact of organosolv lignin from softwood was estimated in 1.85 kg CO_2 eq (17% lower than Kraft lignin). This value falls also in the interval of values estimated by [129], which was 1.4-2.1 kg CO_2 eq per kg dry organosolv lignin from bark. In particular,

the type of solvent used in the organosolv process affects the results significantly. For example, the use of either fossil-based ethanol/methanol or bio-based ethanol/ methanol can lead to a completely different environmental footprint and insights [135]. For instance, one of the insights of Koch et al. (2020) was to recover the fossilbased ethanol used as a solvent as much as possible. However, if bio-ethanol is used, the direction was "to not recover ethanol at all" [135].

Unfortunately, climate change values for lignin obtained from biorefineries are scarce, since in most of the LCAs of biorefineries lignin was neither used internally, nor the product in focus, and the functional unit was not defined in terms of lignin. The only LCA that reported the impact of lignin from the biorefinery was [155], who applied multiple functional units. Modahl et al. estimated an impact of 1.12 kg CO_2 eq per kg of lignin from a mix of timber and wood chips. However, it can be expected that for other biorefineries, the impact is very variable depending on the production process, allocation applied and feedstock used. The other two studies mentioned in section 2.3.2.2 that used multiple functional units did not have lignin as a sold co-product.

2.3.3.2 Climate change performance of lignin-based products

Concerning lignin-based products and the potential reductions of climate change impact that they can allow in the replacement of fossil-based applications, two (conceptually) slightly different approaches are possible i.e. comparing final applications (e.g. asphalt with lignin versus conventional asphalt) or comparing ingredients (e.g. lignin for asphalts and bitumen).

The first approach assesses the two alternative products (with lignin and without lignin) considering the entire life-cycle. In each application, the percentage of lignin used compared to other input materials can be small or large. Based on how much percentage of materials input can be replaced with lignin, the importance of lignin on the final LCA outcome could be low or high. For example, 5% of the weight of asphalts is made of bitumen, which is one of the most environmentally impacting ingredients of asphalts' recipes, and lignin can replace reasonably up to 25% of this bitumen [134].

In the second approach, one of the main aspects that is important to consider is the fact that, for most applications, lignin does not replace other ingredients with a 1:1 mass ratio. For example, 2 kg of lignin can replace 1 kg of carbon fibers or 3 kg of lignin can replace 1 kg of fossil raw materials for the production of tert-butyl catechols [55]. A second aspect is that the use of lignin instead of fossil ingredients often leads to changes in the composition or manufacturing of materials, e.g. using lignin instead of bitumen changes the composition of the asphalt and the energy consumption of the production phase [168]. Landa and Gosselink (2019) published the application of lignin in bio-asphalt showing a lower production temperature of 130°C for this novel asphalt compared to conventional asphalt. If both asphalt composition and processing change significantly due to the use of lignin instead of bitumen, then it will not be possible to directly compare 1 kg of lignin with 1 kg of bitumen (or with a different mass ratio). For example, Arias et al. (2020) assessed bio-based adhesives made from Kraft lignin and organosolv lignin. An interesting finding of the study is that despite organosolv lignin can be used in higher percentage in the adhesive mix than Kraft lignin and its climate change impact was lower than for Kraft lignin, the climate change impact of organosolv lignin adhesives was higher than for Kraft lignin adhesives (15.5 kg CO₂eg versus 8.3 kg CO₂eg per kg of adhesive). The main reason was that the lignin glyoxylation process (required for the functionalization of lignin for this application) requires much (about 2.4 times) higher electricity consumption to process organosolv lignin than to process Kraft lignin. The study of Yuan and Guo (2017) calculated the impact of adhesives from lignosulfonates (hybrid ammonium lignosulfonates). They estimated that 1 kg of adhesives from lignosulfonates lignin generate 0.13 kg of CO₂eq², which is much lower than the impact of the adhesive from Kraft and organosolv lignins calculated by [115].

Given the issues mentioned above about the second approach, most of the LCAs on lignin applications applied the first approach which is more reliable. Figure 7 shows the savings of GHG emissions that are achievable using lignin-based applications to replace conventional petrochemical products estimated by the reviewed LCAs.

As can be observed from Figure 7, there are many applications where lignin can be used which are promising from a GHG emissions perspective. In particular, Obydenkova et al. (2017) reported that deriving a transportation fuel from lignin by pyrolysis that can replace diesel on the market could generate up to 90% of GHG emissions savings. However, the emissions savings vary in the range between 10%

² This value was calculated from 20 kgCO2eq per m3 of finished fiberboard reported in figure 4 of [104] and 154.2 kg/m3 of ammonium lignosulfonate needed for the production of 1 m3 of finished fiberboard [104]

and 90% [106] depending on two critical factors: 1) what source of energy is used in the biorefinery to replace the diverted lignin and 2) what type of allocation method is applied. For example, lignin could be replaced by either natural gas or biomass (e.g. corn stover) as energy sources, and biomass would be preferable from a GHG emissions perspective. However, [106] estimated that the use of corn stover as fuel instead of natural gas would increase the cost by about 30%. Concerning the allocation method, the use of either energy allocation or cut-off allocation (all impact to ethanol) affected significantly the results of Obydenkova et al. (2017). However, a cut-off allocation does not seem fair since lignin cannot be considered a waste in LCA terms [38] according to the waste management framework [170] adopted by the European Union in the Renewable Energy Directive [9].



Figure 7. Savings of climate change impact compared to fossil reference reported in the selected LCAs. PU=Polyurethanes, CF= carbon fibers, WL=wood laminate, PR=phenolic resin. The range of values from Manzardo et al. and Hildebrandt et al. refers to multiple formulations (e.g. varying shares of lignin content within the resin matrix). The range of values from Corona et al. represents the variation of the country where adipic acid is produced along with respective fossil reference and multiple feedstock scenarios. The wide and alternative energy carriers.

Adipic acid also seems a promising application from a GHG emissions perspective, allowing savings between 62% and 78% compared to petrochemical adipic acid [138]. This range represents two different scenarios representing two different possible locations for the adipic acid plant. In particular, the study of Corona et al. was the only self-declared consequential LCA. Accordingly, inside the system boundaries, the unit processes representing the avoided production of heat and electricity internally to the biorefinery were included.

While most of the studies did not analyze the end of life of the products, Manzardo et al. (2019) conducted a full cradle-to-grave LCA of bio-based rigid polyurethane foams and compared their impact with the fossil counterpart. In particular, Manzardo et al. considered three different foams produced from bio-based polyols obtained from lignin. Bio-based polyurethane with lignin showed 6-32% savings of GHG emissions compared to the petrochemical polyurethane foam used as reference [136].

Among the applications that look less promising from a GHG perspective, ligninbased catechol, which is a chemical mainly used for fertilizer but also fine chemicals such as perfumes, shows savings of 2% [140], which is very minor compared to the uncertainty involved. Bio-based asphalts [134] also showed low GHG emissions savings (about 5%) compared to conventional asphalts and this depends also on the percentage of lignin replacing bitumen assumed. On the other hand, the climate change impact per kg of kraft lignin assumed by Tokede et al. was also the lowest shown in Figure 6. Changing methodological assumptions or Kraft mill might lead from a low GHG saving of emissions to higher impact than conventional asphalts.

2.3.3.3 Environmental performance of lignin-based products

In this section, the performance of lignin-based products is discussed considering other environmental impacts in addition to climate change with examples from studies.

Concerning bio-based adhesives derived from lignin, three LCAs were conducted and divergence was found in the insights on the overall performance in comparison with the petrochemical counterparts depending on the type of lignin considered and assumptions made. In particular, Arias et al. (2020) assessed two bio-adhesives used for manufacturing wood panels derived from two different lignins (from Kraft and organosolv) [115]. These adhesives were compared with two alternative bio-based adhesives (from soy and tannin) and three conventional fossil resins (urea-formaldehyde, phenol-formaldehyde and melamine-urea formaldehyde). Nine impact categories were considered and the impacts were compared based on end-point results. On end-point bases, the comparison highlighted that ligninbased adhesives were performing much worse than other bio-based adhesives and conventional adhesives (between 2.5 and 4.5 times higher impact). On the other hand, the preliminary LCA conducted by [104], based on endpoint results, concluded that wood panels made using lignosulfonates-based adhesives are environmentally better than wood panels using urea-formaldehyde. Similarly, also McDevitt and Grigsby concluded that Kraft lignin-based adhesives are environmentally better than urea-formaldehyde adhesives (about 22% lower impact on weighted bases [132]).

With respect to lignin-based polyurethane foams, Manzardo et al. (2019) found better performances compared to the petrochemical foam taken as reference in five out of the eight impact categories considered. In particular, they offer 9-33% savings in photochemical ozone formation, up to 29% in terrestrial eutrophication, 6-43% in freshwater eutrophication, and 14-36% in depletion of abiotic resources (elements).

Concerning lignin-based phenolic resins, the LCA of Hildebrandt at al. (2019) showed that wood-based fiber laminates using lignin-based phenolic resins perform better in nine out of eleven categories (with achievable reduction potentials up to 39% depending on the impact category considered) [131].

Concerning lignin-derived fertilizers, Montazeri and Eckelman assessed ligninbased catechols which are chemicals mainly used for the production of fertilizers. Their assessment showed that lignin-derived catechol, beyond negligible climate change benefits (see Figure 7), potentially offer 7% and 59% environmental impact reductions respectively for ecotoxic effects and depletion of fossil fuels [140]. However, in the other seven environmental impact categories, the fossil route was preferable [140]. In particular, the solvent (Dichloromethane) used in the lignin purification process and electricity for lignin depolymerization were found as the dominant contributors to the environmental impacts of the bio-based route [140]. Krzyżaniak et al. (2019) assessed the final application (cultivation using different fertilizers) and assessed the same impact categories of [140] concluding as well that lignin-based fertilizers are slightly better than mineral fertilizers. Specifically, Krzyżaniak et al. (2019) found that lignin was better than mineral fertilizers in four impact categories (climate change, particulate matter, terrestrial acidification 2

and freshwater eutrophication) while worse in freshwater ecotoxicity, terrestrial ecotoxicity and human toxicity. Concerning the use phase, compared to mineral fertilizers, lignin-based fertilizers showed higher sequestration of organic carbon and lower field emissions in terms of particulate matter and acidification/ eutrophication. For fossil depletion, the impact of lignin used as fertilizers was slightly worse. What appears interesting is that the categories where lignin used as fertilizer and lignin-based catechols (an ingredient for fertilizers) perform better or worse were opposite in the two LCAs (except for climate change) [8,140]. On the other hand, the two products assessed were not directly comparable except for the final use.

These examples show that, while lignin-based products are often preferable for climate change than their fossil counterparts, conversely, trade-offs occur in the other impact categories assessed. It is also not straightforward to summarize for which categories lignin-based products are generally better since it is very case dependent.

2.3.3.4 What fuel to use to replace lignin as an internal energy source?

One of the findings of the review is that the impact of lignin and lignin-based products depends significantly on the type of energy source that is used to replace the burning of lignin in biorefineries and paper mills.

Concerning Kraft lignin, most of the studies found that natural gas used to replace black liquor is the main environmental hotspot for most impact categories. However, Bernier et al. (2013) argue that using natural gas is one of the main drivers to equip old mills with lignin extraction since it is a cheap fuel whose combustion causes much lower local atmospheric emissions than black liquor. Alternatively, the additional steam required caused by lignin extraction can be provided by burning excess hog fuel (if available along with spare boiler capacity) [103]. In existing pulp mills, there is also a fraction of lignin that can be extracted without requiring an increase of natural gas consumption for energy use in the pulp mill (only a minor increase for the lignin extraction process) [100]. This fraction of lignin has a lower impact than the part that requires additional energy for the pulp mill. Secchi et al. (2019) performed an LCA on the effect of lignin extraction on the environmental impact of ethanol produced by a biorefinery and pulp produced by a Kraft mill. In particular, for the biorefinery, 40% of the lignin cake was assumed to be diverted from the internal energy use while for the Kraft mill, 50% of the black liquor was assumed to be removed [119]. Various fossil and biomass sources for energy production were considered to replace the fraction of lignin originally used as fuel and multiple allocation methods were applied (mass, energy and economic values). The results and conclusions were based on single score impacts calculated with ILCD normalization factors [171] combined with equal weighting. The two main outcomes of the study were that 1) the impact of ethanol and pulp does not increase if lignin is extracted and 2) using natural gas to replace lignin as an internal energy source is recommended in biorefineries while cogeneration using biomass is recommended in pulp mills [119].

On the other hand, if the main goal of the biorefinery is the minimization of the CHG emissions and not of the total impact (climate change plus other environmental impacts more than climate change), the use of additional biomass instead of natural gas to compensate the diverted lignin might be preferable [106]. For example, in the case of lignin-derived transport fuels, the use of natural gas does not allow to fulfill the EU 60% GHG savings threshold of policy targets [106] set by the EU renewable energy directive [9,39] and U.S. renewable fuel standard. To fulfill this target, in the example of the biorefinery modeled by [106], the use of corn stover also for internal energy purposes (and not only as feedstock for fuel production) was proposed.

2.3.3.5 Effects of lignin allocation on the LCAs of biorefinery products

In most of the LCAs of biorefinery, the focus was on the main products produced by the biorefinery and not on lignin, which was sometimes used for internal energy needs and some other times marketed for other purposes. This section report on how these LCAs dealt with lignin.

Turk et al. (2020) performed an LCA of nanofibrillated cellulose [172]. In their study, the lignin produced from the biorefinery was considered a waste. Therefore, no impact was apportioned to lignin and the impact of one kg of nanofibrillated cellulose was as high as 800 kg CO_2 eq [172]. In the sensitivity analysis, mass allocation was applied to account for lignin as a by-product instead of waste.

Since Soxhlet extraction and delignification represented a considerable part of the environmental burdens and were also allocated to lignin, the impact of one kg of nanofibrillated cellulose became about 400 kg CO_2 eq [172]. Hence, how the practitioners deal with lignin in assessing such a product has a major effect on the results.

Soam et al. (2016) assessed a second-generation biorefinery producing ethanol from rice straw in India. The study concluded that the ethanol produced offered major GHG emissions savings (77-89%) compared to gasoline. In particular, two assumptions were made: 1) the displaced electricity was coal-based electricity and 2) the carbon emissions from lignin combustion were carbon-neutral. Based on these assumptions the surplus of electricity generated combusting lignin led to major benefits (a credit of 40-45 g CO₂eq per MJ of ethanol over a total impact of about 55) [113]. These same two assumptions were made also by [164] in their assessment of ethanol from common reed produced in China. However, in their study, the credit generated by the replacement of the surplus of electricity was less important (2.5 g CO₂eg per MJ of ethanol over a total impact of 17.5 g CO₂eg). Hence, based on the amount of the surplus of electricity, the surplus of electricity might lead to a major credit or a small credit (this does not only depend on the quantity but also on the electricity mix displaced). One should wonder if, in the cases where a major credit was given, the surplus of electricity was a by-product of the system or the main product of the system. In the second case, the use of substitution would not be appropriate since the principle of physical/economic significance would not be respected.

Nascimento et al. (2016) assessed cellulose nanocrystal from coconut fiber. In the production process, lignin was produced as a by-product and was marketed. The two main environmental hotspots of the process were identified as the production of acetic acid and the electricity required. As an alternative, lignin could be burned for the internal electricity needs of the biorefinery. However, the results of the LCA conducted by Nascimento et al. (2016) showed that the use of lignin as an internal power source led to environmental impact increases in four (climate change, terrestrial acidification, water body eutrophication and marine eutrophication) out of six impact categories assessed. The main reason was that, if lignin were no more a by-product but were internally consumed, the impacts from milling and pulping processes would be attributed to cellulose nanocrystals only and would not be allocated anymore also to lignin. Thus, the benefits from the power generated from

burning lignin were lower than the impact originally allocated to lignin in these four impact categories. However, looking only at the functional unit expressed in terms of nanocrystals and not to the overall system, this conclusion might have been affected by the mass allocation applied. In fact, if economic allocation were applied, a lower impact would have been allocated to lignin since the price of cellulose nanocrystals is higher than lignin [173].

Budsberg et al. (2016) noticed that the production of hydrogen is often the main environmental hotspots of the production of bio-jet fuels. In the biorefineries producing bio-jet fuels, often hydrogen is produced from natural gas and lignin is used as fuel for the internal demand for heat and electricity [148]. Budsberg et al. wondered if, environmentally, this is the best solution or is better to gasify lignin to produce green hydrogen for internal needs. From a climate change perspective, their LCA showed that the current solution is better than using lignin to produce hydrogen (the impact would increase by 10%) due to the GHG emissions caused by the replacement of lignin with natural gas for the production of internal energy needs. However, their LCA showed that if hog fuel could be used instead of natural gas, then using lignin for hydrogen production could lead to important savings of GHG emissions (order of 50%) [148].

2.4 CONCLUSIONS

Lignin, which is a by-product of biorefineries and pulp mills, is currently (mainly) used for bioenergy but can be utilized to produce lignin-based products replacing fossil counterparts in various sectors. In the near future, the electricity mix is expected to be rapidly decarbonized. On the contrary, transport, heat and materials are much harder to decarbonize. Hence, we can expect that the use of lignin for producing bio-based products will start to play a more important role in the next decade. In parallel, the sustainability performance of such products should be monitored using accredited tools. Among them, LCA is the best candidate for sustainability assessment in bioeconomy sectors. Despite LCA is a standardized method, various methodological choices have to be taken by the practitioners leaving room for possible inconsistencies between the results of different studies. Forty-two studies concerning LCAs of lignin and lignin-based products were reviewed to detect the differences (and possible inconsistencies) in the application of the methodology and their influence on the life-cycle environmental impacts. Moreover, the climate change impact reported in LCAs of lignin and the GHG savings allowed by ligninbased products were quantitatively compared. The importance of other impacts in the comparison between lignin-based products and counterparts was also discussed with examples from the studies. The lesson learned from this exercise and possible recommendations are provided in Table 3.

Tahla 3 Laccon	loarnod and	recommendations	Continuacin	the following	two nades
TUDIC J. LC33011	icunicu unu	recommendations.	Continues in	the following	two pages.

Lesson learned	Recommendations
Only a few studies considered the use phase and end of life of the product (see section 2.3.2.1).	The end of life should be considered especially for the comparison between lignin-based products and their fossil counterparts. Realistic, average waste management should be investigated as well as the carbon degradation of lignin during the use phase and waste management.
78% of the LCAs did not specify the type of modeling approach followed i.e. attributional or consequential (see section 2.3.2.1).	The approach followed should be specified since it helps to select properly the unit processes to be included in the system boundaries, the type of data to be used and what type of system expansion method is possible.
Most of the studies adopted a simple functional unit e.g. based on a mass basis (see section 2.3.2.2). This type of functional unit does not state how well each product fulfills the function of the system.	In the definition of the functional unit, how well the function of the product system is fulfilled should be accounted for. Only if the function is fulfilled similarly by the investigated options, a simple functional unit could be used.
While climate change was investigated in all the selected LCAs, other impact categories were often neglected (see section 2.3.2.4). Especially for land use and water use/ depletion, one of the main reasons was probably the absence of consensus on the impact assessment method.	All relevant impact categories should be included. In particular, land use and water use are important for bio-based systems. The assessment method should be selected based on the recommendations from trusted sources (e.g. EU LCA guidelines). If an impact assessment method were not used, at least an estimation of the amounts of land and water needed should be provided.
In almost all LCAs (especially of biorefineries), data were mainly obtained from laboratory and process modeling (see section 2.3.2.5). Few studies used primary (actual) data for kraft lignin production. These studies were also the main sources used in the LCAs that relied on secondary data.	It is important to collect new transparent primary data for lignin production from real operation at a large scale which are currently missing in the public domain.

Lesson learned	Recommendations
Dealing with multifunctionality was identified as the major methodological problem in the assessment of lignin and lignin-based products since lignin is always the result of a multi-output process. Therefore, LCAs of lignin products are affected by higher uncertainties compared to other bio-based products. A standardized method for the selection of the allocation method exists and is provided by ISO 14044:2006. However, there is no shared interpretation in the LCA community and in LCA guidelines. As a result, multifunctionality practices in LCAs of lignin-based products are not harmonised (see section 2.3.2.3).	A consensus on the interpretation of ISO 14044 hierarchy to deal with multifunctionality is urgently needed to have a standardized LCA approach to account for co-products. This is a problem not only of lignin production systems but of all bioeconomy. A ISO-compliant framework that keeps into account the major critical aspects identified during the review (e.g. the application of substitution without a check on physical/economic significance) is needed.
Biogenic carbon dioxide is treated differently in the studies (see section 2.3.2.6). Often, it was treated as carbon-neutral while in other cases a carbon intake was accounted for based on the carbon content of lignin or based on the carbon intake during biomass growth. Moreover, often the carbon credit was integrated into the cradle to gate results for climate change and the accounting of the biogenic carbon intake was a key element for the better performance of lignin- based materials compared to their fossil counterparts. However, recent guidelines e.g. EU PEFCR recommends reporting the biogenic carbon separately in LCAs ending at the gate.	The first recommendation is that the choice regarding biogenic carbon accounting should be stated clearly in the LCA and also next to where the climate change results are shown. This would allow the user of the LCA to have a clear picture and, in case the LCA results were used for other studies, a double counting (or omission) in the assessment of the end-of-life phase would be avoided.
Comparing single lignin-based ingredients (e.g. lignin binder) with fossil-based ingredients (e.g. bitumen) can provide	To have a full picture and a correct estimation of the potential savings that can derive from using lignin-based products to

(e.g. lignin binder) with fossil-based ingredients (e.g. bitumen) can provide an erroneous picture. In fact, the utilities required during the production of the final application might change if lignin is used in the product. Moreover, sometimes, in order to have the same performances, also the other ingredients in the mixture have to be changed (e.g. proportions).

To have a full picture and a correct estimation of the potential savings that can derive from using lignin-based products to replace their petrochemical counterparts, the LCA should compare the final application (e.g. asphalts) rather than the chemical ingredient with its petrochemical counterpart.

Lesson learned	Recommendations
The impact of biorefinery products (e.g. ethanol) is largely affected by how lignin is used in the system and how the practitioners deal with lignin in the LCA.	If lignin is exported as a product from the system, multiple functional units should be used. If the goal of the LCA requires the determination of the impact of a single function, a functional unit should be assigned to lignin and another one to the main product which is the focus of the investigation. Only in this way, the user of the LCA can (easily) understand how lignin was considered in the LCA and the effects of the allocation procedures applied.
Often, there is a trade-off between GHG emissions and economics in the selection of the best fuel to replace lignin in internal uses. Moreover, extracting lignin instead of using it for internal energy needs might affect importantly the environmental performance of biorefinery products.	The LCA should be conducted based on the most probable fuel and possible alternatives should be investigated by sensitivity analysis

Our list of recommendations could promote good practices and increase methodological harmonization in assessing the environmental sustainability of lignin and lignin-based products using LCA. On the other hand, even following these recommendations, conducting an LCA of lignin remains challenging from a methodological perspective. For this reason, the user of the LCA results needs to be very careful in checking the assumptions made by the practitioners. Moreover, using the results from different LCAs that compare lignin-based products and fossilbased products and concluding what option is the best is not straightforward. The reasons are both technical (e.g. using lignin as an ingredient changes also other parameters and lignin can substitute other ingredients with different shares) and methodological (allocation plays a major role, as does the way biogenic carbon storage is accounted). Beyond this, the reviewed LCAs showed that often ligninbased products offer better environmental performances than fossil-based products (especially for climate change), but if lignin is diverted from an energy application, the most probably alternative can have a substantial influence on the overall climate impact.



3

CHAPTER

Reviewing ISO compliant multifunctionality practices in environmental Life Cycle modeling

Published as:

C. Moretti, B. Corona, R. Edwards, M. Junginger, A. Moro, M. Rocco, L. Shen. Reviewing ISO compliant multifunctionality practices in environmental life cycle modeling.

Energies, 13 (2020), p. 3579, 10.3390/en13143579

ABSTRACT

The standard ISO 14044:2006 defines the hierarchical steps to follow when solving multifunctionality issues in Life Cycle Assessment (LCA). However, the practical implementation of such a hierarchy has been debated for twenty-five years, leading to different implementation practices from LCA practitioners.

The first part of this study discusses the main steps where the ISO hierarchy has been implemented differently and explores current multifunctionality practices in peer-reviewed studies. A text-mining process was applied to quantitatively assess such practices in the 532 multifunctional case-studies found in the literature. In the second part of the study, citation network analysis (CNA) was used to identify the major publications that influenced the development of the multifunctionalitydebate in LCA, i.e. the key-route main path. The identified publications were then reviewed to detect the origins of the different practices and their underlying theories.

Based on these insights, this study provides some "food for thought" on current practices to move towards consistent methodology. We believe that such an advancement is urgently needed for better positioning LCA as a tool for sustainability decision making. In particular, consistent allocation practices could be especially beneficial in bioeconomy sectors, where production processes are usually multifunctional, and where current allocation practices are not harmonized yet.

3.1 INTRODUCTION

Life Cycle Assessment (LCA) is supposed to be a standardized methodology to measure the life cycle impacts of products or services. LCA is currently ruled by ISO 14040:2006 and ISO 14044:2006 [38,127]; these standards have been the basis of the LCA methodology for the last two decades. Nevertheless, in the scientific community, some experts wonder if the detail presented in these standards is enough to guide LCA practitioners in practice [57,71,174].

One of the most debated problems in LCA is the so-called "multifunctionality" issue (or commonly, "allocation") [50,83,175]. Multifunctionality issues need to be dealt with when different product systems share a process, e.g. manufacturing processes delivering more than the studied product, or end-of-life activities providing both waste management service and a recovered or recycled product. In these cases, apportioning environmental burdens among the co-products, or rather cofunctions, becomes necessary. According to ISO 14044:2006, multifunctionality should be solved by using the following three-levels hierarchy [38]:

- Avoiding allocation, by subdivision (dividing the unit process into two or more sub-processes) or system expansion ("expanding the product system to include the additional functions related to the co-products");
- 2. Allocation following underlying physical relationships (i.e. an allocation that quantitatively reflects how the inputs and outputs are changed by changes in the amount of each product of the system);
- 3. Allocation (partitioning) based on other relationships (e.g. economic value);

The same hierarchy applies also to "open-loop" recycling i.e. when a material is recycled as a different product because it is no longer suitable to replace the original product directly. Only in such open-loop recycling, ISO 14044:2006 provides further guidance on the third level of the hierarchy, where physical properties (e.g. mass) are preferred to economic value, which in turn is preferred to the number of subsequent uses of the recycled material [38].

The existence of the ISO's multifunctionality hierarchy should avoid the use of inadequate approaches, e.g. determined by the interests of the stakeholders or the ones of the study's commissioner [57]. Nevertheless, the apparent lack of sufficient guidance has fed different implementation practices [68]. Consequently, although most LCAs claim compliance with the two ISO standards, practitioners have applied

different allocation procedures in LCAs assessing the same or similar products [94]. Since the choice of the allocation method typically affects the outcome of the LCA significantly [50,94,96,128,176], this problem has led to different conclusions and therefore low reliability and robustness of the LCA results [177]. Moreover, due to the lack of a shared view in the LCA community, some authors decide not to follow the ISO hierarchy (see [178]), while other authors select the allocation method based on their subjective decision (see e.g. [179] and [180]). Other researchers choose allocation methods that are "commonly" applied in similar case studies in the literature (see e.g. [181]), others calculate also an average allocation parameter considering common parameters (e.g. [182]) or others use "conservative" allocation methods that provide the highest impacts (e.g. see [183]).

This article presents a literature review on the main practices and debates on using ISO 14044:2006 recommendations to solve multifunctionality problems. A critical literature review on multifunctionality methodology development was combined with quantitative analysis of current multifunctionality practices, and a bibliometric review based on citation network analysis (CNA). The quantitative analysis was performed by a text-mining process in 532 multifunctional case-studies found in the literature.

The CNA was used to identify the main knowledge flow on multifunctionality in LCA, also known as "the main path". Tools and software based on the "main path" method are used for many applications: tracking the evolutionary trajectory of a science field or the development of a specific technology, or the evolving changes of legal opinions of courts [184,185]. The "main path" was investigated to detect the historical origins of the different practices currently present in the literature and their underlying theories. The use of such a tool overcomes some limitations of the traditional systematic reviews conducted so far on this topic, which were based on "human" selection of the articles (e.g. through criteria such as the number of citations).

In the literature, the definitions used to characterize the multifunctionality issue are not harmonized. For this reason, we provide an appendix reporting the definitions used in this review to distinguish the different types of products, multifunctional processes, modeling approaches and system expansion approaches.

3.2 METHODOLOGY

Figure 8 summarizes the three main steps followed in this literature review. First, the literature search was performed. Second, a critical review was conducted to identify the main issues and bottlenecks in the LCA literature when implementing the ISO allocation procedures. The critical review was combined with a text mining process to quantitatively assess the current practices in the LCA literature (focusing on all the LCA case studies selected by the query). Third, a bibliometric analysis was performed based on citation network analysis (CNA).



Figure 8. The three steps followed in the literature review

The literature search was based on data collected from the Scopus database in February 2019. The searched publication fields were: title, abstract, and keywords. The search string was characterized by the terms: "Life Cycle Assessment", "LCA", "multifunctionality", "allocation" and "multi-output". Since allocation approaches are also used in other fields (e.g. in business management), the query was first limited to environmental assessment or engineering-related fields. Because of this, the documents were reduced from 1310 documents to 1152. This allowed us to exclude 145 documents belonging to business management, 6 related to veterinary science and 7 others. Our analysis was further refined by considering articles only from the category of Scopus "journals". By applying this last adjustment, the articles resulting from the search became 930. Since only the research articles were analyzed, some relevant books or conference proceedings may have been excluded from the analysis. Nevertheless, often books resume the contributions previously published as articles and some excluded documents might have been considered by some of the reviews reviewed. Figure 9 shows the number of publications per year, highlighting the growing interest in the topic.



Figure 9. The time distribution of the articles on multifunctionality in LCA published in scientific journals per year retrieved from Scopus.

The corpus of documents on which the analyses were performed included the 930 articles retrieved from Scopus and the main LCA guides and standards, i.e. ISO technical reports and standards (also withdrawn ones like ISO 14041:1998) [186–189], the International reference Life Cycle Data system (ILCD) handbook [145], the Product Environmental Footprint (PEF) guide [85] and the Product Environmental Footprint Category Rules (PEFCR) guidance [165]. Out of the 930 documents, 307 studies were identified through their title and abstract as "methodological articles"

(of which 117 were review articles focusing on a specific sector where LCA is applied). These methodological articles focused either on the general methodological debate about multifunctionality procedures, or discussed a specific method, or introduced a new model to solve multifunctionality. The most relevant articles in this group were critically reviewed to understand the main issues when solving multifunctionality in LCA while claiming compliance with ISO. This critical review focused mainly on the articles cited more than 20 times ("most cited ones") and the articles published after 2015 ("recent ones").

The critical review was combined with a text mining process whose aim was to quantify the current practices when solving multifunctionality issues. The text mining process was manually performed on the remaining 532 case studies. These 532 case studies resulted from a further refinement which excluded 91 articles that either did not apply full LCA or were not environmental LCA studies. Concerning the multifunctional case studies retrieved from the literature, we observed that specific parts of the bioeconomy namely agriculture (63 case studies), bioenergy (185), bio-based materials (52) and anaerobic digestion (21), are the ones most affected by the issue of multifunctionality together with related sectors namely aquaculture (14), dairy and meat products (79), fossil counterparts (34) and waste management (50). These sectors together represent 94% of the 532 case studies identified by the query.

A text mining software can detect relevant terms or keywords in the corpus of literature with less time and cost than a person [190]. However, when the keywords represent technical concepts, dedicated software typically achieves low to medium efficiencies (e.g. 25-65%) [190]. For instance, software could not understand when the concept "system expansion" was used as an alternative expression for substitution or for system enlargement. To increase the efficiency of the text mining method, the quantitative estimation was performed directly by the analyst. When the terms representing the concepts of interest (e.g. "allocation") were encountered, the context of their use was assessed by reading the surrounding text.

In the third step, i.e. the bibliometric review, the 930 articles were investigated by CNA. The CNA was performed using Pajek software [191]. Documents are considered "nodes" and the citations are the "links" between these documents. The type of nodes is defined therefore based on the type of document. The "sources" are the documents that are cited but cite no other documents and therefore represent the

origins of the knowledge. The "sinks" are the documents that cite other documents but are not cited and therefore could represent the "current stage" of the knowledge stream. Intermediate documents cite other previous documents and are also cited by more recent documents [184]. Our CNA aimed at identifying the main path of research. This path represents the main knowledge flow in a specific topic, i.e. the major contributions that have influenced the development of the research, which does not mean directly the most cited ones overall [192,193]. The main path was obtained by using an algorithm that computes what citations between articles have been more significant. In particular, such a significance was calculated through the key-route method [194]. This method identifies the main chain of articles by considering the highest transversal count [193,194]. The transversal counts measure the significance of a citation link i.e. by counting the times a citation link is traversed [194]. The transversal count (SPC). The SPC assigns as value to each link the number of paths traversing the link among all possible paths connecting all the sources to all the sinks [184,185].

3.3 THE CRITICAL REVIEW COMBINED WITH TEXT MINING

When critically reviewing the methodological articles on multifunctionality, it emerged that these articles present two main "debates" regarding ISO-compliance practices. These two debates concern the application of system expansion (explained in section 3.3.1 and related sub-sections) and the identification of relevant partitioning criteria (see section 3.3.2). In particular, Pelletier et al. (2015) identified three "schools" distinguished by the way they interpret the ISO hierarchy with respect to these two aspects: (1) the consequential thinking school interprets system expansion as substitution, (2) the natural-science attributional school applies system expansion as enlargement and prioritizes allocation based on a physical parameter, and (3) the socio-economic attributional school applies system expansion as enlargement but prefers economic allocation. According to Pelletier et al. [68], these three schools are "internally consistent" but "mutually exclusive".

3.3.1 Debate on the interpretation of ISO's system expansion

The system expansion debate focuses on how and when should the substitution method be applied. ISO 14044:2006 recommends system expansion as a way to

avoid allocation, but no further specification is provided regarding the differences between enlargement and substitution (see appendix for detailed definitions). and about its implementation in attributional or consequential LCAs. Substitution is often used as a system expansion approach in attributional LCAs (ALCAs), which is not perceived as correct by many LCA experts [51,68,71-73,195]. According to these practitioners, ALCA modeling should not rely on perturbation logic or counterfactual notions, such as substitution or avoidance of other products/ processes (as also highlighted e.g. by Majeau-Bettez et al. (2018). It is argued that the sum of the impacts accounted by attributional LCAs should add up to the worldwide impacts, and this would not be valid anymore if substitution were applied [71,74]. For this reason, Chen et al. (2010) concluded that the "allocation methods, even if perfectible, are still preferable to the system expansion method" (used as synonymous of substitution), because "system expansion does not ensure a global coherency between various LCA studies" [196]. On this bases, the use of substitution as a system expansion method in ALCA is not supported by any of the schools of interpretation identified by Pelletier et al. [68]. Similarly, Bailis and Kavlak, after applying substitution for the by-products of a biofuel, concluded that "the large disparity between system expansion and other methods raises questions about the validity of system expansion" [197]. Concerning system expansion by enlargement, this cannot be applied when the goal of the study requires to obtain the impacts of just one of the co-products or by-products. In these cases, allocation cannot be avoided. For example, "In a milk production system that also produces beef, system expansion without substituting would lead to a system with a function of delivering both milk and beef" [198].

Other authors argue that ISO 14044:2006 does not acknowledge substitution as a system expansion approach. The reason is that ISO refers only to the addition of functions (i.e. enlargement) and not to the substitution of functions [57,71,199– 201]. On these bases, several authors argue that a distinction ALCA/CLCA should be present in future ISO 14044 [68,202] since, for them, this distinction is crucial to select the appropriate system expansion method (enlargement or substitution) (as also pointed out by [61,144]). By contrast, other authors argue the opposite i.e. that substitution is generally recognized as a valid method for avoiding allocation within attributional LCA [203,204]. For many practitioners, substitution is considered as synonymous with system expansion [205,206]. Under this argument and considering the ISO hierarchy, substitution should be preferred to any allocation method [76–
79,207]. Pelletier et al. [68] suggested that the equivalence substitution-system expansion might have been originated in a study of 1994 authored by Tillman et al. [81]. The reason was that Tillman et al. [81] is a frequent citation when justifying the equivalence of substitution with system expansion. However, their study was published previous to the publication of the ISO standards.

3.3.1.1 Current practices in specifying the modeling approach

Although the choice of modeling approach (ALCA or CLCA), which depends on the goal, clearly determines the outcome of an LCA study, our text mining process found that the keywords "attributional" and "consequential" were missing in 75% of the LCAs involving multifunctional systems (see Figure 10 for a detailed breakdown, per product sector). This percentage refers only to the portion of articles published after 2004 when the term consequential LCA was clearly established (see section 3.4.3). There are several possible reasons for this low specification rate of the modeling approach: (1) practitioners could still not be aware of the relevance to differentiate between consequential and attributional approaches (since consequential approaches were still not common), (2) practitioners may not specify the modeling approach because it is a direct consequence of the goal description, (3) they may not agree with a strict distinction between ALCA and CLCA, (4) they may be strictly following current ISO standards (that do not distinguish between the two approaches) or (5) they may have followed the recommendations of a policy directive or national/international guide that does not make such a distinction. Actually, some ALCA studies are emerging that combine attributional modeling with consequential thinking [94]. These approaches aim mainly at accounting for some specific counterfactual effects or credits, and at the same time, limit complexity and uncertainties [35].

The text mining process found that 31% of the self-declared ALCA studies (using the keyword "attributional") used substitution as a system expansion approach to avoid allocation. However, this percentage varies depending on the sector under consideration, ranging from 19% to 45% (see Figure 11). The highest rate of substitution approaches in ALCAs was found in studies related to bio-based materials (45%). On the other hand, the LCAs investigating fossil products that self-declared to be attributional studies are few since substitution is rarely an option for fossil products since they are usually the "substituted ones".



Figure 10. Percentage of articles which applied on the same case study both CLCA and ALCA approaches (Both), self-declared attributional studies (ALCA), self-declared consequential studies (CLCA) and studies which did not declare the approach followed (Not specified). Only case studies published after 2004 were considered (504).



Figure 11. Percentages of self-declared attributional studies (ALCAs) which applied substitution as a system expansion approach. Only the sectors with a significant amount (more than 10) of self-declared attributional studies are included in this graph.

3.3.1.2 The application of substitution as system expansion method

Beyond the use of substitution in ALCA, two other critical aspects of substitution have been discussed in the LCA methodological articles. First, the high uncertainties introduced by the use of the substitution approach since it can lead to different results depending on the choice of substituted and/or substituting by-product [57,208,209]. A sensitivity analysis should be therefore recommended. Second, when substitution is suitable, the substitution of co-products should be avoided by checking the physical/economical significance of the products delivered by the multifunctional process [57,145]. However, some authors (for example [57]) argue that the importance of the co-products' physical significance is not emphasized enough in ISO 14044:2006 and the ILCD handbook. When physical significance is not checked and a by-product is credited for the replacement of co-products, the practitioner could obtain significantly distorted results [57,210]. A common practice to account for physical significance is to select the primary functions based on the main source of revenues [51,211]. In cases where the primary co-function(s) cannot be directly identified, the ILCD handbook proposes that they should be assumed to be those that jointly contribute to more than 50 % of the combined market value of all co-functions of the analyzed multifunctional process [145].

Clear rules for differentiating by-products from the co-products are important, because, in substitution, all the credits from substituting by-products are attributed to the main co-product. If another LCA on the same process is made in which a by-product is considered to be the main product, the impacts of the process "get counted twice", so that the impacts for different products no longer add up to the total for the process (this would be a problem in an attributional model- for further details see the next section).

3.3.1.3 Using substitution as the allocation method

Another point of debate is the use of substitution for allocation (and not as system expansion method). This type of allocation has been mentioned in the literature with different names such as substitution-based allocation [68] and "proxy-based disaggregation" by substitution [51] and various versions of this method have been proposed (e.g. see Hermansson et al. who applied two different versions of this method to assess Kraft lignin [55]). By many practitioners, this option is perceived as the attributional way of using substitution. PEFCR guidance and PEF guide [85,165]

propose that, when a by-product of a multifunctional system directly substitutes another product, such substitution might be considered as an allocation reflecting physical relationships. When this is the case, such substitution has to be based on a direct and empirically demonstrable relationship [85,212]. Pelletier et al. [212] stated that this is different from substitution based on marginal market models applied in consequential LCAs [212]. An example of such a substitution is when "manure nitrogen is applied to agricultural land, directly substituting an equivalent amount of the specific fertilizer nitrogen that the farmer would otherwise have applied" [85,212]. Hence, it is assumed that the impact caused in the system by the production of the substituted by-product corresponds to the impact of the production of the replaced product (as shown in [213]). With substitution, the impact of a byproduct should equal that of the product it substitutes, and so is independent on the actual process that produces it. Moreover, the application of this substitutionbased allocation can lead to a negative impact that in ALCA would mean that the model has been built inconsistently [51]. As an example, this happens when the wrong product is chosen as the main product of the multifunctional system [213] or the substituted product is not a minor product even if representing less than 50% of market value [163]. Even if the substituted co-products are chosen carefully (i.e. they represent small percentages in physical and economical terms), this method sometimes fails in ALCAs assessing multiple impact categories resulting in negative impacts for some of them [128,163]. Moreover, PEFCR guidance and PEF guide [85,165] also allow the possibility of using indirect substitution as a form of allocation based on "other relationship". "Indirect substitution may be modeled as a form of allocation based on some other relationship when a co-product is assumed to displace a marginal or average market-equivalent product via marketmediated processes" [85].

3.3.2 Selection of the ISO allocation criterion

The main discussion on the allocation criterion concerns the nature of the so-called ISO "physical relationships" and "other relationships" [175,198].

The authors in line with the socio-economic school argue that allocation can be based on physical relationships only when the ratio of the output products can be varied since this allows the establishment of physical causality between functional units by mathematical modeling [97,157,214–217]. For example, Bernier et al. [103] assessed the impact of Kraft lignin and applied the physical causality principle to allocate the impact between pulp and lignin "by varying the quantity of lignin precipitated and then observing direct variations in the environmental loads". They also specified that this type of allocation was selected based on ISO standards, which recommends this type of allocation over allocation based on mass, energy or economic values. They, therefore, interpret "physical relationships" as "physical causality relationships" and interpret "other relationships (e.g. economic value)" as "other causal relationships". Accordingly, they consider the allocation by other relationships as the only possible approach when it is not possible to change the ratio of production of the functional outputs of the system [215,217]. The practitioners following the view of this school often argue that, at this level, economic allocation is the recommended option, and only when it is not possible to use economic allocation, the allocation can be based on a physical parameter that should be selected based on the best proxy for economic revenues (e.g. see [175]). For example, this happens when there is a lack of market prices for one specific product [218]. However, the approximations of these causal relationships have always been a debated scientific issue [219]. For example, these relationships can be based on the common function of all co-products (as done by [220]).

The text mining process revealed that only 28% of the LCA case studies selected an allocation method based on ISO relationships that could be interpreted as "causal relationships". The percentage of studies following this interpretation varies significantly depending on the sectors considered (see Figure 12). In particular, it is very low in the studies focusing on anaerobic digestion, bioenergy and bio-based materials (5-16%). On the other hand, this interpretation is largely present in the fossil fuels sector where the allocation of the emissions to the single products is often based on linear programming models calculating the marginal emissions by varying the amount of the functional units [221,222]. Another example where this interpretation is largely present is in the dairy sector. The main reason is that many practitioners assessing dairy products selected their allocation choice based on the recommendations of the International Dairy Federation [223], which adopts this interpretation.

Conversely, the practitioners belonging to the natural-science school often refer to an allocation by physical parameter as ISO-second level allocation by interpreting "physical relationships" as allocation based on a physical parameter e.g. mass or energy value [83,179,224–226]. On this basis, they prefer allocation based on a physical parameter (e.g. mass) over economic allocation because "ISO 14044



Figure 12. The percentage of studies in each research area that used causality as the principle for the allocation choices per (sub-)cluster and overall. The number of studies per sector: anaerobic digestion (21), bioenergy (185), fossil counterparts (34), agriculture (63), aquaculture (14), dairy and meat (79), waste management (50), bio-based materials (52).

standard mentions economic allocation when no other possibility is available" [227]. Economic allocation may be selected by the practitioners following this view if allocation based on physical parameter "result in the attribution of a large proportion of burdens to low-value co-products" [68]. The same school often argue that an allocation based on a physical parameter is preferred over economic allocation, since it is not affected by price fluctuations [225,228]. As a response, authors in line with the socio-economic school argue that the price fluctuation is not the important parameter for the allocation method, but the ratio of prices among all products, which is much less variable because it mainly depends on the fluctuating price of the inputs common to the process [229].

The preference expressed by the natural-science school is adopted by PEFCR guidance and PEF guide, which prefer allocation based on physical keys (e.g. mass or energy) to economic ones [85,165]. In the PEF guide, ISO "physical relationships" might have been interpreted as allocation based on physical parameters (this emerges from our understanding of annex X of PEF guide), leading to the preference for physical allocation keys. On the contrary, the ILCD handbook adopts the interpretation of "ISO physical relationships" from the socio-economic school



Figure 13. The key-route main path of research on LCA multifunctionality (output from Pajek calculations), obtained from the citation network analysis.

and states that only when it is not possible to find clear physical *causal* relationships between the co-functions, allocation based on economic relationships can be used [145]. However, differently from what is usually preferred by the socio-economic school, the ILCD handbook does not give preference to economic allocation over non-causal physical properties such as energy content [145]. The ILCD handbook also adds a footnote to remark that energy allocation is not an allocation based on ISO causal physical relationship but a simplified allocation based on a physical property that is not causal [27]. To make an example of the implication of adopting one interpretation or the other, we can consider a biorefinery example that produces fuels (e.g. ethanol) and chemicals for materials (e.g. lactic acid) [56]. The natural-science school would prefer energy or mass allocation (considered by them as ISO second level) over economic allocation (considered ISO third level). Conversely, the socio-economic school would prefer economic allocation arguing that mass and energy allocations (all considered ISO third level) are meaningless for such a biorefinery because of not representing any causality mechanism. Also, they would argue that it is not appropriate to use energy allocation when not all the co-products are used for their energy content or to use mass allocation when there are energy products among the co-products.

3.4 THE BIBLIOMETRIC REVIEW BASED ON MAIN PATH ANALYSIS

The main path of research identified using CNA is shown in Figure 13 and includes 21 articles. The evolution of the multifunctionality discussion in the scientific community can be divided into four periods, which have been defined as 1) Bilateral beginning, 2) The ISO 14041 influence, 3) Consequential LCA influence, and 4) ISO 14044 application.

3.4.1 Bilateral beginning (1994-1998)

The discussions on the LCA multifunctionality issue were initially developed following two parallel routes (see Figure 13). On the first route, Tillman, Ekvall and their co-authors developed different types of allocation methods for multi-output systems and open-loop recycling [80,81]. It is crucial to notice that at that time, the ISO 14041 was not yet released [188]. Tillman et al. [81] focused their article on the choice of system boundaries based on the purpose of an LCA. They defined three LCA purposes: 1) process tree (PT), today known as ALCA and applied to processes where there are one main product and some by-products, 2) technological whole system (TWS), similar to what today is known as ALCA and applied to processes delivering several co-products, and 3) socio-economic whole system (SWS), similar to current CLCA [81]. In this article, the word *expansion* was used once with respect to SWS, indicating that such a system accounts for economic and social factors and therefore "may lead to further expansion of the system" [81]. In 1994 and 1996, two conferences were held with sessions on allocation and life cycle

inventory. Clift, who was also co-author of the publications in the second parallel route, published the reports of such sessions [230,231]. These reports concluded that allocation must, when possible, be based on causal relationships. Ekvall and Tillman discussed this conclusion, arguing that causal relationships could be either cause-oriented or effect-oriented [80]. An example of the first one is the manufacture of a product that occurs because the company expects customers to be willing to pay for it (cause). An example of an effect-oriented relationship is a system delivering a recycled product, which reduces the amount of virgin product in another system (effect). This second type of relationship resembles the current CLCA thinking. To represent effect-oriented relationships, they argue that the effects of the investigated product on other life cycles can be included in the LCA through the expansion of system boundaries. As the expansion of the system boundaries, they cited the approach developed by Tillman et al. (1994), which today is known as "substitution". Moreover, they argued that when LCA is used as a tool for decision support, the allocation procedure should generally be effect-oriented rather than cause-oriented. Therefore, it is possible to identify the probable origin of the consequential school in the study of 1997 of Ekvall and Tillman [80] and the study of 1994 of Tillmann et al. [81].

The four articles of the second route were authored by Azapagic and Clift. In the first article (1995), they proposed linear programming (LP) modeling to solve the multifunctionality issue and to calculate the optimized environmental impact of plastic resins production, such as polypropylene and polystyrene [232]. The inputs and outputs of the system are then allocated to each of the co- and by-products through marginal changes in its production [232]. The marginal allocation coefficients correspond to the variation of the environmental burdens associated with a marginal variation of one of the co-functions [232]. The second article (1998) focused as well on LP as a tool for solving the problem of allocation and was applied to systems producing borate products. They highlighted that 1) "the main characteristic of this kind of modeling is that it is based on physical and technical relationships between the inputs and outputs [...] describing the underlying physical causation in the system" [157] and 2) that the allocation by causal relationships provided by the model is obtained "by exploring how the burdens change when the quantity of one function is changed with the quantities of all the other functions kept constant" [157]. These changes can be marginal, incremental, or average ones; however, LP can only be applied when system behavior can be linearized, which does not usually happen in average changes (i.e. substantial changes as for example the elimination of a functional output completely) [157].

In 1996, the first draft [233] of the ISO hierarchy for solving multifunctionality was released, as reported by Ekvall and Tillman [80]. This hierarchy was very similar to the one still present in the current ISO 14044:2006. System expansion was indicated in the first level [233]: "by expanding the system boundaries so that inputs, outputs and recycles remain within the system" (retrieved from [80]). From the literal statement, it appears clear that it was intended as an enlargement of the system boundaries to include all the co-functions within the boundaries (see Figure 3A.1). Such an approach is different from the system expansion method (substitution) indicated by Tillman's SWS, where functions are avoided instead of added.

On the second level, it was stated [233]: "where allocation cannot be avoided, the allocation should be based on the way in which the inputs and outputs are changed by quantitative changes in the products or functions delivered by the system" (retrieved from [80]). There was no use of the term "physical relationships" as in ISO 14044:2006. Hence, ISO was proposing allocation methods such as the marginal allocation developed by Azapagic and Clift [232], which are based on quantitative changes in the products or functions delivered by the system.

Finally, the last level allowed the allocation of different functions based on economic relationships, excluding allocation by physical properties. This preference for economic values could be due to their cause-oriented essence (the function is provided because one is willing to pay for it). Based on the analysis of this bilateral beginning and this ISO draft [233], it seems that the socio-economic ALCA school represents the first version of the ISO hierarchy. In fact, they distinguish themselves by applying system expansion by only adding (and not subtracting) functions, and preferring economic allocation to an allocation based on a physical parameter (excluded option by this first draft version).

3.4.2 The ISO 14041 influence (1999-2003)

In the third article of the second route, Azapagic and Clift used the boron product system to examine the different allocation methods recommended by ISO 14041, which was just released in 1998 [188,234]. ISO 14041:1998 introduced the same three levels of the hierarchy of the present ISO 14044:2006. However, the second level included the following clarification: "the resulting allocation will not

necessarily be in proportion to any simple measurement such as mass or molar flows of coproducts". Azapagic and Clift argued that, following ISO 14041, allocation by physical (causal) relationships had to be the result of mathematical system modeling [97,234]. Nevertheless, the ISO 14041 allocation underlying physical relationships allowed also allocations based on the "cause of the limits" of the amount of product output. This aspect emerges from the annex of ISO 14041:1998, where mass or volume allocations are suggested as representing physical relationships for road transportation because the quantity of materials transported is limited by the maximum load that the vehicle can carry [188]. Although these two approaches may at first appear contradictory, they are in fact in line with Azapagic and Clift's work, who also concluded that in some cases (which include the transportation example), allocating based on a physical quantity leads to the same results obtained by marginal allocation [97,234]. In these cases, it may be correct to allocate based on a physical parameter representing the physical causation involved, and therefore, not arbitrarily [97,234].

Azapagic and Clift (1999a) highlighted that system expansion (enlargement) is not always applicable. This approach is not possible when the goal of the study requires to determine the impacts of only one of the products [234]. The reason is that, by expanding the functional unit to include the co-functions, the results at the level of one single product would not be available. They also investigated allocation in heat and power cogeneration plants. Due to lack of data, they could not model the system to represent physical causalities and therefore applied the "avoided burdens approach" (in later research "substitution"). Azapagic and Clift argued that substitution is a conceptually equivalent alternative to system expansion, and is suitable when one co-product displaces its production elsewhere, such as for energy recovery from waste or cogeneration [234].

Actually, annex B of ISO 14041 quoted the same example of system-expansion/ substitution applied to energy from waste incineration [188]. Nevertheless, annex B specified that the expansion of the boundaries like this requires 1) that the goal of the study is aimed at assessing a change, "i.e. a comparison between two alternative scenarios for the same product" and 2) that the modeled change which will actually occur because of the decision supported by the LCA can be predicted with a fair degree of certainty [188]. To apply this type of expansion, the LCA should aim, therefore, to answer the question of what would have been the long-term marginal effect if the service had not been performed [188]. Hence, substitution became a possible system expansion approach in what nowadays is understood as consequential thinking. This annex with allocation examples is no more included in the current ISO 14044:2006.

In the lowest level of the allocation hierarchy of ISO 14041 [188], economic allocation became an example, and no more the only acknowledged allocation method as it was reported in the previous draft version [233]. Hence, in some cases, allocation based on a physical parameter could be preferred to an economic allocation, and this might have given birth to the natural-science ALCA school. Moreover, ISO 14041:1998 specified that the environmental impact should be allocated only to the products causing the release of the emissions (causality principle). ISO 14041:1998 proposed the example of a multi-input incineration process releasing cadmium emissions which should be allocated only to the input wastes that contain cadmium.

The fourth article [97] refers once again to LP-based marginal allocation, stating that this modeling applies "when the functional outputs can be varied independently" i.e. in partial joint production or combined production (see appendix for more details about these definitions). A naphtha cracking was proposed as an example of a system where the outputs can be independently varied (within physical and thermodynamic limits) by changing the operating conditions [97]. When that is not the case (i.e. full joint production; with a fixed ratio of products), "allocation by physical causality cannot be implemented" [97]. Linked to this impossibility, they provided the -many times cited- example of the ratio of sodium hydroxide (NaOH) and chlorine (Cl.) produced by electrolyzing brine, which is fixed by stoichiometry. Other examples that they mentioned about this impossibility are rapeseed oil/ residue (ratio fixed by the chemical structure of the plant) and beef/leather (fixed by the physical structure of the animal) [97]. In these cases, the authors stated that ISO recommended economic allocation because it reflects "the socio-economic demands which cause the multiple-function systems to exist" [97]. They concluded that "allocation on an arbitrary basis, such as mass or energy flow, must be avoided" and "where physical causality between functional units and environmental burdens exists, the allocation should always be based on these causal relationships"[97]. The authors based their methodological choices on the 1997 voting draft of ISO 14041.

The ISO 14041:1998 was complemented by ISO/TR 14049:2000 [186]. This technical report defined system expansion as the addition of functions but lost the concept

of system expansion with substitution when the goal is to assess a change. This is still missing in the current ISO 14044:2006 and ISO/TR 14049:2012 [38,187].

This ISO/TR provided two examples related to the disposal phase of the life cycle. The first example showed how to expand boundaries to compare two processes with different outputs, A and B, using the same inputs. As illustrated in Figure 3A.1d in the appendix, the system boundary for each process needs to be expanded with an alternative process for making the other product. Then the two systems under comparison produce the same functional unit A+B. Moreover, it specified that the added processes shall be those that "would actually be involved when switching between the two analyzed systems" [186]. In the second example, open-loop recycling is solved with a closed-loop procedure that includes the entire recycling processes into the same system boundaries (like case c of Figure 3A.1 in appendix).

Concerning allocation by physical property (e.g. mass or viscosity), ISO/TR 14049:2000 [186] emphasized that this type of allocation should be preferred to economic allocation only when it reflects the way in which the inputs and outputs are changed by quantitative changes in the products, (as, for example, in the transportation example in ISO 14041:1998, quoted above). This had to be proven by varying the ratio of co-products [186].

In 2001, Ekvall and Finnveden published a critical review on allocation in ISO 14041:1998 [217]. Ekvall and Finnveden stated that system expansion (in the form of substitution) could be used in a broader range of LCA goals than the one for which it is recommended by the annex of ISO 14041. For example, it can be used to account for indirect effects [217], similar to how the substitution method is used today in CLCAs.

In the same review, Ekvall and Finnveden (2001) identified the marginal allocation of [97] as a method corresponding to the second level in the ISO hierarchy (the first connection between the two parallel routes in Figure 13). In particular, Ekvall and Finnveden [217] explained that there were two possible interpretations for ISO allocation based on physical-causal relationships. Under the first interpretation, the "environmental burdens allocated to a function should be the burdens avoided if that function is no more delivered while the other functions are unaffected" [217]. This type of allocation is applicable when the environmental burdens are linear with the quantity of each of the functions delivered and, therefore, it is possible to eliminate the functions independently [217]. The second interpretation is that "the environmental burdens allocated to each of the functions should be proportional to the partial derivatives at the point of operation" [217]. This is a generalized description of the LP modeling of Azapagic and Clift.

Concerning the third level of ISO hierarchy, they emphasize that a rigorous interpretation of the standard leads to an allocation based on other causal relationships, e.g. economic value, and not in non-causal relationships (e.g. allocation based on "arbitrary physical property of the products such as mass, volume or energy content") [217]. As aforementioned, this strict approach was also the only one foreseen in the first draft of ISO hierarchy [233] and favored by the ALCA socio-economic school. At this point, the main path stops to be bilateral and starts a period of interconnection that led to the development and definition of what today is categorized as "consequential thinking".

3.4.3 Consequential LCA influence (2004-2008)

In 2004, the keyword *consequential LCA* appeared for the first time [235]. In the same article, Ekvall and Weidema delineated the consequential LCA as commonly defined today. They stated that CLCA avoids allocation by applying substitution-type system expansion, using marginal data [235].

Following the main path of research, we found several articles on CLCA case-studies. In the first article, Thrane conducted a CLCA of fish products [236]. The second article authored by Schmidt and Weidema [237] is focused on how to identify the marginal vegetable oil to be substituted in a CLCA of agricultural systems providing food and oil. Thrane [236] pointed out that, generally, the ISO allocation hierarchy can also be considered valid for CLCA. In fact, when system expansion (either by enlargement or by substitution) or subdivision is not applicable, it is necessary to allocate by physical or other relationships also in CLCA [236]. Dalgaard et al. [238] then performed a CLCA of soybean meal and avoided allocation by applying the substitution of marginal vegetable oil [238]. The fifth article of this period authored by Thomassen et al. compares attributional and consequential LCAs of milk production [89]. They showed that depending on the modeling approach (ALCA or CLCA), the results significantly vary for the same system because of the different ways of dealing with multifunctionality (allocation versus system expansion with substitution). In the middle of this period, ISO 14044:2006 was released.

3.4.4. ISO 14044:2006 application (from 2009)

At this stage, the most cited article identified by the search was published by Finnveden et al. [62]. This article repeated that the underlying physical relationships of ISO14044:2006's second hierarchy level should represent physical, chemical, or biological causation (as also specified before in [157]). Consequently, economic, mass, or energy allocations were intended to be used only as the third level option [62].

Following the main path, we found an LCA on a bio-based plastic product derived from a blood meal [239]. Bier et al. highlighted how different approaches for solving allocation issues in LCAs of bio-based materials could widely vary the results. The next two articles of the main research path discussed the choice of allocation approaches to use in LCAs aimed at informing policy-making. Wardenaar et al. [240] pointed out that methodological uncertainty within ISO leads to significantly different results due to the influence of the allocation approach, and argued that the policy context could benefit from new guidelines [240]. Concerning the ISO hierarchy, they stated that "several authors have argued that substitution is equivalent to system expansion" referring as an example to [80]. However, "conceptually equivalent does not mean that system expansion and substitution provide the same results" because there are "large differences between these two methods" [240]. As a consequence of this assumed conceptual equivalency, some authors "use this implicit argument to choose for substitution, while still claiming compliance to ISO" [240]. Concerning allocation based on a physical parameter, Wardenaar et al. argued that the physical parameter should be the one reflecting the physical characteristics related to "the purpose or use of the product" i.e. the relevant characteristic for which they are sold [240].

Following the main path, we found a study on the Environmental Footprint guidelines published by the European Commission [212]. The study of 2014 of Pelletier et al. [212] highlighted that in ISO's first level system expansion, the functional unit is expanded to include the other co-functions (enlargement), and the impacts are therefore reported at the system level, i.e. at the level of all co-products [212]. This was claimed to be the "literal interpretation of ISO 14044" [68]. Accordingly, the PEF guide [85] does not consider substitution as a system expansion approach, but only enlargement (similarly the more recent PEFCR guidance [165]). However, the ILCD handbook allows system reduction as an option only in CLCA, and for those ALCAs whose aim is to include also the interactions with other systems [145].

The key-route main path analysis allowed us to identify the origins of the "equivalency" substitution-system expansion i.e. the articles on the side of the bilateral beginning period originated by Tillman et al. [81]. The suggestion of Pelletier et al. [68] that this equivalency was originated from the study of 1994 of Tillman et al. [81] was therefore confirmed by our analysis. Nevertheless, the article by Tillman et al. [81] was published before any ISO standard and, therefore, did not refer to the system expansion method as intended by ISO.

In the next publication of the main path, Pelletier et al. (2015) observed that, despite the ISO hierarchy, consistent implementation of this hierarchy in the literature was limited, and presented the three schools of thought (consequential, socioeconomic ALCA and natural-science ALCA) mentioned at the beginning of section 3.3.

The next two articles in the research path are focused on finding allocation parameters for agricultural systems. The first article proposed an allocation based on plant physiological construction cost for plant compounds, which should represent the underlying physical relationships between co-products i.e. the physiological mechanism involved in plant growth [241]. Hence, they concluded that, according to ISO, such a method should be preferred to allocation based on common properties of co-products, such as energy or economic content [241]. Subsequently, Mackenzie et al. [215] studied similar biophysical allocation methods and concluded instead that these methods might not represent the causal physical mechanisms of these systems because they overlook the interconnectivity between co-products [215] as instead, LP would do. Therefore, they concluded that allocation by economics is preferable [215]. Mackenzie et al. also pointed out that many practitioners often choose an allocation based on an arbitrary parameter (e.g. their mass or energy content) also when it does not reflect such a cause-effect mechanism [215].

The last two articles of the main path are focused on how to allocate burdens to byproducts which were previously considered wastes [242,243]. These by-products are scarce wastes that can be converted into valuable products. In particular, Pradel et al. constructed a novel allocation method based on relevant causal relationships obtained by mathematical modeling [242]. This model was applied to wastewater treatment plants delivering sludge (by-product) and clean water (main product) and calculated the allocation factor for sludge and water.

3.5 DISCUSSION AND CONCLUSIONS

Despite the existence of a hierarchy for solving multifunctionality in ISO14044:2006, the complexity of the multifunctionality problem, the lack of sufficient guidance, its difficult interpretation, and the discrepancies in other "ISO-compliant" guides or handbooks have led to a wide variety of allocation procedures in the literature. Such variety is especially present in the system expansion approaches and in the choice of the allocation key.

ISO 14044:2006 does not distinguish between attributional and consequential modeling. For many practitioners, distinguishing between attributional or consequential LCAs is a crucial key to select the method to deal with multifunctionality. For other practitioners, some mixed approaches can be considered advancements in the methodology. We found that only 25% of the LCAs clearly state the approach followed using the terms "attributional" and "consequential". Are practitioners not specifying it because they assume it to be "intrinsically clear" from the goal description, or because they do not agree with such a distinction? Some mixed approaches are also proposed in the literature.

The first major reason for debate on ISO's multifunctionality hierarchy is the application of substitution as a system expansion method in ALCA (found in 31% of the self-declared attributional studies explored through text mining). Such practice is perceived as inappropriate by many practitioners. However, some practitioners who do not acknowledge substitution as system expansion in ALCA recognize the use of substitution as an allocation method for ALCA. Concerning the use of substitution, another aspect that many practitioners pointed out is that a future ISO standard should emphasize more the criterion of physical/economical significance as a prerequisite to apply substitution to avoid incorrect interpretations of the results.

The second reason for the debate is the meaning and application of the "ISO relationships" criterion for the selection of the allocation method. A first interpretation (found in 28% of the case studies) is that the ISO refers to "causal physical relationships" as relationships mathematically modeled, while "other causal relationships" relate to other relationships (e.g. based on physical or economic parameters) selected based on the best proxy for physical relationships. The second interpretation is that allocation by "physical relationships" refers to an allocation by physical parameters (e.g. mass or energy) while "other relationships" refer to economic relationships.

Most (94%) of the LCAs of multifunctional case studies found in the literature search are linked to bioeconomy (agriculture, biofuels, bioenergy, and biomaterials) and its linked sectors (fossil fuels and petrochemical plastic materials and dairy products). This has generated inconsistencies within each area, but also at the boundaries between these sectors, because of their multiple links. As an example, biogas can be produced from the manure of a farm, which produces dairy products with animals that eat dried distillers grains with solubles coming from ethanol fermentation. Such ethanol production may have a pre-treatment process shared with a lactic acid fermentation. This lactic acid may be used to produce poly-lactide, which on the market replaces polypropylene. The biogas above can then be used to generate electricity, that can be partly consumed in the farm, and partially injected in the grid, substituting power from fossil fuels. How many double counting or inconsistencies arise when ISO 14044:2006 is interpreted differently in each of these sectors?

The bibliometric review based on the analysis of the main path obtained tracing the citation network allowed us to 1) reconstruct how the implementation practices of the ISO hierarchy developed in the last 25 years, 2) identify the origin of the different interpretations and their rationales, and 3) understand how the discrepancies found in the critical review were generated. It emerged that, originally, the ISO hierarchy [233] recommended the approach followed by the "socio-economic ALCA school". The socio-economic ALCA school interprets system expansion as enlargement but preferse conomic allocation to allocation based on physical parameters representing a proxy for causality. The origin of the "natural-science ALCA school" was traced to ISO 14041:1998 [188], when, allocation by physical parameter, as well as economic allocation, was permitted as an example of ISO "allocation by other relationships". The natural-science ALCA school interprets system expansion as enlargement and applies allocation based on a physical parameter (for a part of the practitioners subscribing to this view, this choice is justified only when a physical parameter representing causality principles is identified). Its role was promoted by the release of the PEF guide and PEFCR guidance [85,165] which expressed a preference for allocation based on physical parameters over economic ones. Another important view is the one of the "CLCA school" interpreting system expansion as substitution and selecting the allocation method based on causality principles. The birth and development of the CLCA school were found in the annex of ISO14041:1998 and in the publications of Ekvall and co-authors [80,217,235]. They were the first ones (in

the main path) to acknowledge the suitability of the substitution method to avoid allocation and account for counterfactual effects (originally proposed by Tillman et al. [81]) and the assumption of "conceptual equivalency" of substitution with the system expansion method.

Summarizing, following one or the other school of thought, a different method is often preferred for the same system, goal and decision context. Applying these different methods often leads to different conclusions, and sometimes, opposite conclusions.

To increase the consistency and reliability of LCA, we believe that a future revised ISO should:

- State clearly if distinguishing between attributional and consequential LCA is a key principle to implement the hierarchy. If yes, then: it should differentiate the hierarchy for the two approaches and clarify if the hierarchy allows substitution as a system expansion method in attributional LCAs.
- 2. Clarify the meaning of allocation by "physical relationships" and "other relationships" providing more examples and details than the ones reported in ISO 14044:2006 and ISO 14049:2012.

3.A APPENDIX

Type of products

In this article, three definitions are used for different types of products and services: *co-products, by-products and wastes.*

Co-products are the ones satisfying the main (primary) function³ that a production system or process is intended to deliver. Conversely, by-products represent only secondary functions of the system. A by-product is a substance resulting from a production process whose primary function is not the production of that item but either it is inevitably produced or could, in principle, be avoided by the system without altering the main functionality of the process (e.g. a farm offering also tourist accommodation services).

The primary function of a product system is identified by evaluating the purpose of such a system. For example, for the internal combustion engine of a car, the primary product is the mechanical power needed by the car to carry people (primary function). A secondary function of the same engine can be the production of heating (by-product) to keep a proper temperature in the car. Nevertheless, the distinction between primary and secondary functions can be particularly difficult for some unit processes (e.g. sunflower oil vs meal). When such difficulty is encountered, the primary function should be selected by assessing what function of the multifunctional process generates more revenues for the investigated process [51,145,211], within the temporal scope of the LCA. Nevertheless, there are processes whose aim is the generation of several functions of comparable value. In such a case, there can be multiple primary functions. For example, a biorefinery can produce various chemicals and fuels as primary functions (co-products) and provide district heating as a secondary function (by-product).

The shared environmental impact of a process shall be apportioned between coproducts and by-products, but not to wastes [38]. According to ISO 14044:2006, wastes are "substances or objects which the holder intends or is required to dispose of" [38]. There is, however, a fine line between wastes and by-products. For example, manure is nowadays used as feedstock for biogas plants, used cooking oil is used for biodiesel production, and residues of the potato industry are used

³ As highlighted by Majeau-Bettez et al. (2018), co-products have also been defined with the term "primary", "determining", and others.

for animal feed. When these alternative uses make these wastes find a market demand represented by market values, they should be considered, therefore, as by-products. We adopt the distinction waste/by-product provided by the Waste Directive Framework [170]. A "waste" becomes a by-product when the "following conditions are met: 1) further use of the substance or object is certain; 2) the substance or object can be used directly without any further processing other than normal industrial practice; 3) the substance or object is produced as an integral part of a production process; 4) further use is lawful, i.e. the substance or object fulfills all relevant product, environmental and health protection requirements for the specific use and will not lead to overall adverse environmental or human health impacts" [170].

Type of multifunctional processes

As highlighted by several authors, the terminology reported in the literature for distinguishing the different types of multifunctional processes is not harmonized [51,71]. This article follows the terminology defined by Majeau-Bettez et al. [51], who differentiated between *full-joint production*, *partial joint production*, and *combined production*. *Full joint production* takes place when the co-products are produced simultaneously, with a fixed ratio of production (e.g. fixed by the stoichiometry of a chemical reaction, or by natural processes such as the proportions between wheat grains and wheat straw). *Partial joint production* occurs when there is an intermediate level of technological linkage between the different co- and by-products (e.g. an oil refinery as a whole or the production of milk and meat or the transportation of two different products) and *combined production* when there is not technological linkage (e.g. a gasoline station also offering shop services). According to this definition, the ratio of production of the co- and by-products could be varied in every case except for the full joint production.

Type of modeling approaches

The selection of the modeling approach is based on the goal of the study and the decision context. Generally, when the goal of a study is to describe the status of a system, an attributional LCA (ALCA) approach is followed to calculate the environmental impact of providing a specific amount of the functional unit [71]. When the goal is to describe the effect of a change due to a decision, a consequential LCA (CLCA) approach is followed to estimate how this environmental impact would change in response to a *change* in the output of the functional unit (i.e. it is change-oriented) [89]. The current conceptualization of the CLCA approach was first publicly discussed in the 2001 international workshop on electricity data for life cycle inventories [244].

One of the main principles of ALCA is the so-called 100% additivity [51]. This principle means that "results of a separate analysis of all economic activities should add up to the result of an analysis of the total economic activity"[74], so ALCA is suitable for attributing the total impacts to a defined function (product or service), but, for example, it does not indicate to policy makers the impact of policy changes, when these cause an incremental change from the *status quo*.

By contrast, CLCA determines the *change* in impacts due to a *change* in the production of the product or service, or to a change in policy. So it attempts to consider *all* the impacts of the change, also on other sectors that are influenced, for example as a consequence of the use of by-products [51]. CLCA is therefore preferred to ALCA for estimating the impact of policy changes [64]. CLCA usually uses market-driven modeling to forecast what will happen once the product or service of interest is introduced [35]. This means that in CLCA, marginal processes are considered, rather than average ones, including the activities displaced by by-products. This is typically modeled through the so-called substitution approach, whereby CLCA considers only the activities reacting to the change in demand for the functional unit, keeping the total of other services constant. Therefore, the quantification of displaced activities depends on the market characteristics of competing products [177].

Type of system expansion approach and substitution

System expansion means the enlargement of the boundaries of the system under investigation to include additional processes and functions. As mentioned above, expansion of the boundaries can be used to avoid allocation. There are two possible approaches to avoid allocation by expanding the boundaries: enlargement (see Figure 3A.1 for different types of enlargement) and substitution (see Figure 3A.2). By considering the subtraction as "a negative addition" [205], substitution is considered by some LCA practitioners as a form of system expansion used to isolate the impact of just one function from a multifunctional process.

One can apply system enlargement by modifying the functional unit to include all co-functions (case a of Figure 3A.1). This approach is not possible when the goal of the study requires to determine the impacts of only one of the products because the results at the level of one single product would not be available. System enlargement is also often used for comparative assessments. In Figure 3A.1b, the aim is to compare process P1 (providing functions A and B) with process P2 (providing only function A). One needs to add to P2 another process for producing B in order to allow the comparison for the same outputs. Similarly, in Figure 3A.1d. the aim is to compare a process producing A with a process producing B (for example, comparing the impacts of two products which could be made from the same raw material). In this case, one needs to add alternative processes for making both A and B in order to make a meaningful comparison. Even though these processes are not initially multifunctional, system enlargement is applied to allow for a fair comparison. One can also apply system enlargement in openloop recycling systems. In the example of system enlargement from ISO/TR 14049, open-loop recycling is solved with a closed-loop procedure that includes the entire recycling processes into the same system boundaries (like case c of Figure 3A.1).



Figure 3A.1 Different ways to apply system expansion as enlargement-addition of functions. In black: multifunctional process before applying system expansion. In blue: process after the expansion of the boundaries/addition of functions. a) Changing the FU to avoid allocation. b) Adding extra processes (P3, delivering B) to a system (P2, delivering A) that is compared with another system (P1) delivering several functions (A and B). c) Applying closed-loop recycling to a system (P2 represents the intermediate processing of B that allows its re-use). d) Adding extra processes (P3 and P4) to compare systems that provide different functions and that at the beginning were not multifunctional



Figure 3A.2 System expansion by substitution (reduction of functions). The investigated system (P1) delivers two products (A and B). Alternatively, product B can be produced by another system (P2). The substitution method proposes that the impact of producing A (only) by process P1, corresponds to the difference of impact between P1 and P2.





CHAPTER

Kraft lignin as a bio-based ingredient for Dutch asphalts: an attributional LCA

Published as:

C. Moretti, B. Corona, R. Hoefnagels, van Veen M., I. Vural-Gürsel, T. Strating, R. Gosselink, M. Junginger. Kraft lignin as a bio-based ingredient for Dutch asphalts: an attributional LCA

> Sci. Total Environ, 806 (2022), 10.1016/j. scitotenv.2021.150316

ABSTRACT

In the last decade, lignin has received much attention as a feedstock to produce bio-based products. This study investigates the potential benefits of using lignin to mitigate the environmental impact of the road construction sector. An environmental life cycle assessment (LCA) of various top-layer bio-based asphalts using kraft lignin was conducted. From a cradle-to-grave perspective, lignin-based asphalts were compared with conventional asphalts.

The results of the LCA revealed that the climate change impact of lignin-based asphalts could be 30-75% lower than conventional asphalts. For the other ten impact categories, trade-offs were observed. Overall, two key factors to make the environmental impact of lignin-based asphalts lower than conventional asphalts are 1) increasing the amount of bitumen-substituted and 2) using low-grade biomass fuels for process steam in the pulp mill. The substitution of weak filler with lignin was beneficial only for climate change and could lead to a worse overall environmental performance than conventional asphalts. Similarly, higher environmental impacts for lignin-based asphalts could be obtained if the pulp mill consumed natural gas to complete the energy balance to replace the part of the black liquor from which lignin is extracted.

This study also includes an in-depth discussion on methodological choices such as the allocation methods for lignin, functional units, and asphalt layers considered. We believe that such a methodological discussion could be helpful to support future Product Category Rules for asphalt mixtures.

4.1 INTRODUCTION

10% of the GHG emissions of the transportation sector are caused by the construction of roads [245]. This amount corresponds to more than 5% of the total greenhouse gas (GHG) emissions generated in the European Union (EU) [246]. Picturing, this amount is twice as much as the impact contributed by the aviation sector [246]. To contribute to the EU climate change reduction targets, the Netherlands aims to make this sector carbon-neutral by 2030 [247]. Some circular economy practices, such as the recovery and recycling of deteriorated asphalt, are already well-established all around Europe [248,249]. However, such practices alone are not sufficient to achieve carbon neutrality. To achieve such a goal, it is necessary to couple asphalt recycling with alternative construction and production methods as well as alternative (e.g. bio-based) materials and renewable energy sources [134,249].

Projects in the construction sector are often assigned via public tenders. The environmental aspects are becoming more often part of the public tenders in the EU member states [250,251]. With this purpose, the results of environmental Life Cycle Assessments (LCAs) have recently been increasingly used in public tenders of infrastructure projects such as roads, airports and railways [250,251]. LCA, which is standardized by ISO [38,127], is acknowledged as a rigorous methodology and should provide unbiased and comparable environmental calculations. Hence, combining properly the environmental impacts calculated using LCA with project costs should avoid that low-cost materials with high environmental burdens are selected in investments using public funding [251]. With this purpose, tenders in the Dutch construction sector include an environmental cost indicator that simplifies and unites various environmental impacts into a single monetary value score representing the avoided damage cost or shadow cost [250,252].

Bitumen production is one of the most important environmental burdens of asphalts [134,252]. For this reason, some renewable alternatives to bitumen such as plant-based binders, municipal wastes, and live-stock manures are currently under investigation [134,253]. Lignin, a bio-based by-product of pulp mills and lignocellulosic biorefineries, has received attention as a renewable binder to replace bitumen in asphalt mixtures [134,254]. Several roads using lignin-based asphalts have been paved in the Netherlands in the last five years [255]. However, there is a lack of clear evidence supporting the environmental benefits of producing

lignin-based asphalts. To our knowledge, there is only one peer-reviewed LCA [134] of lignin-based asphalts. The results of this LCA suggest that lignin-based asphalts with 25% replacement of the bitumen binder with lignin allow almost a 6% reduction in asphalt production GHG emissions [134]. However, this LCA study [134] presents some research gaps and leaves space for further research. For instance, the use- and end of life- phases were not considered in the study, and it only focused on climate change impacts. Moreover, the life cycle inventory of lignin production lacks transparency for what concerns data, allocation methods and biogenic carbon accounting, which are all crucial aspects to interpret the results of an LCA of a lignin-based application [254].

Our study presents the cradle-to-grave LCA of various top-layer lignin-based asphalts using kraft lignin, assessing 11 impact categories together with an environmental cost indicator. The cradle-to-gate (pulp mill gate) environmental impact of kraft lignin is also discussed. Since the Netherlands is the geographic scope of this LCA, beyond ISO 14040 and ISO 14044 [38,127] and EN 15804 [256], the Dutch Product Category Rules (NL-PCRs) [257], SBK bepalingsmethode 3.0 [258] and Dutch LCA asphalt sector report [252] were used as reference documents to conduct the LCA. Nevertheless, even following the best LCA practices recommended by these methodological documents, multiple sources of methodological uncertainty exist and the results of LCAs should be carefully interpreted [50,146,208]. For this reason, the effect of the allocation methods, functionali units and product systems' definition was broadly discussed through sensitivity analyses.

4.2 MATERIALS AND METHODS

4.2.1. Goal and scope definition

This study aims to assess and compare the environmental impact of lignin-based asphalts with conventional asphalts. The intended audience of this study is made of technology developers and researchers working on lignin-based applications. Given the valuable research conducted concerning LCA methodological implications presented in the discussion section, we believe that this study can be relevant for the LCA community involved in the construction sector. Moreover, the results of this study might be of interest for stakeholders that look for options to reduce the carbon footprint of road construction, e.g. policymakers, the road construction sector. The geographic scope is the Netherlands, and the temporal scope is 5-10 years from now.

In comparative LCAs, a proper functional unit, i.e. the reference unit of comparison, should be carefully selected. Hence, how well a product system fulfills a specific function should be accounted for in the functional unit. In particular, asphalts are required to maintain multiple rheological performances, especially in terms of stiffness/deformation resistance but also workability, durability, permeability and resistance to possible damages caused by moisture or fuels and oils [134,259,260]. Moreover, one of the key mechanisms of failure of asphalts affecting their lifetimes is the occurrence of fatigue cracks (predominately) within the binder of asphalt [134.26].262]. For this reason, various compositions for lignin-based asphalts were tested via standardized methodologies in the laboratory by Asfalt Kennis Centrum (project partner) to guarantee similar functional properties and performances to conventional asphalts. Empirical and functional tests were conducted according to respective documents of EN 13108 series (1 and 20) and EN 12697 series (8-12-23-24-25-26-31-33-35). The empirical tests included run on gyrator samples, hollow space percentage and water sensitivity and were conducted for all asphalt types investigated. The functional tests included performing on gyrator test pieces and rolled test plates, tri-axial test for rutting resistance, stiffness on test beam sawn from the rolled test plates, fatigue on the same beams. The functional tests were applicable only to asphalt concretes. Using these compositions, based on the goal of the study, a functional unit of 1 tonne (t) of top layer asphalt was selected in line with the reference unit used by the Dutch LCA asphalt sector report [252]. Adopting a different functional unit representing the whole asphalt product made of several layers is also considered in section 4.4.2. The baseline calculations refer to stone mastic asphalts (SMAs). Asphalt concretes (ACs) and porous asphalts (ZOABs with the Dutch acronym) were also assessed as alternative product systems. The LCA has a cradle to grave scope. Figure 14 illustrates the sub-division into life cycle stages with coding according to the Dutch reference documents [252,257]. As indicated for environmental product declarations based on the European standard EN 15804 used in the construction sector [263], this LCA is conducted using attributional modeling. By attributional modeling, all the environmental burdens of all processes involved in the life cycle were accounted for with representative average data [68,146]. As mentioned in the introduction, the cradle-to-gate environmental impact of kraft lignin is also discussed in this study. For such a product, a functional unit of 1 kg of dry lignin was adopted.



Figure 14. Flow chart of lignin-based asphalt with cradle-to-grave life cycle stages based on [252]. The unit process named "A1:Lignin production" starts from the cradle to the gate of the pulp mill where lignin is delivered.

The impact categories and respective weighting factors (see Table 4) were selected based on the calculation rules of SBK bepalingsmethode 3.0, which is based on the recommendations from ENI5804 + A2 (2019) [256]. In particular, the weighting factors used in the Netherlands are the so-called MKI weighs [258], which are based on the shadow price method (internationally often referred to as environmental cost indicator). The shadow price corresponds to the cost of the preventive measures for the government to avoid that environmental impact [252].

Impact category	Unit	Weighting factor (€/kg or €/ MJ)
Abiotic depletion	kg Sb eq	0.16
Abiotic depletion (fossil fuels)	МЈ	7.7E-05
Global warming (GWP100a)	kg CO ₂ eq	0.05
Ozone layer depletion (ODP)	kg CFC-11 eq	30.0
Human toxicity	kg 1,4-DB eq	0.09
Freshwater aquatic ecotoxicity	kg 1,4-DB eq	0.03
Marine aquatic ecotoxicity	kg 1,4-DB eq	0.0001
Terrestrial ecotoxicity	kg 1,4-DB eq	0.06
Photochemical oxidation	kg C_2H_4 eq	2.0
Acidification	kg SO ₂ eq	4.0
Eutrophication	kg PO ₄ eq	9.0

Table 4. Environmental impact categories and weighting factors [258].

4.2.2. Life cycle inventory

4.2.2.1 Extraction and processing of raw materials (A1)

The compositions of the asphalts compared were determined with the objective to have similar functionality performances. For this reason, the same type of asphalt (SMAs, ACs or ZOABs) with lignin or without lignin were assumed to have the same lifetime. The compositions of lignin-based and conventional asphalts are reported in Table 5. From this table, it is possible to notice that lignin replaces bitumen and a fraction of weak filler. For ZOABs, lignin replaces only weak filler and not bitumen. Table 5 also reports the background inventory data. As for the Dutch LCA asphalt sector report, the background data were retrieved from the ecoinvent library named "Allocation cut-off by classification" and the Dutch National Environmental Database [264], which contains various datasets representing the environmental performance of buildings in the Dutch context. In particular, the ecoinvent system model "Allocation, cut-off by classification" is an attributional database where the primary (first) production of materials is always allocated to the primary user of a material [265]. Accordingly, if "a material is recycled, the primary producer does not receive any credit for the provision of any recyclable materials", and the recyclable materials are available burden-free to recycling processes and bear only the impacts of the recycling processes [265]. However, in the Dutch asphalt sector, the credits generated outside the system boundaries are allocated to the final environmental impact of the asphalt (see section 4.2.2.6).

4.2.2.1.1 Lignins

There are various technologies and feedstocks from which lignin can be derived from pulp mills or lignocellulosic biorefineries [254]. The kraft process is the dominant process of the pulping industry [122]. With the kraft process, wood is converted into wood pulp, which is used in the production of paper. Black liquor is a by-product of the wood pulping process in a pulp mill. The major component of black liquor is lignin. The black liquor is usually burnt in a recovery boiler to produce internal process energy [103]. Alternatively, kraft lignin can be obtained from black liquor through precipitation and separation processes; for example by the LignoBoost process [266].

Raw material (amounts expressed in kg)	B-SMA	C-SMA	B-AC	C-AC	B-ZOAB	C- ZOAB	Inventory dataset
Recycled content	0	0	288	300	0	0	Burden-free
Cellulosefiber	3	3	0	0	0	0	SBK_ Cellulosevezels [264]
Bitumen	44.5	65	21.1	40	43	43	ESU NL-PCR bitumen [252]. The selection of the bitumen dataset is discussed in section 4.4.1.1.2
Crusher sand	88	75	154	171	106	106	Sand {CH} gravel and quarry operation Cut-off from Ecoinvent 3.6
Natural sand	88	75	76	57	0	0	SBK 296 Industriezand [264]
Crushed stone (Morene)	0	0	411	410	811	811	Gravel, crushed {CH} production Cut-off from Ecoinvent 3.6
Crushed stone (Porfier)	700	710	0	0	0	0	Gravel, crushed {CH} production Cut-off from Ecoinvent 3.6
Weak filler	35	72	23	22	0	40	Limestone, crushed, washed {CH} production Cut-off from Ecoinvent 3.6
Lignin	40	0	24	0	40	0	Modelled based on literature sources (see section 4.2.2.1.1)
Linseed oil	1.5	0	2.9	0	0	0	Linseed seed, at farm {CH} linseed seed production, at farm Cut-off from Ecoinvent 3.6

Table 5. Composition of 1 t of top layer asphalt based on industrial partner's primary data and background inventory datasets used. Lifetimes=15 years for SMAs and ACs and 12 years for ZOABs. B=bio-based, C=conventional.

In our LCA model, kraft lignin is separated from the black liquor and used in asphalts. For the production of kraft lignin, two studies were considered as main data sources: Culbertson et al. (2016) & Bernier et al. (2013). The background data can be found in the Appendix. In these two studies, the precipitation step is carried out by injecting liquid carbon dioxide in the black liquor together with sulfuric acid [100,103]. Three lignin production "scenarios" were modeled, one based on Culbertson et al. (2016) and two based on Bernier et al. (2013):

- "kraft1-BIOM" was modeled based on the inventory data from Culbertson et al.
 (2016). The pulp mill modeled by Culbertson et al. is equipped with a hog fuel boiler, has an internal power plant that sells an electricity surplus to the electric grid and uses natural gas only for the lignin extraction process.
- "kraft2-NG" was modeled based on the pulp mill modeled by [103]. The electricity is instead supplied to the pulp mill by the grid and natural gas is used to produce the part of the energy that is no longer produced from burning black liquor (due to the extracting of lignin). Since the consumption of natural gas (NG) to produce the kraft lignin is much higher in Bernier et al. (2013) than Culbertson et al.(2016), the lignin modeled based on Bernier et al. (2013) will be referred to as "kraft2-NG".
- "kraft2-BIOM" was modeled as well based on the pulp mill modeled by Bernier et al. (2013). According to Bernier et al., the same pulp mill could also potentially use hog fuel (chips of wood bark) instead of natural gas to compensate for the loss of steam production from the recovery boiler. Since the heat source is a key factor affecting the environmental impact of lignin [254], this third scenario represents the use of hog fuel instead of natural gas to compensate the loss of steam from the recovery boiler.
- Details on inventory data can be found in Appendix.

4.2.2.1.2 Allocating the environmental burdens to lignin

Since the production of kraft lignin is the result of a multifunctional process, it is necessary to apply an allocation method to apportion the environmental burdens of the pulp mill between lignin and pulp. In the LCA literature, it is acknowledged that the environmental impact of kraft lignin is importantly affected by the allocation method applied between pulp and lignin [55,254]. For this reason, how the allocation method was selected based on the reference documents is here extensively illustrated. The Dutch Product Category Rules [257] recommends following the directions from EN 15804 [256].

EN 15804:2012+A2:2019 specifies that the allocation between the co-products should be avoided by subdivision every time possible. The other typical option to avoid allocation i.e. system expansion is not allowed by EN 15804:2012+A2:2019. This is a common statement in rules for environmental product declarations to avoid that a substitution approach (one of the methods to perform system expansion) is used as a system expansion method in attributional LCA modeling [267,268]. In fact, the use of the substitution as a system expansion method in attributional LCAs is considered misconduct by many LCA practitioners and often leads to misleading interpretations or erroneous results [51,57,68]. If subdivision is not possible or data for sub-processes are not available, the allocation can be based on the underlying physical causality relationship i.e. reflecting "the way in which the inputs and outputs are changed by quantitative changes in the products or functions delivered by the system" [38].

This type of allocation is often referred to as "physical causality allocation [146,214,215]. A physical causality allocation relies on the mathematical modeling of the changes in operating conditions of the process under investigation to establish a physical causality relationship between functional units⁴. Accordingly, the PCR 2019:14 v.1.0 for construction products [268] and ISO 21930:2017 [269] i.e. the ISO core rules for environmental product declarations of construction products remark that physical causality can be established only if "each of the co-products can be produced without the other(s) or the ratio of the co-products typically varies in normal production", which is the only case when allocation by physical causality can be modeled [97,146,215,216]. According to EN 15804, allocation based on simple physical properties as mass can be used as a proxy for physical causality only if the difference in revenue from the co-products is low (defined as max. 25% difference) [256]. In all other cases, the allocation shall be based on economic values [256].

Accordingly, the pulp mill modeled by Culbertson et al.(2016) generating kraftl-BIOM was subdivided as much as possible. In this way, the lignin extraction process itself was not allocated to the pulp and bleaching was not allocated to lignin. For the other processes, keeping the pulp output constant and extracting kraft lignin leads to a change in the ratio between pulp and the surplus of electricity output produced (and a minor change in the soap output). So, allocation by physical

⁴ Sometimes, such a type of relationship can be reflected by a simple physical parameter for volume allocation in a truck transporting empty packaging or mass allocation in a truck transporting full packaging [187].

causality cannot be established. Moreover, since the difference in revenue from the co-products is not "low", economic allocation was used. To calculate the economic allocation share for lignin, the following prices were used: 535 €/t for kraft lignin [254], 788 €/t for pulp [270], 111 €/t for soap and 0.08 €/kWh for the surplus of electricity [271]. By using these prices, an allocation factor (A_L) of 7.3% to lignin was calculated. Two sensitivity analyses varying the price of kraft lignin and using mass allocation can be found in section 4.4.1.

In the case of the kraft2-NG lignin and kraft2-BIOM, electricity is supplied from the grid. Hence, a physical causality allocation was applied for this pulp mill. This type of allocation is based on the "physical causal relationships between burdens and functional outputs, which in turn requires a model of the system behavior" [157]. Taking the words of Azapagic and Clift, who are the authors that more than others have emphasized the importance of this allocation method [146], "when the causal relationships are represented by a model which describes the real behavior of the product system, the model can be used to allocate burdens between different functions by exploring how the burdens change when the quantity of one function is changed with the quantities of all the other functions kept constant" [157]. In order of magnitude, there are three types of changes that can be modeled, i.e. marginal-, incremental- and average [146,157]. Based on the approach adopted by the data source [103], average changes were used as the base for the allocation. Average changes are substantial changes such as eliminating or adding a functional output [146,157]. In Bernier et al. (2013), this type of allocation is "performed by comparing emissions with and without lignin recovery for the same mill and assigning the differences to lignin".

Both allocation methods preserve the additivity principle of environmental impacts, one of the key aspects of attributional modeling [51,146]. Such a principle for the two allocation methods applied is expressed by equation 1.

$$I_{t} = I_{p} + I_{L} = I_{t}^{*}A_{pp} + I_{t}^{*}A_{Lp} = I_{t}^{*}A_{pe} + I_{t}^{*}A_{Le}$$
 Eq.1

Where:

- I, is the total impact of the pulp mill producing both lignin and pulp
- I_p is the impact of the pulp mill if producing only pulp
- I_L is the additional impact generated by the pulp mill when lignin is extracted compared to when pulp only is produced
- A_{nn} is the physical causality allocation factor for pulp $(=I_n/I_t)$
- $A_{I_{n}}$ is the physical causality allocation factor for lignin (=I₁/I₁)
- A_{ne} is the economic allocation factor for pulp
- A_{lo} is the economic allocation factor for lignin.

4.2.2.2 Transport to the producer (A2) and production of asphalt (A3)

The transportation distances were mostly based on primary information from the asphalt industry. For the transportation of other materials for which primary information was not available, standard values from the Dutch LCA asphalt sector report [252] were used. A summary can be found in Table 6.

Based on primary data collected from one industrial producer (Roelofs Groep), natural gas consumption during the production phase differs between ligninbased and conventional asphalts. Each t of lignin-based asphalts require on average 5.3 Nm³ of natural gas (NG), while conventional asphalts require on average 7.5 Nm³ of natural gas. According to Schwarz et al. (2020), asphalts with recycled material (PR) use more natural gas due to the overheating stage. The overheating stage is responsible for 6% of the total natural gas use of asphalt production, and the overheating of one kg of PR requires 0.0015 Nm³ of natural gas (Schwarz et al., 2020). Therefore, the natural gas usage of each asphalt type can be calculated by using equation 2:

NG usage [Nm³] = average NG usage [Nm³] * (1-0.06) + 0.0015* PR [kg] Eq.2

Filling equation 2, the natural gas requirement is calculated as 5.0 Nm³ for SMA and ZOAB and 5.4 Nm³ for AC. Conventional SMAs and ZOABs require 7.1 Nm³, while ACs require 7.6 Nm³. Moreover, based on primary data from the same industrial producer, per each tonne of asphalt, on average, 5.6 kWh of electricity and 0.17 liters of diesel are needed during the production phase. For electricity, the dataset *Electricity, medium voltage {RER} market group for | Cut-off* from ecoinvent 3.6 was used updating the shares of electricity per source based on the 2030 EU reference scenario for the Netherlands [272].

In the Dutch LCA asphalt sector report [252], for conventional asphalts, natural gas and electricity consumptions are 5-15% higher than assumed in our study based on primary data, while diesel consumption is 40% lower. A sensitivity analysis considering such a relatively reasonable variation of data (probably linked to differences in the plant configurations or younger/older plant) was not conducted. On the other hand, in the Results section, the LCA results of lignin-based asphalts were compared with the calculated impact for conventional asphalts and the impacts for that type of asphalt reported in the Dutch LCA asphalt sector report. In this way, not only the possible differences in utility consumptions but also the ones in the assumed compositions (and transportation distances) were considered.

	km by lorry truck	km inland shipping	km transoceanic shipping	Data source
Cellulosefiber	177	0	0	Standard value from the Dutch LCA asphalt sector report [252]
Bitumen	64	0	0	primary information from the asphalt industry
Crusher sand	7	0	0	primary information from the asphalt industry
Natural sand	86	0	0	primary information from the asphalt industry
Crushed stone (Morene)	25	660	0	Standard value from the Dutch LCA asphalt sector report [252]
Crushed stone (Porfier)	25	53	933	Standard value from the Dutch LCA asphalt sector report [252]
Weak filler	7	0	0	primary information from the asphalt industry
Lignin	146	0	1822	primary information from the asphalt industry
Linseed oil	150	0	0	primary information from the asphalt industry

Tailala	Turner			and a strength of the	6	and the second	and is dening as	un alia mata
IODIE 6	irans	portation	alsiances	assumea	IOLIDE	asphair	mix s ina	realenis
rabie o.	i i ai is	porcación	anstantees	assannea	ion cric	aspriare	1111/2 1119	rearentes

4.2.2.3 Transport to the construction site (A4) and installation (A5)

The distance for the transportation to the construction site was taken to be 50 km by lorry (*Transport, freight, lorry 16-32 metric ton, EURO4 {RER}*/ transport, freight, lorry 16-32 metric ton, EURO4 / Cut-off) based on [252]. During the installation of the product, 0.26 liters of diesel [252] are consumed (*Diesel, burned in building machine {GLO}*/ market for / Cut-off from ecoinvent 3.6).

4.2.2.4 Use phase (B)

The service lifetime has currently no influence on the environmental impact calculations per tonne of asphalt conducted following the Dutch NL-PCR [252]. Based on the Dutch LCA asphalt sector report [252], the only process from the use phase included in this LCA is the leaching of substances from top layers to the soil under the influence of precipitation, which mostly happens in the first years after construction. For this reason, such an effect is considered independently of the lifetime of the asphalt [252]. The leaching emissions of various types of conventional asphalts determined with laboratory tests of factory samples were retrieved from [252]. Based on industrial partners (Asfalt Kennis Centrum and Holding de Vier Ambachten-H4A), there is a piece of the first evidence that the leaching emissions of lignin-based asphalts are in line with the ones from conventional asphalts, but further research is needed. So, for lignin-based asphalts, the same inventory data for leaching were used.

Other environmental burdens during the use phase, such as the ones due to the maintenance of the road surface and repairs, are not included [252].

The NL-PCR Asphalt [257] also considers a 10% mass loss of bitumen in top layers due to the effects of erosion [252]. This material loss occurs in nature as an inert material [252]. Similarly to the bitumen binder, it is assumed that a 10% loss due to erosion is also present in lignin and linseed oil.

4.2.2.5 End of life (C)

The end of life phase of asphalts is made up of several stages. The first stage is the demolition i.e. the removal of the asphalt, which requires an average of 23.0 MJ of diesel [252]. As background data for such a diesel consumption, the process *Diesel, burned in building machine {GLO}| market for | Cut-off* from ecoinvent 3.6 was used. Then, the recovered asphalt is transported for 50 km to the processing site to make road construction mixtures with a percentage of recycled asphalt. For the modeling of this transport step, the process *Transport, freight, lorry 16-32 metric ton, EURO4 {RER}| transport, freight, lorry 16-32 metric ton, EURO4 {RER}| transport, freight, lorry 16-32 metric ton, EURO4 | Cut-off from ecoinvent 3.6 was used. At the asphalt plant, the recovered asphalt is processed via breaking followed by blending/mixing, which requires 13.4 MJ of diesel per tonne of asphalt [252]. As background data, also for such a diesel consumption, the process <i>Diesel, burned in building machine {GLO}| market for | Cut-off* from ecoinvent 3.6

was used. Based on industrial partner information (Holding de Vier Ambachten-H4A), only 1% of the recovered asphalt (at the net of material losses described in section 4.2.2.6) ends in a landfill (*Waste concrete {Europe without Switzerland}*/ *treatment of waste concrete, inert material landfill | Cut-off* from ecoinvent 3.6). Consistently with the selection of the ecoinvent library named "Allocation cut-off by classification", the entire environmental impact of the end of life (C) is apportioned to the primary asphalt. So when an asphalt uses recycled content (secondary material), this material is free of environmental burden in module A1.

4.2.2.6 Benefits and loads beyond the boundaries (D)

Based on the NL-PCR and Dutch LCA asphalt sector report [252,257], module D is "outside the system boundaries" but included in the calculation of the environmental performance of the asphalt. In module D, the benefits resulting from recycling primary raw materials and the loads resulting from the loss of secondary raw materials are accounted.

Therefore, module D includes the benefits generated by the avoidance of the extraction (AI) and the burdens of transport of recycled asphalt granulates to the asphalt plant (A2) but also the loads of material that was obtained via recycling from the previous cycle but doe not get recycled in the current cycle. According to the NL-PCR [257], the amount of materials recovered during recycling is calculated considering two types of material losses. The first loss is due to the fraction of asphalt that leaves the asphalt system with no further use (e.g. a small fraction goes to landfill) or reused in a low-value function other than asphalt e.g. towards foundation layers. In the NL-PCR [257], this fraction accounts for 29% (L). Such a figure was based on the results of the investigation conducted by [273] and applies to all asphalt layers. The second material loss (L,) is represented by the 10% loss of bitumen (and lignin and linseed oil by assumption) due to erosion in the use phase of top layers described in section 4.2.2.4. Moreover, based on the NL-PCR [257], two quality losses were also accounted respectively for bitumen quality (4% quality loss Q₁), which was also applied to lignin, and the quality loss of crushed stones (Q₂=10% in the top layer and 5% for sub-layers). With respect to anti-drip inhibitors, they remain in asphalt but no longer fulfill any function [257]. So, they do not receive any credit in module D.

Equation 3 from NL-PCR [257] shows the mass of each raw material *i* that obtains credits or loads for their AI and A2 impacts. In particular, the loads are represented only by the second part of the equation where m_R is the mass of recycled content and M_{Ri} is the fraction of raw material *i* in the recycled content. As a derogation from NL-PCR [257], the mass of the recycled content was assumed to be the same as the one of the top layer asphalt. In particular, based on the compositions assumed (see Table 5), ACs only have recycled content and therefore a small percentage of loads (minor compared to the benefits) using Equation 3.

$$m_i^*(1-L_{1i})^*(1-L_{2i})^*(1-Q_{1i}-Q_{2i})) - (m_R^*M_{Ri}^*L_{1i}^*(1-L_{2i})^*(1-Q_{1i}-Q_{2i}))_{i\neq R}$$
 Eq.3

4.2.2.5 Biogenic carbon

Lignin-based asphalts contain two bio-based ingredients, which are lignin and linseed oil. The biogenic carbon content of kraft lignin was measured in the lab as 2.4 kg CO_2eq/kg . The biogenic carbon content of linseed oil was assumed as equivalent to 2.9 kg CO_2eq/kg based on [274]. One of the main updates to EN 15804:2012 (A2:2019) is that, in the new version, it is specified that "the biogenic carbon content shall not be allocated but always reflect the physical flow" [263]. Additionally, the NL-PCR [257] specify that "if the submitter can demonstrate based on tests and other technical evidence that the CO_2 is permanently stored in the asphalt product, the CO_2 stored could be included in the calculation of the CO_2 impacts".

The biogenic carbon from the bio-based inputs that is lost as an inert material (or ends to landfill) is assumed not to biodegrade since lignin does not biodegrade under landfill anaerobic conditions [275,276] and is mixed with bitumen in the asphalt matrix. So, it can be considered permanently stored during the life cycle due to losses/landfill since it is expected to remain stored for 100 years. Moreover, the lignin and linseed oil contained in the asphalt that is recycled (the part that is not lost) will go to sub-layers and over a time horizon of 100 years, it will be permanently stored in other asphalt products or e.g. lost as inert material in foundations in the following life cycles. The top layer asphalt contributed to store such an amount of biogenic carbon for a percentage of time, and after 100 years, it will be permanently stored. Hence, the top layer should be entitled to a percentage of the credit for the permanent storage of biogenic carbon over 100 years, which is therefore accounted (for SMAs and ACs, 15% over 100 years, for ZOABs 12%). This credit was included in module D.

4.3. RESULTS

4.3.1 Kraft lignins

The cradle-to-gate climate change impact of the three kraft lignins assessed is shown in Figure 15.



Figure 15. Cradle-to-gate climate change impact (excl. biogenic carbon removal) of 1 kg of kraft lignin. Biogenic carbon content expressed in terms of carbon dioxide equivalent: 2.4 kg CO_2 eq/kg.

As highlighted by the authors of the LCAs from which the data were retrieved [100,103], the climate change impacts of kraft lignin production are mainly caused by the consumption of natural gas (mainly emissions from combustion i.e. 83% of climate change impact) and the production of liquid carbon dioxide. For kraft1-BIOM, other relevant impacts are direct emissions from the calcination reaction (carbon dioxide) and from the combustion of hog fuel (direct emissions of dinitrogen monoxide and biogenic methane and electricity to operate the furnace) and the impact (mainly emissions of biogenic methane) from the treatment of the solid waste generated by the pulp mill. For kraft2 (both -NG and -BIOM), the additional sodium hydroxide, which is necessary to make up for sodium losses caused by lignin extraction, is also important. In particular, beyond natural gas, the differences in impacts are also linked to the different allocation methods. For example, the direct emissions from the calcination reaction are allocated via economic value to lignin in kraft1-BIOM while they were not allocated to kraft 2

using physical causality allocation (since their amount does not change extracting or not lignin). Similarly, the impact of sodium hydroxide is much higher for kraft2 (both -NG and -BIOM) than for kraft1-BIOM since the additional sodium hydroxide that is necessary due to lignin extraction was entirely apportioned to lignin via physical causality allocation. Conversely, given the impossibility to subdivide the process in which it is used, this additional sodium hydroxide was also allocated to the pulp and only a negligible fraction to lignin (via economic allocation see section 4.2.2.1.1).

Figure 16 shows the cradle-to-gate environmental impact of kraft lignin expressed in terms of MKI score. The environmental impact per each impact category can be found in Appendix.

From Figure 16, it is possible to observe that the difference of impact between the kraft lignins assessed is much lower than in the case of climate change (see Figure 15). For all lignins, the reason is that the impacts of carbon dioxide and sulfuric acid production are much higher in terms of MKI scores than for climate change. The reasons (see Figure 16 and details for these categories in Figure 17) can be found in their marine acquatic ecotoxicity and human toxicity impacts. In particular, for liquid carbon dioxide production, the main causes are the materials (mainly copper and steel) used to construct the chemical plant and the production of monoethanolamine (MEA). Similarly, for sulfuric acid production, the main impact is caused by the materials consumed for the construction of the chemical plant.

An important contribution to the impact is these categories is also caused by hog fuel combustion for kraft1-BIOM and kraft2-BIOM. For human toxicity, the main source of impact of hog fuel combustion is made from direct emissions (mainy benzene to air). For marine acquatic ecotoxicity, the electricity for operation of the furnace and materials for the construction of the furnace are the main impacts. For kraft2-NG and kraft2-BIOM, a small negative impact is generated by the reduction of direct emissions from the combustion of black liquor (see Appendix for further information about the meaning of negative emissions using physical causality allocation). Regarding kraft2-NG lignin, the important impact of natural gas is mainly caused by its combustion (76%).



Figure 16. Cradle-to-gate MKI score of 1 kg of kraft lignin. Breakdown per process contribution (top bars) and impact category (bottom bars). In this figure, the MKI score is without biogenic carbon removal, which accounts for 0.12 \in /kg.



Figure 17. Cradle-to-gate impacts of 1 kg of kraft lignin in marine aquatic ecotoxicity (top bars) and human toxicity (bottom bars).

4.3.2 SMAs

Figure 18 shows the climate change impact and environmental profile (in MKI score) for SMAs divided into life cycle stages.

In all SMAs investigated, the extraction and processing of raw materials (AI) make the largest contribution to the total climate change impact and MKI score. What is immediately noticeable is that the environmental impact of the asphalt components (other than lignin) is smaller when lignin is used. The main reason is the replacement of a fraction of bitumen (and for a minor fraction by the biogenic carbon content of linseed oil, which is higher than the climate change impact generated by the production of linseed oil). For climate change, the impact of lignin is negative when hog fuel is used for steam production (kraft1-BIOM and kraft2-BIOM) since the biogenic carbon stored in lignin outperforms the climate change impacts caused by their production. Conversely, in the case of kraft2-NG lignin, the biogenic carbon content of lignin does not outperform the climate change impact caused by the production of lignin but it is almost entirely compensated (almost 95%). Regarding the climate change benefits of accounting module D, these benefits are important for kraft2-NG lignin and conventional SMAs since bitumen (and other materials for a minor fraction) is recycled. However, in the case of kraft]-BIOM lignin and kraft2-BIOM, module D represents a small positive impact since the climate change benefits of recycling bitumen and lignin (and other materials) are lower than the biogenic carbon removal that from the lignin in the assessed asphalt will be incorporated in the recycled asphalt in the following cycle. Overall, all lignin-based SMAs offer a reduction of climate change environmental impacts i.e. 78% using kraft1-BIOM lignin, 34% using kraft2-NG lignin and 75% for kraft2-BIOM.

In terms of MKI score, only SMAs using kratf2-NG shows higher MKI scores than conventional SMAs. The reason can be found in the high environmental impact of the production and combustion of natural gas for kraft1-NG (see section 4.3.1).

The MKI score of conventional asphalt was calculated as 8.5 \in /t, which is 0.3 \in /t lower than calculated in the Dutch LCA asphalt sector report [252] assuming a different composition for conventional SMA and supply chains. Taking this value 8.8 \in /t as a reference would further improve the environmental competitiveness of lignin-based SMAs.



Figure 18. Cradle-to-grave climate change impact (top bars) and MKI score (bottom bars) of 1 t of SMA asphalt. Breakdown per life cycle stage.

4.3.3 ACs

Figure 19 shows the climate change and environmental profile (in MKI score) for ACs divided into life cycle stages.



Figure 19. Cradle-to-grave climate change impact (top) and MKI score (bottom) of 1 t of AC asphalt. Breakdown per life cycle stage.



Figure 20. Cradle-to-grave climate change impact (top bars) and MKI score (bottom bars) of 1 t of ZOAB asphalt. Breakdown per life cycle stage.

Compared to SMAs, the extraction and processing of raw materials (A1) make a lower contribution to the total climate change impact and MKI score since 1) the amount of bitumen in ACs is lower than in SMAs, 2) ACs have a fraction of recycled component (i.e. burdens free) and 3) the amount of lignin is as much lower compared to SMAs (see section 4.2.2.1). Comparing lignin-based and conventional ACs, similar trends to SMAs are observed for climate change and MKI scores. The

MKI score of conventional AC was calculated as $8.3 \notin/t$, which is in line with the range 7.4-8.8 \notin/t reported for several compositions of ACs in the Dutch LCA asphalt sector report [252]. However, considering this broader range instead of $8.3 \notin/t$, it is not possible to claim what asphalt between lignin-based and conventional ACs is better except for the AC using kraft1-BIOM.

4.3.4 ZOABs

Figure 20 shows the climate change (top) and environmental profile (bottom, in MKI score) for ZOABs divided into life cycle stages.

In the case of ZOABs, lignin substituted weak filler instead of bitumen. This means that this type of asphalt has benefits related to the biogenic carbon content of lignin, which is beneficial for climate change (and consecutively for the MKI score). However, for the same amount in mass, the impact of weak filler is negligible compared to bitumen. For this reason, ZOABs show a good performance in terms of reductions of climate change impact (up to 60%) but MKI scores are in line or higher than conventional ZOAB. The MKI score of conventional ZOAB was calculated as $9.2 \notin/t$, which is higher than 7.3-8.5 \notin/t reported in the Dutch LCA asphalt sector report [252]. So, taking this range, it can be concluded that lignin-based ZOABs substituting weak filler with lignin leads to a (slightly) higher MKI score than conventional ZOABs.

4.4 DISCUSSION

4.4.1 Multifunctionality

4.1.1.1 Lignin allocation

Previous literature showed that by-products like lignin are more affected than other products by the LCA issue of multifunctionality due to their lower physical/ economic significance than the main product [57,128,254]. The price of lignin with the same quality specifications (mainly measured in terms of impurities) has been quite stable over time. However, lignin prices can vary significantly depending on its quality specifications (mainly impurities) [117,254].

In our calculations, an average market price of 535 \in /t was assumed for kraft lignin. This price is in line with \$600/t assumed by Dessbesell et al. (2018) but higher than 250\$/t assumed by Abbati De Assis et al. (2018) and Culbertson et al. (2016). For a kraft lignin that meets the quality requirements to be used in asphalt, a reasonable price range is between 370 and 700 \in /t [254]. This price range was used for a sensitivity analysis on the economic allocation applied to the pulp mill producing kraft1-BIOM. To assess the impact of allocation methods, mass allocation was also applied. The results of this sensitivity analysis are shown in Table 7. Based on the results shown in Table 7, climate change impact and MKI scores of asphalts using kraft1-BIOM lignin are affected by the allocation method respectively within the order of ±10% and ±4%. Since the price of pulp in \in /t is higher than the lignin price (see section 4.2.2.1.1), mass allocation results in a higher environmental impact of lignin compared to economic allocation.

Sensitivity	1 kg kraft1- BIOM		1 t SMA kraft1- BIOM		1 t AC kraft1- BIOM		1 t ZOAB kraft1- BIOM	
	GWP 100 (kg CO ₂ eq)	MKI (€)	GWP 100 (kg CO ₂ eq)	MKI (€)	GWP 100 (kg CO ₂ eq)	MKI (€)	GWP 100 (kg CO ₂ eq)	MKI (€)
Min price	0.47	0.13	13.0	7.2	22.7	7.1	23.4	8.7
Baseline (avg. price)	0.58	0.15	14.6	7.5	23.9	7.3	25.0	9.0
Max price	0.67	0.17	16.1	7.7	25.0	7.5	26.6	9.3
Mass allocation	0.71	0.17	16.8	7.8	25.5	7.6	27.2	9.4

Table 7. Sensitivity analysis on the allocation method applied to kraft1-BIOM.

4.4.1.2 Bitumen allocation and dataset

In the baseline calculations, the prescribed process map of the NL-PCR asphalt i.e. ESU NL-PCR bitumen [257] was used. This dataset was based on data from Energie-Stoffe-Umwelt (ESU) consultants, which uses energy allocation [252]. However, bitumen, like lignin for biorefineries, is a low-economic/physical significance product of oil refineries and therefore, its environmental impact is affected strongly by allocation [222,279]. Moreover, applying energy allocation to bitumen seems in contrast with what is recommended by ISO 14049:2012 [187], which is the ISO technical report illustrating how to apply ISO 14044:2006. In such a report, ISO makes an example of bitumen production for which economic allocation is used since no physical parameter ("mass, feedstock energy, thermal conductivity, viscosity, specific mass, etc.") reflects the underlying physical relationship between bitumen and the other co-products.

In literature, there are some alternatives available that LCA practitioners in the construction sector often use. For example, the previous version of the Dutch LCA asphalt sector report [280] was using a dataset for bitumen (SBK bitumen) that was based on data from Eurobitume and economic allocation was applied [252]. Ecoinvent is also often used to retrieve datasets for petroleum products. For bitumen, two possible datasets are *Bitumen, at refinery/RER* and *Bitumen adhesive compound, hot {RER}| production | Cut-off.* Both these datasets use energy allocation [252].

Alternatively, a practitioner could follow a simplified approach and imagine the oil refinery as a black box. In literature, this approach is also referred to as allocation at the aggregate refinery level [222,279,281]. By using this method, the whole upstream and oil refinery emissions are allocated to oil products using allocation shares based on the production volumes (mass allocation), or economic value (economic allocation) or their energy value (energy allocation).

Typical efficiencies (kg of products/kg of petroleum oil input) of oil refineries are between 90 and 94% [281]. This means that for 1 kg of petroleum input (from ecoinvent, *Petroleum {GLO}| market for | Cut-off*), 0.92 kg of petroleum products are produced on average (among them gasoline, diesel and bitumen). The remaining 0.08 kg of petroleum input is burned for refining such products (from ecoinvent, *Refinery gas, burned in furnace/kg/RER*). Based on this simplified modeling, the environmental impact generated by burning fuels in the refinery and producing the upstream crude oil can be allocated to the oil refinery products using the abovementioned allocation shares. In Europe, bitumen represents about 3% of the economic share of oil refineries [282]. Mass and energy allocation shares for bitumen were assumed as 4.8% and 5.0% using petroleum coke as a proxy [222].

Figure 21 shows the climate change and MKI scores of producing 1 kg of bitumen. From Figure 21, it is possible to notice that 1) the dataset for bitumen from the NL-ESU PCR [257] is the one with the highest impact compared to all other alternatives compared, 2) economic allocation is the method that provides the lowest environmental impact for bitumen (in line with the effect of allocation on lignin shown by Table 7) while energy/mass allocation the highest. Moreover, the use of a different dataset and/or a different allocation method can influence the climate change impact up to a factor of 2-3 and the MKI score up to a factor of 4. It also emerges that the difference of data and not only of allocation play an important role in the difference of environmental impacts between the current and previous bitumen datased used by the Dutch LCA asphalt sector report. Figure 22 shows the effect on the comparative analysis between the environmental performances of lignin-based and conventional asphalts taking the lowest environmental impact for bitumen. Consequently, the environmental impacts of both lignin-based and conventional asphalts would be lower since both contain bitumen. On the other hand, assuming a lower impact for bitumen, while the trends observed for climate change in the baseline analysis are conserved, the MKI scores of SMAs and ACs using kraft2-BIOM lignin becomes worse than the ones of the conventional asphalts while they were better in the baseline calculations. In particular, the impact of ACs is the least affected by the allocation applied to bitumen since its bitumen content was the lowest compared to SMAs and ZOABs.



Figure 21. Climate change impact (weighted in terms of MKI score) and MKI score of 1 kg of bitumen using various data and allocation methods. Eco=economic; ene= energy.



Figure 22. Climate change impact (weighted in terms of MKI score) and MKI score of 1 t of top layer asphalt using a different dataset for bitumen (SBK bitumen). C=conventional

4.4.2 Change of functional unit (to 1 m²) and product system

In this LCA, the functional unit was defined as 1 tonne (t) of top layer asphalt. In the LCA literature, it is acknowledged that different functional units could lead to different results for the same product system [50,254,283]. Besides a mass-based functional unit, a common functional unit for asphalts is a surface-based functional unit [249,257,284]. Taking a functional unit of 1 m², it becomes a reasonable option to include in the product system also the middle and base layer asphalts that are under the top layer asphalt. A sensitivity analysis was conducted taking a functional unit of 1 m² of asphalt made of three layers. The compositions, lifetimes, densities and thicknesses of each layer can be found in Appendix. Depending on the type of asphalt, the mass of asphalt made of three layers under a surface of 1 m² has a mass between 0.39 and 0.40 t.

It is necessary to remark what follows to interpret the results of this sensitivity analysis correctly. Sub-layers have important recycled percentages. For ligninbased asphalts, the recycled part was tested with lignin-based recovered asphalt. The recycled material is free of the environmental burden in module A1 (that for lignin was accounted as environmental impact at the net of biogenic carbon



Figure 23. Cradle-to-grave climate change impact (top graph) and MKI score (bottom graph) of baseline (1 t of top layer asphalt) versus alternative functional unit (AFU) i.e. 1 m² of asphalt made of three layers.

removal). Accordingly, asphalt mixtures that already contain secondary material (recycled asphalt) do not include this percentage as environmental benefits in module D. Conversely, they are charged with a fraction of environmental burdens in module D as a result of the 29% loss of these secondary raw materials leaving the product system that cannot be used in the following cycle [257]. These charges are net of the quality losses that these materials would have had in the following cycle in the sub-layers where they would have been utilized. The distance for the asphalt to be recycled to the transportation to the construction site was assumed as 50 km [252]. Leaching does not take place in sub-layers since they are installed above groundwater level and do not come into contact with precipitation [252]. Virgin lignin used in the middle and bottom layer takes a 30% credit on the biogenic carbon that will be stored in the following cycles and was stored by this product system for 30 years.

Figure 23 shows the results of changing functional unit and product system. Changing functional unit does not lead to other trends. On the other hand, 1) the total impact is reduced in terms of surface since 1 m² contains less than 1 t of asphalt and 2) the differences between lignin-based and conventional asphalts are reduced since middle and base layers are 50% made of recycled asphalt (free of burden) and middle layer does not have any virgin lignin and base layer only 1%.

4.5. CONCLUSIONS

This study contributes to the existing research in the field of environmental sustainability of bio-based asphalt products. Various asphalts (stone mastic asphalt, asphalt concrete and porous asphalt) using kraft lignin were assessed using LCA methodology and compared with their conventional counterparts. The major LCA methodological choices were mostly based on the Dutch Product Category Rules (NL-PCRs) for asphalts. The data for kraft lignin were retrieved from two studies from the literature. The effects of allocation methods and defining a different functional unit were broadly discussed as well as the effect of using a different fuel (natural gas or hog fuel) for steam production.

The results of the LCA revealed that using lignin in asphalts could reduce the climate change impact of top-layer asphalt products over their life cycle substantially (order between 30% and 75% depending on the type of asphalt considered). The highest reductions are achieved if the current use of lignin (contained in the black liquor)

for process energy is replaced with low-grade biomass fuels, for example hog fuel. Using natural gas to replace the heating value of lignin for energy production leads to a significant/strong reduction in the climate change benefits.

Considering also other impact categories, on a weighted basis expressed in terms of economic cost (MKI scores), the advantages of using hog fuel instead of natural gas in the pulp mill on the impact of the asphalts are in part mitigated since other categories are less affected by such shift of burdens. Nonetheless, given the important advantages in the climate change category, using hog fuel instead of natural gas also showed advantages in total MKI scores.

For this reason, stone mastic asphalt containing lignin produced from a pulp mill where hog fuel is used to replace the part of the energy that is no longer produced from burning black liquor showed environmental advantages also in terms of MKI scores compared to their conventional counterpart. Conversely, if natural gas is used to deliver the required process heat, the MKI score of lignin-based stone mastic asphalts was worse than the conventional counterpart. In other words, the climate change impact (and, to a lesser extent, overall environmental impact) is largely determined by how additional process heat is produced in the pulp mill. This conclusion is in line with previous research on other lignin-based applications [106,119,254].

For asphalt concretes, it was not possible to identify an "environmental winner" between lignin-based and conventional asphalts since the differences in impact were in line with the uncertainties. For porous asphalts with lignin used only as weak filler, lignin-based asphalts perform worse than conventional asphalts. The reason is that the impact of limestone filler is minor compared to kraft lignin. To maximize environmental benefits, the maximum substitution of bitumen with lignin should be targeted, whereas the substitute of weak filler should be limited.

Regarding the allocation method, using mass allocation instead of economic allocation would lead to much higher environmental impacts for both lignin and bitumen. Using a physical causality allocation avoids that impacts that are typical of pulp production and are unaffected by extracting or not lignin are allocated also to lignin. On the other hand, once the credits for their recycled are accounted for, the effect of the change in allocation method becomes smaller.

Regarding the sensitivity analysis on the functional unit and product system considered, since middle and base layers have a higher percentage of recycled content and a lower amount of bitumen/lignin, once the mix of the three layers is considered as the product system instead of the top layer only, the MKI score difference between the two asphalt options becomes minor. Conversely, looking at climate change only, the difference is still significant given the importance of the biogenic carbon storage for this category.

4.A Appendix

Inventory data

Table 4AI. Life cycle inventory data for the production of 1 t of kraft1-BIOM lignin. Based on Culbertson et al. (2016)

Flow	Data	Unit	Process (ecoinvent)	Multifunctionality
Sulfuric acid for lignin extraction	0.07	t	Sulfuric acid {RER} market for sulfuric acid Cut-off	Subdivison
Natural gas for lignin extraction	1034	MJ	Natural gas, burned in industrial furnace >100kW/RER U	Subdivison
Liquid carbon dioxide for lignin extraction	0.28	t	Based on (Young et al., 2019), 3.56 MJ of heat are necessary to separate the carbon dioxide using MEA from waste gas (burdens free) from ammonia production. For heat from natural gas, the process Natural gas, burned in industrial furnace >100kW/RER was used. The amount of MEA, electricity consumption, losses of MEA to air and water flows of carbon dioxide separation using MEA were retrieved from ecoinvent 3.6 dataset Carbon dioxide, {RER} production Cut-off.	Subdivison
Sodium hydroxide	0.10	t	Sodium hydroxide, chlor-alkali production mix, at plant/RER	Economic allocation
Lime	0.41	t	Lime {RER} market for lime Cut-off	Economic allocation
Natural gas	21029	MJ	Natural gas, burned in industrial furnace >100kW/RER U	Economic allocation
Softwood	49.9	m³	Pulpwood, softwood, measured as solid wood under bark {Europe without Switzerland} market for Cut-off	Economic allocation
Combustion of hog fuel	152040	MJ	Retrieved from Heat, central or small- scale, other than natural gas {CH} heat production, softwood chips from forest, at furnace 50kW, state-of-the-art 2014 Cut-off. Background data for the electricity input for the operation of the furnace updated using 2030 EU reference scenario for the Netherlands (Carpos et al., 2016)	Economic allocation
Waste to landfill	1.07	t	Municipal solid waste {CH} treatment of, sanitary landfill Cut-off	Economic allocation
Sulfur dioxide direct emissions	27	kg	Sulfur dioxide, NL (direct emissions)	Economic allocation
Fossil carbon dioxide direct emissions	1402	kg	Carbon dioxide, fossil (direct emissions)	Economic allocation

Table 4A.2. Life cycle inventory data calculated using physical causality allocation based on average changes for the production of 1 t of "kraft2-NG" and "kraft2-BIOM" [103]. When the name of the flow does not specify the name of the two scenarios, that flow is the same for the two scenarios.

Flow	Data	Unit	Process (ecoinvent)
Sodium hydroxide	0.107	t	Sodium hydroxide, chlor-alkali production mix, at plant/RER
Sulfuric acid	0.23	t	Sulfuric acid {RER} market for sulfuric acid APOS, U
Lime	0.23	t	Lime {RER} market for lime APOS, U
Natural gas (kraft2- NG)	31500	MJ	Natural gas, burned in industrial furnace >100kW/RER U
Natural gas (kraft2- BIOM)	4700	MJ	Natural gas, burned in industrial furnace >100kW/RER U
Hog fuel production (kraft2-BIOM)	1.87	t	Bark chips, wet, measured as dry mass {CH} bark chips production, softwood, at sawmill Cut-off, U
Combustion of hog fuel	26800	MJ	Retrieved from Heat, central or small-scale, other than natural gas {CH} heat production, softwood chips from forest, at furnace 50kW, state-of-the-art 2014 Cut-off. Background data for the electricity input for the operation of the furnace updated using 2030 EU reference scenario for the Netherlands (Carpos et al., 2016).
Liquid carbon dioxide	0.3	t	Carbon dioxide, liquid {RER} market for APOS, U
Water	4.85	t	Tap water {RER} market group for APOS, U
Electricity	10	kWh	2030 EU reference scenario for the Netherlands (Carpos et al., 2016)
Reduction of combustion emissions from black liquor	-1	t	Emissions from avoided combustion of black liquor (Corona et al., 2018)
1			

Environmental impacts

Impact category	Unit	SMA kraft1- BIOM	SMA kraft2- NG	kraft2- BIOM	c-SMA	AC kraft1- BIOM
Abiotic depletion	kg Sb eq	3.92E-01	3.92E-01	3.92E-01	5.72E-01	2.25E-01
Abiotic depletion (fossil fuels)	МЈ	1.58E+03	2.07E+03	1.64E+03	1.96E+03	1.24E+03
Global warming (GWP100a)	kg CO2 eq	1.46E+01	4.31E+01	1.61E+01	6.50E+01	2.39E+01
Ozone layer depletion (ODP)	kg CFC-11 eq	8.52E-06	1.32E-05	9.27E-06	8.71E-06	8.56E-06
Human toxicity	kg 1,4-DB eq	2.53E+01	2.53E+01	2.78E+01	1.69E+01	2.24E+01
Fresh water aquatic ecotox.	kg 1,4-DB eq	1.28E+01	1.07E+01	1.19E+01	6.36E+00	1.34E+01
Marine aquatic ecotoxicity	kg 1,4-DB eq	1.92E+04	1.94E+04	2.05E+04	1.46E+04	1.81E+04
Terrestrial ecotoxicity	kg 1,4-DB eq	6.54E-01	6.55E-01	6.75E-01	1.53E-01	1.29E+00
Photochemical oxidation	kg C2H4 eq	5.27E-02	5.08E-02	5.63E-02	6.69E-02	3.30E-02
Acidification	kg SO2 eq	3.21E-01	3.51E-01	3.64E-01	3.20E-01	2.64E-01
Eutrophication	kg PO4 eq	6.03E-02	5.44E-02	6.54E-02	4.48E-02	6.49E-02

Table 4A.3. Cradle-to-grave environmental impacts of 1 t of top layer asphalt.

It continues

AC kraft2- NG	AC kraft2- BIOM	c-AC	ZOAB kraft1- BIOM	ZOAB kraft2-NG	ZOAB kraft2- BIOM	c-ZOAB
2.25E-01	2.25E-01	4.28E-01	3.79E-01	3.79E-01	3.79E-01	3.79E-01
1.59E+03	1.28E+03	1.70E+03	1.64E+03	2.13E+03	1.70E+03	1.62E+03
4.45E+01	2.49E+01	6.49E+01	2.50E+01	5.36E+01	2.65E+01	6.57E+01
1.19E-05	9.10E-06	9.02E-06	9.51E-06	1.42E-05	1.03E-05	9.28E-06
2.24E+01	2.42E+01	1.64E+01	2.89E+01	2.89E+01	3.14E+01	1.94E+01
1.19E+01	1.27E+01	6.83E+00	1.58E+01	1.36E+01	1.49E+01	1.03E+01
1.83E+04	1.90E+04	1.49E+04	2.48E+04	2.50E+04	2.61E+04	1.92E+04
1.29E+00	1.31E+00	1.29E-01	1.53E-01	1.54E-01	1.74E-01	1.27E-01
3.16E-02	3.56E-02	5.10E-02	5.11E-02	4.92E-02	5.47E-02	4.62E-02
2.86E-01	2.96E-01	2.72E-01	3.26E-01	3.55E-01	3.69E-01	2.86E-01
6.07E-02	6.86E-02	4.80E-02	6.84E-02	6.25E-02	7.35E-02	5.43E-02

Compositions of asphalt mixtures

Table 4A.4 Compositions, densities, thicknesses assumed for SMA (bind and base layer assumed as ACs).

Composition	Lignin	based asp	halt	Bitume	Bitumen based reference		
	Top layer	Bind layer	Base layer	Top Layer	Bind layer	Base layer	_
Cellulosefiber	3			3			kg/tonne
Recycled content		500	500		500	500	kg/tonne
Bitumen 40/60	44.5		5.5		18	18	kg/tonne
Bitumen 70/100		18		65			kg/tonne
Crusher sand	88			75			kg/tonne
Natural sand	88	200	187.5	75	200	200	kg/tonne
Crushed stone (Morene)		267	280		267	267	kg/tonne
Crushed stone (Porfier)	700			710			kg/tonne
Weak filler	35	15	15	72	15	15	kg/tonne
Lignin	40		10				kg/tonne
Linseed oil	1.5		2				kg/tonne
Asphalt density	2250	2375	2375	2255	2375	2375	kg/m³
Asphalt lifetime	15	30	30	15	30	30	Years
Asphalt thickness	3.5	5	8	3.5	5	8	cm

Table 4A.5. Compositions, densities, thicknesses assumed for ACs

Composition	Lignin based asphalt			Bitumen based reference			Unit
	Top layer	Bind layer	Base layer	Top Layer	Bind layer	Base layer	
Recycled content	288	500	500	300	500	500	kg/ tonne
Bitumen 40/60	21.1		5.5	40	18	18	kg/ tonne
Bitumen 70/100		18					kg/ tonne
Crusher sand	154			171			kg/ tonne
Natural sand	76	200	187.5	57	200	200	kg/ tonne

Composition	Lignin based asphalt			Bitume referer	Unit		
	Top layer	Bind layer	Base layer	Top Layer	Bind layer	Base layer	
Crushed stone (Morene)	411	267	280	410	267	267	kg/ tonne
Weak filler	23	15	15	22	15	15	kg/ tonne
Lignin	24		10				kg/ tonne
Linseed oil	2.9		2				kg/ tonne
Asphalt density	2335	2375	2375	2340	2375	2375	kg/m³
Asphalt lifetime	15	30	30	15	30	30	Years
Asphalt thickness	3.5	5	8	3.5	5	8	cm

Table 4A.6. Compositions, densities, thicknesses assumed for ZOABs

Composition	Lignin asphal	based t		Bitume	en based	reference	Unit
	Top layer	Bind layer	Base layer	Top Layer	Bind layer	Base layer	
Recycled content		500	500		500	500	kg/tonne
Bitumen 40/60			5.5		18	18	kg/tonne
Bitumen 70/100	43	18		43			kg/tonne
Crusher sand	106			106			kg/tonne
Natural sand		200	187.5		200	200	kg/tonne
Crushed stone (Morene)	811	267	280	811	267	267	kg/tonne
Weak filler		15	15	40	15	15	kg/tonne
Lignin	40		10				kg/tonne
Linseed oil			2				kg/tonne
Asphalt density	1950	2375	2375	1950	2375	2375	kg/m³
Asphalt lifetime	12	30	30	12	30	30	Years
Asphalt thickness	5	5	8	5	5	8	cm

Comment: about negative emissions using physical causality allocation versus substitution

The second level of ISO 14044:2006 multifunctionality hierarchy recommends an allocation underlying the physical relationships that exist between the products. Physical relationships are described as those relationships that are at the basis of how inputs and outputs change depending on changes in functions delivered. This type of allocation is referred to in the literature as physical causality allocation [82,216,285]. As described by ISO 14044:2006, ISO 14049:2012 and their previous versions, this type of allocation implies modeling some changes in the system under investigation. In this study, an average change was modeled to allocate the impact in one of the two pulp mills (i.e. the one that allowed such type of modeling). Conceptually, emissions based on mathematical modeling of physical causality represent the additional amount of emissions that the system would emit to produce one unit (marginal, incremental or average) more of one co-function. In the case of linear relationships (applicable only to marginal or incremental changes), this amount would also correspond to the emissions that the investigated system would avoid if it could prevent the production of one unit of that co-function. The concepts "marginal", "avoided" and "change" should not make readers link this type of allocation with system expansion by substitution in consequential modeling. The principle of additivity of the emissions typical of attributional modeling is respected by physical causality allocation as for any other allocation, i.e. the sum of the emissions allocated to each product with a physical causality allocation corresponds to the sum of the emissions of the system. This is not the case for substitution used as a system expansion method [286].

Some authors use substitution in attributional modeling to calculate an allocation factor (as a form of "proxy-based disaggregation") instead of as a system expansion method [51]. For example, EU footprint guides [85,165] suggest that substitution can also potentially be used as an allocation method by "other relationships" (ISO third level). Accordingly, Hermansson et al. (2020) applied two substitution-based allocation methods to kraft lignin. This type of allocation often leads to negative impacts, which means this method has been erroneously used and is wrongly representing the physical causalities of that system [286]. This often happens when the wrong product is chosen as the main product of the multifunctional system (Cherubini et al., 2011) and the credit of the subtracted functions (potentially avoided) is higher than the total impact (see Figure 4A.1). For example, in the case

of lignin, this happens if the impact of lignin is calculated as the impact of the total system subtracting the impact of an alternative technique producing a product fulfilling the same functions of pulp [55].

The generation of negative impacts by physical causality allocation has a different meaning [222,287] instead. For example, in the allocation model used in the European legislation for biofuels, negative refining emissions are obtained for heavy fuel oil [222]. The excess of heavy fuel oil is produced due to the need for refining more remunerative products (e.g. gasoline) and not because demanded by the market. So, the extra hydrogen needed by the refinery to desulphurize the extra heavy fuel is caused by gasoline and not by heavy fuel oil. Therefore, the most valuable products take an additional impact because they cause the production of this surplus of low-value products and such additional hydrogen consumption (causality). This additional impact gets a sign minus for the low-value product based on the attributional additivity of emissions calculated with mathematical modeling.



Figure 4AI. Negative emissions by physical causality allocation vs negative emissions by (direct) substitution-allocation. The red line (Total impact) represents the total impact of the system producing A, B and C that need to be allocated. A is the main product, B is a co-product, and C is the by-product with less physical/economical significance. By physical causality allocation, product C may get negative emissions. By substitution-based allocation, product A may get negative emissions after substitution-based allocation. The substitution method does never agree with physical causality impacts in case of negative impact i.e. in such cases, it fails in representing the causality of the system.



5

CHAPTER

Combining biomass gasification and Solid Oxid Fuel Cell for heat and power generation: an early-stage Life Cycle Assessment

Published as:

C. Moretti, B. Corona, V. Rühlin, T. Götz, M. Junginger, T. Brunner, I. Obernberger, L. Shen. Combining biomass gasification and solid oxid fuel cell for heat and power generation: an earlystage life cycle assessment.

Energies, 13 (2020), 10.3390/en13112773

ABSTRACT

Biomass-fueled combined heat and power systems (CHPs) can potentially offer environmental benefits compared to conventional separate production technologies. This study presents the first environmental life cycle assessment (LCA) of a novel high-efficiency bio-based power (HBP) technology, which combines biomass gasification with a 199 kW solid oxide fuel cell (SOFC) to produce heat and electricity. The aim is to identify the main sources of environmental impacts and to assess the potential environmental performance compared to benchmark technologies. Also, the use of various biomass fuels and alternative allocation methods were scrutinized.

The LCA results reveal that most of the environmental impacts of the energy supplied with the HBP technology are caused by the production of the biomass fuel. This contribution is higher for pelletized than for chipped biomass. Overall, HBP technology shows better environmental performance than heat from natural gas and electricity from the European grid. When comparing the HBP technology with the biomass-fueled ORC technology, the former offers significant benefits in terms of particulate matter (about 22 times lower), photochemical ozone formation (11 times lower), acidification (8 times lower) and terrestrial eutrophication (26 times lower). The environmental performance was not affected by the allocation parameter (exergy or economic) used. However, the tested substitution approaches showed to be inadequate to model the environmental impacts of CHP plants under the investigated context and goal.

5.1 INTRODUCTION

Compared to separate production of heat and electricity from fossil fuels, combined heat and power systems (CHPs) can potentially allow for significant reductions of climate change impact [288,289]. In Europe, coupling heat and electricity generation from renewable sources is also one of the most cost-effective decarbonization strategies [290–292]. In particular, solid biomass has attracted increasing interest by policymakers and investors especially due to the high availability of local biomass from forests and wood processing industries in some regions [293]. The environmental performance of biomass-fueled CHPs depends not only on the type of technology but also on the type of biomass, its supply chain and the environmental impact categories in focus [294,295].

Mature CHP technologies using solid biomass as fuels have often shown restricted fuel flexibility, limited electric efficiencies and high particulate matter emissions [296]. To overcome these three limitations, a novel technology has been developed during the H2020 HiEff-BioPower⁵ project [297]. This novel technology (see Figure 24) is based on a fixed-bed updraft gasifier coupled with a novel primary gas treatment zone, a novel gas cleaning unit (GCU) and a solid oxide fuel cell (SOFC). Its current technological readiness level is between 4 and 5 (based on the definition adopted by the European Commission [298]). The biomass fuel is converted into product gas in the gasifier. Syngas derived from biomass (e.g. wood chips) contains HCl, H₂S and tars [299] making it not suitable for direct utilisation in fuel cells [300], which require purified gaseous fuels. Therefore, the syngas from gasifier is first pretreated in a primary gas treatment unit (first tar reforming step) and then purified in the GCU. The GCU is one of the key innovations of this technology. It combines the use of ceramic filter candles and sorbents [297]. Syngas cleaning is processed in five steps: primary tar reforming, high-temperature particle filtration, HCl sorption (after cooling the product gas), H₂S removal by sorbents and tar reforming (after re-heating). After re-heating the product gas is then fed into the SOFC unit to generate electricity. The off-gases from the SOFC unit are then burnt in a catalytic afterburner to recover heat. Most of biomass CHPs are suited for medium and large-scale plants (1-100 MWel). The HBP is available also in small size (about 200 kW of electricity output)[296]. For this size, among biomass technologies, one of the main competitors is the organic Rankine cycle (ORC) technology [301].

⁵ http://www.hieff-biopower.eu/home/



Figure 24. Concept of HBP technology. GCU=Gas cleaning unit. SOFC=Solid oxide fuel cell.

At this stage of the Hieff-BioPower project, the assessment of life-cycle potential environmental impacts of the current design configuration can help the technical development of the HBP, by minimizing the impacts already at an early stage of development. In particular, the literature reports a few Life cycle assessments (LCAs) of heat and power from SOFC-based CHPs, and several ones of CHPs involving biomass gasification processes but no one on their combination⁶. These studies provided the following main findings: (1) the investigated CHPs present lower impact in terms of climate change compared to conventional technologies [289,302] and (2) the biomass fuel production has the highest contribution to total life cycle impacts [303,304]. These studies also highlighted several methodological uncertainties of LCAs that can lead to significantly different results. Such uncertainties are mainly linked to the multifunctional nature of the CHPs. A CHP is a system producing two products, heat and electricity. Depending on the goal of the LCA, it may be necessary to apportion the overall impact of the system to each of the co-products. Finding the right criterion for the allocation of impacts to each co-product is generally understood as a multifunctionality problem [38]. When a multifunctionality issue is encountered, the practitioner has to properly select the functional units and allocation methods [305,306]. The selected criterion could affect the outcome of the LCA significantly and, for this reason, this selection is broadly discussed in the literature [51,56].

The environmental LCA presented in this study has a twofold aim: 1) to identify the main sources of the environmental impact of this new technology and 2) to

⁶ From Scopus database (October 2019): TITLE-ABS-KEY (Ica AND chp AND (gasification OR gasifier) AND sofc) 0 documents, TITLE-ABS-KEY (Ica AND chp AND sofc) 9 documents, TITLE-ABS-KEY (Ica AND chp AND gasification OR gasifier) 21 documents

assess its ecological competitiveness compared to separate production of heat and electricity, and to one of its main competitor, i.e. Organic Rankine Cycles (ORC). Moreover, this case study is used to analytically discuss the influence of the allocation method in the LCA results for CHP plants and provide methodological recommendations for better allocation practices.

5.2 MATERIAL AND METHODS

5.2.1 Goal and scope definition

The LCA has been conducted according to ISO 14040:2006 and ISO 14044:2006 [38,127]. The intended audience of this LCA consists of technology developers, researchers involved in the field of bioenergy and LCA practitioners. An attributional LCA (ALCA) approach is followed since the goal of this study is to identify the activities within HBP causing the highest contribution to the environmental impacts, and not the consequences of changes in these activities [51,68].

Two technologies are considered for environmental comparison: 1) a combination of the electricity mix (EMIX) from the German national grid plus heat provided from a natural gas boiler (NG) and 2) biomass-based organic Rankine cycle (ORC) CHP.

The HBP plant delivers two different functions simultaneously, namely the supply of heat and power. Since the ORC Rankine cycle has a different heat to electricity ratio compared to the HBP, the definition of two functional units was preferred to the definition of a single functional unit with a fixed heat/electricity ratio. Hence, two functional units were defined as follows: 1 kWh of electricity or 1 MJ of heat.

The HBP technology finds one of the main strengths in its fuel flexibility [297] since it can operate with various biomass feedstocks in the forms of chips or pellets. To explore the effect of different feedstocks on the environmental impacts of the HBP CHP technology, this study explored the use of three different types of biomass fuels: wood chips, wood pellets, and Miscanthus pellets. The operation with wood chips was considered as the baseline scenario (WC), while the operation with wood pellets (WP) and the operation with Miscanthus pellets (MP) as alternative scenarios. The baseline scenario with wood chips was also used for comparison with the competing technologies, i.e. the ORC technology (fueled with wood chips as well) and the combination of grid electricity plus natural gas boiler. Additionally, this last competing option was also compared to the WP and MP alternative scenarios.
Figure 25 shows the process diagram of the HBP product system. The system boundaries follow a cradle-to-gate approach. As shown in Figure 25, all the life cycle stages from the extraction of the raw materials to the final dismantling and waste treatment are included. The final distribution and consumption of the products, i.e. heat and electricity, are not included in the LCA. Also, after the power plant is dismantled and parts are recycled, the use of the recycled materials is outside of the system boundaries. Biomass transport stages from the forest to the processing plant and from the processing plant to the HBP plant were included in the study. The transportation of plant components (e.g. the gasifier) from the production site to the power plant location and the construction activities of the plant were not included in the analysis. The exclusion of these activities was based on their expected minor contribution to the total environmental impacts, as also found in similar studies, e.g. [303].



Figure 25. Flowchart of the HBP product system, including system boundaries (dashed lines).

The temporal scope of the study is placed in the near future (the next 5-10 years) when the HBP technology should be commercialized. The HPB is assumed to be installed in Germany, being the country with the maximum potential sales for the

HBP technology in Europe [307]. Nevertheless, some components for the HBP (e.g. the gasifier) might also be manufactured outside Germany (in other EU countries).

Seven mid-point impact categories were selected and the adopted impact assessment models for each impact category were selected following the ILCD recommendations [145] (see Table 8). Climate change (CC) and depletion of mineral, fossil, and renewable resources (MFRD) were chosen because they are considered top priorities in the current societal and political challenges [308]. Particulate matter (PM) and photochemical ozone formation (POF) are selected because of their relevance to the energy sector [309]. Acidification (AC), Terrestrial eutrophication (TE) and Water resource depletion (WRD) were selected because of their relevance for agricultural systems, and therefore for biomass production [42].

Impact Category	Unit	Impact assessment models
Climate change (CC)	kg CO ₂ eq	IPCC 2013 Global Warming Potential 100 years [310]
Particulate matter (PM)	kg PM2.5 eq	Premature death or disability from particulates/respiratory inorganics from [311]
Photochemical ozone formation (POF)	kg NMVOC eq	Potential contribution to photochemical ozone formation for Europe from [312]
Acidification (AC)	molc H+ eq	Accumulated Exceedance (AE) characterizing the change in critical load exceedance of the sensitive area from [313]
Terrestrial eutrophication (TE)	molc N eq	Accumulated Exceedance (AE) characterizing the change in critical load exceedance of the sensitive area from [313]
Water resource depletion (WRD)	m³ water eq	Freshwater scarcity: Scarcity-adjusted amount of water used from Swiss Ecoscarcity 2006 [314]
Mineral, fossil and renewable resource depletion (MFRD)	kg Sb eq	Depletion of resources based on the scarcity model from [315]

Table 8. Selected impact categories and models

To assess the robustness of the results, two sensitivity analyses were conducted. As anticipated in the introduction, a comprehensive sensitivity analysis was performed on the allocation choices to explore their influence in the outcome of the LCA (and as recommended by ISO [38]). The second sensitivity analysis was performed to

explore parameters that are potentially sensitive for the results and that might environmentally improve or make less attractive the technology in the future.

5.2.2 Life cycle inventory

5.2.2.1 Unit processes, data, and assumptions

This study assesses the small scale configuration of the HBP technology, which has a nominal electricity output of the SOFC of 199 kW. Its main characteristics during the average lifetime (assumed 18 years) are reported in Table 9.

For the foreground system, data on the gasifier and the GCU were collected from the industrial partners involved in the H2020 HiEff-BioPower project. For the SOFC, secondary data based on the scientific literature [304] were used due to the unavailability of specific primary data. The background data were largely based on the ecoinvent database (version 3.4). For unavailable data, assumptions were made based on literature (see the following sub-sections for details regarding each phase of the life cycle).

Since the system provides two different products (and functional units), it was necessary to determine an allocation key to partition the overall impact to the two functional units. Allocation by physical causality was not applied under the absence of a representative mathematical model (to model the causality relationships) [128,215,234]. Among the possible remaining allocation methods, the exergy key was chosen because it can represent both quantity and quality of both functional outputs, and is common practice for CHPs (e.g. ecoinvent uses such key [316] and is also recommended by RED II [9]). Table 9 shows the biomass input, intermediate performance indicators, and energy outputs in terms of their exergy and economic value based on the two biomass feedstocks. The exergetic outputs expressed in percentage reported in Table 9 represent also the allocation factors used for the baseline calculations. The economic values are based on three years (2015-2017) average prices for medium size industries without VAT, in Germany, retrieved from Eurostat [317]. The prices were $0.079 \in$ per kWh of industrial electricity and $0.0086 \in$ per MJ of industrial heat.

Flow (unit)	Wood chips (30 wt.%)	Pellets (from wood and Miscanthus) (5 wt.%)
Biomass fuel (kW)	548.5	570.0
Biomass fuel (kg/h)	164.1	115.7
Gross electric power (kW)	170.5	190.0
Thermal power output (kW)	288.5	292.0
Electrical efficiency gross (%)	31	33
Electrical efficiency net (%)	30	32
Thermal efficiency (%)	53	51
Exergy output as heat (%)	24.6	22.7
Exergy output as electricity (%)	75.4	77.3
Economic output as heat (%)	41.8	39.3
Economic output as electricity (%)	58.2	60.7

Table 9. Characteristics of the small scale HBP technology (8000 hours of operations per year). Modeled values.

5.2.2.2 Inventory data for chips and pellets

To model the life cycle of wood chips, the ecoinvent 3.4 dataset "Wood chips, wet, measured as dry mass {CH}| market for | APOS" was used. This dataset includes both wood chips from industrial activities and forest management and represents the average Swiss market (assumed to be a good proxy for Germany). In particular, wood chips from forest management represents an 85% share of the modelled Swiss wood chip market.

For wood pellets, the ecoinvent 3.4 dataset "wood pellet, measured as dry mass {RER}| market for wood pellet | APOS" was used.

For Miscanthus pellets, a similar dataset was not available in ecoinvent. Hence, the inventory data from [318] were used together with the best practices reported in [319]. An average dry yield value of 23.5 t Miscanthus (85% dry matter) per hectare was used to estimate the land requirements to provide enough fuel for the HBP plant for one year. The planting rate of 16'000 Miscanthus per ha was taken from [319]. As Miscanthus is a perennial crop, field preparation activities such as herbicide application, harrowing and plantation, occur only during the first year. The lifetime of the crop was assumed to be 18 years [319] and therefore 1/18 of the impact from field preparation activities was apportioned to one year of operation of the HBP plant. Once the Miscanthus is collected from the field, it is necessary to transport it

to the pelleting plant. The transport distance to the pelleting plant was assumed to be 10 km by tractor [318]. For the chipping of Miscanthus, the energy consumption of the chipper and the amount of lubricating oil were retrieved from the ecoinvent 3.4 datasets "Wood chips, wet, measured as dry mass {CH}| wood chips production, hardwood, at sawmill | APOS". For the pelleting of Miscanthus, the amounts of electricity, heat, lubricating oil and water were retrieved from the ecoinvent 3.4 dataset "Wood pellet, measured as dry mass {RER}| wood pellet production | APOS". The transportation of miscanthus pellets to the HBP plant was assumed to occur by truck and with an average distance of 100 km [318].

5.2.2.3 Inventory data for the manufacturing of the power plant

The HBP manufacturing consists of three sub-processes: the manufacturing of the gasifier, the manufacturing of the SOFC stack and its balance of plant (BoP), and the manufacturing of the GCU. The data for the manufacturing of the gasifier is based on HBP project data and shown in Table 10.

Material	Amount	Process dataset
Steel (low alloyed) (kg)	6770	Steel, low-alloyed {RER} steel production, converter, low-alloyed APOS
Stainless steel (kg)	585	Steel, chromium steel 18/8, hot rolled {RER} production APOS
Iron-nickel-chromium alloy (kg)	220	Iron-nickel-chromium alloy {RER} production APOS
Concrete fireproof (kg)	4480	Concrete block {DE} production APOS
insulating material (kg)	1220	Glass wool mat {CH} production APOS

Table 10. Materials of the gasifier including the primary gas treatment zone

5.2.2.3.1 Inventory for the SOFC stack

The production of the SOFC stack was modeled considering secondary data from scientific literature and, to a lower extent, from ecoinvent database⁷. The literature data was retrieved from studies where the SOFC stacks had a similar power capacity as the HBP technology. The amount of electricity, nickel oxide, solvents, materials for the binder, carbon black, and chromium steel, as well as direct emissions (released during the production of the stack) to the air of carbon dioxide, methyl ethyl ketone and benzyl alcohol were taken from [304] and adjusted proportionally

⁷ https://www.ecoinvent.org/

to the power capacity (factor of 0.793 based on 199 kWe of HBP SOFC versus 250 kWe of SOFC in [304]).

The data for the manufacturing of the anode, cathode, electrolyte, and the required ceramic materials (Lanthanum Strontium Manganite (LSM) and Yittria Stabilised Zirconia (YSZ)) were retrieved from [320].

The other secondary data for the SOFC stack, which were not available in [304,320], were retrieved from the already existing inventory in ecoinvent 3.4 called "Fuel cell, stack solid oxide, 125kW electrical, future {CH}| production APOS" and multiplied times 1.59 to account for the different size (assumption of linear proportionality of materials to the size as before).

For the production of the SOFC's BoP, data for the inputs of steel and energy were retrieved from [304]. The other data were instead retrieved from ecoinvent 3.4 dataset "Fuel cell, solid oxide, 125 kW electrical, future {CH} production | APOS", which was modified as well by multiplying times 1.59.

5.2.2.3.2 Inventory for the GCU

The materials for manufacturing the cage of the GCU were assumed to be similar to the ones of the cage of the external reformer of the SOFC provided in ecoinvent 3.4. The 96 filter candles which are present in the GCU system at the beginning of the operation were included within the manufacturing stage. These candles are made from calcium-magnesium-silicate high-temperature fiberglass. The processes "Calcium borates {GLO}| market for | APOS", "Magnesium {GLO}| market for | APOS" and "Silica sand {DE}| production| APOS" from ecoinvent 3.4 were used as a proxy for CaMgO₄Si. It was further assumed that 1.1 kg of material input would generate 1kg of filter candles. The mass of each candle was derived from the technical sheet of the candles [321].

5.2.2.4 Inventory data for operation and maintenance

The system operation includes all the material and energy inputs needed to operate the plant during one year of service (e.g. gas cleaning sorbents, water), waste outputs (e.g. ash which needs to be disposed of) and direct emissions to the environment (e.g. pollutant gas released to air). Wood chips

Pellets

The resulting direct emissions to air from the HBP are summarised in Table 11. Data for such emissions were only available for wood chips and wood pellets. The emissions from the operation with Miscanthus pellets were assumed to be the same as for wood pellets. Data on the ash formation (grate ash and fly ash) was retrieved from [301].

Maximum values shown	in the table	are used in the L	CI.		
Fuel	Direct emissions, in mg/MJ				
	со	OGC	TSP	NOx	

< 0.01

< 0.01

< 0.01

< 0.01

< 0.01

< 0.01

<20

<20

Table 11. Direct emissions (mg) to air per MJ of overall energy output (heat and electricity).

The operation of the gasifier needs 2.36 kg of natural gas for start-up operations and about 80.0 t of tap water per year for gasification air humidification (based on simulations from project data). According to measurements performed downstream the primary gas treatment zone, i.e. at GCU inlet, the syngas composition during utilisation of wood chips is as follows (in volume percentage): 15.4% CO, 10.6% CO, 1.8% CH, 8.3% H, 21.8% H,O, 41.1% N, During the multiple tests performed, such a composition showed to be stable. After the primary treatment unit, the syngas typically shows contaminant concentrations in the range of 30 ppm for sulphur and 20 ppm for chlorine (on wet basis) when wood chips are used as fuel. The tar concentration at the inlet of the GCU was lower than 2.0 g/Nm³ on dry basis and particulate matter contents (TSP) of about 200 g/Nm³ on dry basis have been determined.

For the operation of the GCU, about 1.2 t of zinc oxides per year are needed for H₂S removal. The GCU also requires 1200 Nm³ of Nitrogen per year for the cleaning of the filter elements. One year of operation of the GCU requires also 4800 kg of dolomite mixed with 900 kg of sodium bicarbonate⁸ as coating materials respectively for CI-sorption. The GCU has been designed to feed the SOFC with a product gas containing less than 5 ppm of chlorine, less than 1 ppm of sulphur, and less than 100 ppm of particulate matter (TPS < 0.1 mg/Nm³, on wet basis). Since the composition of the syngas is expected to be stable (confirmed also by the first test runs), the uncertainty about the simulated electric power output of the SOFC

⁸ From ecoinvent 3.4. Soda ash, dense {GLO}| market for | APOS

is expected to be very low.

The maintenance stage includes all the components which are replaced during the lifetime of the HBP plant. The SOFC stack and the GCU have a shorter lifetime than the average lifetime of the HBP CHP plant. Since the SOFC stack currently investigated for the HBP technology has an estimated lifetime of 5 years, the production of 1/5 extra SOFC stack per year was added to the maintenance stage. The GCU used for the HBP technology has a lifetime of 10 years, therefore, the production of 1/10 extra GCU per year was added to the maintenance stage. All other maintenance inputs (steel components and deionized water consumption for start-ups) for the SOFC were retrieved from [304] and scaled for the capacity of the SOFC under investigation. For the filter candles, an average of 30% of candles is estimated to be replaced each year of operation of the HBP technology. Therefore, the production of 29 extra candles per year was included in the maintenance stage.

5.2.2.5 Inventory data for end-of-life disposal

The main material employed in the components of the HBP is steel and can be recycled at the end of the life of each component. Based on the amount of steel present in the components (and their replacements), it was assumed that about 1900 kg of steel are recycled per average year of operation. The model included the energy for pressing and crushing the steel crap (based on [322]), a recycling efficiency (referred to as RRE in Equation 4) of 88% [323], and a transportation distance of 100 km from [322]. Such transportation was assumed to occur mainly by 16–32 t lorries Euro 3 [322].

A recycling process is a typical example of a multifunctional process fulfilling two functions i.e. the treatment of waste and the production of a recycled product. Based on our goal, the modeling approach (i.e. attributional), and the recommendations by ISO 14044:2006, mass allocation was applied⁹. The impacts arising from transportation (E_{τ}), recycling (E_{RC}) and the extraction and processing of the primary material (E_{ν}) were therefore allocated by mass between this life cycle and the following one (see equation 4 expressing the allocated impact to our functions). The resulting mass allocation factor (1/(1+RRE)) was 53% (1/1.88). The second part of equation 4 related to the virgin material takes into account the fact

⁹ ISO 14044:2006 prioritizes allocating by a physical property for open-loop recycling over the economic value or number of uses (among ISO third level allocations i.e. by other relationships). Additionally, system expansion cannot be applied since we want to isolate the first function (first use of the material which led to its treatment) from the second function (next use or cycle of the material).

that the primary production was already accounted entirely in the manufacturing phase, and therefore the corresponding burdens (e.g. extraction of raw material) that belong to the following life cycle should be subtracted.

$$E_{steel\ disposal} = RRE(E_T + E_{RC})\frac{1}{1+RRE} - \frac{RRE}{1+RRE} E_{v}$$
 Eq.4

There are some precious metals (e.g. used as catalytic materials) used in the power plant that, depending on the recovery efficiency and initial concentration, might be economically convenient to recover, though e.g. hydrometallurgical treatment [324]. Nevertheless, such specific recovery processes were not modeled because of the unavailability of LCI data. Materials other than the steel used in the power plant components consist of hazardous waste (24 kg) and inert waste (10t per year in the chips scenario, and 20t per year for the pellets scenarios). The treatment of the hazardous waste was modeled through the ecoinvent dataset *Hazardous waste, for underground deposit [DE]* treatment of hazardous waste, underground deposit [*DE*] treatment of materials for sorbents and was modeled through the ecoinvent 3.4 dataset *Inert waste, for final disposal [CH]* market for inert waste, for final disposal | APOS.

5.2.2.6 Inventory data for the competing technologies

For the comparative analysis, the ecoinvent 3.4 datasets "1 MJ Heat, district or industrial, other than natural gas {CH}| heat and power co-generation, wood chips, 2000 kW, state-of-the-art 2014 | APOS" and "1 kWh Electricity, high voltage {CH}| heat and power co-generation, wood chips, 2000 kW, state-of-the-art 2014 | APOS" were used for the ORC. This dataset represents a state of the art ORC co-generation plant equipped with an electrostatic precipitator for particulate emission reduction and includes the infrastructure. For the separate production of heat and electricity, the ecoinvent 3.4 datasets "1 MJ Heat, central or small-scale, natural gas {CH}| heat production, natural gas, at boiler condensing modulating <100kW | APOS" and "1 kWh Electricity, medium voltage {DE}| market for | APOS" were used.

Following the description provided in ecoinvent 3.4 for the ORC ecoinvent dataset, the capacity of the ORC plant is 1000 kW thermal, and 200 kW electric (similar to the electric output of the HBP technology). This information was used to estimate the exergy allocation factor of 46% for heat (assumed district heating provided at 90°C as for HBP) and 54% for electricity. Based on 2015-2017 average prices for

Germany, the economic allocation shares for ORC would be 66% for heat and 34% for electricity. Since the total power input (as wood chips) is 2000 kW, this ORC plant has an overall energy efficiency of 60%, i.e. 10% electrical efficiency plus 50% thermal efficiency.

5.3 RESULTS

5.3.1 Environmental impact of the HBP technology

5.3.1.1 Baseline (wood chips)

Figure 26 shows the breakdown of the impact of the HBP technology for the seven investigated impact categories (see appendix for absolute values).



Figure 26. Main contributions to the cradle-to-grave environmental impact of producing heat and electricity with the HBP technology using wood chips as biomass fuel. The presented breakdown is valid for both functional units. CC=Climate change, PM=Particulate matter, POF=Photochemical ozone formation, AC=Acidification, TE=Terrestrial eutrophication, WRD=Water resource depletion, MFRD=Mineral, fossil and renewable resource depletion.

The main contributions to the cradle to grave environmental impact are the wood chips used, followed by the maintenance phase and manufacturing phase.

The impact of wood chips is made of two components i.e. their transportation and their production. The impact of transporting wood chips (based on the Swiss supply chain assumed by the dataset retrieved from ecoinvent) represents 18-27% of the impact of wood chips for all impact categories, except for photochemical ozone formation (9%) and water depletion (2%). In all impact categories, except water depletion, the impact of the production of wood chips is mainly caused by the production and combustion of diesel and petrol (60-80%) used in power sawing machines, skidders and chippers. The production of the lubricants used in the three processes mentioned above causes about 2-10% of the impact of wood chips production in all impact categories except for water resource depletion for which it represents 80% of the impact. The water depletion impact of wood chips is mainly due to the fraction of vegetable oils used for lubricating the chains during power sawing activities (in absolute terms, this impact is quite low, see Figure 27 and Figure 31).

The main impact of the manufacturing stage is due to the production of the SOFC system, which contributes to 63%-100% of the impacts in this stage (depending on the category). Within the SOFC system, the production of the SOFC stack and the inverter. This is mainly due to the large electricity consumption during the manufacturing of the stack (as also highlighted by Rillo et al. [304]) and the manufacturing of chromium steel (mainly caused by the production of ferrochrome [304]).

Concerning the maintenance impacts, the maintenance of the SOFC system contributes to 63-100% of the impact depending on the impact category. The major contributor (95-99%) to the impact of the maintenance of the SOFC system is the replacement of the SOFC stack, which requires the production of a new SOFC stack every five years of operations

The operation phase is dominated by the operation of the GCU (mainly zinc oxide used and sodium bicarbonate) expect for water depletion whose impact is mainly caused by the water used for the operation of the gasifier. The contribution of direct emissions is negligible in all impact categories. The particulate matter caused by the operation of the HBP technology was only 1% of the total particulate matter impact.

5.3.1.2 Alternative scenarios (wood and Miscanthus pellets)

Figure 27 shows the environmental impact of the baseline scenario in comparison to the alternative scenarios.



Figure 27. HBP technology fuelled with various biomass fuels (the same graph applies to both 1 MJ heat or 1kWh electricity). Values are normalized taking as 100% the impacts of the most impacting scenario. CC=Climate change, PM=Particulate matter, POF=Photochemical ozone formation, AC=Acidification, TE=Terrestrial eutrophication, WRD=Water resource depletion, MFRD=Mineral, fossil and renewable resource depletion. WC=wood chips, WP=wood pellets, MP=Miscanthus pellets.

The results show that, in all impact categories, the total life cycle impact is the lowest for the operation with wood chips compared to the other two biomass scenarios (wood pellets and Miscanthus pellets). Since the inventories for manufacturing and maintenance are the same, the main difference between the three scenarios is the production of the biomass fuel (wood chips have lower environmental impacts than the two pellets).

The impact of the WC scenario is between 10% and 70% lower than for the WP scenario (with the highest impact difference for water depletion and particulate matter). For water depletion, the impact of wood pellets is almost entirely caused by the electricity consumption of the pellet factory. For particulate matter, the shaving process accounts for about 54% of the impact of producing wood pellets. Shaving is, therefore, the main cause of the significantly higher particulate matter impact in the production of wood pellets compared to wood chips. The impact of shaving is mainly caused by its drying process (87%), which leads to high particulate emissions due to the combustion of industrial wood.

Except for particulate matter and photochemical oxidant formation, the Miscanthus scenario presents higher environmental impacts than the wood pellets scenario. The characterized results indicate between 18% and 28% lower impacts for the wood pellets scenario than for the Miscanthus scenario in the categories of acidification, climate change, resource depletion (mineral, fossil and renewables) and terrestrial eutrophication. The particulate matter impact is lower (-27%) in the case of Miscanthus pellets because the shaving process -that was the main source of impact for wood pellets- is not used to produce Miscanthus pellets.

The difference in impact is even higher for the water depletion category, which scores 97% lower in the wood pellets scenario than in the Miscanthus pellets scenario. The irrigation needed during its cultivation is the main cause of the significantly higher water depletion in the scenario with Miscanthus pellets (see Figure 28). Other activities that are an important source of impacts for Miscanthus pellets are the electricity for pelleting, the diesel burnt during the harvesting stage and the emissions caused by fertilizing (see Figure 28 for the single contributions in each impact category).

Similar to the baseline case of wood chips, direct emissions have a negligible impact for the operation with wood pellets and Miscanthus pellets. This aspect is particularly important in the case of biomass technologies installed in heavily populated areas.

Since Miscanthus is an energy crop, it is important to assess the impacts due to land use. As for the other impact categories, the selection of the method was based on ILCD recommendations [145]. Accordingly, the carbon deficit caused by land use was assessed using the Soil Organic Matter model of [325]. This model accounts for the changes in soil quality caused by the occupation and transformation of the land. Land occupation generates changes in soil quality which depend on the amount of area occupied and the duration of such an occupation. Land transformation generates changes in soil quality which depend on the extent of changes in land properties and the area affected. In this model, the deficits in soil organic matter content are assessed and expressed by an indicator whose unit is kilograms of carbon deficit¹⁰.

¹⁰ These deficits are caused by the effects of agricultural practices on degradation rates. The changes can also be additions of soil organic matter. For example, these additions can be caused by the application of manure or crop residues. It should be observed that this modeling of land use impacts does not account for the counterfactual effects caused by land use changes modelled in consequential LCAs of bioenergy.

The production of 1 MJ of heat using Miscanthus pellets generates a 0.86 kg C deficit. Such an impact is much higher than for 1 MJ of heat generated using wood chips (0.12 kg C deficit) and using wood pellets (0.13 kg C deficit). The reason is that Miscanthus is an energy crop. Hence, differently from the feedstock for wood chips and pellets, it requires dedicated cultivation.



Figure 28. Main contributions to the environmental impact of Miscanthus pellets supplied to the HBP CHP plant. CC=Climate change, PM=Particulate matter, POF=Photochemical ozone formation, AC=Acidification, TE=Terrestrial eutrophication, WRD=Water resource depletion, MFRD=Mineral, fossil and renewable resource depletion

5.3.2 Benchmarking with competing technologies

5.3.2.1 Comparison with ORC technology (both fueled with wood chips)

Figure 29 shows the comparison between HBP technology and ORC technology both fueled with wood chips.



Figure 29. Comparison of HBP technology with ORC technology for 1 MJ of heat. The graph for 1kWh electricity shows some minor differences due to a slightly different Carnot factor assumed for ORC techology. Values are normalized with respect to the impacts of the most impacting scenario. CC=Climate change, PM=Particulate matter, POF=Photochemical ozone formation, AC=Acidification, TE=Terrestrial eutrophication, WRD=Water resource depletion, MFRD=Mineral, fossil and renewable resource depletion. WC=wood chips.

The heat co-generated by the HBP technology shows lower environmental impact compared to the same amount of heat produced by the ORC technology allowing -42% impact on climate and -87/-96% impact in terms of particulate matter, photochemical ozone formation, acidification and terrestrial eutrophication. These differences can be explained by two main advantages of the HBP technology: 1) the HBP has higher energy and exergy efficiencies, and therefore less biomass is needed for producing the same amount of energy and exergy as outputs and 2) HBP avoids the external combustion of biomass occurring in the ORCs, and therefore releases less particulate emissions (the particulate matter impact of the ORC technology is for 97% caused by direct emissions of particulates). Although the HBP technology has the same thermal efficiency as the ORC technology, its electric efficiency is three times higher. On the other hand, the HBP technology shows higher water depletion (+38%) and resource depletion (+79%). For water depletion, the high impact is caused by the replacements of the SOFC stack. For depletion of resources (minerals, fossil and renewables), the main cause can be found in the production of the SOFC stacks. All these components are not present in the case of an ORC.

Similar results were obtained when comparing electricity production from HBP and ORC. The HBP shows a lower impact of -45% for climate change, -96% for particulate matter, -91% for photochemical ozone formation, -88% acidification and -96% for terrestrial eutrophication. On the other hand, the HBP technology shows an increased impact in terms of water (+31%) and depletion of resources (MFRD) +70%.

5.3.2.2 Comparison with conventional production of heat and electricity

Figure 30 and Figure 31 shows the comparison between the HBP technology operating with the three investigated biomass fuels and conventional separate productions of heat and electricity.



Figure 30. Comparison of HBP technology with competing technologies (for 1 MJ heat). Values are normalized taking as 100% the impacts of the natural gas boiler. CC=Climate change, PM=Particulate matter, POF=Photochemical ozone formation, AC=Acidification, TE=Terrestrial eutrophication, WRD=Water resource depletion, MFRD=Mineral, fossil and renewable resource depletion. WC=wood chips, WP=wood pellets, MP=Miscanthus pellets.



Figure 31. Comparison of HPB technology with competing technologies (for 1 kWh electricity). Values are normalized taking as 100% the impacts of the German electricity mix. CC=Climate change, PM=Particulate matter, POF=Photochemical ozone formation, AC=Acidification, TE=Terrestrial eutrophication, WRD=Water resource depletion, MFRD=Mineral, fossil and renewable resource depletion. WC=wood chips, WP=wood pellets, MP=Miscanthus pellets.

The heat co-generated by the HBP technology shows a lower environmental impact compared to the heat produced by a condensing boiler burning natural gas. Even considering the least preferred fuel scenario in each impact category, the impact differences are at least -94% in terms of climate change, -70% in photochemical ozone formation, -37% in acidification, -43% in terrestrial eutrophication and -22% in depletion of resources. In particular, the significant difference in climate change is mainly generated by the biogenic carbon dioxide emissions (which are assumed to be carbon neutral) instead of fossil ones.

On the other hand, the HBP technology causes +28% impacts in particulate matter in the WP scenario (caused by the high particulate matter released when producing wood pellets) and significantly higher water depletion for the MP scenario (+13000%), due to the water used for irrigation in the cultivation of Miscanthus (the only scenario with irrigation). When wood chips are fed instead of pellets, the HBP shows a much lower impact in terms of particulate matter (-59%) but still a relatively higher impact in water depletion (+19%), due to indirect water consumption in different life cycle activities.

The electricity co-generated by HBP technology shows a lower environmental impact compared to electricity produced by the German electricity mix (EMIX). In particular, even considering the worst fuel scenario, the differences of impact are at least -86% for climate change, -43% for photochemical ozone formation, -56% for acidification and -63% for terrestrial eutrophication. Nevertheless, the HBP using Miscanthus pellets as fuel can lead to an increase in particulate matter (+7%) and water resource depletion (+146%; caused by irrigation of Miscanthus) and the same impact in terms of depletion of mineral and fossil resources. When operating with wood chips, the HBP shows a much lower impact than the EMIX, leading for example to -66% impacts in particulate matter, -98% in water depletion and -54% in depletion of resources (MFRD).

5.3.2.3 Comparing with other LCAs of SOFC CHPs

In the literature⁶, 8 LCAs of SOFC CHPs have been conducted along with a review of LCAs on SOFC systems. In most of these LCAs, the fuels used in the SOFC CHPs assessed were natural gas and biogas and the capacity of the SOFC was only a few kilowatts (1-20 kW) of electricity. Among these 8 LCAs, LCAs on SOFC CHPs of larger capacity (comparable to the one of the HBP) were conducted by [326–328]. Our results for the climate change impact of the HBP technology (0.03-0.09 kgCO2eq/ kWh_{el} depending on the fuel considered) indicate considerably lower impacts than for the SOFC CHPs assessed by these LCAs.

These lower impacts are especially found for the SOFC CHPs using natural gas as fuel because of the avoidance of direct emissions of fossil CO_2 allowed by the HBP which is fueled with a biofuel instead of fossil fuel. In particular, among the LCAs of SOFC CHPs whose size is comparable to the HBP and operating with natural gas, Strazza et al. [326] assessed a 230 kW_{el} SOFC CHP with electric efficiency of 53.4%. The resulting impact was 0.47 kgCO₂eq per kWh of electricity, which is at least 5 times higher than for the HBP. An older study [328] assessing a 125 kW_{el} SOFC CHP operating with natural gas, calculated an impact of 0.9-1.0 kgCO₂eq per kWh of electricity, which is at least 10 times higher than for the HBP. Staffell et. [320] assessed a 1 kW_{el} micro-SOFC CHP fueled with natural gas and calculated an impact of 0.32-0.37 kgCO₂eq/kWh_{el}, which is a least 3-4 times higher than for the HBP.

The impact of the HBP is also at least 44% lower than for SOFC CHPs operating with biogas. In this case, the main reason can be found in the different fuel production

processes and composition of the fuel used and consequent different composition of the direct emissions (e.g. methane emissions released with the exhaust gases). In particular, Strazza et al. [326] calculated an impact of 0.16 kgCO₂eq for a 230 kW_{el} SOFC CHP with 52.2% electric efficiency operating with biogas from sewage sludge. For the same type of system but with a capacity of 125 kW_{el}, Sadhukhan [327] calculated an impact of about 0.19 kgCO₂eq per kWh_{el}. Concerning this last figure, Sandhukhan used a different functional unit (1 ton of sewage sludge processed through anaerobic digestion) and we derived it by applying exergy allocation on the energy outputs.

Similarly to the HBP, for multiple impact categories, Strazza et al. found that the impact of this system, independently on the fuel considered (natural gas or biogas) was dominated by the production of the fuel. The only exception was the climate change impact of the operation with natural gas, whose impact was mainly caused by the operation phase (mainly direct emissions of fossil CO₂ of the system).

5.3.3 Alternative methods for solving multifunctionality

Exergy allocation was used to partition the total environmental impact between heat and electricity, as explained in section 5.2.2. In the literature, the two most applied alternative approaches to address the multifunctionality of SOFC CHPs are system expansion (enlargement) and economic allocation [306]. The first approach was only applied in studies where it was not necessary to differentiate between the impacts of heat and those of electricity.

Although the substitution method has a clear link with consequential analyses, it has been often applied in the literature for attributional LCAs with goals similar to the one of this study [57,71,329]. By the substitution method, the impact of the main product is obtained by subtracting the impact of the marginally avoided secondary products from the impacts of the overall system [51,213]. In particular, the main product is defined as the one providing the highest share of revenues within the analysed product system (physical/economical significance) [51].

A sensitivity analysis was performed to understand the influence of the method on the results of the study. This analysis explored the variation of the results, when applying economic allocation and the substitution method for the WC scenario, ORC scenario, and separate productions of heat and electricity. When applying substitution, the first step is identifying the main product. Based on the economic heat/electricity ratio (see Table 9), electricity is the main product for the HBP. The production of heat by the HBP technology can marginally avoid the production of heat from natural gas on the market (Heat, central or small-scale, natural gas {CH}| heat production, natural gas, at boiler condensing modulating <100kW | Conseq from ecoinvent 3.4).

On the other hand, the HBP heat could also avoid the production of heat by an average biomass boiler (Heat, district or industrial, other than natural gas {CH}| heat production, softwood chips from forest, at furnace 300kW | APOS from ecoinvent 3.4). The choice of a biomass boiler as substituted technology can be considered as an "alternative activity allocation" i.e. a form of "proxy-based disaggregation" [51]. This type of allocation is performed through the subtraction of impacts but differs from the substitution performed in consequential LCAs because it is not based on modeling of marginality [51]. Instead, this allocation takes as substituted processes the ones providing "primary productions of identical products and not of products that fall under different categories" [51]. This approach might, therefore, be an option also in attributional LCAs when reflecting the underlying physical relationship between the main and subsidiary products [51]. This sensitivity analysis considered both approaches, the substitution of a marginal activity (heat from a natural gas condensing boiler) and the substitution of an alternative activity (heat from a biomass boiler, marked as (a) in Figure 32 and Figure 33).

Based on the economic heat/electricity ratio of the ORC technology (see Table 9), heat is the main product for the ORC technology. In the case of the ORC, the electricity produced from the ORC avoids the production of marginal electricity from the electricity mix (this process is represented in the model by the ecoinvent dataset Electricity, high voltage {DE}| market for | Conseq).

The sensitivity analysis (see Figure 32 and Figure 33) indicated that, compared to exergy allocation, the economic allocation method apportions more impacts on heat (+70% in every category) while it decreases by 23% the impacts of electricity. The same applies to the ORC technology (+40% and -36% respectively for heat and electricity). For CHPs, it is therefore important to show the impacts for both heat and electricity when an allocation method is applied, so that a full picture of its environmental impacts is provided.



Figure 32. Sensitivity on allocation method for the generation of 1 MJ with HBP technology and competing technologies. Boiler running with natural gas taken as 100%. ORC=Organic Rankine Cycle, NG=Natural gas boiler, (a)= substitution of heat from a biomass boiler.

On the other hand, the conclusions of the comparative assessment did not change when applying exergy or economic allocation methods. This was true for all three comparisons: 1) between HBP with wood chips and ORC with wood chips, 2) between the three different biomass fuels scenarios and 3) between the HBP and the separate productions. For instance, the impact of the HBP per MJ of heat with both allocation methods was lower than for ORC in climate change, particulate matter, photochemical ozone formation, acidification and terrestrial eutrophication, but it was higher in the two categories concerning the depletion of resources (see Figure 32). On the other hand, the percentages of potential environmental impact savings or intensifications compared to separate production can change significantly. For example, for climate change, the savings of impact of the HBP compared to ORC was 42% when using exergy allocation while it was decreased to 30% with economic allocation. However, for particulate matter, there was no difference.



Figure 33. Sensitivity on allocation method for the generation of 1 kWh with HPB technology and competing technologies (1 kWh electricity). Electricity mix taken as 100%. ORC=Organic Rankine Cycle, EMIX=Electricity mix, (a)= substitution of heat from a biomass boiler.

Concerning substitution (see Figure 32 and Figure 33), the variations compared to other allocation methods were small or large depending on the impact category considered and the type of substitution applied. Moreover, both types of substitution approaches and the alternative activity method led to negative results in some impact categories. This last aspect highlights that the modeling was not consistent with the attributional goal of the study, which is not aimed at assessing a change in demand, and therefore, it should provide negative emissions for a single product of a multifunctional process whose overall impact is positive [51]. When a physically/economically significant product (the substituted function was 42% of total revenues for HBP and 34% for ORC) is substituted in attributional LCAs (by assuming that its impact corresponds to the one that would be replaced in the market), the results are often not aligned with other allocation methods and contrasts with the attributional aim of the LCA. This aspect emerges clearly when multiple impact categories are investigated in the same LCA study resulting in conclusions in contrast with other allocation methods and of difficult interpretation.

5.3.4 Sensitivity analysis on potentially sensitive parameters

5.3.4.1 Internal parameters

The results of the analysis indicated that the production of the biomass fuel (23-78% for the baseline scenario WC, depending on the category) and the SOFC stacks (10-43%) have a high contribution to the total impacts.

The ecoinvent dataset used for wood chips included both wood chips obtained as by-products of sawmill activities (15%) and from forest management (85%). To reduce the environmental impact, a scenario with only sawmill wood chips as fuel could be used. This type of wood chips presents a lower impact compared to wood chips from forest management because an important percentage of the impact of the upstream activities occurring in the forests is allocated to the main products of the sawmills. This scenario was assessed by sensitivity analysis to estimate the potential variation in the impact of the HBP (see the second column of Table 12). By using only sawmill wood chips, the environmental impact of HBP technology can be significantly reduced (indicatively by 10-40%).

Table 12. Sensitivity analysis on the reduction of the environmental impact of the HBP technology by either increasing the SOFC stack lifetime or using only wood chips produced as industrial by-products.

Impact category	Industrial wood chips only (measured in % variation)	Longer SOFC stack lifetime (measured in % variation)
Climate change (CC)	-37%	-4%
Particulate matter (PM)	-15%	-6%
Photochemical ozone formation (POF)	-47%	-2%
Acidification (AC)	-16%	-7%
Terrestrial eutrophication (TE)	-37%	-4%
Water resource depletion (WRD)	+8%	-10%
Mineral, fossil and renewable resource depletion (MFRD)	-10%	-3%

Alternatively, the impact of HBP technology could be improved by acting on the SOFC stack. Since the stack needs to be replaced every five years, the environmental impact could be improved by increasing the SOFC stack lifetime and therefore reducing the number of replacements over the plant lifetime. The second column

of Table 12 shows the reduction of the environmental impact that could potentially be achieved by increasing the lifetime of the SOFC from five to seven years. This would lead to a decrease between 2% and 10% of the impacts of the wood chips scenario (baseline).

5.3.4.2 External parameters

Since the technology will be deployed after 2025, it is important to explore how the comparative evaluation will change taking into account the current trends of decarbonization, which should lead to a decrease in the share of coal-produced electricity by shifting to renewables. In particular, the expected decarbonization of the European electricity grid will diminish the environmental benefits of HBP technology.

Impact category	EMIX Germany, ecoinvent 3.4		Future EMIX Germany, EU reference scenario 2030		Future EMIX EU, IEA current policy scenario 2030	
	1 kWh electricity	Savings (%) HBP vs EMIX	1 kWh electricity	Savings (%) HBP vs EMIX	1 kWh electricity	Savings (%) HBP vs EMIX
Climate change (kg CO ₂ eq)	6.41E-01	-95%	5.61E-01	-95%	3.96E-01	-92%
Particulate matter (kg PM2.5 eq)	7.87E-05	-66%	7.84E-05	-66%	6.74E-05	-60%
Photochemical ozone formation (kg NMVOC eq)	6.02E-04	-49%	6.06E-04	-50%	4.98E-04	-39%
Acidification (molc H+ eq)	1.58E-03	-80%	1.35E-03	-77%	1.12E-03	-72%
Terrestrial eutrophication (molc N eq)	4.33E-03	-89%	3.86E-03	-87%	3.49E-03	-86%
Water resource depletion (m ³ water eq)	2.75E-03	-98%	1.32E-04	-54%	4.47E-04	-87%

Table 13. Sensitivity analysis on the savings of environmental impacts of 1 kWh of electricity produced by the HBP compared to the grid electricity mix (EMIX).

Impact category	EMIX Germany, ecoinvent 3.4		Future EMIX Germany, EU reference scenario 2030		Future EMIX EU, IEA current policy scenario 2030	
	1 kWh electricity	Savings (%) HBP vs EMIX	1 kWh electricity	Savings (%) HBP vs EMIX	1 kWh electricity	Savings (%) HBP vs EMIX
Mineral, fossil and renewable resource depletion (kg Sb eq)	7.47E-06	-54%	7.89E-06	-56%	7.41E-06	-54%

To assess this variation, the electricity mix based on two future scenarios for 2030 were considered: the EU reference scenario for Germany [272] and the IEA current policy scenario [330]. Due to the unavailability of the IEA current policy scenario for Germany, the IEA average mix of 2030 for the EU was taken as a proxy. This second scenario represents a more decarbonized electricity sector and includes other countries where the HBP could be commercialized. In particular, the IEA current policy scenario has only 13.7% coal and 44.7% renewables. The future savings of environmental impact allowed by the HBP is shown in the two columns on the right in Table 13. The environmental savings from the HBP technology will be only slightly affected (order of 5% overall) by the change expected in the electricity mix for 2030.

5.4 CONCLUSIONS

This article presented the first life cycle assessment (LCA) of a novel technology integrating biomass gasification and SOFC technologies. This technology is currently under development in the H2020 HiEff-BioPower (HBP) project and allows for the use of various biomass types as feedstock. This LCA assessed the environmental impacts when operating the technology with three different fuels: wood chips, wood pellets and Miscanthus pellets. The impact of producing heat and electricity with the HBP technology was compared to the state of the art competing technologies. The results showed that most of the impacts of producing heat and electricity with the HBP technology are generated during the production (including transportation) of the biomass fuels (between 23% and 99%).

of the total impacts depending on the category and the fuel). The use of wood chips as fuel generates much lower impacts per functional unit than the operation with wood pellets (11-70% lower) and Miscanthus pellets (9-99% lower), in all impact categories. The next highest contributor to the life cycle environmental impacts is the SOFC stack, due to both the high energy intensity (especially in electricity consumption) and material intensities of its manufacturing processes, and its short lifetime (the stack should be replaced every 5 years). Beyond increasing the fuel efficiency of the technology and therefore reducing the consumption of biomass fuels, the main recommendation to technology developers would be to increase the lifetime of the SOFC stack. Increasing the SOFC stack lifetime could decrease the environmental impacts of 2-10%, depending on the category.

The comparison of the HBP technology with separate productions of heat and electricity (from natural gas condensing boilers and the German electricity grid) indicated significantly lower impacts for the HBP technology, especially in climate change (86/94% lower), photochemical ozone formation (-43/-70%), acidification (-37/-56%) and terrestrial eutrophication (-43/-63%). Overall, HBP showed also better performance than ORCs, since they have higher exergy efficiencies and almost zero particulate emissions resulting in 86-96% lower impact in the category particulate matter.

The sensitivity analysis on the allocation method for heat and electricity provided useful insights for the choice of allocation methods in CHP plants, and led to the following recommendations: 1) the attributional LCAs of CHPs should always provide the results for both heat and electricity to allow for better interpretation of results, independently of the allocation method, 2) LCA results from different CHP plants should not be compared if they assumed different allocation approaches and 3) substitution is not recommended in attributional LCA (especially if the substituted product is not a minor by-product) because it provides results which are not in line with the attributional aim (e.g. negative emissions) and lead to conclusions in contracts with the ones from applying allocation methods which are proven to be a good proxy of physical causality for CHPs and therefore preferable.

5.A APPENDIX

Table 5A1 presents the characterized environmental impacts per functional unit of heat and electricity for each fuel scenario.

Table 5A1. Cradle-to-grave environmental impacts per functional unit and biomass fuel. WC=wood chips, WP=wood pellets, MP=Miscanthus pellets.

	1 MJ of Heat			1 kWh of electricity		
Impact Category	Wood chips	Wood pellets	Miscanthus pellets	Wood chips	Wood pellets	Miscanthus pellets
Climate change (kg CO ₂ eq)	1.52E-03	3.42E-03	4.36E-03	3.05E-02	6.87E-02	8.77E-02
Particulate matter (kg PM2.5 eq)	1.34E-06	4.19E-06	3.07E-06	2.70E-05	8.41E-05	6.17E-05
Photochemical ozone formation (kg NMVOC eq)	1.51E-05	1.71E-05	1.66E-05	3.04E-04	3.43E-04	3.33E-04
Acidification (molc H+ eq)	1.57E-05	2.87E-05	3.49E-05	3.15E-04	5.78E-04	7.02E-04
Terrestrial eutrophication (molc N eq)	2.46E-05	5.72E-05	7.96E-05	4.93E-04	1.15E-03	1.60E-03
Water resource depletion (m ³ water eq)	3.01E-06	1.02E-05	3.36E-04	6.03E-05	2.04E-04	6.76E-03
Mineral, fossil and renewable resource depletion (kg Sb eq)	1.71E-07	2.80E-07	3.71E-07	3.43E-06	5.63E-06	7.46E-06



6

CHAPTER

Attributional and consequential LCAs of a novel bio-jet fuel from Dutch potato by-products

Published as:

C. Moretti, I. Vera, M. Junginger, A. López-Contreras & L. Shen. Attributional and consequential LCAs of a novel bio-jet fuel from Dutch potato by-products.

Sci. Total Environ (2022). 813 (2022). 10.1016/j. scitotenv.2021.152505

ABSTRACT

To mitigate the climate change impact of aviation, jet fuels from bio-based byproducts are considered a promising alternative to conventional jet fuels. Life cycle assessment (LCA) is a commonly applied tool to determine the environmental impacts of bio-jet fuels. This article presents both attributional and consequential LCA models to assess an innovative bio-jet fuel produced from potato by-products in the Netherlands. The two models led to opposite conclusions regarding the overall environmental performance of this bio-jet fuel. The attributional LCA showed that this bio-jet fuel could offer about a 60% GHG emissions reduction compared to conventional jet fuel. In comparison, the consequential LCA estimated either a much lower climate change benefit (5-40%) if the potato by-products taken from the animal feed market are replaced with European animal feed or a 70% increase in GHG emissions if also imported soybean meals are used to replace the feed. Contrasting conclusions were also obtained for photochemical ozone formation. Conversely, the attributional and consequential LCAs agree on acidification, terrestrial eutrophication and depletion of fossil fuels. Although the consequential LCA was affected by higher uncertainties related to the determination of the actual product displaced, it allowed understanding the consequence of additional animal feed production. This process was not included in the system boundaries of the attributional LCA.

6.1 INTRODUCTION

The substitution of petrochemical fuels with low-carbon fuels is crucial for reducing anthropogenic greenhouse gas (GHG) emissions. Although the direct emissions from the aviation sector were responsible for only 2% of pre-covid-19 worldwide GHG emissions [331], the sector is expected to continue growing [332]. The role of biomass for the aviation sector is essential as there are limited options for decarbonization, especially in the next twenty years [333,334]. Unlike road transport, it is challenging to equip aircrafts with electric-powered engines for long distances [334]. Other alternatives such as hydrogen or other fuels based on renewable electricity and CO₂ are at the early stages of development [335]. Therefore, the decarbonization of aviation is still far away from practical implementation.

In the short term, the most promising option for GHG emissions reduction is using drop-in jet fuels from sustainable biomass feedstocks [333,334]. Bio-jet fuels can be blended with petroleum fuels and used in existing engines without modifications [47]. Despite that, to date, hydroprocessed esters and fatty acids (HEFA) is the only commercial technology available for bio-jet fuels [333]. However, HEFA bio-jet fuel production volumes are still considerably limited, accounting for less than 1% of total jet fuels worldwide [336]. The high production costs (2 to 6 times higher than conventional jet fuels) and strict sustainability requirements to be incentivized (60-65% GHG emissions' savings depending on the country) result in unfavorable industry development conditions (de Jong et al., 2015; IRENA, 2017; O'Connell et al., 2019). Currently, used cooking oil is the only alternative applied on industrial levels that is near cost-competitive and delivers low life cycle GHG emissions (O'Connell et al., 2019; Pavlenko et al., 2019). Nevertheless, the availability of used cooking oil is limited. Used cooking oil is also demanded for road and marine transport fuels and chemicals [128,341,342].

Many emerging bio-jet fuels have been recently certified by the American Society for Testing and Materials [343]. Among them, Fischer-Tropsch (FT) and alcohol-to-jet (ATJ) pathways are the production routes closest to commercialization [334].

For policy decision-making, it is crucial to assess the potential environmental benefits/impacts of bio-jet fuels compared to petrochemical jet fuels. Generally, these assessments are carried out using Life Cycle Assessment (LCA), a method for environmental assessments standardized by ISO 14041 and ISO 14040 [38,127]. This article investigates the environmental performance of an innovative ATJ

route developed in the Netherlands to use local potato by-products from the food processing industry. This abundant low-price carbohydrate by-product is currently used (mainly) as animal feed but can be potentially transformed into valuable biobased products such as biofuels, materials and chemicals [344–346].

Despite being a standardized approach, the LCA results can be affected by significant variability. The option of different methodological assumptions can steer the life-cycle environmental impact of a process/product, even when assessing the same fuel and feedstock [47,48]. A recent review of LCAs for HEFA biofuels [47] identified the method used to deal with multifunctional processes and the inclusion/exclusion of land-use change (LUC) emissions as the most important sources of variability. For example, the GHG emission intensity of a bio-jet fuel could increase by 2-3 times using either a different allocation method or making a different assumption for LUC [47]. Similar findings for bio-jet fuel pathways were also reported by Capaz et al. (2020). However, note that these issues also apply to other biofuels in general [45,46].

Multifunctionality practices and the inclusion of land-use changes depend on the specific LCA's goal and scope and consequently, the choice of modeling approach [45,347]. In the literature, two main modeling approaches are distinguished: attributional and consequential LCAs [244]. The attributional approach attempts to quantify the portion of global burdens associated with the specific products under assessment [59]. The environmental impact of such products is determined by analyzing the production system using representative average data [51,68]. Via attributional LCAs (ALCAs), the environmental impacts of each co-product are obtained by distributing the burden based on allocation parameters such as energy or market values [57,68,128]. In this way, the so-called additivity principle of ALCAs is respected (i.e. the sum of the attributional LCAs of all worldwide products corresponds with the total environmental burdens worldwide [59]. In attributional LCAs, looking at the status-quo and not to what has happened in the past or the future, LUC is generally not included. However, when LUC is included, only direct LUC is addressed and indirect LUC is not considered [45,63,64]. Consequential LCAs (CLCAs) focus on modeling the relative changes in the entire techno-sphere when the decision supported by the LCA is adopted. Hence, the CLCA provides information on both direct and indirect environmental burdens that occur due to the changes in demand for a product caused by such a decision [59]. Therefore, a CLCA allows assessing all the causal-effect relations within the market by changing product demand using marginal data [48,51,177]. In CLCAs, multifunctionality is addressed by a substitution approach (or often referred to as "displacement method"), and both direct and indirect LUC are included in the life cycle inventory [45,63,64].

While there is plenty of methodological literature regarding the effects of the modeling approach on the LCA result, the practice of applying both modeling approaches to the same product system before drawing recommendations about a product system is rare. No more than 8% of the peer-reviewed LCAs on bioenergy products in the Scopus database applied both modeling approaches to their case study [146]. Only 3% of peer-reviewed LCAs on bioenergy products declared to adopt a consequential modeling and the remaining studies used an attributional or unspecified modeling approach [146]. A recent review of the 100 most cited LCAs of bioenergy products highlighted that ambiguous results in these LCAs are mainly due to choices in the inventory modeling approach that are inconsistent with the goal of the study [94].

To our knowledge, there is only one peer-reviewed LCA applying both modeling approaches to a bio-jet fuel [348]. In their study, the CLCA resulted in an environmental benefit mainly due to the substitution of the power surplus. Their ALCA using allocation instead of substitution could not confirm such a benefit. While there is only one study applying both modeling approaches to a bio-jet fuel, there are several comparisons between the two modeling approaches for the case of animal feed [89,237]. Among them, a recent study by van Zanten et al. (2018) applied attributional and consequential modeling approaches to understand the environmental advantages of replacing soybean meal with an alternative protein source (peas or rapeseed meal) in pig diets. While their ALCA concluded that this practice could lead to environmental advantages, the CLCA concluded the opposite [349].

Given the lessons learned by the abovementioned literature regarding the effects of the modeling approach on LCA results, this study presents both attributional and consequential LCAs of this bio-jet fuel. In particular, the attributional LCA aimed to investigate the environmental impacts caused by the production of this jet fuel starting from the total environmental impact of the production system (i.e. the full pie). The attributional LCA of the bio-jet fuel attributes a piece of the pie to the bio-jet fuel. Technically, other practitioners could have a different aim leading to investigate one of the co-product of the same production system. These attributional LCAs attribute the remaining shares of the pie to the other system's co-products. The sum of each functional output's environmental impact corresponds to the total impact of the product system (additivity of these ALCAs). Hence, the ALCA modeling serves to answer the research question: how much each of the functions, as entailed in the different product outputs, is responsible for the production system's total environmental impact? The company producing such bio-based products will be interested in consulting these LCAs. The results of these ALCAs allow understanding the environmental performance of the entire spectrum of bio-based products produced, usually benchmarking with their petrochemical counterparts, as well as possible process improvements.

While attributional modeling has a producer perspective, consequential modeling has a policy perspective instead. CLCA aimed to answer the following research question: how could introducing the new bio-jet fuel to the market potentially change the overall environmental impact of the supply chain of bio-jet fuel production and the economic sectors affected by the change? For example, the consequential LCA in our study aimed to understand the environmental impact of this novel bio-jet fuel at the net of the effects of displacing the potato by-products from the animal feed market.

6.2. MATERIALS AND METHODS

6.2.1. Goal and scope definition

Both attributional and consequential LCAs aimed to assess the environmental impact of a novel bio-jet fuel. Based on this goal, a functional unit of 1 MJ of bio-jet fuel was defined. The results of the two LCAs were compared with the environmental impact of conventional jet fuel (assumed kerosene). Such a comparison was used to evaluate if the two modeling approaches lead to similar findings regarding the environmental performance of this novel bio-jet fuel. This exercise aimed not only to provide a comprehensive picture of environmental impacts of the bio-jet fuel using a food processing by-product, but also to join the current LCA debate about the influence of the type of modeling approach used on the environmental performance of bio-based fuels and materials using by-products with low economic significance or waste as resources [45,146]. Hence, the intended audience of this LCA includes both technology developers and the LCA community.

Since this novel bio-jet fuel has been developed in the Netherlands from a local feedstock, the geographic scope of the LCA is the Netherlands. The temporal scope is the year 2030, when this fuel could be commercialized on a large scale.

A cradle-to-grave scope was adopted in the two LCAs of the bio-jet fuel. The respective process flow diagrams are shown in Figure 14. For both ALCA and CLCA, the common unit processes included are: acetone-ethanol-butanol (ABE) fermentation, swing adsorption, alcoholic condensation, hydrotreatment, distribution and combustion. The production of the aircrafts and the biorefinery plant and their decommissioning were neglected in both models. The exclusion of capital goods (including infrastructures) and their end of life is common practice for LCAs of biofuels in the EU. In fact, the well-to-wheel approach, which is the main tool adopted for the EU biofuels, excludes the production of the vehicles and plants and their decommissioning from the LCA scope [9,37]. The reason is that the contribution of capital goods on the total life cycle of fuels is expected to be small and very similar for biofuels and conventional fuels [36,350]. The exclusion of capital goods from this LCA is also in line with the update of the EU Product Environmental Footprint (PEF) recommending the exclusion of capital goods from the LCA "unless there is evidence from previous studies that they are relevant" [351].

In the attributional LCA, the flow diagram starts with potato cultivation and the potato processing industry generating the potato by-products. In the CLCA modeling, the unit processes representing the production of the feedstock (cultivation and potato industry) are not included because they are not affected by the change that the CLCA aims to assess. In fact, if the potato by-products were not used for this application, they would be produced anyway. The use of potato by-products for this application changes neither the raw potatoes inputs nor the utilities utilized by the factory per tonne of the main food product. Hence, introducing this bio-jet fuel as a substitute for petrochemical kerosene (i.e. the change assessed) does not affect these processes.

In the Netherlands, the potato by-products that can be used for this fuel are made of potato peels (80%), grey starch (15%) and press-pieces (5%) [347]. Most potato byproducts (potato peels and potato pieces) are currently used in animal feed. Grey starch might be used for other purposes [352]. Since grey starch represents only 15% of potato by-products, we can assume that taking away the potato by-products used as feed affects mainly the animal feed supply. If the animal feed demand is to
be maintained, additional feed needs to be supplied from other sources. For this reason, the flow diagram of the CLCA includes the unit process named "marginal animal feed production", representing the production of such additional animal feed from other sources.

Furthermore, attributional modeling was also used to determine the environmental impact of the co-products in the value chain, namely animal feed (from fermentation residue), bio-based hydrogen, bio-based carbon dioxide and biolubricants. In this way, it is possible to transparently present how the environmental impact of each process contributes (i.e. is allocated) to the environmental impact of each function delivered by the system. The impacts between the main product (jet fuel) and these by-products are partitioned by physical or economic relations. The allocation choices between the co-products can be found in the next section. For CLCA, system expansion is applied to these by-products. The by-products are assumed to replace the marginal productions of the same or similar products (see Section 6.2.2); therefore, the environmental impacts of the individual co-product cannot be quantified.

In the ALCAs, the environmental impacts of each co-products were compared with their current market benchmarks. This comparison is in the interest of the producer of such bio-based products. The company wants to avoid burden-shifting between their co-products i.e. producing one bio-based product with a good environmental performance at the expense of a bad environmental performance for the other bio-based products produced. In case of opposite conclusions between the two LCAs of the bio-jet fuel for certain impacts, it is possible to use the results of the other ALCAs to immediately understand if the main reason is linked to the allocation method. For example, we could notice if most of the environmental impact in that category is caused by a certain unit process and was allocated mainly to a co-product instead of the bio-jet fuel. If this is not the case, we can exclude this effect of allocation and look for other reasons not linked with the allocation applied in the ALCA. For example, a reason can be the inclusion of a different process in the system boundaries (e.g. animal feed displacement effect) or marginal versus average data.

The selection of the market benchmarks for the ALCAs and the substituted products in the CLCA can be found in sections 6.2.2.5 and 6.2.2.7).

For the attributional LCAs of the bio-based co-products of the jet-fuel, the functional units were defined as 1 kg of each product, i.e. 1 kg of animal feed or 1 kg of hydrogen or 1 kg of carbon dioxide or 1 kg of lubricants (see Table 16 and Table 17 for the actual production ratios). Since these co-products are delivered at the bio-jet fuel production process gate and are usually not combusted, their LCA follows a cradle-to-gate scope. Given the additivity principle of ALCA, summing the environmental impact of the functional units of all ALCAs conducted leads to the total environmental impact of the system shown in Figure 34 on the left.



Figure 34. Cradle-to-grave process flow diagram detailing inputs and co-products for attributional (left) and consequential (right) LCAs of the investigated bio-jet fuel.

Five midpoint impact categories were selected (see Table 14). Besides climate change and depletion of fossil fuels, which are priority indicators in the current environmental decision-making [308], other four impact categories were considered. Among them, photochemical ozone formation, acidification and terrestrial eutrophication are recognized as key environmental problems linked to

nitrogen oxides in the Netherlands [353]. These three categories are acknowledged as important impacts for bio-based products' policy and investment decisions. Previous work showed that a fair comparison with petrochemical products with reasonable uncertainties is possible for these five impact categories [43,354]. As for the sustainability assessment of bio-based fuels used in the European Union [9,37,350], the direct biogenic carbon dioxide emitted to the atmosphere from the combustion biofuels was considered carbon neutral. Based on the European legislation for biofuels and supporting documents (mentioned above), the only exception regards the consequences of land use changes on the soil organic carbon (biogenic). Accordingly, the carbon removal from the original biomass matter was neglected except for net effects related to LUC (for details regarding the data used for LUC modeling, please refer to section 6.2.2.6.

Impact Category	Unit	Impact assessment models
Climate change	kg CO ₂ eq	[310]
Photochemical ozone formation	kg NMVOC eq	[312]
Acidification	molc H+ eq	[313]
Terrestrial eutrophication	mol N eq	[355]
Depletion of fossil fuels	МЈ	[315]

Table 14. Environmental impact categories and impact assessment models.

6.2.2. Life cycle inventory

6.2.2.1 Feedstock production (ALCA) and transportation

Potato by-products are a low economic significance by-product of the potato processing industry. In the flow diagram of the ALCA (see Figure 34), the product system starts with the cultivation of potatoes. After the raw potatoes are harvested, they are sent to the potato industry to be processed into food products. The main inputs to the potato industry are electricity and heat. The quantities are reported in

Table 15. Economic allocation was applied to the potato industry unit process (for details, see section 6.2.2.4).

The dewatered potato by-products are then transported by lorry (16-32 t) to the refinery, which is assumed in Rotterdam. Based on the location of potato processing factories in the Netherlands, the potato by-products need to be transported on average for 105 km [347].

Table 15. Mass and energy inputs of the potato industry per tonne of potato by-products and background data sources for these inputs. Processed potatoes output based on confidential data from the industry and not disclosed.

Activity level data	Quantity per tonne of potato byproduct	Foreground data sources	Background data
Inputs			
Raw potatoes, wet (t)	4.6	[356]	Agri-footprint 5.0 (Potatoes, market mix, at regional storage/NL) This dataset already includes the average transportation of potatoes in the Netherlands.
Heat (GJ)	12.4	[356]	Heat, district or industrial, natural gas {Europe without Switzerland} heat production, natural gas, at boiler condensing modulating >100kW APOS, from ecoinvent 3.6.
Electricity (kWh)	558.9	[356]	For electricity, the dataset Electricity, medium voltage {RER} market group for APOS from ecoinvent 3.6 was used updating the shares of electricity by fuel mix based on the 2030 EU reference scenario for the Netherlands [272].

6.2.2.2 From potato by-product to bio-jet fuel

At the refinery location, the potato by-products are processed via ABE fermentation, which produces a mix of valuable alcohol/ketone made of butanol and acetone with a minor fraction of ethanol. Table 16 reports on the ABE fermentation process's inputs and output, which uses pervaporation for in-situ butanol recovery, increasing the ABE yield [357,358].

The mix of ABE alcohols is then processed through alcohol condensation [360], avoiding an important part of energy consumption to separate the alcohols and acetone to purity [357,361]. The water produced by alcohol condensation is recycled back to a second pervaporation membrane for water separation, which needs to be treated before being discharged to the environment. In parallel, the fermentation gases (a mix of carbon dioxide and hydrogen) are separated via swing adsorption, with 344 kWh of electricity consumed per t of carbon dioxide separated [362]. The alcoholic condensate is then further deoxygenized via hydrotreatment using part of the separated bio-hydrogen [347,360]. Table 17 reports on the inputs and outputs of the thermochemical upgrading to bio-jet fuel.

Flow	Amount	Comment/ background data
Inputs		
Potato by-products (t) wet	46.6	As the output from the potato processing industry. Corrected based on dry base content for the data source [356].
Enzymes (kg)	2.2	Electricity (1.9 kWh per kg of enzyme) and steam (4 MJ per kg of enzyme) to produce -amylase enzymes based on [359]. For electricity in the attributional modeling, the dataset Electricity, medium voltage {RER}] market group for APOS from ecoinvent 3.6 was used to update the shares of electricity per source based on the 2030 EU reference scenario for the Netherlands [272]. For electricity in the consequential modeling, Electricity, medium voltage {NL} market for Conseq was used. For steam, the dataset Heat, from steam, in chemical industry {RER} market for heat, from steam, in chemical industry APOS or Conseq from ecoinvent 3.6 was used.
Potassium hydroxide (t)	0.1	Potassium hydroxide {RER} production APOS or Conseq from ecoinvent 3.6.
Electricity (kWh)	1000	For electricity in the attributional modeling, the dataset Electricity, medium voltage {RER} market group for APOS from ecoinvent 3.6 was used to update the shares of electricity per source based on the 2030 EU reference scenario for the Netherlands [272]. For electricity in the consequential modeling, Electricity, medium voltage {NL} market for Conseq was used. The marginal electricity dataset, which was based on 2014 data, was not modified since marginal technologies are more stable in time than data based on average technologies [211].
Nitrogen (kg)	2.8	Nitrogen, liquid {RER} market for APOS or Conseq from ecoinvent 3.6
Co-outputs (Details	about alloca	ation are reported in section 6.2.2.4)
Organic residue (t)	3.5	Sold as animal feed.
Part of fermentation gases made of hydrogen (t)	0.05	Partly used for hydrotreatment while the surplus is sold.
Part of fermentation gases made of biogenic carbon dioxide (t)	1.6	Sold.

Table 16. Mass and energy inputs and outputs of ABE fermentation (including in-situ recovery pervaporation) per t pure ABE from [347] and background data sources for the LCA.

Table 17. Mass and energy inputs and outputs of the thermochemical upgrading (alcohol condensation plus hydrotreatment) per t of bio-jet fuel from (Moretti et al., 2021) and background data sources for the LCA.

Flow	Data	Background data
Inputs		
Pure ABE	1.5 t	
Heat	4.6 GJ	Heat, district or industrial, natural gas {RER} market group for APOS or Conseq from ecoinvent 3.6.
Cooling energy	1.7 GJ	Cooling energy {CH} from natural gas, at cogen unit with absorption chiller 100kW APOS or Conseq from ecoinvent 3.6.
Bio-hydrogen	36.5 kg	From swing adsorption.
Co-outputs		
Lubricants	41.7 kg	Mass allocation. Details in section 6.2.2.4.
Wastes		
Wastewater	0.6 t	Wastewater, average {Europe without Switzerland} market for wastewater, average APOS from ecoinvent 3.6

6.2.2.3 Distribution and combustion

For the distribution of the bio-jet fuel and petrochemical kerosene to the "tanks", a distance of 65 km was assumed, representing the transportation from Rotterdam to Amsterdam. As background data, the process Transport, freight, lorry >32 metric ton, EURO4 {RER}| transport, freight, lorry >32 metric ton, EURO4| APOS or Conseq from ecoinvent 3.6 was used. The technology under assessment is similar to the ATJ-synthetic paraffinic kerosene pathway where isobutanol or ethanol are upgraded to jet fuel via alcohol dehydration and condensation reaction followed by hydrotreatment [363]. For this reason, the combustion emissions for the bio-jet fuel were assumed to be the same as ASTM-certified synthetic paraffinic kerosene. The carbon dioxide from the bio-jet fuel is biogenic and was assumed carbon neutral as currently assumed in biofuels' sustainability calculations according to the EU legislation [9,37]. The inventory for distribution and combustion can be found in Table 18. The lower heating value (LHV) of the bio-jet fuel was assumed to be 44.1 MJ/kg, which is typical of synthetic paraffinic kerosene [364]. An LHV of 43.2 MJ/ kg was assumed for petrochemical kerosene [364]. Combustion emissions were based on a single-aisle passenger aircraft operating on an average distance and with an average payload for both jet fuels. With these assumptions, the so-called payload fuel energy intensity is 8.62 kJ/kg_{payload}-km_{great-circle distance} for the bio-jet fuel [364] and 8.65 kJ/kg_{pavload}-km_{oreat-circle distance} for conventional jet fuel.

Emissions to air	Data (grams/kg) for bio-jet fuel	Data (grams/kg) for petrochemical kerosene
CO ₂	3100	3151
СО	4.0	4.0
N ₂ O	0.01	0.01
NO _x	14.5	15.5
SO _x	0	1.4
CH	0.005	0.005

Table 18. Inventory of combustion emissions per kg of jet fuel retrieved from [364]

6.2.2.4 Multifunctionality in the attributional model

In attributional modeling, the allocation method was based on each unit process's "causality mechanism" as recommended by ISO 14044:2006. The purpose of the potato industry is to produce revenues selling potato food products. Hence, an economic allocation was used. Based on [356], approximately 1% of the environmental impact of the potato industry was allocated to the potato by-products (based on the five-year average price).

For the ABE fermentation process, energy allocation was applied because the process is driven to produce an energy product, even though the fermentation residue used for animal feed has a higher mass content compared to the alcohols. The following allocation factors were retrieved from [347]: 34.4% to ABE, 6% to the fermentation gases and 59.6% to the animal feed.

The swing adsorption process aims to remove the fermentation gases continuously and separate carbon dioxide and hydrogen. Between these two co-products, carbon dioxide is more significant from both economic and mass perspectives. Given that this unit process's main product is not an energy product, economic allocation was also applied for this multi-output process. As a result, allocation shares of 59.4% and 40.6% were calculated respectively for carbon dioxide and hydrogen [347].

Together with the bio-jet fuel, the hydrotreatment process also delivers a minor coproduct (i.e. lubricants). The goal of the hydrotreatment process is to generate biojet fuel (energy). Energy allocation should be applied to the hydrotreatment process. Since the lower heating value of the lubricants is unknown, a mass allocation was used as a proxy for energy allocation resulting in 4% of the environmental impact allocated to lubricants. Since lubricants are a minor product, the allocated environmental impact to each t of fuel is only slightly affected by the applied allocation method. Conversely, the impact of the lubricants is more affected by the allocation method based on how different is the actual lower heating value of the lubricants compared to the one of the bio-jet fuel.

6.2.2.5 Multifunctionaility in the consequential model

6.2.2.5.1 By- and co-products

The consequential modeling included avoiding the production of all bio-jet fuel's co-products in the system boundaries through the so-called system expansion by substituting marginal production. For most commodity products, the marginal production data were taken from the Ecoinvent Consequential datasets. Lubricants were substituted with Lubricating oil {RER}| production | Conseq. The surplus of hydrogen was substituted with Hydrogen (reformer) E from PlasticsEurope since nowadays hydrogen production mainly comes from steam reforming of natural gas [365]. The current market demand for carbon dioxide is mainly for the promotion of plant growth, the creation of inert environments or as a heat transfer medium, as a refrigerant, or as a chemical for the production of a variety of other chemicals [366,367]. For these markets, commercial carbon dioxide is obtained as a "waste gas" from ammonia and hydrogen production processes [366,367]. In the ammonia production process, an intermediate mix of gases made of hydrogen, nitrogen and carbon dioxide is produced. However, only the first two gases are required to make ammonia. So, carbon dioxide needs to be removed from the gas stream. A gas with a similar composition that needs to be separated is also generated in hydrogen production from reforming natural gas.

The impact associated with CO₂ production is led by extraction and purification of the abovementioned "waste" gas (that comes burdens-free) [367]. In our LCA, it was assumed that the CO₂ is extracted and purified from the waste gas of ammonia production using monoethanolamine (MEA), which is a common practice [367,368]. Based on [368], 3.56 MJ of heat are necessary to separate 1 kg of CO₂ from waste gas from ammonia production using MEA as extraction solvent. For heat from natural gas, the process *Heat*, *district or industrial*, *natural gas* {*Europe without Switzerland*]/ *heat production*, *natural gas*, *at industrial furnace* >100kW / *Conseq* from ecoinvent 3.6 was used. The amount of MEA, losses of MEA to air and water flows of carbon dioxide separation using MEA were retrieved from ecoinvent 3.6 dataset *Carbon dioxide*, *{RER}| production | Conseq*. This modeling could be considered also a good proxy for the purification of carbon dioxide using MEA from gases from other industrial processes and not only from ammonia production. The electricity input for carbon dioxide liquefaction was not considered since the carbon dioxide delivered by the swing adsorption process is in a gas state.

6.2.2.5.2 Additional animal feed to replace potato by-products

Concerning the additional animal feed production, it was necessary to identify the mix of marginal technologies that will fulfill such marginal demand. A decision tree is typically used to determine the marginal technologies, i.e. the technologies affected by a small change in market demand [211]. Since the additional production of animal feed affects a market more than a specific process, it is necessary to understand the trend in the volume of the affected market [211].

Potato peels are a balanced feed with good fibre, starch and protein contents [369]. The dry content and starch content (on a dry basis) of this mix of potato byproducts were measured as 12.3% and 55.7%, respectively [347]. It could be argued that potato peels are also used to provide calorific values of the feed for its high content of starch. Due to commercial confidentiality, it is not possible to disclose the exact feed markets of potato peels supplied by the specific potato processing industry in the project. We used generic feed market information about potato peel-based feed to determine the types of displacement feed. In the baseline analysis, the protein content is used to quantify the equivalent function of different types of animal feeds because the nutrient value of feed is determined by the protein content.

The protein content (on a dry basis) was measured only for the sample of potato peels as 16.0%. In this study, the protein content of potato peels (being the main fraction) was assumed for the entire mix of potato by-products. Potato by-products are a minor amount in the Dutch animal feed market currently used mainly for pigs [370,371] and cattle [369,372]. In the Netherlands (and nearby countries), the compound feed production for pigs is declining and for cattle is stagnating [373,374]. Therefore, there is a potential to provide a marginal increase in production since the production system is not saturated, and thus the current market is the marginal technology. The animal feed produced in Europe is made mostly of grains produced in Europe (71%) and imported oil meals (24%) [373]. The

grains are mostly made of wheat (32%), corn (30%) and barley (25%), while soybean meal makes 58% of all the oil meals used in the EU and is mainly imported from Argentina (58%) and Brazil (33%). These major crops were considered to build the marginal market that can be found in Table 19. The sensitivity of the choices made the marginal feed production is discussed in Section 6.4.2.

Table 19. Animal feed actual/average (used for the comparison conducted in the ALCA) and marginal market proportions (used for the CLCA) based on [373] and background sources; n.a. = not applicable.

Flow	Actual/ Average market (%)	Marginal (%) market based on the rescaling of the major crops	Amount assumed (kg) for the marginal animal feed modeling in the CLCA*	Background data sources for ALCA (benchmark used for comparison) and CLCA (marginal animal feed used in system expansion)
Barley grain (EU)	17.8%	23.9%	1.5	Barley grain {DE} barley production APOS or Conseq
Corn gran (EU)	21.3%	28.6%	1.8	Maize grain, Swiss integrated production {CH} production APOS or Conseq
Wheat grain (EU)	22.7%	30.5%	1.9	Wheat grain {DE} wheat production APOS or Conseq
Other grains	9.2%	0%	0	Oat grain {FI} oat production APOS
Total grains	71%	83%	5.2	n.a.
Soybean meal (AR)	8.1%	10.9%	0.7	Soybean {AR} soybean production APOS or Conseq. Soybean production used as proxy for soybean meal.
Soybean meal (BR)	4.6%	6.2%	0.4	Soybean meal {BR} market for soybean meal APOS or Conseq
Other oil meals	11.3%	0%	0	Rapeseed, at farm/NL Economic from Agri- footprint 5.0. Rapeseed production as a proxy for rapeseed meal.
Total oil meals	24%	17%	1.1	n.a.
Neither gains nor oil meals	5.0%	0%	0	Neglected

* to obtain 1 kg of total protein considering the dry matter and protein content of the dataset used as background data.

6.2.2.6 Land use changes

There are still many concerns around the sustainability of biofuels' production; many of these are related to direct or indirect carbon stock changes from landuse transitions [375]. The induced direct carbon stock changes from bio-jet fuel production are proven to play a critical role in their environmental performance [376]. Simultaneously, the role of carbon stock changes becomes increasingly relevant when the impact on other lands from the displacement effect of biomass production is accounted for (indirect land-use change) [377]. Therefore, it is paramount to include such direct and indirect effects of land-use change (iLUC) from biofuels systems.

In our study, LUC impacts were retrieved from the Ecoinvent database (version 3.6) and Agri-footprint database (version 5.0). Under these databases, LUC is assessed following the methodological principles recommended by PAS2050 [167]. A 20-year time horizon for carbon pools to reach equilibrium is considered [378]. Carbon stock changes are addressed over the four main carbon pools [379]: aboveground biomass, belowground biomass, dead organic matter and soil organic carbon. Note that the LUC Ecoinvent model follows the WFLDB Quantis adapted version (country-level perspective) of the Blonk tool to assess direct LUCs [380]. The LUC model of Agri-footprint 5.0 used for the direct land-use change of potato product relies on the Blonk tool [380].

To illustrate, carbon stock change effects from potato production within the same country are accounted for directly as they occur on the same land as the potato production land use. In the ALCA model, no indirect effects are generated from the current potato production as there is no effect of using land already dedicated for this purpose.

In contrast, in the CLCA model, animal feed production's displacement effects outside the production country's boundaries are accounted as indirect. The upstream displacement effects of the potato by-products attributed to the marginal animal feed production are assumed with a direct casual-effect relationship. Consequently, the displacement effect from the marginal feed (corn, wheat, barley, and soybean) production and derived carbon stock changes in the EU, Brazil and Argentina are attributed 100% to this displacement effect and accounted for entirely in this supply chain. The data of carbon stock exchanges of the marginal feed produced in the EU, Brazil and Argentina are obtained from ecoinvent version 3.6.

6.2.2.7 Petrochemical kerosene and other reference products

For petrochemical kerosene, the dataset *Kerosene {Europe without Switzerland}| kerosene production, petroleum refinery operation* from the libraries of ecoinvent 3.6 named APOS and Conseq respectively for the ALCA and CLCA. This dataset represents the European production of kerosene at the factory gate of the oil refinery. Distribution and combustion were modeled as detailed in section 6.2.2.3. The process flow diagram detailing inputs and co-products for petrochemical kerosene can be found in the appendix.

For the other co-products, the comparison was conducted with the following reference products:

- for hydrogen, the dataset Hydrogen (reformer) E from PlasticsEurope [381];
- for carbon dioxide, the inventory was modeled as illustrated for the marginal carbon dioxide using average data i.e. 1) assuming that 3.56 MJ of heat are necessary (*Heat, district or industrial, natural gas {Europe without Switzerland}*/ heat production, natural gas, at industrial furnace >100kW / APOS from ecoinvent 3.6) and 2) retrieving the amount of MEA, losses of MEA to air, water flows of carbon dioxide separation using MEA from the ecoinvent 3.6 dataset Carbon dioxide, {RER}/ production / APOS.
- for lubricants, the dataset Lubricating oil {RER}| production | APOS from ecoinvent 3.6;
- for animal feed, the average European mix has been considered and can be found in the second column of Table 19.

6.3. RESULTS

The results section is structured as follows. First, the environmental impact hotspots of the ALCA of the bio-jet fuel are presented. Second, the environmental hotspots of the CLCA of the bio-jet fuel are presented. Third, the environmental impacts of the bio-jet fuel assessed with both modeling approaches are compared to petrochemical kerosene. Afterward, the results of the ALCAs of the bio-based co-products are presented, along with a comparison with market benchmarks. The last section of the Results (section 6.3.4.4) reports on the "attributional pie", showing how much each co-product contributes to the environmental impact of the entire product system.

6.3.1 ALCA: identification of the environmental hotspots

The overall breakdown of the cradle-to-grave environmental impact of 1 MJ of biojet fuel is illustrated in Figure 35 (numerical results can be found in the appendix).



Figure 35. Breakdown of the cradle-to-grave attributional environmental impact of 1 MJ of bio-jet fuel per key-unit process.

From Figure 35, it can be observed that feedstock production is a relevant environmental hotspot (23-30% of the cradle-to-grave environmental burden) in all categories except for photochemical ozone formation. Both biochemical conversion and alcohol condensation generate about 30% of climate change and depletion of fossil resources, adding up to 60% of the environmental impact for these two categories. The bio-jet fuel combustion dominates the photochemical ozone formation, acidification and terrestrial eutrophication impacts (60-80% of the cradle-to-grave impact in these categories).

Heat from natural gas used for feedstock production contributes to 45% of the impact share for climate change and 56% for the depletion of fossil fuels. The impact of electricity for feedstock production is relatively small for all categories (less than 18%) and minor (3%) for acidification. Conversely, for all categories, raw potato

production represents a significant impact share of feedstock production. This share ranges from 25% in depletion of fossil fuels to up to 94% in acidification. The main environmental impact sources from raw potato are the cultivation stage and their transport to the processing industry. The transport process is responsible for 20% of climate change impact and 35% of fossil fuel depletion. For climate change, 45% of the impacts in the cultivation stage are caused by dinitrogen monoxide emissions to air. These emissions result from applying fertilizers, manure and crop residues. In addition, 20% of the impact is caused by diesel use in agricultural machinery and 14% by calcium ammonium nitrate production. The acidification and terrestrial eutrophication from feedstock production are related to ammonia emissions from fertilizer application. The diesel burned in agricultural machinery also represents 38% of the depletion of fossil fuels caused by feedstock production. Regarding the small impact (5%) of the feedstock production on the cradle-tograve photochemical ozone formation, it is mainly caused by the transportation of the potatoes to the processing food industry and by the diesel burned in the agricultural machinery during the cultivation phase. The direct carbon stock change caused by the cultivation of potatoes processed in the Netherlands has a negligible impact since there is no displacement effect. Long-term established agricultural land is used in the EU for the production of potatoes. Thus, the carbon stock changes within this land are relatively small as the land is already dedicated for agricultural purposes.

In all categories, biochemical conversion is important and has high impact especially for climate change and fossil fuel depletion. The environmental impact of this process is caused mainly by the production of electricity (44-72%) and potassium hydroxide (26-51%). The impact of alcohol condensation represents 30% of climate change and depletion of fossil fuels, respectively, and 12% of photochemical ozone formation. For these impact categories, the main sources of impact are the production of steam (55-56%) and cooling energy (43-45%). The production of steam impact is related to the combustion of natural gas. Swing adsorption impact is minor and generated by the allocated electricity (1-5%). Since hydrogen is the only consumable of hydrotreatment and comes from a closedloop flow, hydrotreatment has no impact. The environmental impact of hydrogen production is already accounted for in the biochemical conversion and the swing adsorption process. Feedstock transport is an important environmental hotspot for climate change (9%), photochemical ozone formation (18%) and use of fossil resources (9%), mainly due to the production and combustion of diesel. The distribution of the bio-jet fuel to the tank has a negligible contribution to the cradle-to-grave environmental burden in all categories. The combustion of the bio-jet fuel releases nitrogen oxide emissions resulting in high photochemical ozone formation, acidification and terrestrial eutrophication.



6.3.2 CLCA: potential environmental impacts due to changes in demand

Figure 36. Breakdown of the cradle-to-grave consequential environmental impact of 1 MJ of bio-jet fuel per key-unit process. Numerical results can be found in the appendix (see Table 20 for LUC contribution).

Figure 36 shows the breakdown of the cradle-to-grave CLCA. Similar to the results from the ALCA, the environmental impacts caused by the utilities in the biochemical conversion are high in most of the impact categories. Similarly, the impact of alcoholic condensation is high in climate change and depletion of fossil resources (marginal data did not change that). The environmental impact caused

by the combustion of the bio-jet fuel remains important also for the CLCA for photochemical ozone formation, acidification and terrestrial eutrophication.

Different from ALCA, the consequential modeling leads to the result that the additional amount of animal feed is one of the most important contributors of impact for all impact categories. The impact of the additional animal feeds can be divided into biomass production and LUC (see Table 20).

Impact category	Biomass (o	rop) pro	duction			LUC		Total
	Barley grain (EU)	Maize grain EU	Soybean from Argentina	Soybean from Brazil	Wheat grain EU	Soybean from Argentina	Soybean from Brazil	_
Climate change (%)	11	9	3	4	16	46	11	100
Photochemical ozone formation (%)	18	16	17	9	23	13	5	100
Acidification (%)	10	52	12	4	14	6	2	100
Eutrophication, terrestrial (%)	18	40	5	0	27	7	2	100
Depletion of fossil fuels (%)	26	21	12	2	36	0	0	100

Table 20. Breakdown of the environmental impacts of the additional feed needed to replace the potato by-product, contributed by biomass (crop) production and land use changes.

For biomass production, producing fertilizers and resulting field emissions from their application are the major environmental impacts. For barley, the impact of producing nitrogen fertilizers is high in all categories, ranging from 32% for photochemical ozone formation to 79% for acidification. Similarly, for wheat grain, the production of nitrogen fertilizers contributes between 39% (photochemical ozone formation) and 93% (acidification) of the environmental impact. Tillage is also an important source of environmental impact for barley in three categories i.e. photochemical ozone formation (26%), acidification (19%) and depletion of fossil fuels (13%). The impact of tillage is also high in these categories for wheat grain (but with slightly lower percentages for wheat grains). For maize production, direct field emissions cause 33% of climate change due to dinitrogen monoxide and carbon dioxide released into the air. They also cause 86% of acidification and 80% of terrestrial eutrophication (80%), mainly due to ammonia to air.

The effects of LUC are relevant for climate change, photochemical ozone formation, acidification and terrestrial eutrophication. The LUC impacts are attributed primarily

to soybean production. LUC is responsible for 94% and 74% of the climate change impacts of Argentinian and Brazilian soybeans, respectively. The LUC climate change impact of the Argentinean soybeans is higher than that of the Brazilian soybeans, given the larger share of high carbon stock lands converted to arable land. For Argentina, the land-use transition to arable land involves mainly removing forest land, from which 85% is secondary and 8% is primary. A lower share of forest removal characterizes the land-use transition in Brazil, with 75% clear-cutting of primary forest. In addition, 5% of the land-use transition is attributed to the removal of grasslands. Grasslands contain considerably lower carbon stocks than forests.

For photochemical ozone formation, 18% of the impact of marginal animal feed is caused by LUC. Of this 18%, 13% is caused by clear-cutting of secondary forest to arable land in Argentina and 4% by clear-cutting of primary forest to arable land in Brazil while the remaining 1% by clear-cutting of primary forest to arable land in Argentina. For clear-cutting of secondary forests in Argentina (and similarly for Brazil with slightly different shares), the photochemical ozone formation impact is caused by air emissions of ethene (29%), nitrogen oxides (21%), formaldehyde (11%) and propene (9%) due to the burning of vegetation, for 7% by diesel combustion in tractors and for 5% by petrol combustion in power sawing.

The small acidification and terrestrial eutrophication impacts caused by LUC is mainly due to Argentinian soybeans (see Table 20). More than 90% of acidification and terrestrial eutrophication related to Argentinian soybeans' LUC is due to the clear-cutting of secondary forests converted into arable land. The acidification due to such clear-cutting is caused by ammonia emissions to air (61%) and nitrogen oxides (26%) due to the burning of vegetation. For terrestrial eutrophication, these percentages become 64% for ammonia emissions and 34% for nitrogen oxides.

The marginal production of carbon dioxide (given the high amount of carbon dioxide produced) is the main environmental credit for climate change, depletion of fossil fuels and photochemical ozone formation. In these three categories, the credit is mainly caused by avoiding the production and combustion of natural gas with smaller credits for the avoidance of the production of the MEA solvent used for carbon dioxide separation.

The credit for substituting the surplus of hydrogen is lower than the one for carbon dioxide in climate change and depletion of fossil fuels, because the quantity of the surplus H_2 is much smaller than that of CO₂... Nevertheless, the surplus H_2 offers

environmental credits for photochemical ozone formation (98% caused by avoided nitrogen oxides emissions to air from fuel combustion) and acidification (70% nitrogen oxides and 30% sulfur dioxide to air from fuel combustion).

However, in the future, the production of hydrogen and carbon dioxide and other chemical products might become less carbon-intensive (see section 6.4.1 and section 6.4.2 for discussion regarding the uncertainties on future benchmark technologies).

The credit for the substitution of lubricants is high only for photochemical ozone formation and depletion of fossil fuels. 90% of such credit is caused by the avoidance of direct air emissions of non-methane volatile organic compounds (NMVOC) during the production of the lubricants. For depletion of fossil fuels is caused by avoided production of crude oil (75%).

6.3.3 Comparison with petrochemical kerosene

As shown by Figure 37, the environmental performance of the bio-jet fuel compared to conventional jet fuel is very different depending on the LCA modeling selected.

Based on both ALCA and CLCA results, the bio-jet fuel assessed offers a lower fossil fuel depletion impact compared to petrochemical kerosene (55% reduction based on ALCA and 49% based on CLCA).

For climate change, the ALCA and CLCA models lead to different conclusions. The ALCA model shows benefits for the bio-jet fuel (58% lower impact), whereas the CLCA calculated a 68% higher climate change impact than conventional jet fuels. In the CLCA model, LUCs due to the additional animal feed production is influential for climate change (47% of the cradle-to-grave climate change) but not significant for the depletion of fossil fuels (see Table 20).

For photochemical ozone formation, the differences between the bio-jet fuel and the petrochemical jet fuel are insignificant (1% difference) based on the ALCA model, whereas based on the CLCA model, a 50% higher impact is observed for the bio-jet fuel compared to petrochemical kerosene. In the CLCA model, the credits for photochemical ozone formation from co-products substitution are only 30% of the impact from the additional production of animal feed. Moreover, the impact of the transportation of potato by-products becomes 3 times higher in all categories

(with slightly different percentages 2.9-3.1 times given the shift from average to marginal data). The reason is twofold: 1) this impact is no more allocated to the coproducts but entirely apportioned to the bio-jet fuel and 2) this unit process does not have co-products to substitute.





Consequential modeling

Figure 37. Comparing the cradle-to-grave environmental impacts of 1 MJ bio-jet fuel from potato by-products: the attributional model (top) and the consequential model (bottom). Characterized impacts can be found in the appendix.

The bio-jet fuel has a higher impact on acidification than the petrochemical jet fuel (10% in ALCA and 75% higher in CLCA). In the CLCA model, the benefits from coproduct substitution do not compensate for the impact of the additional animal feed production i.e. the benefits are only 17% of the additional impact from animal feed production). Similar patterns are observed for terrestrial eutrophication between the results of ALCA and CLCA.

Based on the comparison of the results obtained with CLCA and ALCA approaches (see Figure 37), it can be concluded that the recommendations for producing this bio-jet fuel would be very different depending on the approach used with respect to climate change and photochemical ozone formation impacts. In fact, for these categories, the bio-jet fuel has a significantly higher impact once the additional production of marginal animal feed is considered. The attributional modeling overlooks this aspect.

6.3.4 ALCA: Environmental performance of the other biobased products

This section presents the ALCA comparisons between the environmental impacts of the co-products of the bio-jet fuel production process and their reference products.

6.3.4.1 Animal feed (organic residue)

Figure 38 shows the cradle-to-gate comparison between the organic residue from ABE fermentation sold as animal feed and the average European animal feed consumed (see Table 19). In four out of five impact categories, environmental benefits ranging between 65% and 87% were observed. A 38% higher impact was observed for the depletion of fossil fuels (see Figure 38 for the sources of impact). Figure 38 also shows the shares of the contribution of the various unit processes to the cradle-to-gate environmental impact of the animal feed co-product. We can observe that, since the allocation is applied only after subdividing the process as much as possible, the animal feed is not responsible for any environmental impact caused by either the swing adsorption process or alcohol condensation process. Similar considerations also apply to the other co-products. For this reason, the environmental impact of each co-product has its own "recipe" of environmental burdens' contributors, which differ among them.



Figure 38. Cradle-to-gate comparison between the animal feed by-product (left bar) obtained from the bio-jet fuel production process (with process contributions highlighted) and the average European animal feed. Values are normalized taking the most impacting value per each category as the reference.

6.3.4.2 Bio-based carbon dioxide and bio-based hydrogen

Figure 39 shows the cradle-to-gate comparison between the carbon dioxide separated via swing adsorption and conventional carbon dioxide (see section 6.2.2.7). In three out of five impact categories, environmental benefits ranging between 16% and 65% were observed. However, the acidification and terrestrial eutrophication caused by the bio-jet fuel are higher than for conventional jet fuel (see Figure 39 for the impact sources).



Figure 39. Cradle-to-gate comparison between the bio-based carbon dioxide (left bar) delivered by this innovative process (with process contributions highlighted) and conventional carbon dioxide. Values are normalized taking the most impacting value per each category valued as 100%.

Figure 40 shows the cradle-to-gate comparison between the bio-based hydrogen separated via swing adsorption and conventional hydrogen from reforming (see section 6.2.2.7). In four out of five impact categories, environmental benefits ranging between 45% and 78% were observed compared to reforming hydrogen (see Figure 40 for process contributions). For bio-hydrogen, the breakdown of the environmental impact corresponds to the one of bio-based carbon dioxide since they are both co-products leaving the system from the swing adsorption process.



Figure 40. Cradle-to-gate comparison between the bio-based hydrogen (left bar) delivered by this innovative process (with process contributions highlighted) and reforming hydrogen. Values are normalized by taking the most impacting value per category valued as 100%.

6.3.4.3 Biolubricants

Figure 41 shows the cradle-to-gate comparison between the biolubricants produced from hydrotreatment and petrochemical lubricants (see section 6.2.2.7 for petrochemical lubricants). In four out of five impact categories, environmental benefits ranging between 12% and 88% were observed compared to petrochemical lubricants. For climate change, the impact of biolubricants was 18% higher than petrochemical lubricants (see Figure 41 for process contributions). For bio-hydrogen, the breakdown of the environmental impact corresponds to the one of bio-based carbon dioxide since they are both co-products leaving the system from the swing adsorption process.



Figure 41. Cradle-to-gate comparison between the biolubricants (left bar) delivered by this innovative process (with process contributions highlighted) and petrochemical lubricants. Values are normalized taking the most impacting value per each category valued as 100%.

6.3.4.4 The "attributional pie"

By summing up the environmental impacts of each bio-based product (i.e. each piece of the pie), it is possible to visualize their contributions to the total pie i.e. the environmental impact of the entire "attributional" product system shown in Figure 34. Figure 42 shows the whole pie with its pieces.

The bio-jet fuel is the main cause of the environmental impact of the investigated product system. In fact, being responsible for the existence of the entire system product, the bio-jet fuel is the product that got allocated the environmental impact of all unit processes in the system (at least partially). The animal feed obtained as the organic residue also received a significant fraction of the environmental impact of the product system. The reason is that it got allocated most of the environmental impact at the level of the ABE fermentation due to its significant mass ratio compared to the other co-products (e.g. >5 kg of animal feed/kg of bio-jet fuel). The other co-products have minor "responsibilities" instead.



Figure 42. The attributional environmental impact of the overall production system and its allocation to each of the bio-based co-product.

6.4. DISCUSSION

6.4.1 Uncertainties in the attributional model

The ALCA has modeling uncertainties that are usually related to the application of a different allocation method or the price fluctuation for economic allocation. Applying energy or mass allocation at the level of the potato processing industry would mean a significant increase of the environmental impact of the potato by-products since they have physical characteristics similar to the main food products but much lower market price. As we argued in our earlier work [347], such a type of allocation leads to distorted results and would not respect the ISO causality principle that the allocation criterion should reflect. Regarding the price fluctuation of potato by-products, the market trend of the last five years showed a small variation (order of 3%) that would have a negligible effect on the results of the ALCA. The economic allocation factor applied at the level of the swing adsorption unit is also affected by uncertainties related to the prices of carbon dioxide and hydrogen. For both these products, novel technologies with a lower carbon footprint and/or using greener energy are expected to take place in the next decade. Carbon capture and storage (CCS) might also play a role in many chemical and energy processes in the longer term (e.g. the next two decades). Despite the deployment of CCS has been very slow so far (accounting for less than 0.5% of global investment in clean energy), robust climate targets could increase CCS investments [382]. Greener energy and CCS could affect the price of carbon dioxide and/or hydrogen, affecting the allocation share applied. Uncertainties also apply to the allocation factors regarding the fermentation residue sold as feed and lubricants, but they are not affected by future price fluctuations since based on energy and mass values respectively. The environmental impact of petrochemical kerosene is also influenced by the allocation applied at the level of the oil refinery, but with a minor effect on the overall environmental impact, for two reasons: 1) combustion emissions dominate the environmental impact of conventional kerosene (70-90% of cradle-to-grave impact) in all categories except depletion of fossil fuels and 2) there is small difference in energy, mass or economic shares for European kerosene over the total EU refinery system output [222].

6.4.2 Uncertainties in the consequential model

It is still challenging to address (without significant uncertainties) the historical question [89] "which feed ingredient will meet the increased protein demand?". A different answer to this question can significantly impact the final numerical results of the LCA [349].

For our study, taking different animal feeds to replace potato by-products does change the conclusion for climate change (see Figure 43). In fact, the climate change impact of the bio-jet fuel could become between 4% and 23% lower than petrochemical kerosene if the company currently utilizing the potato by-products replaced them with European animal feed only. Conversely, the climate change impact of the bio-jet fuel could double than calculated for the baseline scenario (and therefore significantly higher than petrochemical kerosene) importing soybean from South America to replace the potato by-products diverted from the European animal feed market. For the other four categories, the conclusions of the CLCA regarding the comparison between the bio-jet fuel and petrochemical kerosene are unaffected by the market displacement from the animal feed market. However, the numerical results could change significantly depending on the type of animal feed assumed and the impact category considered.



Figure 43. Consequential cradle-to-grave comparison between the innovative bio-jet fuel from potato by-products and petrochemical kerosene for 1 MJ of fuel varying the type of marginal animal feed (m.a.f.). Values are normalized taking the most impacting value per each category between the bio-jet fuel with baseline calculations and petrochemical kerosene (i.e. 100% taken as Figure 37).

Moreover, in our study, the displacement of animal feed with potato by-products was made on a protein basis. Alternatively, an energy basis could be assumed since animals do not need only proteins from the feed but also energy (the same apply to humans, who are the final user of the food products). The gross energy of potato by-products is 17.2 MJ/kg_{drv} [370]. Keeping the same composition of the marginal mix, the mass of animal feed to be substituted would increase by 10% mainly because soybean meals have much higher protein content than potato by-products but only slightly higher energy content. Alternatively, if one type of animal feed only were consumed as an alternative for potato by-products on the market, it should be considered what follows. Assuming that the primary function of potato by-products used as animal feed is to provide calorific energy and not protein to animals, it would be incorrect to compare 1 MJ of soybean meal only with 1 MJ of potato by-products. In fact, what matters is the primary function and the causality mechanism behind the provision of such a function [41,223]. For the same amount of energy, soybean meal would also provide a much higher amount of nutrients (proteins) to animals than the original potato by-products (hence, the primary function of soybean meal is to provide proteins). This means that the product's primary function that is included in the system boundaries would no more correspond to the primary function of the displaced product. Figure 44 shows



the sensitivity analysis results where the market displacement was assumed based on energy as the primary function.

As for the previous sensitivity analysis, what emerges is that the outcome of the CLCA on the better or worse performance between the bio-jet fuel and conventional jet fuel was affected only for climate change. However, the environmental performance of the bio-jet fuel for photochemical ozone formation, acidification and terrestrial eutrophication can change significantly. The case of potato by-products displaced by maize is the best scenario for climate change and depletion of fossil resources. In particular, almost 40% reduction of climate change impact could be achieved compared to petrochemical kerosene. On the other hand, the acidification and eutrophication impacts would increase in the case of maize compared to barley and wheat grains.

Given the temporal scope (the year 2030), the production method of the products substituted in the CLCA based on market trends such as the surplus of hydrogen and carbon dioxide are affected by uncertainty. In fact, in the future, environmental impact reductions could be achieved if carbon dioxide and hydrogen production will be produced on a large scale using renewable energy (see discussion in section 6.4.1 for future effects on benchmark technologies). CLCAs with a future

Figure 44. Sensitivity analysis results based on energy as the primary function. Consequential cradle-to-grave comparison for 1 MJ of fuel varying the type of marginal animal feed (m.a.f.). Values are normalized taking the most impacting value per each category between the biojet fuel with baseline calculations and petrochemical kerosene (i.e. 100% taken as Figure 37).

scope need to be revised if a significant market shift occurs and such shift was not accounted. Otherwise, such LCAs would consider an outdated production method instead of the one actually affected [237]. In the results of our CLCA, the credits for substituting carbon dioxide and hydrogen were (much) lower than the environmental impact of the marginal production of animal feed in four out of five categories. However, there is high uncertainty in the impacts of the additional animal feed (see Figure 43 and Figure 44). In the future, more sustainable animal feeds might be marketed e.g. food waste enriched with proteins using insects [383], or the market demand for meat may decline due to a switch in diet choices [384]. Furthermore, we assumed that the entire fraction of potato by-products are taking away feed from the animal feed supply. However, as detailed in section 6.2.1, 15% of potato by-products are not used as animal feed and would need waste treatment or can be used for lower-value applications.

6.4.3 Indirect land use change

The iLUC impacts from the displacement effects from potato by-products accounted for almost 45% of the climate change impact category in the baseline calculations. The results suggest that iLUC impacts steer to a large extent the biofuel environmental performance. However, iLUC impacts are subject to high uncertainty [385] and were significant only for imported soybeans. Carbon stock changes from land-use transitions were directly retrieved from publicly available LCA databases and not modeled directly, given the scope of the study. Therefore, impacts from carbon stock changes are conditioned by the assumptions and methods carried out in such databases.

Ecoinvent (WFLDB Quantis-adapted version of the Blonk tool) assumes that mainly forests are converted to arable land. For Brazil, 75% of the total converted land corresponds to forests. However, recently it's been suggested that the effect of soybean as an (in)direct driver from forest loss in Brazil is more significant than previously understood [386]. Therefore, if a larger share of forest loss were accounted for, the share increase would lead to higher carbon losses and overall CO_2 emissions.

Forest contains considerably higher carbon stocks than other land categories such as grasslands, shrublands, or cropland [378]. Contrastingly, the change from any of the mentioned categories towards soybean would result in a lower iLUC impact. Inherently, iLUC occurs somewhere else where biomass is produced and often with a significant time-related lag effect [387]. Therefore, attributing causality from displacement effects is extremely challenging.

In this paper, we attributed 100% of the carbon stock change impacts to the displacement effect. However, several economic, social, and environmental variables affect such direct casual effect attribution. Therefore, in reality, the displacement effect between shifting the land use in one location to the marginal animal feed production in another one is submitted to numerous conditions such as market conditions [63]. Thus, the causal-effect relationship might not be as direct as assumed in this study, and the potential impacts from carbon stock changes could decrease. In addition, land-use carbon stock changes are highly location and context-specific. These conditions can vary considerably driven by biophysical characteristics (e.g. temperature), management practices (e.g. land intensification), and socio-economic conditions, which can vary significantly even within the same region [388,389]. Thus, the real iLUC impact from bio-jet fuel production due to displacement effects is difficult to determine. Still, when accounted for in biofuels, iLUC generally results in an unfavorable environmental performance [390]. Note that iLUC processes and impacts are valid for any landbased service. Future research should focus on including adequate measures to estimate the percentage of attribution from iLUC process that help to reduce the uncertainty nature from these processes and understand better the overall performance of biofuel supply chains.

6.4.4 Advantages and disadvantages of attributional or consequential LCAs

There has been an open discussion, for more than 20 years, on which modeling approach (attributional or consequential) is better for environmental product labeling and policy making [146,177]. Currently, two (or more) well-established "internally consistent but mutually exclusive schools" exist [68]. Each of these schools claims that there is "general agreement in the literature" [391] that supports their modeling choices over the rest [391,392]. However, other researchers believe that both modeling approaches are necessary and both approaches should be kept well distinguished [68,146,393,394]. As shown by our LCA investigation and highlighted by previous literature [47,48,64], attributional and consequential LCAs of the same product or system could lead to different conclusions.

Table 21 shows a summary of major differences between attributional and consequential LCAs in general and in our case study. In our case, the two modeling approaches led to contrasting results for climate change and photochemical ozone formation (see section 6.3.3). The ALCA showed that the bio-jet fuel and its co-products offer environmental impact reductions in most categories compared to their conventional counterparts. A similar conclusion was drawn by Djomo et al. (2008), who concluded that "using potato steam peels to produce hydrogen along with feeding animals with its by-products offer more environmental benefits than using the potato steam peels directly for animal fodder". However, like our ALCA, the LCA of [395] and other peer-reviewed LCAs investigating bio-based products from potato by-products [347,396,397] overlooked the effects of diverting potato by-products from the animal feed market. However, the fact that we do not know precisely which animal feed market the potato peels are mainly sold for and with what feed would be probably replaced is a major uncertainty in our study. On the other hand, checks on the consequences linked with the indirect effects of biofuels production (and resulting environmental impacts) are needed for a policy perspective aiming to avoid unintended counterfactual effects.

From our case study, we have learned what follows. Contradictory trends in the outcome of attributional and consequential LCAs of a fuel from a bio-based by-product can be expected in certain impact categories if the three circumstances are in place. The first one is that the by-product feedstock is already marketed to be utilized by another process. The second is that the process from which the by-product feedstock is diverted is not part of the processes delivering the final bio-based product (the bio-jet fuel in our case). In our case study, the two LCAs would have led to less contradictory outcomes, e.g. if the potato industry itself would be the current user of the potato by-products they generate e.g. to produce biogas used internally. In that case, the potato industry would need to replace that biogas with an alternative energy input that both types of LCA would include. Third, the displacement of a by-product from its current use leads to more contradictory results between the two LCAs if their conversion process does not have a high yield. Consequently, i.e. requiring a large amount of by-product feedstock per t of the final product (the bio-jet fuel in our case).

Table 21. Summary of aims, proc	luct systems, mi	ultifunctionality,	uncertainty ai	nd application
of attributional and conseque	ntial modeling	approaches in	general and	in our study.
(continues to the next page)				

Most critical difference	Attributional LCA (ALCA)	Consequential LCA (CLCA)
Aim		
General	"Provide information on what portion of global burdens can be associated with a product (and its life cycle). In theory, if one were to conduct attributional LCAs of all final products, one would end up with the total observed environmental burdens worldwide" [398].	"Provide information on the environmental burdens that occur, directly or indirectly, as a consequence of a decision (usually represented by changes in demand for a product)"[398].
In our case study	Investigate the environmental impacts caused by the production of an innovative bio-jet fuel starting from the total environmental impact of the production system (i.e. the full pie). LCAs of the other co-products of the bio-jet fuel aimed to understand how much each of the co-product is responsible for the production system's total environmental impact.	Investigate how introducing an innovative bio-jet fuel to the market could potentially change the overall environmental impact of the supply chain of bio-jet fuel production and the economic sectors affected by the change. Hence, the aim was to understand the environmental impact of this novel bio-jet fuel at the net of all the displacement effects i.e. the potato by-products from the animal feed market and the products potentially replaced by the co-products of the bio- jet fuel.
Product syster	n definition	-
General	"The systems analysed ideally contain processes that are actually directly linked by (physical, energy, and service) flows to the unit process that supplies the functional unit or reference flow". [398].	"The systems analysed in these LCAs are made up only of processes that are actually affected by the decision, that is, that change their output due to a signal they receive from a cause-and- effect chain whose origin is a particular decision". [398].
In our case study	The most critical aspect was the feed production within the syste Conversely, the ALCA accounted by-products but did not include	choice to incorporate additional animal m boundaries in the consequential LCA. for the impact of producing the potato the effect on the animal feed market.
Multifunction	ılity	
General	Allocation of the inputs and outputs of processes among co-products based on certain allocation keys.	The approach aims to reflect cause- and-effect chains including the expansion of the system to include affected unit processes outside the supply chain and substitution i.e. avoided burden effect due to the effects of introducing co-products in the market leading to the displacement of conventional market products.

Most critical difference	Attributional LCA (ALCA)	Consequential LCA (CLCA)
In our case study	Various allocation methods were applied at unit process-level (for details, see section 6.2.2.4).	Our setup is driven by the change in potato by-products' final use. Currently, potato by-products are used as animal fed. However, shifting the use for bio-jet fuel production results in a supply-side deficit, which requires to be covered by producing additional feed. Substitution was applied to the co-products.
Uncertainty rela	ted to product system definition a	nd multifunctionality
General	The modeling uncertainties in ALCAs are often related to the application of a different allocation method or the price fluctuation for economic allocation.	The fact that CLCAs are more sensitive to uncertainties than ALCAs due to the inclusion of market prospects is already broadly acknowledged by the literature [35,89,348].
In our case study	For this specific case study, the u on the outcome of the study can section 6.4.1). The inclusion of the additional ar boundaries of the CLCA was affect both the type and functionality of see section 6.4.2) as well as uncer changes (for further details, see s	ncertainty of the allocation practices be considered minor (for details, see minal feed production in the system cted by high uncertainties depending on f the animal feed assumed (for details, rtainties regarding indirect land use section 6.4.3).
Applications		
General	Attributional LCAs have been broadly applied for ecolabeling and policy support since ALCA results are usually less sensitive to assumptions and have lower uncertainties [94,391,399]. Regarding policy application, results from attributional LCAs have been used in EU legislation to place thresholds on GHG emissions savings.	Consequential LCA have been applied mainly for policy support on understanding consequences of possible policy decisions especially for biofuels. The production of biofuels and current drastic changes in the energy and materials sectors are occurring due to policy interventions. Although the consequences of a decision might be uncertain, policy interventions could be supported to prevent unwanted effects from happening.
In our case study	Consequential modeling become that might lead to significant en- could not prevent the chance tha of the aviation sector could push increasing the impacts of meat p feed market. For this reason, whil market regulation, both ALCA an making. For example, if the curre towards maize as animal feed to savings can still be reached by th jet fuel. The directives of such a s	es a key tool to avoid unintended effects vironmental damages. In fact, an ALCA at diminishing the environmental impact up the impact somewhere else e.g. vroduction due to shifts in the animal le ALCA could be a proper tool for d CLCA should be used to support policy nt user of potato by-products shifts replace potato by-products, 40% GHG is bio-jet fuel compared to conventional hift are in the hands of policy making.

6.5. CONCLUSIONS

The main goal of this article was to compare the results of attributional and consequential LCAs to evaluate a future bio-jet fuel produced in the Netherlands.

For this specific case study, in all impact categories assessed, the environmental burdens were higher when using consequential modeling than attributional modeling, leading to contrasting conclusions in this fuel's environmental performance than conventional jet fuels. The reason was that, besides the major environmental hotspots related to the bio-jet fuel conversion processes, the impact of the production of additional animal feed could be much higher than the credits from co-product displacement in the consequential LCA.

So, even if the results of consequential LCAs by including market prospects and indirect land-use changes are more sensitive to uncertainties than attributional LCAs, we believe that both LCAs are necessary for decision making to mitigate possible indirect effects on the affected markets. In this specific case, our consequential LCA highlights the environmental issues arising if the potato by-products diverted from the European animal feed market to produce the biojet fuel are replaced (in part) with imported soybean meals from South America. Such an aspect was instead overlooked using attributional modeling. To mitigate indirect environmental impacts on the animal feed market, it is necessary that the market is steered with a holistic perspective and both ALCA and CLCA become necessary.

The technology investigated in our study is at a pilot scale and was successfully tested using potato by-products. However, other bio-based residual streams could be converted via ABE fermentation. Further research may provide data for assessing the operation of this innovative technology with other feedstocks.

6.A APPENDIX

Figure 6A.1. Process flow diagram conventional jet fuel (kerosene)

Conventional jet fuel



Table 6A.1. Cradle-	to-grave attri	ibutional environn	nental impact c	of 1 MJ of bio ₋	jet fuel with nur	merical breakdo	umo	
Impact category	Feedstock production	Feedstock transportation	Biochemical conversion	Swing adsorption	Alcohol condensation	Distribution	Combustion	Sum
Climate change	1.05E-02	3.68E-03	7.99E-03	1.60E-03	1.07E-02	1.33E-04	2.05E-04	3.49E-02
Photochemical ozone formation	2.28E-05	1.91E-05	1.77E-05	2.82E-06	1.07E-05	7.57E-07	3.54E-04	4.28E-04
Acidification	1.19E-04	1.80E-05	3.30E-05	3.55E-06	1.24E-05	6.79E-07	3.02E-04	4.88E-04
Eutrophication, terrestrial	5.21E-04	6.72E-05	9.42E-05	1.06E-05	3.14E-05	2.54E-06	1.50E-03	2.22E-03
Resource use, fossils	1.37E-01	5.61E-02	1.35E-01	2.81E-02	1.69E-01	2.08E-03	0.00E+00	5.26E-01
Table 6A.2. Cradle-	-to-grave con:	sequential enviror	nmental impac	t of 1 MJ of bi	io-jet fuel with n	umerical break	umop	
Impact category	Trans feeds dewa	sportation of stock (incl. tering)	Biochemical conversion	Swir adsc	ng A orption co	lcohol ondensation	Hydroge substitu	en ition
Climate change	1.08E-	-02	1.26E-02	1.63E	E-03 2.	48E-02	-9.61E-0	20
Photochemical oz formation	one 5.81E-	-05	4.08E-05	8.06	E-06 1.4	40E-05	-2.78E-0	ß
Acidification	5.21E-	05	9.40E-05	9.80	E-06 8.	20E-06	-2.93E-0	2
Eutrophication, terrestrial	2.03E	-04	2.83E-04	3.98	E-05 9.	65E-06	-1.16E-04	
Resource use, foss	ils 1.61E-	Ю	1.48E-01	2.42	E-02 3.	65E-01	-9.11E-02	

Impact category	Transportation of feedstock (incl. dewatering)	Biochemical conversion	Swing adsorption	Alcohol condensation	Hydrogen substitution
Climate change	1.08E-02	1.26E-02	1.63E-03	2.48E-02	-9.61E-03
Photochemical ozone formation	5.81E-05	4.08E-05	8.06E-06	1.40E-05	-2.78E-05
Acidification	5.21E-05	9.40E-05	9.80E-06	8.20E-06	-2.93E-05
Eutrophication, terrestrial	2.03E-04	2.83E-04	3.98E-05	9.65E-06	-1.16E-04
Resource use, fossils	1.61E-01	1.48E-01	2.42E-02	3.65E-01	-9.11E-02
It continues					
---	---	---	--	-----------------	--------------------------------
Carbon dioxide substitution	Additional feed	Lubricants substitution	Distribution	Combustion	Sum
-1.57E-02	1.16E-01	-1.38E-03	1.35E-04	2.05E-04	1.39E-01
-1.71E-05	2.54E-04	-2.50E-05	7.62E-07	3.54E-04	6.60E-04
-1.89E-05	3.31E-04	-6.71E-06	6.46E-07	3.02E-04	7.42E-04
-5.29E-05	2.10E-03	-1.21E-05	2.53E-06	1.50E-03	3.95E-03
-2.65E-01	3.04E-01	-6.05E-02	2.08E-03	0.00E+00	5.88E-01
	•	LCA		CLCA	
Results for ALCA a average and margi	nd CLCA (baseline o nal data led only to c	alculations). Foi 1 minor difference	a conventional product li ce for acidification impacts.	ke petrochemico	I kerosene, the change between
Impact category	@	tio-jet fuel	Petrochemical kerosene	Bio-jet fuel	Petrochemical kerosene
Climate change	M	.49E-02	8.28E-02	1.39E-01	8.28E-02
Photochemical ozo	ne formation 4	.28E-04	4.36E-04	6.60E-04	4.36E-04
Acidification	4	.88E-04	4.42E-04	7.42E-04	4.24E-04

1.71E-03 1.16E+00

3.95E-03 5.88E-01

1.71E-03 1.16E+00

2.22E-03 5.26E-01

Eutrophication, terrestrial Depletion of fossil fuels



7

CHAPTER

Environmental Life Cycle Assessment of polypropylene made from used cooking oil

Published as:

C. Moretti, M. Junginger, L. Shen. Environmental life cycle assessment of polypropylene made from used cooking oil.

Resour. Conserv. Recycl., 157 (2020), p. 104750, 10.1016/j.resconrec.2020.104750

ABSTRACT

Used cooking oil (UCO) has received much attention as feedstock for the production of renewable fuels and bio-based materials. This study aims to assess the environmental impact of UCO-based polypropylene (PP) by a cradle-to-factory gate Life-Cycle Assessment (LCA). 16 impact categories were assessed. The results were interpreted with normalization and weighting steps. For several multi-output processes, different allocation procedures were scrutinized.

On a normalized and weighted basis, the environmental impacts of UCO-based PP are dominated by climate change (28%), fossil resource use (23%) and water use (11%). The following environmental hotspots are identified: the polymerization process (38%), the production of hydrogen (21%), the production of LPG (18%) and the combustion of LPG (8%). Compared to petrochemical PP, cradle to factory gate impact reductions of 40-62% for climate change and 80-86% for fossil fuel resource use can be achieved by UCO-based PP, depending on the allocation approach chosen. Moreover, if renewable propane and methane produced from the biorefinery could be used to replace LPG, the overall weighted environmental impact of UCO-based PP could be potentially further reduced by 34%.

7.1 INTRODUCTION

It was 1954 when Natta and his research group first polymerized propylene to a crystalline isotactic polypropylene (PP) [400]. Already at that time, Natta realized that the properties of polypropylene could have introduced new trends in the world of plastics [401]. Nowadays, PP represents one of the most widely used plastics in Europe (nearly 20% of the total consumed plastics) [402]. In 2018, the global production of polypropylene resin was 56 Mt. The demand is projected to increase to 88 Mt by 2026 [403]. The high demand for polypropylene is due to its versatility, which allows its use in many applications such as food packaging, construction pipes, and automotive parts. The properties that especially make polypropylene multipurpose are associated with its high melting point, low density, excellent stiffness and strength [404].

In the last decade, due to the growing demand for plastics and related concerns about their climate change impact, bio-based plastics have attracted attention as a possible option to replace petrochemical plastics. Many studies have highlighted that bio-based plastics could potentially offer a lower carbon footprint [7,56,405–407]. However, their environmental performance depends on the type of polymer, the impact category in focus, the selected system boundaries, the type of biomass feedstock and its final application and the supply chain [407–410]. In 2017, the global production capacity of bio-based plastic materials reached 2.05 Mt and was expected to increase to 2.44 Mt by 2022 [411]. In particular, bio-based PP entered the market in 2019 with a production capacity of about 19 kt, which is predicted to increase by about six folds by 2024 [412]. The three main synthesis routes for bio-based PP are: 1) using bio-ethanol from sugar fermentation [413–415], 2) using bio-syngas [415,416] and 3) using hydrotreatment of used cooking oil (UCO) [417].

To our knowledge, only two peer-reviewed environmental life cycle assessments (LCAs) of bio-based PP are publicly available [415,418], and no peer-reviewed LCA has been conducted for the third route, i.e. UCO-based PP. Mayumi et al. performed a cradle-to-factory gate LCA of biomass-derived PP and polyethylene (PE) at the design stage [418]. In the study, they quantified greenhouse gas (GHG) emissions of polyolefins made from the waste wood-syngas route. They found out that bio-based PP and PE could lead to higher GHG emissions compared to the petrochemical counterparts. The impacts of the bio-based polymers are dominated by biomass production and conversion processes. Kikuchi et al. (2017) investigated PP and

PE made from sugarcane-ethanol and woody-biomass-derived syngas based on data from demonstration plants and simulations. Defining one liter of ethanol as functional unit, they found that the highest GHG reduction can be obtained by using ethanol for PE production (a saving of about 50% compared to petrochemical PE), followed by use of ethanol as transport fuel (a saving of about 40% compared to gasoline), and PP production (a saving of about 15-20%). They concluded that PP production from bio-syngas offers relatively limited GHG emission reduction.

This study aims to investigate the environmental impacts of bio-based PP obtained through the third route, i.e. via hydrotreatment of UCO. UCO is a waste and thus does not compete with food and feed. It also avoids potential land use changes. In the EU-28, the total potential UCO available from the gastronomy sector, food processors and households is estimated at around 4 Mt per year [419]. These features make it an attractive feedstock for future transportation fuels and material production [341,420]. Primary data were collected from Neste Oyj based on a new commercial production facility in Europe [421].

The primary objective of this study is to provide a full picture of the environmental footprint of UCO-PP. Unlike the published LCAs of bio-based PP, which focused on climate change only, this study assesses 16 mid-point environmental impact categories following the recommendation of impact category selection by the European Commission's Product Environmental Footprint Category Rules (PEFCR) guidance (draft version 6.3) [165]. Such a wide selection of indicators has been rarely reported for the environmental assessment of bio-based chemicals and products [7,42]. For bio-based materials, environmental indicators such as land use, eutrophication, and acidification should not be ignored before investment or a policy decision is made [42,407]. Since there are several multifunctional processes involved, the second objective of the study is to gain insights into how the allocation choices could influence the LCA results. Despite that LCA is a standardized methodology, multifunctionality is one of the main remaining issues in LCA [50]. The findings of this LCA should be used both to understand the full environmental impacts, originated from both resources and emissions, and also to provide recommendations for future EU policy directives on the allocation choices for innovative bio-based plastics. The policy-level allocation recommendation is so far only available for renewable transport fuels and bioenergy [9]. The insights gained from this case study could help to reveal some complexity and demonstrate the influence of the allocation decisions in the LCAs of innovative bio-based products.

7.2 MATERIAL AND METHODS

7.2.1 Goal and scope definition

Life Cycle Assessment (LCA) is a widely applied standardized method used to assess the environmental impacts of a product or a service. In particular, this LCA study focuses on UCO-based PP and has been developed within the EU Bio-SPRI¹¹ project. The LCA was conducted according to ISO 14040 and ISO 14044. The draft PEFCR guidance was used as practical guidance when the guidance from the ISO standards were insufficient [38,127,165]. The main goal of this LCA is to assess the impact of UCO-based PP to identify the environmental hotspots of the life cycle. The second goal of this LCA is to add further valuable research to the open debate of solving multifunctionality of biorefineries (as detailed in section *Multifunctionality*). The results of this study are intended to be used by the industry for further process improvement, policy makers and the LCA community.

Based on the goal of the study, the functional unit (FU) is defined as 1 kg of polypropylene. A cradle-to-factory gate scope and an attributional approach are adopted in the LCA. Figure 45 shows the process diagram of the production of UCO-based PP. Used cooking oil is converted into high value hydrotreated vegetable oil (HVO) by Neste NEXBTL technology. Together with the HVO renewable diesel grade product, a renewable HVO naphtha grade product is obtained from the hydrotreatment process. This study focuses on the cracking of this "bio-naphtha" to obtain propylene via a process equivalent to petrochemical steam cracking.

Accordingly to the location of Neste Oyj's biorefinery (Rotterdam), the geographic scope is defined as the Netherlands. This reflects the specific situation of UCO collected from the Netherlands and nearby regions and all the major processes for the conversion into polypropylene occurring in the Netherlands as well with the exception of steam cracking occurring in a neighbor country. Nevertheless, when a specific inventory for the Netherlands was not available, or a specific process occurs in another EU country, average European data have been used.

The temporal scope is current (2018) to the near future (5-10 years), and the technological scope is defined as the status-quo technology which is ready for commercialization (technology readiness level 8, based on definition reported in [298]).

 $^{^{\}rm n}\,$ BIOSPRI (Bioeconomy: Support to Policy for Research and Innovation) project funded by DG RTD of the European Commission.

Sixteen mid-point impact categories are selected to analyze the full environmental footprint. The adopted impact assessment models for each impact category are listed in

Table 22. Their selection is based on the recommendation of PEFCR draft guidance (version 6.3), which was the version available at the time when the study was conducted. Differently, from the PEFCR guidance, particulate matter (PM) and land use are assessed using the methods recommended by the PEF guide [85].

Due to the many impact categories considered, normalization and weighting are applied to identify the *overall* environmental hotspots. To ensure that the same impact assessment models are used for characterization and normalization, the following selection has been done. For water use and resource depletion categories, the normalization factors are retrieved from PEFCR guidance v.6.3. For all the other impact categories, per capita, EU 27 normalization factors (2010) are retrieved from ILCD 2015 [171].

Table 22 reports the normalization and weighting factors applied in this study.



Figure 45. Flow diagram of the production of polypropylene from used cooking oil (UCO). The dashed box represents the primary production of vegetable oil, which is a unit process that is out from the system boundaries in the baseline analysis.

Impact Category	Unit	Impact assessment models	Normalization factors EU 27 per person [165,171]	Weighting factors with toxicity [165,171]
Climate change	kg CO ₂ eq	IPCC 2013, GWP 100a with carbon climate feedback [310]	9.22E+03	21.06
Ozone depletion	kg CFC-11 eq	[422]	2.16E-02	6.31
Human toxicity, non- cancer effects	CTUh	USEtox (recommended + interim) [423]	5.33E-04	1.84
Human toxicity, cancer effects	CTUh	USEtox (recommended + interim) [423]	3.69E-05	2.13
Particulate matter	kg PM2.5 eq	[311]	3.80E+00	8.96
lonizing radiation Human Health (HH)	kBq U235 eq	[424]	1.13E+03	5.01
Photochemical ozone formation	kg NMVOC eq	[312]	3.17E+01	4.78
Acidification	molc H+ eq	[313]	4.73E+01	6.20
Terrestrial eutrophication	molc N eq	[313]	1.76E+02	3.71
Freshwater eutrophication	kg P eq	[355]	1.48E+00	2.80
Marine eutrophication	kg N eq	[355]	1.69E+01	2.96
Freshwater ecotoxicity	CTUe	USEtox (recommended + interim) [423]	8.74E+03	1.92
Land transformation	kg C deficit	Soil Organic Matter model [325]	7.48E+04	7.94
Water use	m ³	AWARE factors [425]	1.15E+04	8.51
Resource use, minerals and metals	kg Sb eq	[315]	5.79E-02	7.55
Resource use, fossil fuels	MJ	[315]	6.53E+04	8.32

Table 22. Selected impact categories and models, normalization and weighting factors

7.2.2 Life cycle inventory modeling

7.2.2.1 Unit processes, data, and assumptions

For the foreground system, primary data were collected from Neste. Those are site-specific data. The background data were largely based on the Ecoinvent database (version 3.4) and PlasticsEurope's Eco-profiles [404]. For unavailable data, assumptions were made based on literature and/or validated by Neste. Important assumptions have been scrutinized by sensitivity analyses in the discussion section (*Data uncertainty*).

The production of UCO-based PP starts from the collection of UCO. In the baseline calculations, UCO has been considered as waste and, therefore, entering the system "free of burdens" (cut-off approach). A sensitivity analysis of this approach can be found in the discussion section (*Model uncertainty: multifunctionality*). The impacts of the collection are taken into account in the LCA. UCO is sourced from restaurants and commercial buildings mainly in the nearby regions of the bio-refinery country of location. The collection of UCO from restaurants and other users to the biorefinery is carried out by trucks and by water and distances assumed based on Neste's specific supply chain.

During the NEXBTL process, the oil is pre-treated and deoxygenized under high pressure to transportation fuel quality using hydrogen. The triglycerides of UCO are converted to saturated straight and branched-chain hydrocarbons and oxygen (of the triglycerides eliminated). In the chemical reaction shown by equation 5, a triglyceride with formula $C_{57}H_{104}O_6$ is taken as an example. Hydrogen is produced by steam reforming of natural gas and fuel gas.

$C_{57}H_{104}O_6 + 15 H_2 \rightarrow 3 C_{18}H_{38} + C_3H_8 + 6 H_2O$ Eq.5

The main inventory data related to the NEXBTL unit process are collected in Table 23. For each ton of pre-treated oil, 1.02 t of UCO are needed [426]. Steam is produced in the refinery while the Dutch national grid supplies the electricity. The amounts of steam reported in the inventory tables are equivalent kilograms, recalculated to keep into account the different energy content between the actual steam flows and the chosen dataset (equivalent kilograms on an energy basis). The chosen dataset represents the production of 1 kg of steam used for heating in the chemical industry. The heat is produced with the average fuel mix used in the European chemical industry [427]. The process releases wastewater that is treated on-site.

Flow	Data	Process	References on which foreground data are based on
Inputs			
Collected UCO	49.0 kg		
Phosphoric acid	28.0 g	Phosphoric acid, industrial grade, without water, in 85% solution state {RER} purification of wet-process phosphoric acid to industrial grade, product in 85% solution state APOS	[36]
Process chemicals	0.1 kg	Chemical, inorganic {GLO} production APOS	
Water	5.0 kg	Water, decarbonised, at user {RER} water production and supply, decarbonised APOS	[36,426]
Sodium hydroxide	48.0 g	Sodium hydroxide, chlor-alkali production mix, at plant/RER (PlasticsEurope. Industry data 2.0 project)	[36,428]
Electricity	6.0 MJ	Electricity, medium voltage {NL} market group for APOS	[36,426]
Steam	10.0 kg	Steam, in chemical industry {RER} production APOS	[36,426]
Hydrogen	1.7 kg	Hydrogen (reformer) E (PlasticsEurope. Industry data 2.0 project)	[36,381,426]
Nitrogen	1.6 g	Nitrogen, liquid {RER} market for APOS	[36,426]
Outputs			
Bio-based naphtha,	1.0 kg		Along with HVO diesel and propane whose percentages are not disclosed.
Wastes			
Wastewater (output to technosphere: waste to treatment)	8.8 L	Wastewater, average {Europe without Switzerland} market for wastewater, average APOS	[36,426]
Solid waste going to incineration	0.5 kg	Final waste flow, waste to incineration	

Table 23. Input and output data per 1 kg of bio-based naphtha from UCO (NEXBTL unit process-hydrotreatment plus pre-treatment)

After hydrotreatment, bio-based naphtha is transported by train¹² to the steam cracking unit (distance based on Neste's specific case). During steam cracking, biobased naphtha is diluted with steam and cracked into smaller hydrocarbons such as propylene and ethylene. The reaction is highly endothermic; therefore the feed is heated in a furnace burning fuel gas. The mass and energy inputs of the steam cracking unit in real operation are kept confidential and therefore the process has been modeled based on literature and is not Neste- specific. Process data from Karimzadeh et al. 2009 were modified considering the same feed capacity as the real operating pilot. The direct emissions released by the combustion of propane and butane are retrieved from the US Environmental Protection Agency dataset for industrial boilers [429]. In the calculations, The mix of 50% propane and 50% butane is assumed according to the dataset used for the production of LPG [430]. A selection of the main inventory data of this unit process is shown in Table 24.

The last process is polymerization of the propylene to obtain PP. The polymerization of bio-based propylene is identical to that of petrochemical propylene. The polymerization process has been extrapolated from the most recent PlasticsEurope's Eco-profiles [404] as follows: assuming that 1.02 kg (based on [432]) of propylene are required to produce 1 kg of polypropylene, the impact of the polymerization process has been obtained by subtraction. The PlasticsEurope's Ecoprofile [433] has been widely used as benchmarks for comparison. However, they are often regarded as "black box" data because of a lack of transparency constrained by confidentiality.

As a consequence, it is challenging to interpret the impacts of the polymers fully. This can be considered a limitation of the study because of the restricted interpretation of the polymerization process due to the use of this dataset (see section *Data uncertainty*). Nevertheless, polymerization has also been modeled according to the Matter (MATerials Technologies for CO_2 Emission Reduction) study of 1998 for sensitivity analysis, and when necessary the comparison has been used as a validation test [432]. Despite the Matter study is twenty years old, it has been selected since it is the only transparent dataset for PP polymerization available in the public domain.

¹² From Ecoinvent 3.4: Transport , freight train {Europe without Switzerland}| market for | APOS

Flow	Foreground data	Adopted background processes from Ecoinvent or modeled direct emissions	References on which these data are based on
Inputs			
Bio-based Naphtha	2.67 kg		Naphtha from hydrotreatment. [431]
LPG	0.63 kg	Liquefied petroleum gas {CH} market for APOS	[429,431]. Notice that only 0.43 kg of this LPG is burnt while the rest pumped with the feed.
Steam	2.9 kg		[431]. This steam is the sum of diluted steam (66%) and boiling feed water (34%).
Outputs			
Propylene	1.0 kg		
Steam	5.1 kg	Steam, in chemical industry {RER} production Conseq (dataset used for net production)	[431] Steam conditions: 520 °C and 112 bar
Other cracked gases and heavier products	Not disclosed		Ethylene is the major co-product and is accompanied by several other by-products (e.g. hydrogen, methane, benzene) whose percentages are not disclosed.
Direct Emissions			
Nitrogen oxides	1.3E-03 kg	Emissions to air due to LPG burning	[429,431].
Carbon dioxide of fossil origins	1.3E+00 kg	Emissions to air due to LPG burning	[429,431].
Carbon monoxide	7.5E-04 kg	Emissions to air due to LPG burning	[429,431].
Methane	1.9E-05 kg	Emissions to air due to LPG burning	[429,431].
Dinitrogen monoxide	8.5E-05 kg	Emissions to air due to LPG burning	[429,431].
Particulate <2.5	5.7E-05 kg	Emissions to air due to	[429,431].

LPG burning

Table 24. Selected inventory data of the steam cracking process per 1 kg of bio-based propylene made from bio-based Naphtha.

um

Flow	Foreground data	Adopted background processes from Ecoinvent or modeled direct emissions	References on which these data are based on
Particulates, > 2.5 um, and < 10um	1.9E-05 kg	Emissions to air due to LPG burning	[429,431].
VOC, volatile organic compounds	2.5E-05 kg	Emissions to air due to LPG burning	[429,431].
Sulphur oxides	1.6E-06 kg	Emissions to air due to LPG burning	[429,431].

7.2.2.2 Multifunctionality

According to ISO 14044, multifunctionality (or commonly also referred to as "allocation") should be solved using the following hierarchy [38]:

1) Avoiding allocation by subdivision or system expansion (i.e. expanding the product system to include the additional functions related to the co-products);

2) Allocation underlying physical relationships (i.e. an allocation that reflects how the inputs and outputs are changed by quantitative changes in the products of the system); and

3) Allocation (partitioning) based on other relationships (e.g. economic allocation).

In this context, the framework followed in this LCA to identify the impact of the choice of allocation approach is detailed. In the product system studied, there are two processes where more than one useful product is delivered: hydrotreatment and steam cracking.

The hydrotreatment process mainly delivers three products: hydrotreated vegetable oil (renewable diesel), propane and bio-based naphtha whose percentages are not disclosed. In this study, bio-based naphtha is the precursor of propylene while the other two co-products are sold. In this LCA, the problem of how to assign the environmental impact to the multiple products of hydrotreatment has been solved through energy allocation. Concerning the hierarchy above, the allocation is not avoided due to the inapplicability of subdivision (the process cannot be further subdivided). System expansion, both enlargement and reduction approaches, is also not possible. In fact, according to the goal and scope of the study, the functional unit cannot be enlarged to include all the co-functions. Concerning the inapplicability of system reduction, bio-based naphtha is a non-dominant byproduct and therefore, the criterion of physical significance is not respected (i.e. the mass ratio of bio-naphtha is in the magnitude of a few percents to that of HVO, see Table 23). Allocation by physical causality is not applied because this would require a mathematical model (commonly based on linear programming) that is not available [215,222,234]. Among the possible remaining allocation methods, energy has been chosen according to RED recommendations when dealing with transportation fuels [9,434]. Although naphtha is not used as transportation fuels, it is a by-product of fuel production (HVO) and therefore, RED recommendations are followed to respect the energy balances.

Nevertheless, it should be noticed that the RED focuses only on greenhouse gas emissions while 16 impact categories are analyzed in this study. For this reason, the authors have performed a sensitivity analysis on all possible allocation approaches used in the study (see the discussion section 7.4.2). Energy allocation has been adopted for the hydrotreatment process.

The steam cracking unit delivers two main products, i.e. propylene and ethylene, other cracked gases and steam (see Figure 45 and Table 24). Propylene is in the focus of this study. Nevertheless, ethylene represents the biggest mass fraction among the cracked gases. The steam cracking process also delivers 2.2 kg (6 MJ) of net industrial steam per kg propylene. Similar to the hydrotreatment unit process, subdivision, system expansion, and physical causality are not applicable. Avoiding allocation by substituting all propylene co- and by-products is not possible because a non-dominant product is in focus. For this unit process, a hybrid method has been chosen. Steam is directly used for other processes of the same biorefinery, and, otherwise, should be produced as marginal production of refinery steam from Ecoinvent (Steam, in chemical industry {RER}] production | Conseq). Direct substitution has been therefore applied to the net production of high pressure (HP) steam as, in this case, it may represent physical causality better than other arbitrary allocations (as highlighted by the PEFCR draft guidance). Energy allocation has been applied to all the other co-(by-) products resulting in 20% of the remaining impacts allocated to propylene.

7.2.2.3 Biogenic carbon accounting

The carbon content of UCO-based PP originates from the CO_2 sequestered by biomass. According to the PEFCR draft guidance, only biogenic carbon emitted

later than 100 years after its absorption shall be considered permanent storage [165]. Permanent storage results in a carbon credit to be assigned to the biobased product. When biogenic carbon is emitted earlier than 100 years, no carbon credits must be assigned for temporally carbon storage or delayed emissions [165]. In particular, the PEFCR guidance recommends to not assign carbon credits for cradle-to-user assessments [165]. Nevertheless, the biogenic carbon content at the factory gate "shall be reported as additional information" [165]. Accordingly, the authors have therefore proceeded as follows: the climate change impact is reported both with and without biogenic carbon removals while only the second one has been considered for the weighted results. Considering the chemical formula of propylene C_3H_6 , per kg PP, the biogenic carbon removal corresponds to 3.14 kg CO_2 eq.

7.3 RESULTS

7.3.1 Impact assessment and interpretation

The cradle-to-factory gate environmental impact results are shown in Table 25, while the breakdown results of each unit process are illustrated in Figure 46 (see Appendix A for numerical values behind the figure).

From Table 25, it can be seen that climate change (28%), fossil resource use (23%) and water use (11%) as the most important environmental impacts of UCO-based PP. shows that the NEXBTL process (hydrotreatment plus pre-treatment), steam cracking and polymerization are the three most significant key-unit processes in terms of environmental impact. On a weighted basis, polymerization contributes 38%, steam cracking 26% and hydrotreatment 29% of the total cradle-to-factory gate impacts¹³. The impact is almost entirely caused by the production (18%) and combustion of LPG (8%) and the production of hydrogen (21%) respectively for steam cracking and hydrotreatment (environmental hotspots). In particular, 7% out of 8% share of LPG combustion is caused by releasing GHG emissions. UCO collection and transportation of naphtha account only for 5% and 2% respectively. Pre-treatment represents the remaining 5%.

¹³ These shares of impact highlighted by the weighted results are not affected by the choice of applying weighting with toxicity instead of without toxicity. Indeed, changing weighting factors, only the shares of steam cracking and polymerization vary to 28% and 36% respectively.

Impact Category	Unit	Value	Normalized and weighted scores (Total 100%)
Climate change (without biogenic carbon removal (BCR))	kg CO ₂ eq	0.63	28%
Climate change, with biogenic carbon removal (BCR)	kg CO ₂ eq	-2.51	Not applicable
Ozone depletion	kg CFC-11 eq	9.0E-08	1%
Human toxicity, non-cancer effects	CTUh	1.1E-08	1%
Human toxicity, cancer effects	CTUh	5.5E-09	6%
Particulate matter	kg PM2.5 eq	1.2E-04	5%
lonizing radiation HH	kBq U235 eq	6.4E-02	6%
Photochemical ozone formation	kg NMVOC eq	1.9E-03	6%
Acidification	molc H+ eq	2.1E-03	5%
Terrestrial eutrophication	molc N eq	6.0E-03	2%
Freshwater eutrophication	kg P eq	8.7E-06	0%
Marine eutrophication	kg N eq	5.6E-04	2%
Freshwater ecotoxicity	CTUe	2.2E-01	1%
Land transformation	kg C deficit	1.1	2%
Water use	m ³	7.4E-01	11%
Resource use, minerals and metals	kg Sb eq	3.2E-07	1%
Resource use, fossil fuels	Д	9.3	23%

Table 25. Cradle-to-factory gate environmental impacts of 1 kg UCO-based PP

The hydrotreatment process has a significant contribution (20-40%) in the following impact categories (see): climate change, human toxicity without cancer effects, human toxicity with cancer effects, particulate matter, photochemical ozone formation, acidification, terrestrial eutrophication, marine eutrophication, and resource use categories. For almost all the above impact categories, hydrogen production from steam reforming is the most (88-97%) relevant source of environmental impacts of the hydrotreatment process. The only exception is human toxicity without cancer effects, where most of the impact is due to producing electricity (37%) and phosphoric acid (31%).



Figure 46. Breakdown of cradle-to-factory gate environmental impact of UCO-based PP per key-unit process (climate change shown without biogenic carbon removal (BCR))

Since a PlasticsEurope's "black box" Ecoprofile has been used for polymerization (see section 7.2.2.1), an interpretation at the activity level is not possible. Nevertheless, it is possible to identify the five impact categories where polymerization contributes with the highest share (on a weighted basis) on the total impact (see Appendix A): water use (10%), climate change (7%), resource use of fossil fuels (5%), human toxicity (cancer) (3%) and ionizing radiation (3%).

From, it is possible to see that steam cracking shows negative impacts in two categories: *human toxicity without cancer effects* and *freshwater eutrophication*. The "credit" received from the substitution of the net steam produced during the steam cracking overcompensates the two impacts caused by LPG production and combustion. The steam cracking process is responsible for a significant share of environmental impact in several impact categories: namely, climate change (ca. 30%), ozone depletion (ca. 75%), ionizing radiation HH (40%), land use (80%) and fossil fuels resource use (ca.50%). In particular, LPG production and its combustion account respectively for 18% and 82% of the cradle-to-factory gate impact on climate change. The production of LPG alone is instead entirely responsible for the impact caused in the other four impact categories.

7.4 DISCUSSION 7.4.1 Data uncertainty

This section reports several sensitivity analyses related to data uncertainty. The first assumption that is discussed is related to UCO collection. UCO was assumed to be sourced locally in the nearby regions of the bio-refinery, based on Neste's specific case. This assumption led to a small overall impact (5%) from UCO collection. Nevertheless, UCO has attracted much attention as one of the bio-based feedstocks to achieve European renewable energy and greenhouse gas (GHG) emissions targets. This has led to an increase in demand and the international trade of UCO [435]. Accordingly, it is reasonable that in the case of larger scale production of biobased PP, UCO may be collected globally. Globally sourced UCO has been modeled considering the macro-areas, which together account for more than 90% of the 2017 Dutch consumption: Asia (40%), US (25%), West EU (20%) and Netherlands (15%) [436]. According to a market study by [437], China has the highest UCO collectable potential. We assume that UCO is shipped¹⁴ from China (port Shanghai) to represent the imported Asian UCO. For the US and Western Europe, New York and London have been considered for calculating the transportation distances, while the collection in the Netherlands has been modeled assuming the same distances used for the baseline calculations. The first column of Table 26 shows the changes in the results of environmental impact results when UCO is globally sourced. Compared to the baseline analysis, the change of UCO sources leads to a significant increase in all 16 impacts. The most affected categories are particulate matter (+60%), photochemical ozone formation (+48%), acidification (+78%), terrestrial and marine eutrophication (+55-57%) and human toxicity without cancer effect (+22%). Overall, on a weighted basis, the environmental impact increases by 19%.

Another important assumption is related to the dataset used for hydrogen production, which is among the main environmental hotspots of this route. In the investigated system, hydrogen is produced from steam reforming of natural gas. According to the selected dataset [381], the production of 1 kg of hydrogen generates 9.4 kg CO_2 eq. Nevertheless, literature reports GHG emissions from steam reforming of natural gas ranging from 8.9 to 12.9 kg CO_2 eq./kgH₂ [438]. Accordingly, the climate change impact of bio-based PP production could vary in the range

¹⁴ From Ecoinvent 3.4: Transport, freight, sea, transoceanic ship {GLO}| market for | APOS)

from -2% to 12% considering other datasets. In the future, environmental impact reductions could be achieved if hydrogen could be produced using renewable energy, i.e. via electrolysis powered by photovoltaics [439]. In such a case, the climate change impact of UCO-PP could be reduced by a third. However, as green hydrogen is not expected to be commercially viable within the next decade, it is out of scope for our LCA, and therefore not investigated in more detail.

Table 26. Variation of cradle-to-factory gate impact assessment results by different assumptions on UCO collection and process optimization of using renewable propane and methane

Impact Category	Increase of impact (%) shifting from UCO locally sourced to sourced globally	Increase of impact (%) (negative values stand for a decrease) changing from conventional LPG in the baseline to the renewable propane and methane scenario
Climate change	10%	-42%
Ozone depletion	11%	-86%
Human toxicity, non-cancer effects	22%	-44%
Human toxicity, cancer effects	7%	2%
Particulate matter	60%	-31%
Ionizing radiation HH	6%	-41%
Photochemical ozone formation	48%	-5%
Acidification	78%	-17%
Terrestrial eutrophication	57%	4%
Freshwater eutrophication	12%	-15%
Marine eutrophication	55%	4%
Freshwater ecotoxicity	6%	-12%
Land transformation	13%	-95%
Water use	1%	0%
Resource use, minerals and metals	4%	2%
Resource use, fossil fuels	10%	-64%
Total weighted results	19%	-34%

LPG production and combustion has been identified as the second environmental hotspot of this route. The composition of LPG is variable depending on the location where it is sourced. For example, it can be 25% propane/75% butane in Italy and 95%/5% in Sweden [440]. In the baseline, we assumed that petrochemical LPG

is used to produce the UCO-based PP. However, it is possible to optimize the process energy requirement by using the propane from UCO obtained from the hydrotreatment to meet the need of LPG for dilution of the steam cracking and by using the methane produced in the steam cracking as energy source (see Figure 45). The environmental impact of UCO-based PP would vary as shown in the third column of Table 26. The overall environmental footprint of UCO-based PP would be reduced by 34%. In particular, these reductions of impact are high for climate change (-42%), ozone depletion (-86%), human toxicity without cancer effects (-44%), particulate matter (-31%), ionizing radiation HH (-41%), land use (-95%) and fossil resources (-64%). It is possible to notice that these reductions are even higher than the percentage of the impact caused by steam cracking on the overall environmental impact. This is consistent with the substitution approach used for steam produced. In fact, for these impact categories, the impact caused by steam cracking becomes negative when renewable propane and methane are used instead of LPG. The reason is that the impact caused by the production and combustion of renewable propane and methane (not the allocation of impact to methane produced by steam cracking) becomes lower than the credit for steam production (substitution).

Another important assumption is related to the datasets selected to model polymerization. As highlighted in the previous section, the impact of polymerization is particularly significant in five impact categories (on a weighted basis). For these impact categories, the share on the total impact has been validated using the data from the Matter study [432]. According to the Matter study, PP polymerization requires 2.1MJ of electricity¹⁵ and 1.3MJ of steam ¹⁶ (averaged¹⁷) per kg of polypropylene. Using the inventory data from Matter study instead of PlasticsEurope (see section *Life cycle inventory modeling*), the share of polymerization would shift from 96% to 88% for water use, from 25% to 46% for climate change, from 22% to 40% for resource use of fossil fuels, from 54% to 48% from human toxicity (cancer) and from 54% to 59% for ionizing radiation. Overall, considering all the impact categories, the total share of polymerization would shift from 38% to 48% on a weighted basis. The authors consider the 9% difference in line with the different temporal scope of the two datasets (due to improvements in process efficiencies that have occurred

¹⁵ Dataset used from Ecoinvent 3.4: Electricity, medium voltage {RER}| market group for | APOS

¹⁶ Dataset used from Ecoinvent 3.4: Steam, in chemical industry {RER}| production | APOS

 $^{^{\}prime\prime}$ Ranging between 0.8 and 1.8 MJ depending if the polymerization occurs in liquid phase, gas phase or suspension

over 20 years). Figure 47 summarises the results of the sensitivity analyses on data uncertainty on weighted bases. The variations are shown using the baseline values as 100%.



Figure 47. Sensitivity analysis on data uncertainty (normalized and weighted results with baseline values taken as 100%). Steam cracking using renewable propane and methane is shown as a negative impact, as their production and combustion have a lower impact than the credit for steam production (substitution). For more details, see the main text.

7.4.2 Model uncertainty: multifunctionality

This section provides a sensitivity analysis of the allocation approaches that were selected for the baseline calculations. The first sensitivity is related to UCO, which has been considered as a waste, being its use promoted in the European Union [9]. Accordingly, UCO has been treated with a cut-off approach in the baseline calculations. Nevertheless, the increasing demand for UCO for renewable diesel production has driven the high price of UCO in the past decade [435]. Accordingly, it might be argued that UCO should be considered as a by-product rather than a waste and a part of the impact caused in the first life (e.g. vegetable oil production) should be assigned to the recycled function (e.g. PP). For this sensitivity analysis, the 50/50 method instead of the cut-off approach is used. The 50/50 method assigns the credits and the burdens due to recycling to both previous (50%) and subsequent life cycle (50%). Accordingly, we allocate 50% of the impact of the primary production of vegetable oil to UCO. This method is the most conservative and it is usually applied for open-loop recycling when it is not known whether the use of the recycled material should be promoted [71].

Table 26 shows the variation of the environmental impact when the 50/50 method is applied to UCO. UCO is an oil waste derived from the use of oils and fats in cooking activities. For this sensitivity analysis, palm oil and soybean oil, whose data were available in the Ecoinvent database, were considered¹⁸.

When 50% of the environmental impacts of the production of the vegetable oils are allocated to the second life, the impacts of UCO-PP are significantly increased. On a weighted basis, the impacts of UCO-based PP would increase from 25% to 160% depending on the types of primary vegetable oil. In particular, larger variations are obtained when UCO origins from soybean oil. For climate change and the use of fossil resources, the environmental impact would increase by 17%-58% and 10-28%, respectively.

Impact Category	Increase (%) (negative values stand for a decrease) 50/50 Method palm oil	Increase (%) (negative values stand for a decrease) 50/50 method soybean oil
Climate change	17%	58%
Ozone depletion	6%	28%
Human toxicity, non-cancer effects	-12%	-3701%
Human toxicity, cancer effects	16%	166%
Particulate matter	113%	485%
Ionizing radiation HH	4%	12%
Photochemical ozone formation	42%	164%
Acidification	37%	127%
Terrestrial eutrophication	40%	155%
Freshwater eutrophication	37%	1282%
Marine eutrophication	55%	789%
Freshwater ecotoxicity	20%	3150%
Land transformation	3%	2496%
Water use	26%	20%
Resource use, minerals and metals	38%	347%
Resource use, fossil fuels	10%	28%
Total weighted results	25%	160%

Table 27. Variation of cradle-to-factory gate impact assessment results by using 50/50 method on UCO open-loop recycling

¹⁸ From Ecoinvent 3.4, Palm oil, crude {GLO}| market for | APOS, Soybean oil, crude {GLO}| market for | APOS.

The second sensitivity is related to the multifunctionality of hydrotreatment and steam cracking. In the baseline calculations, energy allocation was used for hydrotreatment, while direct substitution for net steam and energy allocation is applied for the steam cracking process. Alternatively, different allocation methods could have been followed:

- Energy allocation (only, not combined with direct substitution). In this approach, energy allocation is applied to all the co-products of steam cracking including steam. All the products have been valued with their lower heating values (LHVs) while the energy value of steam has been considered to be its enthalpy. Unlike the hybrid method applied for the steam cracking process in the baseline, no credits for steam substitution have been assigned (strictly attributional LCA and consistent with RED).
- Exergy allocation (only, not combined with direct substitution). Compared to
 the baseline, the only difference is that exergy allocation has been applied to
 the steam cracking unit (for hydrotreatment using exergy or energy allocation
 key is indifferent). The reason behind this choice is that exergy can account for
 different quality in energy carriers. The superheated conditions of the steam
 released by steam cracking are 520 °C and 110 bars. This flow has been valorized
 with the exergetic value of this steam at such conditions. This means that it is
 assumed that this steam is entirely recovered and directly used, e.g. as process
 steam input in other refinery processes. The other co-products are energy
 and chemical products whose exergy value has been approximated with their
 LHVs. For the steam in input to the steam cracking unit, Ecoinvent database
 has been used¹⁶.
- *Cut-off.* Differently, from the baseline calculations, a cut-off approach is applied to the hydrotreatment unit process for bio-based naphtha. Due to the minor production share of bio-based naphtha compared to renewable diesel, all the environmental burdens of hydrotreatment are assigned to the renewable diesel. In this case, bio-based naphtha comes into the system as an 'emissionsfree' input and, therefore, no impact has been apportioned for UCO collection and NEXBTL process. This is consistent with the model for this conversion route developed by the Joint Research Centre (JRC) of the European Commission [350]. In their model, HVO was the investigated product and the JRC applied a cut-off approach, neglecting the very small fraction of naphtha. The hybrid approach used for the steam cracking process remains unchanged.

- *Cut-off & energy allocation.* In this case, the cut-off approach is applied to the hydrotreatment unit process. Differently from the previous case, solely energy allocation is applied to the steam cracking unit.
- By-products substitution. In this case, system expansion followed by substitution is applied to all the by-products of hydrotreatment and steam cracking (but not to the co-products, i.e. diesel and ethylene). All the credits are assigned to the co-products diesel/naphtha and propylene/ethylene that is then partitioned by energy allocation. In the hydrotreatment unit, renewable propane has been assumed to replace petrochemical propane¹⁹. For the steam cracking unit, the following by-products are substituted with the conventional processes which would be avoided: steam, hydrogen, bio-methane, bio-based benzene, and bio-butadiene²⁰. The other by-products (C5, C7, and C8) that have not been substituted have been considered as neutral, i.e. neither burdens or related credits are caused by them. For the baseline calculations, they were instead accounted in the energy allocation shares.

Figure 48 shows that energy and exergy allocation worsen the environmental footprint of bio-based PP. The only exception is resource use (minerals and metals) with energy allocation, although the difference is minimal (1%). This increase in impacts is caused by disregarding the credit from the substitution of steam in the baseline. Moreover, energy and exergy allocation lead to very similar results but impacts are higher in the case of exergy allocation. The reason behind this is that the exergy value of steam is lower than the enthalpy value. The weighted impact increases by 35%. Hence, it is concluded that exergy allocation is the most conservative among the assessed approaches.

The cut-off and by-product substitution approaches provide a significantly lower environmental footprint for bio-based PP. On the other hand, the by-products substitution approach is also the one assigning the highest impact for ionizing radiation HH and resource use of minerals and metals. The reason for this is that the credits for direct substitution of the by-products above do not compensate for the higher amount of impact apportioned to propylene. Moreover, negative impact results are obtained when system expansion followed by substitution is applied in

¹⁹ From Ecoinvent 3.4, Propane {GLO}| market for | Conseq

²⁰ From Ecoinvent 3.4, Steam, in chemical industry {RER}| production | Conseq, Natural gas, from medium pressure network (0.1-1 bar), at service station {GLO}| market for | Conseq. Butadiene {RER}| production | Conseq. From PlasticsEurope's Ecoprofiles, Hydrogen (reformer) E from PlasticsEurope [381]), Benzene, at plant/RER based on PlasticsEurope Industry 2.0 database [449]

the following impact categories: human toxicity without cancer effects, particulate matter, freshwater eutrophication, freshwater ecotoxicity and resource use of fossil fuels. This is caused by the by-products of hydrotreatment (propane) and steam cracking (especially bio-methane and benzene), which displace products that have high impacts for these five categories.

Conversely to other methods, the weighted impact decreases by (-) 113% by using the by-products substitution method (overall negative impact) compared do the baseline values. These negative flows violate the desirable characteristics of an attributional LCA [51]. This negative impact means that the perturbation logic of substitution has created "links between emissions and activities that are not mediated by product or service flows" [51].



Figure 48. Sensitivity analysis of the allocation approaches for hydrotreatment and steam cracking units (numerical values in Appendix B). Baseline results expressed as 100%.

Moreover, the results presented in Figure 48 confirm the findings of Sandin et al. [57] that when a non-dominant product is in focus: 1) the results are sensitive to the choice of the allocation method and 2) the substitution method provides results in contrast with other allocation methods.

These two findings apply only when non-dominant products are in focus. Thamsiriroj and Murphy, who studied the same route but with renewable diesel (HVO) in focus (the physically dominant product), concluded that UCO HVO is relatively unaffected by allocation methodology (energy allocation, substitution, and cut-off approaches) [441]. Moreover, these results show that the RED statement that energy allocation provides results generally in line with the substitution method fails when a non-dominant product is investigated and cannot be extended to more impact categories than climate change.

The authors, therefore, recommend avoiding the use of the substitution method in attributional studies because its application can lead to results in contrast with other methods. The only exception can be the use of direct substitution for byproducts when this can represent physical causality. Nevertheless, this should be based on a direct and empirically demonstrable relationship [212]. This would also be in line with ISO 14044 recommending allocation by-physical causality shall be preferred to other allocation methods. When a mathematical model is not available, the use of direct substitution shall be validated by comparing it with other allocation (partitioning) methods. For this case study, the baseline results, where direct substitution has been applied, have been validated by comparison to energy, exergy, and cut-off allocation methods. Moreover, the baseline results can be considered the most representative being an average among the possible allocation methods for all 16 impact categories and on a weighted basis.

7.4.3 Environmental benchmarking of UCO-based PP

The lack of harmonization in LCA method limits the direct comparison between bio-based and petrochemical materials studies, especially when considering multiple impact categories [7]. Specifically, such a comparison is often reliable only for climate change impacts [7]. In particular, this issue emerges when different black box datasets for petrochemical PP are compared (see Appendix C). From the analysis in Appendix C, it is possible to consider also resource use (fossil fuels) along with climate change as a reliable impact category (less than 10% variation). From cradle to factory gate, 1 kg of petrochemical PP causes 1.65-1.78 kg CO_2 eq GHGs and the resource use (fossil fuels) ranges between 67 -74 MJ (see Appendix C). To be conservative, the two lower values have been used for comparison. Hence, UCO-based PP (baseline) has a 62% lower impact on climate change and 86% for resource use (fossil fuels) compared to petrochemical PP.

Moreover, these reductions are almost unaffected when UCO is globally imported (58% and 85% respectively). By using the 50/50 method, these reductions are reduced to the minimum of 40% and 82% respectively for climate change and fossil resources use when UCO is derived from soybean oil. When exergy allocation is used for hydrotreatment and steam cracking, these environmental impact reductions are instead reduced to a minimum of 45% in terms of climate change and to 80% as resource use of fossil fuels. From these results, it can be concluded that UCO-based PP is a favorable alternative option to petrochemical polypropylene in terms of climate change and fossil fuel resources.

It should be kept in mind that these impact reductions are for the cradle-to-factory gate scope and the differences above in climate change impact do not account for BCR. Thus, UCO-based PP has the potential advantages to act as a biogenic carbon sink if the material is recycled. Moreover, biogenic carbon emissions are released when burned in a waste-to-energy system. These advantages are out of the scope of this LCA and therefore not estimated.

Comparing our results with the LCA published by Kikuchi et al. [415], it is found that UCO-based PP shows about 80/90% lower impact on climate change compared to bio-based PP made from sugarcane and woody biomass at factory-gate.

It should be recognised that UCO is a very limited feedstock. The European Commission already promotes its use for renewable diesel as a second-generation biofuel [434]. UCO-based PP is developed from the bio-based naphtha, which is a by-product of this renewable diesel. From an environmental perspective, it would be therefore interesting to assess what is more attractive between replacing petrochemical diesel or polypropylene. "Sidestream naphtha" case was chosen as baseline due to the current strong market demand for renewable diesel (main product). Nevertheless, the cracking of all HVO diesel and naphtha would be technically feasible. Increasing the mixture of bio-based HVO naphtha and diesel used for bio-based PP would not lead to a different environmental impact compared to the baseline calculation due to the slightly different LHVs

assumed for bio-based naphtha and diesel (energy allocation). It is known that UCO renewable diesel could lead to GHG emissions saving up to 88%, which are much higher compared to 40-62% savings by other biodiesels [435]. These savings could appear higher than the ones allowed by UCO-based PP. Nevertheless, we are not able to answer this question properly without proper modeling for the end of life of polypropylene.

7.5 CONCLUSIONS

The first objective of the study is to identify the major environmental burdens in the cradle-to-factory gate life cycle of UCO-based PP. The environmental footprint of UCO-based PP is dominated by the polymerization process (38%), the production of hydrogen (21%), the production of LPG (18%) and the combustion (8%) of LPG. Climate change (28%), fossil resource use (23%) and water use (11%) have resulted in being the most important environmental impacts of UCO-based PP.

It is found that if the renewable propane and methane from UCO could be used instead of LPG, the environmental footprint of UCO-based PP could be potentially further reduced by 34%. Bio-based PP from UCO might also offer significantly better environmental performances in terms of climate change compared to bio-based PP from sugarcane and woody biomass (-80/90% at factory gate).

Compared to petrochemical PP, UCO-PP offers substantial impact savings for climate change (62%) and for fossil fuel resource use (86%). These savings remain substantial even in the cases of 1) globally imported UCO (58% and 85% respectively), 2) when UCO is considered a by-product instead of waste (40% and 82% respectively based on the 50/50 method) or 3) when a different allocation approach is used for hydrotreatment and steam cracking (savings of 45% and 80%, respectively). From these results, it can be concluded that bio-based PP from UCO is a promising alternative option to replace petrochemical polypropylene in terms of climate change and fossil fuel resources.

It should be reminded that the comparisons made above are for the scope of cradle to factory gate. UCO-based PP has the further advantage of having a 100% biogenic carbon content embedded in the product, potentially for the long term (e.g. in a durable application). The full biogenic carbon balances should be accounted for in a future cradle to grave LCA when a final product made from UCO-PP is analysed. The second objective of the study is to scrutinize how the allocation procedures used to solve multifunctionality affect the results of this LCA. It is found that exergy allocation leads to an increase of 35% of the environmental footprint of UCO-based PP compared to the baseline in which a hybrid direct substitution and energy allocation is applied. Conversely, the environmental footprint would become negative by using system expansion followed by substitution. The negative footprint obtained by using substitution is because bio-based naphtha and propylene, which are the precursors of UCO-based PP, are two physically non-dominant output-products of multifunctional processes. Such a negative impact is a clear violation of the desirable characteristics of an attributional LCA. It is recommended to avoid the use of system expansion followed by substitution in attributional studies. The only exceptional cases are the ones where direct substitution of by-products can represent physical causality like applied to the steam produced during steam cracking. This steam produced is directly used by other processes of the same biorefinery, and, otherwise, should be produced as marginal production of refinery steam. In fact, this would result in line with ISO 14044 recommending allocation by physical causality shall be preferred to other allocation methods. As a mathematical model is not available to model physical causality relationships, the use of direct substitution for steam has been validated by comparing with other allocation methods, showing alignment in all the impact categories assessed.

It is concluded that economic significance should be considered as an important requirement to fulfill before applying substitution. When this requirement is not respected, direct substitution shall be avoided.

Last but not least, UCO has been used as feedstock for a variety of applications of chemicals and fuels. The increasing demand for UCO has driven the price up in the past years. In this study, the impact of considering UCO as a by-product instead of a waste was assessed. It is found that the LCA results could significantly vary depending on the type of original vegetable oils as well as how the allocation is performed (e.g. based on the 50/50 approach). The results provided in this study should be used to elicit the discussion in the context of assessing the impacts of products in a future circular and bio-based economy.

7.A APPENDIX: DETAILED BREAKDOWN PER LIFE CYCLE STAGE AND INTERPRETATION

Table 7A.1 provides the numerical breakdown of the environmental impact assessment per unit process. Table 7A.2 shows the same results on weighted basis.

Table 7A.1. Breakdown of cradle-to-factory gate impact assessment results (1 kg of bio-based PP)

Impact Category	Unit	UCO- collection	NEXBTL	Transport of naphtha	Steam cracking	Polymerization
Climate change (w/o BCR)	kg CO ₂ eq	4%	40%	2%	29%	25%
Ozone depletion	kg CFC- 11 eq	5%	7%	0%	74%	14%
Human toxicity, non-cancer effects	CTUh	32%	38%	5%	-71%	96%
Human toxicity, cancer effects	CTUh	4%	36%	6%	1%	54%
Particulate matter	kg PM2.5 eq	10%	40%	4%	6%	39%
Ionizing radiation HH	kBq U235 eq	3%	4%	1%	40%	54%
Photochemical ozone formation	kg NMVOC eq	7%	37%	4%	21%	31%
Acidification	molc H+ eq	6%	39%	4%	9%	43%
Terrestrial eutrophication	molc N eq	7%	46%	5%	19%	23%
Freshwater eutrophication	kg P eq	5%	26%	6%	-59%	121%
Marine eutrophication	kg N eq	7%	45%	4%	19%	24%
Freshwater ecotoxicity	CTUe	24%	16%	1%	9%	50%
Land transformation	kg C deficit	8%	6%	1%	80%	5%
Water use	m³	0%	1%	0%	0%	99%
Resource use, minerals and metals	kg Sb eq	29%	28%	2%	14%	27%
Resource use, fossil fuels	MJ	4%	25%	1%	47%	22%

Table 7A.2. Breakdown of cradle-to-factory gate impact assessment results on weighted basis

Impact Category	Unit	UCO- collection	NEXBTL	Transport of naphtha	Steam cracking	Polymerisation
Climate change (w/o BCR)	kg CO ₂ eq	1%	11%	0%	8%	7%
Ozone depletion	kg CFC- 11 eq	0%	0%	0%	0%	0%
Human toxicity, non- cancer effects	CTUh	0%	0%	0%	-1%	1%
Human toxicity, cancer effects	CTUh	0%	2%	0%	0%	3%
Particulate matter	kg PM2.5 eq	1%	3%	0%	0%	2%
Ionizing radiation HH	kBq U235 eq	0%	0%	0%	2%	3%
Photochemical ozone formation	kg NMVOC eq	0%	2%	0%	1%	2%
Acidification	molc H+ eq	0%	2%	0%	0%	2%
Terrestrial eutrophication	molc N eq	0%	1%	0%	0%	1%
Freshwater eutrophication	kg P eq	0%	0%	0%	0%	0%
Marine eutrophication	kg N eq	0%	1%	0%	0%	0%
Freshwater ecotoxicity	CTUe	0%	0%	0%	0%	0%
Land transformation	kg C deficit	0%	0%	0%	2%	0%
Water use	m ³	0%	0%	0%	0%	10%
Resource use, minerals and metals	kg Sb eq	0%	0%	0%	0%	0%
Resource use, fossil fuels	MJ	1%	5%	0%	11%	5%

7.B APPENDIX: MULTIFUNCTIONALITY SENSITIVITY ANALYSIS

Table 7B.1 provides the numerical results of the sensitivity analysis on multifunctionality.

Table 7B.1. Numerical values of sensitivity analysis on multifunctionality

Impact Category	Unit	Energy allocation	Cut off & energy allocation	Exergy allocation	Baseline	Cut-off baseline	By-products substitution
Climate change (w/o BCR)	kg CO ₂ eq	7.6E-01	4.9E-01	9.1E-01	6.3E-01	3.6E-01	2.8E-01
Ozone depletion	kg CFC-11 eq	9.9E-08	9.0E-08	1.1E-07	9.0E-08	8.0E-08	5.1E-08
Human toxicity, non-cancer effects	CTUh	2.3E-08	1.7E-08	2.9E-08	1.1E-08	4.4E-09	-1.7E-08
Human toxicity, cancer effects	CTUh	5.9E-09	3.9E-09	6.2E-09	5.5E-09	3.4E-09	3.1E-09
Particulate matter	kg PM2.5 eq	1.7E-04	1.1E-04	2.2E-04	1.2E-04	6.2E-05	-1.5E-04
lonizing radiation HH	kBq U235 eq	6.4E-02	6.1E-02	6.7E-02	6.4E-02	6.0E-02	7.0E-02
Photochemical ozone formatior	kg NMVOC eq	2.1E-03	1.3E-03	2.3E-03	1.9E-03	1.1E-03	4.2E-04
Acidification	molc H+ eq	2.6E-03	1.7E-03	3.0E-03	2.1E-03	1.2E-03	-4.8E-04
Terrestrial eutrophication	molc N eq	6.7E-03	3.7E-03	7.3E-03	6.0E-03	2.9E-03	2.9E-03
Freshwater eutrophication	kg P eq	1.5E-05	1.3E-05	1.6E-05	8.7E-06	6.4E-06	-9.7E-06
Marine eutrophication	kg N eq	6.1E-04	3.4E-04	6.7E-04	5.6E-04	2.7E-04	2.8E-04
Freshwater ecotoxicity	CTUe	2.4E-01	1.6E-01	2.6E-01	2.2E-01	1.4E-01	-2.4E-02
Land transformation	kg C deficit	1.2E+00	1.1E+00	1.3E+00	1.1E+00	9.6E-01	8.3E-01
Water use	m ³	7.4E-01	7.3E-01	7.4E-01	7.4E-01	7.3E-01	5.6E-01
Resource use, minerals and metals	kg Sb eq	3.1E-07	1.6E-07	3.2E-07	3.2E-07	1.5E-07	4.6E-07
Resource use, fossil fuels	MJ	1.1E+01	8.5E+00	1.3E+01	9.3E+00	6.6E+00	-1.6E+01
Weighted	р	6.0E-03	4.3E-03	6.9E-03	5.1E-03	3.4E-03	-6.8E-04

7.C APPENDIX: PETROCHEMICAL POLYPROPYLENE?

Large differences in the most of the impact categories can be noticed when comparing PlasticsEurope²¹ and Thinkstep Gabi²² datasets for petrochemical PP. The result of this comparison is presented in Figure 7C.1. Due to the large variations, these dissimilarities cannot be justified only by the different geographic scope (Europe and Germany).



Figure 7C.1. Variation of impact for petrochemical PP comparing different datasets.

²¹ Polypropylene, PP, granulate, at plant/RER

²² DE: Polypropylene granulate (PP) mix ts


8

CHAPTER

Summary, conclusions and recommendations

8. SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

There is a growing interest in converting local bio-based "residues" streams into high-value products i.e. chemicals, energy and materials. Compared to bio-based products from dedicated crops, this type of bio-based product can limit potential trade-offs such as land competition, potential biodiversity damages and food security issues. Moreover, local bio-based residues do not require significant imports of bio-based products and biomass from outside the EU. Furthermore, avoiding dedicated cultivation, products from residual flows are expected to have a much lower environmental impact than bio-based products from dedicated crops.

However, the literature has previously emphasized that calculating the life cycle environmental impact of products from bio-based residues is challenged by several methodological uncertainties. In particular, one of the most relevant uncertainties regards the so-called LCA multifunctionality issue, i.e. the allocation of environmental burdens among the process co-products. In fact, a variation in the allocation type or in the input data used for determining allocation factors has a much higher effect on the environmental burdens of low-economic/physical significance streams than for the main products.

On top of it, selecting the multifunctionality approach is challenged by various inconsistencies in the reported interpretation of the so-called ISO multifunctionality hierarchy in ISO-compliant LCA guides. Since different guides are prominent references in different bioeconomy sectors or countries, inconsistent multifunctionality practices are claimed as "ISO-compliant" in LCAs of bio-based products. Consequently, significantly different results for products with similar life cycle inventory data are reported in the literature. The literature has acknowledged such inconsistency several times. However, the origins of such inconsistent interpretations are not fully understood and the magnitude of such inconsistency among practices applied in different bioeconomy sectors has never been quantified so far.

For this reason, the first research question (RQ) that this thesis aimed to answer was: what multifunctionality practices are adopted in LCAs of bio-based products and how can the consistency of the life cycle inventory model be improved?

In this thesis, current practices in implementing ISO 14044 multifunctionality

recommendations in LCAs of bio-based products have been quantitively reviewed for the first time. The same review also contains an investigation of the origins and rationales of ISO multifunctionality hierarchy's interpretations. This investigation was fundamental to select the multifunctionality practices adopted in the LCA case studies presented in this thesis by considering only those fully "ISO compliant". While the ISO-guidelines review (Chapter 3) quantified multifunctionality practices for products in various bioeconomy value chains, the lignin–use review (Chapter 2) looked at those applied in LCAs related to one of the feedstocks of interest, i.e. lignin.

Besides reviewing the LCAs of various lignin applications, four LCAs were conducted for bio-based asphalts utilizing lignin, bioenergy from wood chips, a bio-jet fuel from potato by-products and polypropylene from used cooking oil. The selection of these feedstocks prioritized 1) bio-based by-products that are locally available in the EU (with a major focus on the Netherlands) and 2) those for which the collection of primary data for modeling the respective innovative conversion technologies was possible.

These LCAs were conducted to answer this thesis's second research question (RQ2) is: What is the environmental impact of novel bio-based products made from byproduct/waste streams compared to their fossil counterparts?

All the investigated high-value bio-based products are produced using innovative emerging technologies to convert agro-industrial and food-processing by-products. This thesis presents the life cycle environmental impact of several innovative technologies for the first time as well as their environmental performance compared to their petrochemical counterparts. More than just climate change, this thesis aims to look at the broad picture of the environmental impacts of innovative bio-based products from residual streams and their environmental hotspots in multiple impact categories. In this way, it was possible to understand if the lower impact of residue-based products compared to bio-based products from dedicated crops also applies to other environmental issues than climate change. Accordingly, environmental impacts that are generally higher for bio-based products than petrochemical products, such as eutrophication and acidification, were considered. The analysis of the environmental hotspots was fundamental to recommend possible ways to reduce the overall environmental footprint of each product robustly. For each proposed action, potential trade-offs in burdensshifting were analyzed. Furthermore, several comparisons between products from bio-based by-products and comparable bio-based products from dedicated crops were also performed.

Since the scientific literature highlighted the magnitude of multifunctionality uncertainty on the reported climate change impacts of bio-based products from residual streams, this thesis aims to answer the third RQ: *How are the LCA results of such products influenced by the approach adopted to deal with the multifunctionality issue?*

Various sensitivity analyses on the multifunctionality approach were conducted for the LCAs performed. Previous literature assessing bio-based by-products mainly performed this type of analysis on climate change and rarely on a broad spectrum of environmental indicators. Conversely, this thesis aims to understand the environmental impact categories for which the multifunctionality uncertainties hamper the comparison with the petrochemical counterparts, making it inconclusive.

8.1 RESEARCH QUESTION 1

1. What multifunctionality practices are adopted in LCAs of bio-based products and how can the consistency of the life cycle inventory model be improved?

The first part of this research question i.e. *what multifunctionality practices are adopted in LCAs of bio-based products*? was provided in two reviews. The review presented in Chapter 2 focused on the effect of multifunctionality practices in comparing the results from 42 peer-reviewed LCAs regarding lignin and derived products.

The review presented in Chapter 3 looked at practices in implementing ISO 14044 multifunctionality recommendations in LCAs of products belonging to diverse bioeconomy sub-sectors and LCAs of the petrochemical products used as the benchmark to calculate their environmental performance.

The second part of RQI i.e. how can the consistency of the life cycle inventory model be improved? was answered based on the review presented in Chapter 3 using key-route main path analysis to identify the major publications that influenced the debate under the interpretation of the ISO-multifunctionality hierarchy. This exercise allowed us to detect the origins of the different interpretations, their underlying theories and not-(fully) ISO compliant practices. These insights were fundamental to 1) select the allocation method in all LCA case studies reported in the following chapters and 2) prioritize fully ISO-compliant methods as alternative methods applied in the sensitivity analyses.

8.1.1 Lignin use-review

Lignin is the second most abundant natural biopolymer on Earth and accounts for about 30% of the organic carbon in the biosphere [56,266]. The complexity of the lignin molecule and its physical characteristics make the conversion into highervalue products interesting and challenging at the same time. Chapter 2 presented a review of 42 peer-reviewed LCAs regarding lignin and derived products. Most emerging high-value lignin-based applications presented considered by these 42 studies seem to offer promising climate change performances. From a climate change perspective, transportation fuels, adipic acid and propylene seem the most promising applications currently at the development stage to utilize lignin.

Lignin, which is currently mainly used as low-value on-site fuel, is always a byproduct from a multifunctional production process of pulp mills or lignocellulosic biorefineries. As shown by the summary of the adopted multifunctionality practices in the 42 reviewed LCAs illustrated in Figure 4, a wide variety of multifunctionality practices were applied. This fact is not a problem per se but is a problem if the rationale for their selection is inconsistent or they derive from inconsistent application of ISO 14044:2006 recommendations. Mass allocation was the most adopted method to deal with lignin multifunctionality (13 out of 42 studies). However, this method has the limit of assigning the same impact per kg of product to lignin and the main products of the multifunctional system (pulp mill or lignocellulosic biorefinery). This does not reflect the value of lignin (e.g. energetic or economic) and the fact that lignin is a by-product and not the main product of pulp mills and lignocellulosic biorefineries. Three studies also applied the cut-off method. However, a cut-off allocation does not seem appropriate if lignin cannot be considered a waste in both ISO LCA and EU legislation terms (e.g. see the EU waste framework directive and the updated EU Renewable Energy Directive). This circumstance applies to lignin with a market price that cannot be treated as a waste. Moreover, if lignin is no more internally combusted and a different fuel is used, the environmental impact of pulp or the bio-based products from the lignocellulosic biorefineries is affected.

Substitution was adopted as a system expansion approach by four studies. However, this method could also lead to numerically negative impacts for lignin if the main product is erroneously substituted. Regarding such a result, some authors of the articles reviewed in Chapter 2 referred to inconsistent results caused by doubtful interpretation of ISO 14044:2006. Accordingly, they state that "to avoid such pitfalls, it is recommended that LCA practitioners, sustainability scientists, and the chemicals industry collaborate to form a consensus on a standardized LCA approach to account for co-product flows for bio-based chemicals" [65]. However, applying substitution to obtain the impact of lignin by subtracting an alternative production of pulp is conceptually not correct and not an option allowed by ISO 14044:2006. In fact, even in consequential LCAs, only by-products can be substituted and not the main product. A check on co-products' economic significance is necessary before applying substitution to avoid erroneous application (as remarked before also by several authors [51,211] and the ILCD handbook [70]). However, some authors [57] remarked that ISO does not emphasize this aspect enough. Conversely, the substitution method would be applicable for those LCAs focusing on the main product of a system co-producing lignin. In those cases, lignin could be substituted if it is not an economically significant product i.e. a by-product.

Even for climate change only, allocation practices could significantly affect the environmental impact of lignin and derived applications. Taking the example of lignin-derived transportation fuels, the allocation method applied was one of the two reasons for the significant change of the emissions savings compared to diesel ranging between 10% and 90% (the energy source i.e. biomass or fossil fuels, was the second main factor). The cut-off allocation (lignin considered a by-product free of burden) led to the highest climate change savings for lignin-based transportation fuels.

Applying allocation to the biogenic carbon was one of the main comparability issues for the reported climate change impacts of lignins in the literature. However, the actual biogenic carbon content of the final products should be preferred to an allocated biogenic carbon, since there is no allocation parameter (physical or economic) that can reflect how the carbon from biomass ends in the co-products. For example, bio-based products with no carbon content can be derived from biomass feedstock with an important carbon content (e.g. bio-based hydrogen could be obtained from wood chips). The reviewed case studies showed promising climate change performances and trade-offs in other impact categories with some conflicting results for similar lignin-based applications. For example, divergence was found in the outcome of comparisons with petrochemical counterparts of three LCAs of lignin-based adhesives [104,115,132] for the same end-point categories i.e. damage to human health, ecosystem guality and resources. Arias et al. [115] concluded that ligninbased adhesives have between 2.5 and 4.5 times higher impact than conventional adhesives. At the same time, the other two LCAs showed a better performance for lignin-based adhesives in the same categories. However, the study from Arias et al. lacks transparency on the allocation practices and handling of biogenic carbon for lignin, limiting a full understanding of the reasons leading to the different outcomes. The same lack of transparency on these critical methodological choices was reported in the only LCA of lignin-based asphalts found and reviewed in Chapter 2 [134]. Since dealing with allocation and biogenic carbon significantly affects the outcome of the LCAs of bio-based products, it is strongly recommended to illustrate their modeling transparently.

8.1.2 ISO guidelines -review

Regarding substitution, the review of Chapter 3 clarified that the concept of equivalency between substitution and system expansion originally illustrated by Tillman and Ekvall [80,81] does not mean that they are equivalent in practice. In fact, system expansion and substitution do not provide the same results (as remarked already remarked before by several authors [240]). Moreover, when the first ISO hierarchy was drafted, system expansion as enlargement and not as substitution was the only option. The ISO compliance of system expansion in the form of substitution originated from the publication of the annex of ISO 14041. However, this annex stated that substitution is possible only if the following conditions are met: 1) the LCA study's goal is assessing the long-term marginal effect of a change and 2) the change modeled can be predicted with low uncertainty. This annex was then removed from the current ISO 14044:2006. However, nowadays, the LCA goal and modeling described in that annex are well-known under "consequential LCA". However, this term became recognized much later than when this annex was published.

Accordingly, in the attributional LCAs presented in this thesis, system expansion was applied only via enlargement, i.e. as literally explained in ISO 14044: "expanding

the product system to include the additional functions related to the co-products," e.g. modifying the functional unit to include all co-products. Such expansion by enlargement is applicable in both attributional and consequential LCAs. Still, it cannot be applied if the study's goal requires obtaining the impacts of just one of the co-products or by-products because such results would not be available (e.g. also remarked by [234,442]). In these cases, ALCA cannot avoid allocation while applying substitution as a system expansion method is instead possible for CLCA.

On this basis, distinguishing ALCA/CLCA and using substitution as a system expansion method only in CLCA could be a way to increase consistency in future LCAs of bio-based products (and others). Comparing ALCA results only with ALCA results and not with CLCA results would increase the comparability of the results in more categories than climate change. The text mining process applied in Chapter 3 highlighted that the practice of strictly distinguishing between attributional and consequential modeling is already implemented in 25% of the LCAs in the scientific literature with a highly variable percentage among bioeconomy subsectors (see Figure 10). Some guides [85,267,443] specify their modeling approach as attributional and accordingly do not allow substitution as a system expansion method. However, some authors [68,202] recommend that the distinction ALCA/CLCA should also be present in future ISO 14044 to avoid substitution in attributional LCAs.

The review presented in Chapter 3 also clarified the nature of the so-called ISO "physical relationships" since they have been interpreted uniformly by the studies belonging to the main path. <u>Physical relationships used to define the allocation criterion in ISO's second level should represent "causal relationships" instead of relationships based on any physical parameter.</u> Such relationships can be modeled only if the ratio of the output products can be varied i.e. the functional outputs can be varied independently. This allows establishing physical causality between functional units via mathematical modeling by changing the operating conditions (a concept already remarked by several authors [97,157,214–217]). <u>Hence, a future ISO 14044 should remark on this concept.</u> However, this interpretation is currently present only in 28% of LCAs on multifunctional bioeconomy systems (see Figure 12 in Chapter 3).

Conversely, such interpretation was adopted far more widely (see Figure 12 in Chapter 3) in LCAs of petrochemical products (>60%). Oil refining processes have been historically the most cited example of applying physical causality allocation

based on linear programming (LP). For this reason, it is still common practice to use this method for oil products. For example, LP-based marginal refinery emissions are incorporated in the so-called fossil fuel comparator i.e. the reference value of carbon emissions of fossil fuels used to calculate the GHG emissions savings of biofuels in the EU legislation [9,37]. The selection of the allocation method based on physical causality is also common practice in LCAs of dairy and meat products (\approx 40-45%). In this case, the reason is that physical relationships are interpreted as physical causal relationships by a leading guide developed by the International Dairy Federation (IDF) [223]. Physical parameters can seldom represent physical causality. The most cited exception is the transportation of heavier products that linearly increase the truck's fuel consumption. As remarked by ISO 14049:2012 (i.e. the ISO report on applying ISO 14044 to the inventory analysis), the reason is the "linear interdependence of the fuel consumption and the load mass" independently if the additional mass is of product A or B. Such linear independence implies that mass allocation quantitatively reflect how the input (fuel) is changed by changes in the amount of each product of the system i.e. the transported load (mass of e.g. product A).

Except for these rare cases, if detailed modeling (e.g. LP) is not possible/too timeconsuming or the ratio between products is fixed, it is necessary to allocate by "other relationships". The third-level ISO allocation is predominant in bioeconomy LCAs, but the selection procedure adopted is often not transparent with missing sensitivity analyses [83,94,285].

The review of Chapter 3 clarified that "other relationships" should be interpreted as other causal relationships whose allocation key (e.g. economic value, energy content, or mass basis) should be based on a proxy of physical causality (e.g. earlier remarked by [201,215,222,232]). Hence, the frequent allocation practice of using an arbitrary parameter for which no causality can be justified should be avoided. If the co-product ratio cannot be varied, no physical property can be used as a proxy for physical causality (on this last point, there seems to be a shared consensus by experts on LCA multifunctionality, as remarked recently by [444]). If the ratio can be varied, the guidance for justifying the "other relationship" selection is given (insufficiently) only by ISO 14049:2012 and, unfortunately, not in the more known main text of ISO 14044:2006. How to approximate causal relationships in this last case has been a debated scientific issue for two decades. According to ISO 14049:2012, economic allocation is often the preferred option (see the example of bitumen). Economic allocation well represents why the products are produced (i.e. to generate revenues). However, the preference for a physical parameter is often defended if the goal of the LCA is to "understand and manage the environmental implications of the material and energy flows associated with efforts to meet human needs through economic activity" [444]. In this way, it is possible to directly link inputs, outputs, and impacts via allocation by physical parameters. However, material flow analysis more than LCA is the correct tool to provide the information required by this goal definition (as remarked recently by [444]).

On the other hand, there are also a small number of cases when economic allocation does not well represent (other) causal relationships. In this case, the choice of the allocation parameter remains value-laden. A typical example is a waste management process requiring the allocation of its environmental burdens and benefits to the different waste inputs. Waste management is not a human activity meant only to generate revenues selling a product e.g. energy or a recycled bottle, but also a necessary service to treat various wastes and/or give them a new life. Suppose the treated products are uniform, e.g. an average mix of plastic wastes with food contamination. It is necessary to allocate the environmental burdens (and benefits) from their treatment to the original plastic products and food products. The price of the original food product contaminating the plastic waste might be much higher than that of the plastic item e.g. a packaging film. In that case, the environmental impact and energy generated by incinerating such wastes can be assumed to be linearly proportional to the energy contents of the treated plastic wastes (input 1) and food contamination (input 2). At the same time, there is a much lower proportionality to the economic values of food and plastics. Suppose the same waste flow is treated via recycling. In that case, the environmental impact of the recycling process and the amount of recycled plastic varies linearly with the mass of plastic waste (input 1) and not with the mass of food contamination (input 2). For this reason, ISO 14044 states that mass/energy allocation, the number of useful cycles and economic allocation are all possible options for open-loop recycling.

8.1.3 List of recommendations

Based on the lessons learned from Chapters 2 and 3, the following summary of recommendations to properly deal with the multifunctionality issues in LCAs of bio-based products can be formulated:

- 1. The rationale of the multifunctionality approach should always be transparently illustrated;
- Allocation parameters not reflecting the ISO causality principle should be avoided;
- 3. By-products with a market value should not be considered wastes, and as such cut-off allocation should be avoided.
- 4. LCA results of a bio-based and conventional product should be compared based on allocation methods following the same principle (i.e. ISO causality);
- 5. Such an allocation should reflect the aim of the process of producing the main product(s) and not the by-product.
- 6. Using the distinction between attributional and consequential modeling becomes an important way to increase the consistency and comparability of results further, and ISO should provide their definitions and uses since those proposed by SETAC are not universally accepted (as recently remarked by Schaubroeck et al. [59]); On this basis, consequential LCA modeling should be used complementarily to attributional LCAs and not using hybrid modeling for a correct picture of the environmental impacts.
- 7. The use of substitution should be linked with a specific goal of the LCA, i.e. assessing a change via consequential modeling;
- Before applying substitution in consequential LCAs, a check on the economic significance of co-products is necessary. Such a check would avoid an erroneous generation of negative impacts by erroneously substituting main products instead of by-products
- 9. Allocating the biogenic carbon content of bio-based materials should be avoided, preferring the tracking of the physical biogenic carbon content in the final products. In fact, there is no single allocation parameter (physical or economic) that can reflect physical causality for the carbon content of coproducts.

8.2 RESEARCH QUESTION 2

What is the environmental impact of novel bio-based products made from byproduct/waste streams compared to their fossil counterparts?

8.2.1 Lignin-based asphalts

Given lignin's high biogenic carbon content, there is interest in developing a durable and circular application that could store such biogenic carbon permanently. For this purpose, roads with bio-based asphalts using lignin to replace bitumen are currently tested worldwide, e.g. from the Netherlands [255] to Australia [134]. Chapter 4 investigated the environmental impact of various bio-based asphalts with kraft-lignin in the Dutch context. A cradle-to-grave attributional LCA was conducted, i.e. from the extraction of all asphalt's ingredients to its recycling at the end of life, for a functional unit of 1 t of top-layer asphalt. The environmental impact was calculated for 11 impact categories and was weighted to obtain an environmental cost indicator, i.e. a single monetary value score representing the avoided environmental damage cost or shadow cost [257]. Various top-layer asphalts were considered. i.e. stone mastic asphalts, asphalt concretes and porous asphalts. Both natural gas and biomass hog fuel were considered as possible steam sources to replace the part of black liquor no more burnt in a recovery boiler.

For both climate change and the weighted environmental damage, the production of the raw materials, i.e. bitumen and lignins (excl. biogenic carbon intake), were the main environmental impacts of the asphalt. After recycling, the benefits of their second life largely compensated for the impacts of lignin and bitumen productions. Such benefits were calculated using the formula proposed by the respective Dutch Product Category Rules. Regarding kraft lignin production, steam production with natural gas dominated the climate change impact. Together with steam production (with natural gas or hog fuel), liquid carbon dioxide and sulfuric acid were the other main source of environmental impact for lignin production in the other ten categories. On the life cycle of the asphalt, another main source of environmental impact was the energy for the production process (i.e. natural gas and electricity).

The results of the LCA revealed a climate change impact reduction between 30% and 75% for top-layer asphalts using lignin compared to conventional asphalts. Hence, the cradle-to-grave LCA of bio-based asphalts showed that storing the

biogenic carbon for 100 years in a durable application that is recycled over time could be a way to mitigate the climate change impact of the road construction sector. The percentage of lignin replacing bitumen and the steam source used by the pulp mill to replace the fraction of black liquor were the two main factors influencing bio-based asphalts' overall environmental cost performance.

A sensitivity analysis was conducted on the assumed functional unit and product system. A functional unit of 1 m² of asphalt of three layers (top, middle and base) instead of 1 t of top-layer asphalt was selected in such a sensitivity analysis. Changing the functional unit, i.e. from 1 t to 1 m² and product system, i.e. from top-layer only to the entire asphalt block made of three layers, did not alter the outcome of the comparison between bio-based and conventional asphalts. On the other hand, the difference in weighted environmental damage between lignin-based and conventional asphalts became minor since lower layers have a higher percentage of recycled content (burdens-free) and a lower amount of bitumen/ lignin (less binder needed in lower layers).

8.2.2 Bioenergy from wood chips

Chapter 5 of this thesis presented the LCA of a novel CHP technology (close to commercialization) integrating biomass gasification with a 199 kW solid oxide fuel cell (SOFC). This technology can produce heat and electricity from wood chips from sustainable forest management and sawmills.

Wood pellets and Miscanthus pellets were also assessed as feedstocks. For the seven impact categories considered, the bioenergy produced using this technology utilizing these three fuels was compared with conventional energy technologies i.e. heat from natural gas and electricity from the German grid.

The production of biomass fuels was the main source of environmental impacts of the energy produced using this technology. Depending on the category and the fuel, biomass fuels' production (including transportation) represented between 23% and 99% of the total impacts. For all categories, energy from wood chips as fuel generates much lower impacts than energy from wood pellets (11–70% lower depending on the category with the highest impact difference for water depletion and particulate matter) and Miscanthus pellets (9–99% lower depending on the category with the highest impact difference for water depletion). This confirms that utilizing bio-based by-products as a feedstock usually leads to a

lower environmental impact than dedicated crops (i.e. Miscanthus); also for other categories than climate change.

Besides the production of the feedstock, another important contributor to the environmental impact was identified in the SOFC stack due to both the high energy intensity and material consumption of its manufacturing process and its short technical lifetime (5 years).

Compared to heat from natural gas and electricity from the German grid, this technology showed significantly lower impacts for climate change (86/94% lower), photochemical ozone formation (-43/-70%), acidification (-37/-56%) and terrestrial eutrophication (-43/-63%). Since this technology achieves high exergy efficiencies and almost zero particulate emissions, it is also performing environmentally better than organic Rankine cycles using wood chips.

8.2.3 Bio-jet fuel from potato by-products

Food processing by-products also have high availability all around Europe and are largely unexploited for innovative bio-based products. Potato by-products are used as low-price animal feed but could be potentially transformed into higher value bio-based products [383]. The Netherlands is the fourth country in EU28 for potato production [445], with 20% of the processed potatoes becoming a by-product [446]. For this reason, potato by-products have been recently tested as a feedstock to produce bio-jet fuels to partly substitute the 4000kt/y of jet-fuels consumed in the Netherlands [347].

Chapter 6 investigated the environmental performance of the innovative bio-jet fuel derived from Dutch potato by-products mentioned above. Both attributional and consequential LCA models were applied. Besides climate change, photochemical ozone formation, terrestrial eutrophication, acidification and depletion of fossil fuels were investigated.

The bio-jet fuel's climate change impact was 60% lower than conventional jet fuel when applying attributional modeling. A lower climate change benefit (5-40% reduction) was calculated by consequential modeling, assuming that European animal feed would replace the potato by-products on the animal feed market. In an extreme case, assuming the use of imported soybean meals to replace the potato by-products, the climate change impact of the bio-jet fuel could become

three times the one of petrochemical kerosene due to the high land-use change impact of soybean.

Opposite outcomes between the two LCAs were obtained for photochemical ozone formation. Conversely, both modeling approaches confirmed that the investigated bio-jet fuel causes higher acidification and terrestrial eutrophication impacts than petrochemical kerosene but lower fossil fuel depletion.

8.2.4 Polypropylene from used cooking oil

Used cooking oil from the gastronomy sector, food industry and households could provide 4 Mt of used cooking oil per year in EU 27 [419]. Used cooking oil is already largely exploited worldwide for renewable diesel. However, together with renewable diesel grade products, other grades can be obtained from the hydrotreatment process, mainly naphtha and propane fractions [421,447]. A new commercial production facility converting such bio-based naphtha fraction into bio-based polypropylene has recently started operation [421].

Chapter 7 presented a cradle-to-gate LCA of polypropylene (PP) from used cooking oil considering 16 impact categories. Once normalization and weighting are applied to these 16 impact categories, the environmental impact of UCO-based PP was dominated by its climate change impact (28%), fossil resource use (23%) and water depletion (11%). The following overall environmental hotspots (with respective contributions) were identified: the polymerization process (38%), the production of hydrogen (21%), the production of LPG (18%) and the combustion of LPG (8%). Compared to petrochemical PP, significant environmental impact reductions can be achieved for climate change (40-62%, depending on the allocation method used at the process level) and fossil fuel resource use (80-86%). Moreover, these climate change impact reductions are for the cradle-to-factory gate scope without biogenic carbon removal. So, further benefits could be achieved by storing such carbon. The PlasticsEurope's Ecoprofile for petrochemical PP was used as the benchmark for comparison. PlasticsEurope provides mainly "black box" data due to confidentiality. As a consequence of the lack of transparency, it was challenging to fully interpret the impacts of petrochemical PP. So, it was not possible to provide robust conclusions for categories such as toxicity, ozone depletion and freshwater eutrophication.

Compared to bio-based PP made from sugarcane and woody biomass, PP from

UCO shows about 80/90% lower impact on climate change, from cradle to gate, without taking into account biogenic carbon removals. So, for climate change, biobased PP from UCO can be considered a better alternative to both petrochemical and bio-based PP from dedicated crops. The reductions on impact in the climate change category were almost unaffected if UCO is globally imported instead of locally sourced. Moreover, compared to petrochemical PP, PP from UCO has the further advantage of having an embedded biogenic carbon content. Storing such carbon in a durable application combined with recycling could lead to further environmental benefits from a cradle-to-grave perspective. Using this advantage, bio-based PP from UCO could reach the reduction of climate change impact allowed by renewable diesel from UCO compared to oil diesel, which is up to 88% and is as well higher than the reductions allowed by other biodiesels (40–62% savings).

For both UCO-based diesel and PP, <u>additional environmental impact reductions</u> <u>could be achieved using renewable hydrogen</u> locally produced via electrolysis powered by photovoltaics or wind power. For PP from UCO, <u>renewable propane</u> <u>and methane</u> produced as co-products could be used to (partially) replace LPG consumption for steam cracking instead of being marketed. <u>Greener electricity</u> <u>and steam</u> used for polymerization could lead to further benefits for PP.

8.2.5 Summary of lessons learned and recommendations of RQ2

This section summarizes lessons learned regarding the life cycle environmental impacts of emerging products from bio-based by-products ("residues").

1. <u>Bio-based products from by-products ("residues" streams) have usually lower</u> <u>climate change impact than their fossil counterparts.</u>

Figure 49 shows a summary of the reduction of climate change impacts from using bio-based by-products compared to the fossil counterpart was observed in this thesis. Climate change impact reduction between 30 and 70% can be achieved by lignin-based asphalts (details in Chapter 4) and between 86% and 96% by producing energy (heat or electricity) from wood chips via an innovative technology combining gasification and SOFC (details in Chapter 5). An innovative bio-jet fuel from potato by-products showed a 5-60% lower climate change impact than conventional jet fuel (details in Chapter 6). A climate change reduction in the range

of 40–62% was calculated for PP from used cooking oil compared to petrochemical PP (details in Chapter 7). Moreover, the lignin-use review also showed a reduction of climate change impacts of 62-78% for lignin-based adipic acid and 6–32% for lignin-based polyurethane (details in Chapter 2). It is worth remarking that these climate change performances are specific to the product assessed with respective technologies and feedstocks and cannot be straightforwardly generalized to the product category, e.g. bio-jet fuels or bio-based asphalts. Even for the same product e.g. lignin-based asphalt, various factors and environmental optimization play a role in achieving a positive environmental performance e.g. type of lignin, supply chain, the composition of the asphalt etc.



Figure 49. Climate change benefits (expressed as a percentage compared to the fossil counterparts taken as 100%) calculated in the LCAs conducted in this thesis. Variations represent both data uncertainties e.g. future energy mixes and methodological uncertainties e.g. caused by allocation choices. The LCA of polypropylene from used cooking oil had a cradle-to-gate scope and excluded potential benefits of biogenic carbon removal if stored in a durable application. All other LCAs had a cradle-to-grave scope.

2. Moreover, climate change impacts are not the only relevant environmental impacts. In most cases, the savings in terms of climate change impacts were also reflected by savings of similar magnitude for the depletion of fossil resources. Trade-offs with conventional (fossil) products occur instead in other categories. Previous LCA literature has widely observed this for bio-based products from dedicated crops. This thesis confirms that this also applies to products from bio-based by-products. For example, the investigated biobased products performed significantly worse than their fossil counterparts for acidification and terrestrial eutrophication. These two impacts are mainly due to atmospheric pollution caused by fuel combustion (e.g. in tractors) and fertilizers volatilization releasing Nitrogen (N) and Sulphur (S) emissions. Hence, they are strictly linked with agricultural activities. The allocation of even a small percentage of e.g. fertilizer-related impacts to the bio-based byproduct feedstock leads to higher acidification and eutrophication impacts than petrochemical products (not requiring fertilizers). Since forestry residues/ wood chips assessed in Chapter 5 do not require any cultivation, higher eutrophication and acidification impacts were not observed for this byproduct feedstock. Furthermore, the pretreatment of bio-based feedstocks often requires acids or other chemicals with high environmental impacts in toxicity categories. For example, this was the case for sulfuric acid and liquid carbon dioxide solvent for lignin.

- 3. The decision-maker has to be careful if by-products ("residues") are already utilized to a large degree for other purposes. In such a case, particular attention is needed, especially when converting a low-grade bio-based feedstock into a high value-added product has a low yield, leading to a high feedstock consumption. This was the case of potato by-products converted via ABE fermentation that are otherwise used mainly as animal feed (see Chapter 6). Their current user would be affected and should not replace them with a much less environmentally sustainable product. This was also the case for the overall performance of lignin-based asphalts. The part of the black liquor from which lignin is extracted would normally be combusted in the boiler of the pulp mill. If the pulp mill consumes natural gas to complete the energy balance, lignin-based asphalts could potentially lead to a higher overall environmental impact than conventional asphalts.
- 4. Process energy plays an important role in the climate change reduction achievable using bio-based products. <u>If the goal is to maximize climate change</u> <u>benefits, low-value biomass (e.g. hog fuel) or renewable gaseous fuels (e.g.</u> <u>renewable propane or biogas) are key choices.</u> However, this choice often has an economic trade-off with natural gas or LPG (see Chapters 2 and 4 and 7 for details regarding the entities of the environmental burden shifting using one fuel or the other). <u>Similar considerations apply to low carbon intensity</u>

<u>chemicals and the use of renewable electricity.</u> Electricity generally has a much higher relevance for the environmental impact of bio-based chemicals and fuels than their petrochemical counterparts.

5. For durable applications, the modeling of biogenic carbon involves high uncertainties. Since durable applications like asphalt or plastics (e.g. PP) can be recycled multiple times, accounting properly for the biogenic carbon storage is crucial for better climate change performance compared to their petrochemical counterparts. While most guidelines, e.g. the Dutch-product category rules for the construction sector, consider the biogenic carbon storage as permanent after 100 years, the new recommendation of the EU PEF [351] of not accounting for credits for permanent (and temporary) carbon storage could penalize this type of bio-based products.

8.3 RESEARCH QUESTION 3

How are the LCA results of such products influenced by the approach adopted to deal with the multifunctionality issue?

8.3.1 Lignin-based asphalts

In Chapter 4, lignin is used as an alternative ingredient to bitumen in asphalts. For asphalts using kraft lignin, an allocation is necessary at the level of the pulp mill to apportion the environmental impact between the main product (pulp) and lignin. Applying mass allocation instead of economic allocation to the unit processes of the pulp mill that lignin shares with pulp showed a much higher environmental impact for lignin.

Bitumen is as well a product from a multi-output process, i.e. the distillation of crude oil. Hence, the environmental impact of bitumen is also the result of an allocation procedure. As for kraft lignin, the sensitivity analysis showed that shifting from economic allocation to mass allocation at the level of the oil refinery leads to a higher environmental impact allocated to bitumen.

The cradle-to-gate environmental impacts of both asphalts were significantly affected by the allocation method chosen for the lignin and bitumen production processes. Hence, <u>the cradle-to-gate comparison between lignin-based asphalts</u> and conventional asphalts was highly affected by the allocation method applied to lignin or bitumen.

However, once the credits for recycling lignin and bitumen are accounted, a high fraction of the environmental impact of lignin and bitumen is credited. Such a credit makes the type of allocation method (i.e. economic, mass or energy) less influential over the cradle-to-grave life cycle. Hence, <u>the cradle-to-grave comparison between</u> the environmental impacts of the two asphalts was less affected by the allocation applied to lignin and bitumen (if the same method is applied to both). This applied to both the comparison per impact category or on a weighted basis.

8.3.2 Bioenergy from wood chips

Chapter 5 presented an LCA of an innovative CHP combining gasification and SOFC technologies. As for most CHPs producing heat and electricity in similar magnitudes, the allocation between these two products can significantly change the allocated environmental impact in most impact categories. Scrutinizing the effect of the allocation method applied between heat and electricity, the importance of providing the LCA results for both heat and electricity produced by CHPs emerged. In this way, it is possible to understand the effects of the allocation applied and increase transparency to allow for the right interpretation.

The comparison between the innovative SOFC CHP and an ORC CHP using the same fuel (i.e. wood chips) showed the <u>importance of comparing LCA results</u> <u>between CHP technologies based on the same allocation method.</u> In fact, different CHPs can produce heat and electricity with very different ratios. Only if the same allocation parameter (independently if exergy or economic) is applied to both CHPs, heat from the investigated technology will have a lower impact than heat from ORC in five categories (climate change, particulate matter, photochemical ozone formation, acidification, and terrestrial eutrophication) and higher in the two related to resource depletion. Conversely, if exergy allocation is applied to one CHP and economic allocation to the other CHP, the outcome could be misleading in some categories.

Testing the use of substitution in attributional modeling provided negative environmental impacts in several impact categories that 1) conflicts with the attributional aim and 2) cannot be replicated using exergy and economy allocation methods, i.e. allocation practices that the LCA literature has broadly acknowledged as a good proxy of causality for CHPs. This fact confirms the importance of the recommendation provided in section 8.1.3 that the use of substitution should be linked with a specific goal of the LCA, i.e. assessing a change via consequential modeling;

8.3.3 Bio-jet fuel from potato by-products

Chapter 6 investigated the effect of applying a different modeling approach (i.e. attributional or consequential) to investigate an innovative bio-jet fuel. Regarding the allocation method applied in the attributional LCA, the effect on the environmental impact of the price fluctuation on the allocation share between potato by-products and potato food products played a minor role. However, applying energy or mass allocation at the level of the potato processing industry would significantly change the environmental impact of the potato by-products. These two last methods would allocate impacts of similar magnitude to the potato by-products and the potato food products since they have similar physical characteristics (while the economic allocation reflected the much lower market price of the potato by-products). Nevertheless, these two types of allocation would not respect the ISO causality principle that should be reflected by the allocation criterion (i.e. the potato processing industry works to generate revenues and not energy products). So, energy and mass allocations should not be used in this case.

An attributional LCA was also conducted to determine the environmental impact of the co-products of the bio-jet fuel, namely animal feed (from fermentation residue), bio-based hydrogen, bio-based carbon dioxide, and biolubricants. In this way, it was possible to understand how physical or economic relations partition the environmental impact of each process and how the sum of the environmental impact of each product delivered by the system composes the total environmental impact of the system.

As often emphasized in LCA literature, consequential modeling has a much higher uncertainty than attributional modeling. A deep investigation was conducted on <u>the marginal production of animal feed assumed</u> since it <u>could significantly</u> <u>change the consequential LCA outcome</u>. The attributional LCA cannot capture any <u>displacement effects on the animal feed market</u>. In fact, the system analyzed in attributional LCAs aims only to assess the processes directly linked by (physical, energy, and service) flows to the unit process supplying the functional unit and not at all processes affected by the decision based on a cause-and-effect chain. However, <u>the additional animal feed production in the system boundaries of the</u> consequential LCA led to major uncertainties linked to the specific type of animal feed assumed to be used as a substitute for potato by-products. Furthermore, the displacement of the animal feed itself presents two other major sources of uncertainties: 1) the functionality assumed for animal feed, i.e. provision of proteins or energy to the animals and 2) indirect land-use changes if soybean is part of the displaced animal feed. Other major uncertainties of the consequential modeling arise from the future production method of the co-products substituted (surplus of hydrogen, lubricants and carbon dioxide). Since the investigated biojet fuel will take some years before being marketed and major changes towards decarbonization are expected in the EU chemical sector, forecasts of future markets can significantly differ from the future markets as assumed. So, the substitutions applied in the CLCA of the bio-jet fuel could lead to significant changes in the results.

8.3.4 Polypropylene from used cooking oil

The production of polypropylene from used cooking oil analyzed in Chapter 7 is obtained via two main multi-output processes, i.e. hydrotreatment and steam cracking. Polypropylene from used cooking oil at the production gate already showed better climate change and fossil fuel resource use than petrochemical PP.

The allocation method used at the process level can vary the climate change mitigation potential range between 40% and 62% and fossil fuel resource use between 80% and 86%. Such uncertainties consider both UCO as a by-product instead of waste or a different allocation approach used for hydrotreatment and steam cracking. Despite multifunctionality uncertainty being relatively small for these two impact categories, it was much higher in other categories. For example, by using the so-called 50/50 method on UCO open-loop recycling, the total weighted impact could increase between 25% and 160% and e.g. become up to 4 times higher for particulate matter impact depending on the types of primary vegetable oil.

Regarding the allocation method for hydrotreatment and steam cracking units, exergy allocation was apportioning the highest impact to bio-based PP in most categories compared to energy allocation or cut-off allocation. Conversely, as already experienced with the bioenergy case study presented in Chapter 5, <u>the substitution method would lead to a negative footprint for bio-based PP.</u> This

negative footprint is due to the low physical (and economical) significance of biobased naphtha and propylene, i.e. UCO-based PP's precursors. However, such a negative impact should be considered the result of a wrong methodological choice and not part of the uncertainty since physical significance should be checked before applying substitution to avoid substituting by-products.

8.3.5 Summary of lessons learned and recommendations of RQ3

This section summarizes lessons learned regarding the uncertainty caused by the approach adopted to deal with the multifunctionality issue in the LCAs of emerging products from bio-based by-products ("residual flows") assessed in this thesis.

- 1. Applying mass or energy allocation to the unit process delivering the by-product feedstock often leads to a much higher environmental impact for by-product feedstocks. This was the case for lignin in Chapter 4 (and also its fossil counterpart i.e. bitumen in the same chapter) and potato by-products in Chapter 6. Nevertheless, these two types of allocation do not reflect ISO causality relationships. In fact, both the potato processing industries and pulp mills do not operate to produce lignin and potato by-products but to generate revenues selling pulp and food products, and the most desired product reflects "why these processes exist". There might be cases when the most desired product for the same process might change in the future and a change of economic allocation share would reflect this.
- 2. In comparative LCAs between bio-based and conventional products, the practitioner can expect that multifunctionality uncertainty can make the comparison with conventional products inconclusive for some categories (e.g. see the comparison between the LCA impacts of bio-based PP from UCO and petrochemical PP in Chapter 7). In attributional LCAs, this means that the environmental performance of a bio-based product in several categories might depend on the adopted allocation method. Hence, an allocation key might apportion a higher share of a unit process' impact to a co-product than another co-product and another key would have done the opposite. Accordingly, the allocation applied in the attributional LCA of a certain co-product can affect the outcome of the comparison with its fossil counterpart. For this reason, increasing transparency in illustrating allocation choices and respective data (e.g. prices assumed) is the only way to allow for the

right interpretation (as strongly remarked under RQ1). Moreover, it is worth remarking that <u>attributional LCA results of all co-products of a system should</u> <u>always sum up to the total impact of the system (e.g. see the deep illustration</u> of this mechanism for the case of heat and electricity produced by SOFC-CHP in Chapter 5 and bio-jet fuel and its co-products in Chapter 7).

- 3. For the same product, the cradle-to-gate LCA results might have relatively high multifunctionality uncertainty, but the cradle-to-grave LCA might have low multifunctionality uncertainty (or vice-versa). For example, Chapter 4 showed that the environmental impacts of lignin and bitumen have high multifunctional uncertainties. However, when used in asphalt production, the uncertainties associated with the productions of lignin and bitumen are less significant because unit processes such as the asphalt production process are responsible for a significant percentage of the overall impacts. Therefore, summing a high-impact monofunctional process like asphalt production lowers the relevance of the multifunctionality uncertainty of some ingredients in the overall impact of the product system. Furthermore, asphalt recycling allows the recovery of a significant fraction of lignin and bitumen. So, the total net mass (consumed minus recovered) of lignin and bitumen allocated to the first life cycle is lower once recycling is included in the system boundary, decreasing their relevance on the overall environmental impact of the asphalt. Hence, large credits received due to recycling can lower the effect of the choice of the allocation method at the process level. However, the credits regarding products from multifunctional processes should rely on the same allocation method for correctness. For example, suppose the impact of oil refining was allocated to bitumen using a certain allocation approach. In that case, the credited impact for recovered bitumen should be based on the impact of bitumen calculated via the same allocation. The importance of comparing LCA results based on consistent allocation methods was strongly remarked under RQ1. This should also apply to credits from recycling activities when crediting the virgin production from a multifunctional process.
- 4. The case studies presented in this thesis highlighted that the environmental impact of by-products/residues used as feedstock for bio-based products are generally not trivial in the life cycle impact of the bio-based products and are highly affected by multifunctionality uncertainty. In fact, a small change in the allocation share (e.g. due to a sudden and substantial increase in demand) of the main product can significantly change the allocation share of the by-product.

- 5. Consequential LCA modeling is a key tool for policy support if the bio-based residues used as feedstock already have other (low-value) applications. Their utilization for a different use may, in turn, cause high indirect environmental effects. Using a consequential LCA, these potential burden shifts with overall potentially negative side effects can be foreseen/mitigated in advance and monitored, e.g. through policy measures. This was the case of potato by-products used for bio-jet fuel and currently used as animal feed (see Chapter 6). Since the current use of potato by-products is not part of the supply chain of the potato by-products, the attributional LCA neglecting this aspect showed a 60% climate change mitigation potential for the investigated bio-jet fuel. The consequential LCA estimated no or much lower climate change benefit (5-40%).
- 6. Testing substitution-based allocation in attributional modeling provided negative environmental impacts in several impact categories for bio-energy from SOFC CHP in Chapter 5 and polypropylene from used cooking oil in Chapter 7. The allocation of negative impacts to these products was not in line with the results generated by other allocation methods that are broadly acknowledged as a good proxy of causality relationships. Hence, such negative impacts conflict with the attributional aim, confirming once more the importance of using substitution only under specific LCA goals requiring the assessment of a change in demand and consequent adoption of consequential modeling.

8.4 OVERARCHING LIMITATIONS OF THIS THESIS AND FUTURE RESEARCH

This thesis aimed to analyze the life cycle environmental performance of several innovative bio-based products from agro-industrial and food-processing by-products and methodological complexities in conducting their environmental LCA. A major focus of the chapters of this thesis was on LCA multifunctionality, outlying unresolved issues in the implementation of ISO recommendations, and their effect on the environmental impacts of bio-based products from residual streams. As a result, this thesis provided recommendations to improve the consistency of life cycle inventory models of multifunctional product systems delivering bio-based products.

This thesis does not presume to have solved and covered all issues related to multifunctionality practices and LCAs were conducted only for a number of LCA cases for bio-based products from residual streams. Besides the research conducted in this thesis regarding the multifunctionality issue in LCAs of products from bio-based residual steams, future research is still necessary to investigate LCA multifunctionality for other relevant products from bio-based residues. When more than one allocation method is suitable for these products, evaluating the related uncertainty varying allocation methods is necessary.

Moreover, the case studies analyzed in this thesis and collected data could have been suitable to cover (broadly) other major aspects behind the life cycle impacts of bio-based products of high relevance nowadays. In particular, the following are two alternative research questions on two major aspects currently under debate by the scientific literature on the life cycle impacts of bio-based products.

The first aspect to investigate is: *what are the implications of the temporal scope on the data used for prospective LCAs*? In this thesis, all LCAs regarded innovative bio-based products that have just or not reached commercialization yet. So, even if all LCAs were based on primary data as much as possible, primary data does not mean directly high data quality for emerging technologies. The chapters of this thesis looked at various technologies at an early stage of development to assess their environmental performance and provide the necessary guide for investment and research. However, pilot plants might significantly differ from future commercialized technology. Major reasons behind higher data uncertainties for innovative products than commercialized products are due to: 1) potential process design changes, 2) size scaling effects, 3) process synergies that could be optimized, 4) future technological learning and 4) external factors (e.g. a future infrastructural change of the electricity mix) [448]. All these aspects have been considered in the performed LCAs, but a higher focus could have been given.

A second research question to investigate could be: what are the most environmentally friendly and cost-effective ways of using bio-based by-products as feedstocks? In fact, bio-based by-products ("residues") have limited availability and their best use should be preferred. Furthermore, products from bio-based by-products generally have a (much) higher production cost than petrochemical products and their production needs support from national and international policy initiatives. The chapters of this thesis provided only a limited overview of biobased products that could be obtained from the same feedstock and respective reductions of environmental impacts compared to their fossil counterparts. Moreover, techno-economic analysis was not applied in the chapters of this thesis. So, cost implications were completely neglected in the comparison between biobased based and petrochemical alternatives.

Despite the challenges faced by LCA practice, LCA is a powerful tool to understand the environmental impacts of products and services. An increasing number of governments and companies consider LCA the key environmental management decision-support tool. The strength of LCA is its capacity to detect burden-shifting from one impact category (e.g. climate change) to another (e.g. human toxicity) and from one life cycle stage (e.g. production process) to another (e.g. combustion during fuel use). Such issues cannot be detected by other available metrics such as green chemistry metrics, circularity metrics, and process-related parameters (e.g. consumption of energy or water) and are crucial in the environmental comparisons to support decision making.



CHAPTER

English summary

9. ENGLISH SUMMARY

In the last decade, the need to reduce dependence on fossil resources has led to the emergence of numerous bio-based alternatives in European sectors traditionally dominated by petrochemical products. The *European Green Deal* and *EU bioeconomy strategy* could further accelerate the growing trend of innovative bio-based commodities worldwide. This significant growth in the short term should be accompanied by scientific evidence on the environmental impacts of innovative bio-based products. In fact, to achieve the targets of bioeconomy policies, investments should be guided towards environmentally sustainable biobased products. Science-based evidence can confirm if a certain bio-based product achieves the expected lower environmental footprint than its petrochemical counterpart.

Locally sourced *bio-based "residues"* are a key feedstock to extend the amount of bio-based products produced sustainably in the EU. This feedstock does not generate concerns about food security and land competition, is usually cheaper than dedicated crops, and does not require transoceanic imports. Products from this type of feedstock were the major focus of this thesis. In particular, one of three main research questions was: *what is the environmental impact of innovative biobased products made from by-product/waste streams compared to their fossil counterparts*?

This thesis explored the conversion of various local bio-based residues into a heterogeneous range of innovative products from an environmental perspective. Among the explored feedstocks were *food processing residues*, i.e. potato by-products and used cooking oil, *forestry residues*, i.e. wood chips, and a by-product from *paper/pulp industries*, i.e. lignin.

The life cycle environmental impacts of innovative bio-based products from these agro-industrial and food-processing by-products were compared to their petrochemical counterparts using *Life Cycle Assessment* (LCA) methodology. LCA is an internationally standardized method to assess products and services' life cycle environmental impacts, from raw materials extraction to waste management. Various policy decision instruments and regulation mechanisms already rely on LCA results to assess the environmental performance of bio-based products.

In particular, this thesis presents for the first time the LCAs of *four* emerging products:

- 1. polypropylene from used cooking oil via hydrotreatment,
- 2. *bioenergy* from a solid oxide fuel cell (SOFC) combined heat and power plant (CHP) relying on gasification of *wood chips*,
- 3. *a bio-jet fuel* via acetone-butanol-ethanol fermentation of *potato by-products* and,
- 4. *bio-based asphalts* with *lignin* from the kraft pulping process used as the binder.

These four LCAs were based on primary data for the respective innovative conversion technologies. All four LCAs scrutinized a broad range of environmental impact indicators.

LCA modeling needs to be applied consistently on both sides to perform a meaningful comparison between the life cycle impacts of bio-based products and their petrochemical counterparts. Although LCA practice is standardized, one of the historically most debated points with lower convergence of views is the so-called *multifunctionality issue*. The multifunctionality issue occurs if a process in the production system under assessment provides more than one function or product. The hierarchical steps to follow when solving multifunctionality in LCA are based on the recommendations of ISO 14044:2006.

However, the *interpretation* of such recommendations has been debated for twenty-five years leading to different implementation practices in the LCA scientific literature and ISO-compliant LCA guides. Since the production processes of bio-based products are frequently multifunctional, a criterion for allocating environmental impacts between the co-products is often necessary when conducting LCAs of bio-based products. Hence, the lack of a shared interpretation of ISO recommendations for selecting multifunctionality approaches and consequent lack of consistent allocation practices closely affect LCAs of bio-based products. So, a major research question of this thesis was: *what multifunctionality practices are adopted in LCAs of bio-based products and how can the consistency of the life cycle inventory model be improved*?

In particular, all feedstocks explored in this thesis are by-products obtained through the production process of a more economically valuable product. Hence, the feedstock originates from a multifunctional process for all four case studies. Based on the experience gained by previous LCAs of more established products from bio-based residues, multifunctionality is a major source of LCA *uncertainty* for products from this type of feedstock. In fact, for such products, not only the modeling of the production process but also the feedstock itself is affected by the LCA multifunctionality issue. So, for an accurate interpretation of the LCA results, it is necessary to understand *how the LCA results of such products are influenced by the approach adopted to deal with the multifunctionality issue.*

Accordingly, different multifunctionality practices both at the level of the feedstock and at the processing level were deeply discussed in the four LCAs presented. The multifunctionality issue in LCAs of bio-based products was further investigated via two reviews. A first review scrutinized multifunctionality practices and their effect on LCA results' comparability for one of the feedstocks of interest, i.e. lignin. A second review focused on the existing interpretations of the multifunctionality solutions recommended by ISO 14044:2006, the related debates and their historical origins. In the same review, a text mining process was adopted to quantify ISO-compliant multifunctionality practices in all LCAs of bio-based products in the literature. The knowledge gained from these two reviews allowed detecting the different interpretations' origins and underlying rationales. Such knowledge was fundamental to select only (fully) ISO-compliant multifunctionality practices in the LCAs of the four investigated bio-based products. Based on the lessons learned from the two reviews, nine key recommendations to deal with the multifunctionality issues of biobased products were formulated. These recommendations relate to the context of the use of substitution approaches, ISO-causality as the key principle for selecting allocation methods, transparency in the rationale for their selection and a strict distinction between attributional and consequential modeling approaches.

The LCAs of all four products from bio-based residues showed a generally better *climate change* performance than their fossil counterparts. The climate change impact reductions were quantified as 30-70% for top-layer lignin-based asphalts, 86-96% for energy (heat or electricity) from wood chips via gasification-SOFC CHP, 5-60% for an innovative bio-jet fuel from potato by-products and 40–62% for polypropylene from used cooking oil. Moreover, the review of LCAs of lignin-based products showed promising climate change impact reductions of 62-78% for lignin-based adipic acid and 6–32% for lignin-based polyurethane. *Low carbon-intensity process energy* was confirmed to play an important role in achieving high climate change reductions for all these products. Hence, low-quality biomass or renewable gaseous fuels (e.g. renewable propane or biogas) and renewable electricity are key choices to maximize climate change benefits.

For all four products, the reduction of climate change impacts compared to petrochemical counterparts was also reflected by similar reductions for the *depletion of fossil resources*. Trade-offs with conventional (fossil) products were instead observed for other impact categories. The allocation of even a small percentage of e.g. fertilizer-related impacts to *agricultural by-products* led to higher *acidification and eutrophication* impacts than their petrochemical counterpart for products from this type of feedstock. Moreover, if the pretreatment of the feedstock to be converted into high-value bio-based products requires chemicals like sulfuric acid and liquid carbon dioxide solvent for lignin, higher impacts than their petrochemical counterparts can be observed in *toxicity categories*.

Besides multifunctionality, the modeling of *biogenic carbon storage* for durable bio-based products and the modeling approach (*attributional or consequential*) for by-products utilized to a large degree for other purposes may lead to high uncertainties. Accordingly, in such cases, selecting a different multifunctionality or modeling approach can make the comparative LCAs between bio-based and conventional products inconclusive for some impact categories. Despite these methodological challenges, the case studies investigated in this thesis highlighted once more the strength of LCA as a key tool to provide decision support on the environmental impacts of bio-based products to avoid undesirable shifts of environmental burdens from one ecological issue to another one.



CHAPTER

Nederlandse samenvatting
10. NEDERLANDSE SAMENVATTING

In het afgelopen decennium heeft de noodzaak om de afhankelijkheid van fossiele grondstoffen te verminderen geleid tot de opkomst van tal van biobased alternatieven in Europese sectoren die traditioneel worden gedomineerd door petrochemische producqten. The European Green Deal en EU bioeconomy strategy kunnen deze groeiende trend voor innovatieve bio-based goederen wereldwijd verder accelereren. Deze significante groei zou op korte termijn gepaard moeten gaan met wetenschappelijk onderbouwing van de milieu-impacts van innovatieve bio-based producten. Om de doelen van het bio-economie beleid te behalen, moeten investeringen gericht gestuurd moeten worden met betrekking tot de duurzamheid van bio-based producten . Wetenschappelijk bewijs kan bevestigen óf verschillende bio-based producten de ecologische voetafdruk reduceren ten opzichte van de petrochemische tegenhanger.

De inzet van Lokale bio-based 'residuen' als grondstoffen zijn belangrijk om meer bio-based producten duurzaam te kunnen produceren. Deze grondstoffen zijn geen grond tot zorg met betrekking tot voedselveiligheid en land competitie. Daarnaast zijn ze gewoonlijk goedkoper dan - gewassen, en import is vaak niet noodzakelijk. Producten uit residuen van dit type grondstof stonden centraal in dit proefschrift. Een van de drie hoofdonderzoeksvragen was: Wat is de milieu-impact van innovatieve bio-based producten gemaakt van bijproduct/afvalstromen in vergelijking met hun fossiele tegenhangers?

Dit proefschrift onderzoekt de conversie van verschillende lokale biobased reststoffen naar een heterogeen aanbod van innovatieve producten vanuit een milieuoogpunt. Tot de onderzochte grondstoffen behoorden voedselverwerkingsresiduen, dat wilt zeggen aardappelbijproducten en gebruikt frituurvet, bosbouwresiduen, houtsnippers, en een bijproduct van de papier-/ pulpindustrie, namelijk lignine.

De milieueffecten over de levenscyclus van innovatieve bio-based producten van deze agro-industriële en voedselverwerkende bijproducten werden vergeleken met hun petrochemische tegenhangers met behulp van Life Cycle Assessment (LCA)methodologie. LCA is een internationaal gestandaardiseerde methode om de milieueffecten van de levenscyclus van producten en diensten te beoordelen, van de winning van grondstoffen tot afvalbeheer. Verschillende beleidsinstrumenten en reguleringsmechanismenzijn deels gebaseerd op LCA-resultaten om de milieuprestaties van bio-based producten te beoordelen.

In het bijzonder presenteert dit proefschrift voor het eerst de LCA's van vier opkomende producten:

- 1. olypropyleen uit gebruikt frituurolie via waterstofbehandeling,
- 2. bio-energie uit een vasteoxidebrandstofcel (SOFC) gecombineerd met warmtekrachtkoppeling (WKK) op basis van vergassing van houtsnippers,
- 3. bio-kerosine via aceton-butanol-ethanolfermentatie van aardappelbijproducten, en
- 4. biobased asfalt met lignine uit het kraftpulpproces als bindmiddel.

Deze vier LCA's waren gebaseerd op primaire gegevens voor de specifieke innovatieve conversietechnologieën. In het kader van de vier LCA's is een breed scala aan milieu-impact indicatoren onder de loep genomen.

LCA-modellering moet van beide kanten consistent worden toegepast om een zinvolle vergelijking te maken tussen de levenscycluseffecten van bio-based producten en hun petrochemische tegenhangers. Hoewel de LCA-praktijk gestandaardiseerd is, is een van de historisch meest besproken punten (met tevens een lage mate van convergentie van standpunten) het zogenaamde multifunctionaliteitsprobleem. Het multifunctionaliteitsprobleem doet zich voor als een proces in het te beoordelen productiesysteem meer dan één functie of product levert. De te volgen hiërarchische stappen bij het oplossen van multifunctionaliteit in LCA zijn gebaseerd op de aanbevelingen van ISO 14044:2006.

Erwordt echteral vijfentwintigjaar gedebatteerd over de interpretatie van dergelijke aanbevelingen, wat heeft geleid tot verschillende implementatiepraktijken in de LCA-wetenschappelijke literatuur en ISO-conforme LCA-gidsen. Omdat de productieprocessen van biobased producten vaak multifunctioneel zijn, is bij het uitvoeren van LCA's van biobased producten vaak een criterium nodig voor de toerekening van milieueffecten tussen de co-producten. Vandaar dat het ontbreken van een breed gedeelde interpretatie van ISO-aanbevelingen voor het selecteren van multifunctionaliteitsbenaderingen en het daaruit voortvloeiende gebrek aan consistente toewijzingspraktijken nauw van invloed zijn op LCA's van biogebaseerde producten. Een belangrijke onderzoeksvraag van dit proefschrift was daarom: welke multifunctionaliteitspraktijken worden toegepast in LCA's van bio-based producten en hoe kan de consistentie van het levenscyclusinventarisatiemodel worden verbeterd?

In het bijzonder zijn alle grondstoffen die in dit proefschrift worden onderzocht, bijproducten die worden verkregen door het productieproces van een economisch waardevoller product. De grondstof is dus afkomstig uit een multifunctioneel proces voor alle vier casestudies. Op basis van de ervaring die is opgedaan met eerdere LCA's met meer gevestigde producten uit bio-based residuen, is multifunctionaliteit een belangrijke bron van -onzekerheid voor producten uit dit type grondstof. In feite wordt voor dergelijke producten niet alleen de modellering van het productieproces, maar ook de grondstof zelf beïnvloed door het LCAmultifunctionaliteitsprobleem. Voor een nauwkeurige interpretatie van de LCAresultaten is het dus noodzakelijk om te begrijpen hoe de LCA-resultaten van dergelijke producten worden beïnvloed door de aanpak die is gevolgd om het multifunctionaliteitsprobleem aan te pakken.

Dienovereenkomstig werden verschillende multifunctionaliteitspraktijken zowel op het niveau van de grondstof als op het verwerkingsniveau uitvoerig besproken in de vier gepresenteerde LCA's. Het multifunctionaliteitsprobleem in LCA's van bio-based producten is verder onderzocht via twee reviews. Een eerste review onderzocht multifunctionaliteitspraktijken en hun effect op de vergelijkbaarheid van LCA-resultaten voor een van de van belang zijnde grondstoffen, namelijk lignine. Een tweede review richtte zich op de bestaande interpretaties van de multifunctionaliteitsoplossingen aanbevolen door ISO 14044:2006, de gerelateerde debatten en hun historische oorsprong. In dezelfde review werd een tekstdelvingsproces gebruikt om ISO-conforme multifunctionaliteitspraktijken in alle LCA's van bio-based producten in de literatuur te kwantificeren. De kennis die uit deze twee reviews werd verkregen, maakte het mogelijk om de oorsprong en onderliggende beweegredenen van de verschillende interpretaties te detecteren. Dergelijke kennis was van fundamenteel belang om alleen (volledig) ISO-conforme multifunctionaliteitspraktijken te selecteren in de LCA's van de vier onderzochte bio-based producten. Op basis van de lessen die uit de twee beoordelingen zijn getrokken, zijn negen belangrijke aanbevelingen geformuleerd om de multifunctionaliteit van bio-based producten aan te pakken. Deze aanbevelingen hebben betrekking op de context van het gebruik van substitutiebenaderingen, ISOcausaliteit als het belangrijkste principe voor het selecteren van allocatiemethoden, transparantie in de reden voor hun selectie en een strikt onderscheid tussen attributie- en consequentiële modelleringsbenaderingen.

De LCA's van alle vier producten uit biobased residuen lieten over het algemeen betere prestaties zien met betrekking tot klimaatveranderings dan hun fossiele tegenhangers. De reductie van de impact op klimaatverandering werd gekwantificeerd als 30-70% voor de toplaag op lignine gebaseerd asfalt, 86-96% voor energie (warmte of elektriciteit) uit houtsnippers via vergassing-SOFC WKK, 5-60% voor een innovatief bio-kerosine uit aardappelbijproducten en 40–62% voor polypropyleen uit gebruikt frituurvet. Bovendien toonde de review van LCA's van op lignine gebaseerde producten veelbelovende verminderingen van de impact op klimaatverandering van 62-78% voor op lignine gebaseerd adipinezuur en 6-32% voor op lignine gebaseerd polyurethaan. Er werd bevestigd dat procesenergie met een lage koolstof intensiteit een belangrijke rol speelt bij het bereiken van een sterke vermindering van de klimaatverandering voor al deze producten. Daarom zijn het gebruik van biomassa van lage kwaliteit, hernieuwbare gasvormige brandstoffen (bijv. hernieuwbaar propaan of biogas) en hernieuwbare elektriciteit belangrijke keuzes om de reductie van klimaatverandering te maximaliseren.

Voor alle vier de producten werd de vermindering van de gevolgen van klimaatverandering in vergelijking met petrochemische tegenhangers ook weerspiegeld in vergelijkbare verminderingen voor de uitputting van fossiele hulpbronnen. Voor andere impactcategorieën werden daarentegen trade-offsen met conventionele (fossiele) producten waargenomen De toekenning van zelfs een klein percentage van bijvoorbeeld kunstmest-gerelateerde impacts aan agrarische bijproducten leidden tot hogere effecten op verzuring en eutrofiëring dan hun petrochemische tegenhanger voor producten van dit type grondstof. Bovendien, als de voorbehandeling van de grondstof die moet worden omgezet in hoogwaardige bio-based producten chemicaliën vereist zoals zwavelzuur en vloeibaar kooldioxide-oplosmiddel voor lignine, kunnen grotere impacts worden waargenomen in toxiciteitscategorieëndan hun petrochemische tegenhangers.

Naast multifunctionaliteit kan het modelleren van biogene koolstofopslag voor duurzame bio-based producten en de modelleringsaanpak (attributioneel of consequentieel) voor bijproducten die in hoge mate voor andere doeleinden worden gebruikt, tot grote onzekerheden leiden. Dienovereenkomstig kan in dergelijke gevallen het selecteren van een andere multifunctionaliteit of modelleringsaanpak de vergelijkende LCA's tussen bio-based en conventionele producten onbeslist maken voor sommige effectcategorieën. Ondanks deze methodologische uitdagingen, benadrukten de case studies die in dit proefschrift werden onderzocht eens te meer de kracht van LCA als een belangrijk hulpmiddel om beslissingsondersteuning te bieden over de milieueffecten van bio-based producten om ongewenste verschuivingen van milieubelasting van het ene ecologische probleem naar het andere te voorkomen.

REFERENCES

- [1] European Commission. What is bioeconomy? 2020. https://ec.europa.eu/research/ bioeconomy/index.cfm (accessed October 2, 2020).
- [2] Sturm V, Banse M. Transition paths towards a bio-based economy in Germany: A model-based analysis. Biomass and Bioenergy 2021;148. https://doi.org/10.1016/j. biombioe.2021.106002.
- [3] Vogelpohl T. Transnational sustainability certification for the bioeconomy? Patterns and discourse coalitions of resistance and alternatives in biomass exporting regions. Energy Sustain Soc 2021;11. https://doi.org/10.1186/s13705-021-00278-5.
- [4] Brandão AS, Gonçalves A, Santos JMRCA. Circular bioeconomy strategies: From scientific research to commercially viable products. J Clean Prod 2021;295. https://doi. org/10.1016/j.jclepro.2021.126407.
- [5] Oguntuase OJ, Adu OB. Bioeconomy as Climate Action: How ready are African Countries? African Handb Clim Chang Adapt 2020:1–15. https://doi.org/10.1007/978-3-030-42091-8_82-1.
- [6] Van Renssen S. A bioeconomy to fight climate change. Nat Clim Chang 2014;4:951–3. https://doi.org/10.1038/nclimate2419.
- [7] Spierling S, Knüpffer E, Behnsen H, Mudersbach M, Krieg H, Springer S, et al. Bio-based plastics A review of environmental, social and economic impact assessments. J Clean Prod 2018. https://doi.org/10.1016/j.jclepro.2018.03.014.
- [8] Krzyżaniak M, Stolarski MJ, Warmiński K. Life cycle assessment of poplar production: Environmental impact of different soil enrichment methods. J Clean Prod 2019;206:785– 96. https://doi.org/10.1016/j.jclepro.2018.09.180.
- [9] European Commission. Proposal for a Directive of the European Parliament and of the Council on the promotion of the use of energy from renewable sources. Off J Eur Union 2016.
- [10] Kuosmanen T, Kuosmanen N, El-Meligi A, Ranzon T, Gurria P, Lost S, et al. How big is the bioeconomy? Reflections from an economic perspective. 2020. https://doi. org/10.2760/144526.
- The German Bioeconomy Council. Bioeconomy policies and strategies established by 2017. Diagram prepared by the German Bioeconomy Council (Bio€okonomierat – BO ¨ R), Berlin. 2018.
- [12] Stegmann P, Londo M, Junginger M. The circular bioeconomy: Its elements and role in European bioeconomy clusters. Resour Conserv Recycl X 2020;6. https://doi.org/10.1016/j. rcrx.2019.100029.
- [13] European Commission. Communication from the Commission: The European Green Deal. COM(2019) 640 Final 2019:p24.
- [14] Fritsche U, Brunori G, Chiaramonti D, Galanakis C, Hellweg S, Matthews R, et al. Future transitions for the Bioeconomy towards Sustainable Development and a Climate-Neutral Economy - Knowledge Synthesis Final Report 2020.
- [15] European Commission. Chemicals Strategy for Sustainability Towards a Toxic-Free Environment. 2020.
- [16] Ronzon T, M'Barek R. Socioeconomic indicators to monitor the EU's bioeconomy in transition. Sustain 2018;10. https://doi.org/10.3390/su10061745.

- [17] Ronzon T, Piotrowski S, Tamosiunas S, Dammer L, Carus M, M'barek R. Developments of economic growth and employment in bioeconomy sectors across the EU. Sustain 2020;12. https://doi.org/10.3390/su12114507.
- [18] European Commission. Brief on jobs and growth of the EU bioeconomy 2008-2017 2020.
- [19] M'barek R, Philippidis G, Ronzon T. Alternative Global Transition Pathways to 2050: Prospects for the Bioeconomy: An application of the MAGNET model with SDG insights. 2019. https://doi.org/10.2760/594847.
- [20] Panchaksharam Y, Kiri P, Bauen A, vorn Berg C, Puente A, Chinthapalli R, et al. Roadmap for the Chemical Industry in Europe towards Bioeconomy. WwwRoadtobioEu/Uploads/ Publications/Roadmap/RoadToBio_strategy_documentPdf 2019:40.
- [21] Nita V, Benini L, Ciupagea C, Kavalov B, Pelletier N. Bio-economy and sustainability: a potential contribution to the Bio-economy Observatory. 2013. https://doi. org/10.2788/78614.
- [22] Mai-Moulin T, Visser L, Fingerman KR, Elbersen W, Elbersen B, Nabuurs GJ, et al. Sourcing overseas biomass for EU ambitions: assessing net sustainable export potential from various sourcing countries. Biofuels, Bioprod Biorefining 2019;13:293–324. https:// doi.org/10.1002/bbb.1853.
- [23] Mandley SJ, Daioglou V, Junginger HM, van Vuuren DP, Wicke B. EU bioenergy development to 2050. Renew Sustain Energy Rev 2020;127. https://doi.org/10.1016/j. rser.2020.109858.
- [24] EPURE. European renewable ethanol key figures 2019. 2020.
- [25] Spekreijse J, Lammens T, Parisi C, Ronzon T, Vis M. Insights into the European market for bio-based chemicals. Eur Comm 2019;19:1–162. https://doi.org/10.2760/673071.
- [26] Herrmann R, Jumbe C, Bruentrup M, Osabuohien E. Competition between biofuel feedstock and food production: Empirical evidence from sugarcane outgrower settings in Malawi. Biomass and Bioenergy 2018;114:100–11. https://doi.org/10.1016/j. biombioe.2017.09.002.
- [27] Lathuillière MJ, Miranda EJ, Bulle C, Couto EG, Johnson MS. Land occupation and transformation impacts of soybean production in Southern Amazonia, Brazil. J Clean Prod 2017;149:680–9. https://doi.org/10.1016/j.jclepro.2017.02.120.
- [28] Callegari A, Bolognesi S, Cecconet D, Capodaglio AG. Production technologies, current role, and future prospects of biofuels feedstocks: A state-of-the-art review. Crit Rev Environ Sci Technol 2020;50:384–436. https://doi.org/10.1080/10643389.2019.1629801.
- [29] Searchinger T, Edwards R, Mulligan D, Heimlich R, Plevin R. Do biofuel policies seek to cut emissions by cutting food? Science (80-) 2015;347:1420–2. https://doi.org/10.1126/ science.1261221.
- [30] Philippini RR, Martiniano SE, Ingle AP, Franco Marcelino PR, Silva GM, Barbosa FG, et al. Agroindustrial Byproducts for the Generation of Biobased Products: Alternatives for Sustainable Biorefineries. Front Energy Res 2020;8. https://doi.org/10.3389/ fenrg.2020.00152.
- [31] Silalertruksa T, Gheewala SH. A comparative LCA of rice straw utilization for fuels and fertilizer in Thailand. Bioresour Technol 2013;150:412–9. https://doi.org/10.1016/j. biortech.2013.09.015.

- [32] Bakker R, Elbersen W, Poppens R, Lesschen J. Rice straw and Wheat straw Potential feedstocks for the biobased economy. NL Agency Minist Econ Aff 2013.
- [33] Khoo HH, Tan RBH, Chng KWL. Environmental impacts of conventional plastic and bio-Based carrier bags. Int J Life Cycle Assess 2010;15:284–93. https://doi.org/10.1007/s11367-010-0162-9.
- [34] de Jong S, Antonissen K, Hoefnagels R, Lonza L, Wang M, Faaij A, et al. Life-cycle analysis of greenhouse gas emissions from renewable jet fuel production. Biotechnol Biofuels 2017. https://doi.org/10.1186/s13068-017-0739-7.
- [35] Giuntoli J, Bulgheroni C, Marelli L, Sala S. Brief on the use of Life Cycle Assessment (LCA) to evaluate environmental impacts of the bioeconomy 2019.
- [36] Edwards R, Padella M, Giuntoli J, Koeble R, O'Connell A, Bulgheroni C, et al. Definition of input data to assess GHG default emissions from biofuels in EU legislation, Version 1c. 2017. https://doi.org/10.2790/658143.
- [37] European Parliament. FQD (2015). Fuel Quality Directive 2015/652/EC 2015.
- [38] ISO. ISO 14044, Environmental management Life cycle assessment Requirements and guidelines. International Standard Organization. Iso 14044 2006.
- [39] EPA. Overview for Renewable Fuel Standard. WwwEpaGov 2017.
- [40] Air Resources Board of the California Environmental Protection Agency. CEPA (2009) Proposed Regulation to Implement the Low Carbon Fuel Standard - Volume I 2009.
- [41] FAO. Greenhouse Gas Emissions from the Dairy Sector A Life Cycle Assessment. FOOD Agric Organ UNITED NATIONS Anim Prod Heal Div 2010. https://doi.org/10.1016/S0301-4215(01)00105-7.
- [42] Broeren MLM, Zijp MC, Waaijers-van der Loop SL, Heugens EHW, Posthuma L, Worrell E, et al. Environmental assessment of bio-based chemicals in early-stage development: a review of methods and indicators. Biofuels, Bioprod Biorefining 2017. https://doi. org/10.1002/bbb.1772.
- [43] European Commission. Environmental impact assessments of innovative bio-based product. Task 1 of "Study on Support to R&I Policy in the Area of Bio-based Products and Services" - Study. 2019. https://doi.org/10.2777/251887.
- [44] Hermann BG, Blok K, Patel MK. Twisting biomaterials around your little finger: Environmental impacts of bio-based wrappings. Int J Life Cycle Assess 2010;15:346–58. https://doi.org/10.1007/s11367-010-0155-8.
- [45] Plevin RJ. Assessing the Climate Effects of Biofuels Using Integrated Assessment Models, Part I: Methodological Considerations. J Ind Ecol 2017;21:1478–87. https://doi. org/10.1111/jiec.12507.
- [46] Stratton RW, Wong HM, Hileman JI. Quantifying variability in life cycle greenhouse gas inventories of alternative middle distillate transportation fuels. Environ Sci Technol 2011. https://doi.org/10.1021/es102597f.
- [47] Zemanek D, Champagne P, Mabee W. Review of life-cycle greenhouse-gas emissions assessments of hydroprocessed renewable fuel (HEFA) from oilseeds. Biofuels, Bioprod Biorefining 2020;14:935–49. https://doi.org/10.1002/bbb.2125.
- [48] Capaz RS, Posada JA, Osseweijer P, Seabra JEA. The carbon footprint of alternative jet fuels produced in Brazil: exploring different approaches. Resour Conserv Recycl 2020. https://doi.org/10.1016/j.resconrec.2020.105260.

- [49] Pawelzik P, Carus M, Hotchkiss J, Narayan R, Selke S, Wellisch M, et al. Critical aspects in the life cycle assessment (LCA) of bio-based materials - Reviewing methodologies and deriving recommendations. Resour Conserv Recycl 2013. https://doi.org/10.1016/j. resconrec.2013.02.006.
- [50] Reap J, Roman F, Duncan S, Bras B. A survey of unresolved problems in life cycle assessment. Part 1: Goal and scope and inventory analysis. Int J Life Cycle Assess 2008;13:290–300. https://doi.org/10.1007/s11367-008-0008-x.
- [51] Majeau-Bettez G, Dandres T, Pauliuk S, Wood R, Hertwich E, Samson R, et al. Choice of allocations and constructs for attributional or consequential life cycle assessment and input-output analysis. J Ind Ecol 2018;22:656–70. https://doi.org/10.1111/jiec.12604.
- [52] Schrijvers DL, Loubet P, Sonnemann G. Critical review of guidelines against a systematic framework with regard to consistency on allocation procedures for recycling in LCA. Int J Life Cycle Assess 2016;21:994–1008. https://doi.org/10.1007/s11367-016-1069-x.
- [53] Cherubini F, Strømman AH. Life cycle assessment of bioenergy systems: State of the art and future challenges. Bioresour Technol 2011. https://doi.org/10.1016/j. biortech.2010.08.010.
- [54] Gordillo V, Rankovic N, Abdul-Manan AFN. Customizing CO2 allocation using a new non-iterative method to reflect operational constraints in complex EU refineries. Int J Life Cycle Assess 2018;23:1527–41. https://doi.org/10.1007/s11367-017-1380-1.
- [55] Hermansson F, Janssen M, Svanström M. Allocation in life cycle assessment of lignin. Int J Life Cycle Assess 2020. https://doi.org/10.1007/s11367-020-01770-4.
- [56] Vera I, Hoefnagels R, van der Kooij A, Moretti C, Junginger M. A carbon footprint assessment of multi-output biorefineries with international biomass supply: a case study for the Netherlands. Biofuels, Bioprod Biorefining 2020. https://doi.org/10.1002/ bbb.2052.
- [57] Sandin G, Røyne F, Berlin J, Peters GM, Svanström M. Allocation in LCAs of biorefinery products: Implications for results and decision-making. J Clean Prod 2015. https://doi. org/10.1016/j.jclepro.2015.01.013.
- [58] Tonini D, Schrijvers D, Nessi S, Garcia-Gutierrez P, Giuntoli J. Carbon footprint of plastic from biomass and recycled feedstock: methodological insights. Int J Life Cycle Assess 2021;26:221–37. https://doi.org/10.1007/s11367-020-01853-2.
- [59] Schaubroeck T, Schaubroeck S, Heijungs R, Zamagni A, Brando M, Benetto E. Attributional & Consequential Life Cycle Assessment: Definitions, Conceptual Characteristics andModelling Restrictions. Sustainability 2021. https://doi.org/10.3390/ sul3137386.
- [60] Heijungs R, Allacker K, Benetto E, Brandão M, Guinée J, Schaubroeck S, et al. System Expansion and Substitution in LCA: A Lost Opportunity of ISO 14044 Amendment 2. Front Sustain 2021;2. https://doi.org/10.3389/frsus.2021.692055.
- [61] Corrado S, Ardente F, Sala S, Saouter E. Modelling of food loss within life cycle assessment: From current practice towards a systematisation. J Clean Prod 2017;140:847–59. https:// doi.org/10.1016/j.jclepro.2016.06.050.
- [62] Finnveden G, Hauschild MZ, Ekvall T, Guinée J, Heijungs R, Hellweg S, et al. Recent developments in Life Cycle Assessment. J Environ Manage 2009. https://doi.org/10.1016/j. jenvman.2009.06.018.

- [63] Schmidt JH, Weidema BP, Brandão M. A framework for modelling indirect land use changes in Life Cycle Assessment. J Clean Prod 2015;99:230–8. https://doi.org/10.1016/j. jclepro.2015.03.013.
- [64] Plevin RJ, Delucchi MA, Creutzig F. Using Attributional Life Cycle Assessment to Estimate Climate-Change Mitigation Benefits Misleads Policy Makers. J Ind Ecol 2014. https://doi.org/10.1111/jiec.12074.
- [65] Nessi S, Bulgheroni C, Garbarino E, Garcia-Gutierrez P, Orveillon G, Sinkko T, et al. Environmental sustainability assessment comparing through the means of lifecycle assessment the potential environmental impacts of the use of alternative feedstock (biomass, recycled plastics, CO2) for plastic articles in comparison to using current feeds. Draft Rep Stakehold Consult (Part 2) 2018.
- [66] Wilfart A, Gac A, Salaün Y, Aubin J, Espagnol S. Allocation in the LCA of meat products: is agreement possible? Clean Environ Syst 2021;2:100028. https://doi.org/10.1016/j. cesys.2021.100028.
- [67] Montazeri M, Zaimes GG, Khanna V, Eckelman MJ. Meta-Analysis of Life Cycle Energy and Greenhouse Gas Emissions for Priority Biobased Chemicals. ACS Sustain Chem Eng 2016;4:6443–54. https://doi.org/10.1021/acssuschemeng.6b01217.
- [68] Pelletier N, Ardente F, Brandão M, De Camillis C, Pennington D. Rationales for and limitations of preferred solutions for multi-functionality problems in LCA: is increased consistency possible? Int J Life Cycle Assess 2015;20:74–86. https://doi.org/10.1007/ s11367-014-0812-4.
- [69] Brankatschk G, Finkbeiner M. Application of the Cereal Unit in a new allocation procedure for agricultural life cycle assessments. J Clean Prod 2014;73:72–9. https://doi. org/10.1016/j.jclepro.2014.02.005.
- [70] European Commission. International Reference Life Cycle Data System (ILCD) Handbook - General guide for Life Cycle Assessment - Detailed guidance. 2010. https:// doi.org/10.2788/38479.
- [71] Schrijvers DL, Loubet P, Sonnemann G. Developing a systematic framework for consistent allocation in LCA. Int J Life Cycle Assess 2016;21:976–93. https://doi.org/10.1007/ s11367-016-1063-3.
- [72] De Camillis C, Brandão M, Zamagni A, Pennington D. Sustainability assessment of future-oriented scenarios: a review of data modelling approaches in Life Cycle Assessment. Towards recommendations for policy making and business strategies. 2013. https://doi.org/10.2788/95227.
- [73] Sandin G, Peters GM, Svanström M. Life cycle assessment of construction materials: The influence of assumptions in end-of-life modelling. Int J Life Cycle Assess 2014;19:723–31. https://doi.org/10.1007/s11367-013-0686-x.
- [74] Heijungs R. Economic drama and the environmental stage: Formal derivation of algorithmic tools for environmental analysis and decisionsupport from a unified epistemological principle. Leiden University, 1997.
- [75] Marinković S, Dragaš J, Ignjatović I, Tošić N. Environmental assessment of green concretes for structural use. J Clean Prod 2017;154:633–49. https://doi.org/10.1016/j. jclepro.2017.04.015.
- [76] Manninen K, Koskela S, Nuppunen A, Sorvari J, Nevalainen O, Siitonen S. The applicability of the renewable energy directive calculation to assess the sustainability of biogas production. Energy Policy 2013;56:549–57. https://doi.org/10.1016/j.enpol.2013.01.040.

- [77] Karlsson H, Börjesson P, Hansson PA, Ahlgren S. Ethanol production in biorefineries using lignocellulosic feedstock - GHG performance, energy balance and implications of life cycle calculation methodology. J Clean Prod 2014. https://doi.org/10.1016/j. jclepro.2014.07.029.
- [78] García CA, Fuentes A, Hennecke A, Riegelhaupt E, Manzini F, Masera O. Life-cycle greenhouse gas emissions and energy balances of sugarcane ethanol production in Mexico. Appl Energy 2011;88:2088–97. https://doi.org/10.1016/j.apenergy.2010.12.072.
- [79] Li X, Mupondwa E. Life cycle assessment of camelina oil derived biodiesel and jet fuel in the Canadian Prairies. Sci Total Environ 2014;481:17–26. https://doi.org/10.1016/j. scitotenv.2014.02.003.
- [80] Ekvall T, Tillman AM. Open-loop recycling: Criteria for allocation procedures. Int J Life Cycle Assess 1997;2:155–62. https://doi.org/10.1007/BF02978810.
- [81] Tillman AM, Ekvall T, Baumann H, Rydberg T. Choice of system boundaries in life cycle assessment. J Clean Prod 1994;2:21–9. https://doi.org/10.1016/0959-6526(94)90021-3.
- [82] Flysjö A, Cederberg C, Henriksson M, Ledgard S. How does co-product handling affect the carbon footprint of milk? Case study of milk production in New Zealand and Sweden. Int J Life Cycle Assess 2011;16:420–30. https://doi.org/10.1007/s11367-011-0283-9.
- [83] Muench S, Guenther E. A systematic review of bioenergy life cycle assessments. Appl Energy 2013;112:257–73. https://doi.org/10.1016/j.apenergy.2013.06.001.
- [84] van der Harst E, Potting J, Kroeze C. Comparison of different methods to include recycling in LCAs of aluminium cans and disposable polystyrene cups. Waste Manag 2016;48:565–83. https://doi.org/10.1016/j.wasman.2015.09.027.
- [85] EC. Product Environmental Footprint (PEF) Guide 2012:154. https://doi.org/ Ares(2012)873782 - 17/07/2012.
- [86] Pelletier N, Ardente F, Brandão M, De Camillis C, Pennington D. Rationales for and limitations of preferred solutions for multi-functionality problems in LCA: is increased consistency possible? Int J LCA 2014. https://doi.org/10.1007/s11367-014-0812-4.
- [87] Famiglietti J, Guerci M, Proserpio C, Ravaglia P, Motta M. Development and testing of the Product Environmental Footprint Milk Tool: A comprehensive LCA tool for dairy products. Sci Total Environ 2019;648:1614–26. https://doi.org/10.1016/j. scitotenv.2018.08.142.
- [88] Görkem F. The Environmental Life Cycle Assessment of Dairy Products. Food Eng Rev 2019. https://doi.org/https://doi.org/10.1007/s12393-019-9187-4.
- [89] Thomassen MA, Dalgaard R, Heijungs R, De Boer I. Attributional and consequential LCA of milk production. Int J Life Cycle Assess 2008;13:339–49. https://doi.org/10.1007/ s11367-008-0007-y.
- [90] Desjardins RL, Worth DE, Vergé XPC, Maxime D, Dyer J, Cerkowniak D. Carbon footprint of beef cattle. Sustainability 2012;4:3279–301. https://doi.org/10.3390/su4123279.
- [91] Baldini C, Gardoni D, Guarino M. A critical review of the recent evolution of Life Cycle Assessment applied to milk production. J Clean Prod 2017;140:421–35. https://doi. org/10.1016/j.jclepro.2016.06.078.
- [92] Finnegan W, Yan M, Holden NM, Goggins J. A review of environmental life cycle assessment studies examining cheese production. Int J Life Cycle Assess 2018. https:// doi.org/10.1007/s11367-017-1407-7.

- [93] Cederberg C, Stadig M. System Expansion and Allocation in Life Cycle Assessment of Milk and Beef Production. Int J Life Cycle Assess 2003. https://doi.org/10.1007/ BF02978508.
- [94] Agostini A, Giuntoli J, Marelli L, Amaducci S. Flaws in the interpretation phase of bioenergy LCA fuel the debate and mislead policymakers. Int J Life Cycle Assess 2019. https://doi.org/10.1007/s11367-019-01654-2.
- [95] European Bioplastics. Bioplastics. Facts and figures. 2020.
- [96] Klöpffer W. The critical review of life cycle assessment studies according to ISO 14040 and 14044. Int J Life Cycle Assess 2012;17:1087–93. https://doi.org/10.1007/s11367-012-0426-7.
- [97] Azapagic A, Clift R. Allocation of environmental burdens in multiple-function systems. J Clean Prod 1999;7:101–19. https://doi.org/10.1016/s0959-6526(98)00046-8.
- [98] Parajuli R, Knudsen MT, Birkved M, Djomo SN, Corona A, Dalgaard T. Environmental impacts of producing bioethanol and biobased lactic acid from standalone and integrated biorefineries using a consequential and an attributional life cycle assessment approach. Sci Total Environ 2017;598:497–512. https://doi.org/10.1016/j. scitotenv.2017.04.087.
- [99] Ekvall T, Andrae ASG. Attributional and consequential environmental assessment of the shift to lead-free solders. Int J Life Cycle Assess 2006;11:344–53. https://doi.org/10.1065/ lca2005.05.208.
- [100] Culbertson C, Treasure T, Venditti R, Jameel H, Gonzalez R. Life cycle assessment of lignin extraction in a softwood kraft pulp mill. Nord Pulp Pap Res J 2016;31:30–40. https://doi.org/10.3183/npprj-2016-31-01-p030-040.
- [101] Boerjan W, Ralph J, Baucher M. Lignin biosynthesis. Annu Rev Plant Biol 2003;54:519– 46. https://doi.org/10.1146/annurev.arplant.54.031902.134938.
- [102] Ragauskas AJ, Beckham GT, Biddy MJ, Chandra R, Chen F, Davis MF, et al. Lignin valorization: Improving lignin processing in the biorefinery. Science (80-) 2014;344. https://doi.org/10.1126/science.1246843.
- [103] Bernier E, Lavigne C, Robidoux PY. Life cycle assessment of kraft lignin for polymer applications. Int J Life Cycle Assess 2013;18:520–8. https://doi.org/10.1007/s11367-012-0503-y.
- [104] Yuan Y, Guo M. Do green wooden composites using lignin-based binder have environmentally benign alternatives? A preliminary LCA case study in China. Int J Life Cycle Assess 2017;22:1318–26. https://doi.org/10.1007/s11367-016-1235-1.
- [105] Balaguera A, Carvajal GI, Albertí J, Fullana-i-Palmer P. Life cycle assessment of road construction alternative materials: A literature review. Resour Conserv Recycl 2018;132:37–48. https://doi.org/10.1016/j.resconrec.2018.01.003.
- [106] Obydenkova S V., Kouris PD, Hensen EJM, Heeres HJ, Boot MD. Environmental economics of lignin derived transport fuels. Bioresour Technol 2017;243:589–99. https:// doi.org/10.1016/j.biortech.2017.06.157.
- [107] Khan TA, Lee JH, Kim HJ. Lignin-based adhesives and coatings. Lignocellul Futur Bioeconomy 2019:153–206. https://doi.org/10.1016/B978-0-12-816354-2.00009-8.
- [108] Czaikoski A, Gomes A, Kaufmann KC, Liszbinski RB, de Jesus MB, Cunha RL da. Lignin derivatives stabilizing oil-in-water emulsions: Technological aspects, interfacial rheology and cytotoxicity. Ind Crops Prod 2020;154. https://doi.org/10.1016/j.indcrop.2020.112762.

- [109] Hao W, Björnerbäck F, Trushkina Y, Bengoechea MO, Salazar-Alvarez G, Barth T, et al. High-performance Magnetic Activated Carbon from Solid Waste from Lignin Conversion Processes. Part I: Their Use as Adsorbents for CO2. Energy Procedia 2017;114:6272–96. https://doi.org/10.1016/j.egypro.2017.08.033.
- [110] Park S, Choi MS, Park HS. Nitrogen-doped nanoporous carbons derived from lignin for high CO2 capacity. Carbon Lett 2019;29:289–96. https://doi.org/10.1007/s42823-019-00025-z.
- [111] Hernández-Ramos F, Fernández-Rodríguez J, Alriols MG, Labidi J, Erdocia X. Study of a renewable capping agent addition in lignin base catalyzed depolymerization process. Fuel 2020;280. https://doi.org/10.1016/j.fuel.2020.118524.
- [112] Cordeiro-Junior PJM, Kronka MS, Goulart LA, Veríssimo NC, Mascaro LH, Santos MC dos, et al. Catalysis of oxygen reduction reaction for H2O2 electrogeneration: The impact of different conductive carbon matrices and their physicochemical properties. J Catal 2020;392:56–68. https://doi.org/10.1016/j.jcat.2020.09.020.
- [113] Soam S, Kapoor M, Kumar R, Borjesson P, Gupta RP, Tuli DK. Global warming potential and energy analysis of second generation ethanol production from rice straw in India. Appl Energy 2016;184:353–64. https://doi.org/10.1016/j.apenergy.2016.10.034.
- [114] Axelsson E, Olsson MR, Berntsson T. Increased capacity in kraft pulp mills: Lignin separation and reduced steam demand compared with recovery boiler upgrade. Nord Pulp Pap Res J 2006;21:485–92. https://doi.org/10.3183/npprj-2006-21-04-p485-492.
- [115] Arias A, González-García S, González-Rodríguez S, Feijoo G, Moreira MT. Cradle-to-gate Life Cycle Assessment of bio-adhesives for the wood panel industry. A comparison with petrochemical alternatives. Sci Total Environ 2020;738. https://doi.org/10.1016/j. scitotenv.2020.140357.
- [116] Carvajal JC, Gómez Á, Cardona CA. Comparison of lignin extraction processes: Economic and environmental assessment. Bioresour Technol 2016;214:468–76. https:// doi.org/10.1016/j.biortech.2016.04.103.
- [117] Hodásová L, Jablonský M, Škulcova A, Aleš H. Lignin, potential products and their market value. Wood Res 2015;60:973–86.
- [118] Gosselink R. Lignin as a renewable aromatic resource for the chemical industry. 2011.
- [119] Secchi M, Castellani V, Orlandi M, Collina E. Use of Lignin side-streams from biorefineries as fuel or co-product? Life cycle analysis of bio-ethanol and pulp production processes. BioResources 2019;14:4832–65. https://doi.org/10.15376/biores.14.2.4832-4865.
- [120] Cheremisinoff NP, Rosenfeld PE. Sources of air emissions from pulp and paper mills. Handb Pollut Prev Clean Prod 2010;2:179–259. https://doi.org/10.1016/b978-0-08-096446-1.10006-1.
- [121] Viikari L, Suurnäkki A, Grönqvist S, Raaska L, Ragauskas A. Forest Products: Biotechnology in Pulp and Paper Processing. Encycl Microbiol 2009:80–94. https://doi. org/10.1016/B978-012373944-5.00123-1.
- [122] Bajwa DS, Pourhashem G, Ullah AH, Bajwa SG. A concise review of current lignin production, applications, products and their environment impact. Ind Crops Prod 2019;139. https://doi.org/10.1016/j.indcrop.2019.111526.
- [123] Radotić K, Mićić M. Methods for Extraction and Purification of Lignin and Cellulose from Plant Tissues 2016:365–76. https://doi.org/10.1007/978-1-4939-3185-9_26.

- [124] Vural Gursel I, Dijkstra JW, Huijgen WJJ, Ramirez A. Techno-economic comparative assessment of novel lignin depolymerization routes to bio-based aromatics. Biofuels, Bioprod Biorefining 2019;13:1068–84. https://doi.org/10.1002/bbb.1999.
- [125] Renders T, Van Den Bosch S, Koelewijn SF, Schutyser W, Sels BF. Lignin-first biomass fractionation: The advent of active stabilisation strategies. Energy Environ Sci 2017;10:1551–7. https://doi.org/10.1039/c7ee01298e.
- [126] Tarabanko VE, Tarabanko N. Catalytic oxidation of lignins into the aromatic aldehydes: General process trends and development prospects. Int J Mol Sci 2017;18. https://doi. org/10.3390/ijms18112421.
- [127] ISO. ISO 14040: Environmental management Life Cycle Assessment Principles and Framework. 2006.
- [128] Moretti C, Junginger M, Shen L. Environmental life cycle assessment of polypropylene made from used cooking oil. Resour Conserv Recycl 2020;157:104750. https://doi. org/10.1016/j.resconrec.2020.104750.
- [129] Yadav P, Athanassiadis D, Antonopoulou I, Rova U, Christakopoulos P, Tysklind M, et al. Environmental Impact and Environmental Cost Assessment of a Novel Lignin Production Method. J Clean Prod 2020:123515. https://doi.org/10.1016/j.jclepro.2020.123515.
- [130] Yiin CL, Yusup S, Quitain AT, Uemura Y, Sasaki M, Kida T. Life cycle assessment of oil palm empty fruit bunch delignification using natural malic acid-based low-transitiontemperature mixtures: a gate-to-gate case study. Clean Technol Environ Policy 2018;20:1917–28. https://doi.org/10.1007/s10098-018-1590-7.
- [131] Hildebrandt J, Budzinski M, Nitzsche R, Weber A, Krombholz A, Thrän D, et al. Assessing the technical and environmental performance of wood-based fiber laminates with lignin based phenolic resin systems. Resour Conserv Recycl 2019;141:455–64. https://doi. org/10.1016/j.resconrec.2018.10.029.
- [132] McDevitt JE, Grigsby WJ. Life Cycle Assessment of Bio- and Petro-Chemical Adhesives Used in Fiberboard Production. J Polym Environ 2014;22:537–44. https://doi.org/10.1007/ s10924-014-0677-4.
- [133] Liao Y, Koelewijn SF, van den Bossche G, van Aelst J, van den Bosch S, Renders T, et al. A sustainable wood biorefinery for low-carbon footprint chemicals production. Science (80-) 2020;367:1385–90. https://doi.org/10.1126/science.aau1567.
- [134] Tokede OO, Whittaker A, Mankaa R, Traverso M. Life cycle assessment of asphalt variants in infrastructures: The case of lignin in Australian road pavements. Structures 2020;25:190–9. https://doi.org/10.1016/j.istruc.2020.02.026.
- [135] Koch D, Paul M, Beisl S, Friedl A, Mihalyi B. Life cycle assessment of a lignin nanoparticle biorefinery: Decision support for its process development. J Clean Prod 2020;245. https://doi.org/10.1016/j.jclepro.2019.118760.
- [136] Manzardo A, Marson A, Roso M, Boaretti C, Modesti M, Scipioni A, et al. Life Cycle Assessment Framework to Support the Design of Biobased Rigid Polyurethane Foams. ACS Omega 2019;4:14114–23. https://doi.org/10.1021/acsomega.9b02025.
- [137] Isola C, Sieverding HL, Numan-Al-Mobin AM, Rajappagowda R, Boakye EA, Raynie DE, et al. Vanillin derived from lignin liquefaction: a sustainability evaluation. Int J Life Cycle Assess 2018;23:1761–72. https://doi.org/10.1007/s11367-017-1401-0.
- [138] Corona A, Biddy MJ, Vardon DR, Birkved M, Hauschild MZ, Beckham GT. Life cycle assessment of adipic acid production from lignin. Green Chem 2018;20:3857–66. https:// doi.org/10.1039/c8gc00868j.

- [139] Van Duuren JBJH, Brehmer B, Mars AE, Eggink G, dos Santos VAPM, Sanders JPM. A limited LCA of bio-adipic acid: Manufacturing the nylon-6,6 precursor adipic acid using the benzoic acid degradation pathway from different feedstocks. Biotechnol Bioeng 2011;108:1298–306. https://doi.org/10.1002/bit.23074.
- [140] Montazeri M, Eckelman MJ. Life Cycle Assessment of Catechols from Lignin Depolymerization. ACS Sustain Chem Eng 2016;4:708–18. https://doi.org/10.1021/ acssuschemeng.5b00550.
- [141] Das S. Life cycle assessment of carbon fiber-reinforced polymer composites. Int J Life Cycle Assess 2011;16:268–82. https://doi.org/10.1007/s11367-011-0264-z.
- [142] Akmalina R, Pawitra MG. Life cycle assessment of ethylene production from empty fruit bunch. Asia-Pacific J Chem Eng 2020;15. https://doi.org/10.1002/apj.2436.
- [143] Kumaniaev I, Navare K, Crespo Mendes N, Placet V, Van Acker K, Samec JSM. Conversion of birch bark to biofuels. Green Chem 2020;22:2255–63. https://doi.org/10.1039/ d0gc00405g.
- [144] Nguyen TLT, Hermansen JE. System expansion for handling co-products in LCA of sugar cane bio-energy systems: GHG consequences of using molasses for ethanol production. Appl Energy 2012;89:254–61. https://doi.org/10.1016/j.apenergy.2011.07.023.
- [145] European Commission. ILCD Handbook General guide on LCA Detailed guidance. Constraints 2010;15:524–5. https://doi.org/10.2788/38479.
- [146] Moretti C, Corona B, Edwards R, Junginger M, Moro A, Rocco M, et al. Reviewing ISO compliant multifunctionality practices in environmental life cycle modeling. Energies 2020;13:3579. https://doi.org/10.3390/en13143579.
- [147] Raman JK, Gnansounou E. LCA of bioethanol and furfural production from vetiver. Bioresour Technol 2015;185:202–10. https://doi.org/10.1016/j.biortech.2015.02.096.
- [148] Budsberg E, Crawford JT, Morgan H, Chin WS, Bura R, Gustafson R. Hydrocarbon bio-jet fuel from bioconversion of poplar biomass: Life cycle assessment. Biotechnol Biofuels 2016;9. https://doi.org/10.1186/s13068-016-0582-2.
- [149] Ecoinvent. ecoinvent 2020. https://www.ecoinvent.org/ (accessed September 11, 2020).
- [150] Rahman MM, Canter C, Kumar A. Well-to-wheel life cycle assessment of transportation fuels derived from different North American conventional crudes. Appl Energy 2015;156:159–73. https://doi.org/10.1016/j.apenergy.2015.07.004.
- [151] Soam S, Kapoor M, Kumar R, Gupta RP, Puri SK, Ramakumar SSV. Life cycle assessment and life cycle costing of conventional and modified dilute acid pretreatment for fuel ethanol production from rice straw in India. J Clean Prod 2018;197:732–41. https://doi. org/10.1016/j.jclepro.2018.06.204.
- [152] González-García S, Gullón B, Rivas S, Feijoo G, Moreira MT. Environmental performance of biomass refining into high-added value compounds. J Clean Prod 2016;120:170–80. https://doi.org/10.1016/j.jclepro.2016.02.015.
- [153] Ojeda K, Ávila O, Suárez J, Kafarov V. Evaluation of technological alternatives for process integration of sugarcane bagasse for sustainable biofuels production-Part 1. Chem Eng Res Des 2011;89:270–9. https://doi.org/10.1016/j.cherd.2010.07.007.
- [154] Shinde PN, Mandavgane SA, Karadbhajane V. Process development and life cycle assessment of pomegranate biorefinery. Environ Sci Pollut Res 2020;27:25785–93. https://doi.org/10.1007/s11356-020-08957-0.

- [155] Modahl IS, Brekke A, Valente C. Environmental assessment of chemical products from a Norwegian biorefinery. J Clean Prod 2015;94:247–59. https://doi.org/10.1016/j. jclepro.2015.01.054.
- [156] Klöpffer W. The critical review of life cycle assessment studies according to ISO 14040 and 14044. Int J Life Cycle Assess 2012. https://doi.org/10.1007/s11367-012-0426-7.
- [157] Azapagic A, Clift R. Linear programming as a tool in life cycle assessment. Int J Life Cycle Assess 1998;3:305–16. https://doi.org/10.1007/BF02979340.
- [158] Laurent A, Clavreul J, Bernstad A, Bakas I, Niero M, Gentil E, et al. Review of LCA studies of solid waste management systems - Part II: Methodological guidance for a better practice. Waste Manag 2014;34:589–606. https://doi.org/10.1016/j.wasman.2013.12.004.
- [159] Beck T, Bos U, Wittstock B, Baitz M, Fischer M, Seldbauer K. Land Use Indicator Value Calculation in Life Cycle Assessment – Method Report. 2010.
- [160] Hélias A. Comments on the international consensus model for the water scarcity footprint (AWARE) and proposal for an improvement. Sci Total Environ 2020;709. https://doi.org/10.1016/j.scitotenv.2019.136189.
- [161] Benali M, Ajao O, Jeaidi J, Gilani B, Mansoornejad B. Integrated Lignin-Kraft Pulp Biorefinery for the Production of Lignin and Its Derivatives: Economic Assessment and LCA-Based Environmental Footprint 2016:379–418. https://doi.org/10.1007/978-981-10-1965-4_13.
- [162] Patel AD, Meesters K, Den Uil H, De Jong E, Blok K, Patel MK. Sustainability assessment of novel chemical processes at early stage: Application to biobased processes. Energy Environ Sci 2012. https://doi.org/10.1039/c2ee21581k.
- [163] Moretti C, Corona B, Rühlin V, Götz T, Junginger M, Brunner T, et al. Combining biomass gasification and solid oxid fuel cell for heat and power generation: An early-stage life cycle assessment. Energies 2020;13:2773. https://doi.org/10.3390/en13112773.
- [164] Shuai W, Chen N, Li B, Zhou D, Gao J. Life cycle assessment of common reed (Phragmites australis (Cav) Trin. ex Steud) cellulosic bioethanol in Jiangsu Province, China. Biomass and Bioenergy 2016;92:40–7. https://doi.org/10.1016/j.biombioe.2016.06.002.
- [165] European Commission. PEFCR Guidance document, Guidance for the development of Product Environmental Footprint Category Rules (PEFCRs), version 6.3, December 14 2017. 2018.
- [166] EPA. GUIDANCE FOR DETERMINING BEST AVAILABLE Guidance for Determining Best Available Control Technology for Reducing Carbon Dioxide Emissions from Bioenergy Production. 2011.
- [167] BSI. PAS 2050: 2011 Specification for the assessment of the life cycle greenhouse gas emissions of goods and services. 2011. https://doi.org/978 0 580 71382 8.
- [168] van Vliet D, Slaghek T, Giezen C, Haaksman I. Lignin as a green alternative for bitumen 2017. https://doi.org/10.14311/ee.2016.159.
- [169] Landa and Gosselink. Lignin-based bio-asphalt, 2019.
- [170] EU. Directive 2008/98/EC of the European Parliament and of the Council of 19 November 2008 on waste and repealing certain directives. 2008. https://doi.org/2008/98/EC.; 32008L0098.
- [171] Benini L, Mancini L, Sala S, Schau EM, Manfredi S, Pant R. Normalisation method and data for Environmental Footprints JRC Technical Report Normalisation method and data for Environmental Footprints 2014.

- [172] Turk J, Oven P, Poljanšek I, Lešek A, Knez F, Malovrh Rebec K. Evaluation of an environmental profile comparison for nanocellulose production and supply chain by applying different life cycle assessment methods. J Clean Prod 2020;247. https://doi. org/10.1016/j.jclepro.2019.119107.
- [173] Nascimento DM do, Dias AF, Araújo Junior CP de, Rosa M de F, Morais JPS, Figueirêdo MCB de. A comprehensive approach for obtaining cellulose nanocrystal from coconut fiber. Part II: Environmental assessment of technological pathways. Ind Crops Prod 2016;93:58–65. https://doi.org/10.1016/j.indcrop.2016.02.063.
- [174] Zamagni A, Buttol P, Porta PL, Buonamici R, Masoni P, Guinée J, et al. Critical review of the current research needs and limitations related to ISO-LCA practice. 2008. https:// doi.org/10.1590/S0102-88392004000200002.
- [175] Weidema B. Avoiding Co-Product Allocation in Life-Cycle Assessment. J Ind Ecol 2000;4:11–33. https://doi.org/doi:10.1162/108819800300106366.
- [176] Lloyd SM, Ries R. Characterizing, propagating, and analyzing uncertainty in life-cycle assessment: A survey of quantitative approaches. J Ind Ecol 2007. https://doi.org/10.1162/ jiec.2007.1136.
- [177] Brando M, Martin M, Cowie A, Hamelin L, Zamagni A. Consequential Life Cycle Assessment: What, How, and Why? Encycl Sustain Technol 2017:277–84. https://doi. org/10.1016/B978-0-12-409548-9.10068-5.
- [178] Benetto E, Jury C, Kneip G, Vázquez-Rowe I, Huck V, Minette F. Life cycle assessment of heat production from grape marc pellets. J Clean Prod 2015;87:149–58. https://doi. org/10.1016/j.jclepro.2014.10.028.
- [179] Zaimes GG, Khanna V. The role of allocation and coproducts in environmental evaluation of microalgal biofuels: How important? Sustain Energy Technol Assessments 2014;7:247–56. https://doi.org/10.1016/j.seta.2014.01.011.
- [180] Śliwińska A, Burchart-Korol D, Smoliński A. Environmental life cycle assessment of methanol and electricity co-production system based on coal gasification technology. Sci Total Environ 2017;574:1571–9. https://doi.org/10.1016/j.scitotenv.2016.08.188.
- [181] Bava L, Bacenetti J, Gislon G, Pellegrino L, D'Incecco P, Sandrucci A, et al. Impact assessment of traditional food manufacturing: The case of Grana Padano cheese. Sci Total Environ 2018;626:1200–9. https://doi.org/10.1016/j.scitotenv.2018.01.143.
- [182] Esteves VPP, Esteves EMM, Bungenstab DJ, Feijó GLD, Araújo O de QF, Morgado C do RV. Assessment of greenhouse gases (GHG) emissions from the tallow biodiesel production chain including land use change (LUC). J Clean Prod 2017;151:578–91. https:// doi.org/10.1016/j.jclepro.2017.03.063.
- [183] Pa A, Craven JS, Bi XT, Melin S, Sokhansanj S. Environmental footprints of British Columbia wood pellets from a simplified life cycle analysis. Int J Life Cycle Assess 2012;17:220–31. https://doi.org/10.1007/s11367-011-0358-7.
- [184] Liu JS, Chen HH, Ho MHC, Li YC. Citations with different levels of relevancy: Tracing the main paths of legal opinions. J Assoc Inf Sci Technol 2014. https://doi.org/10.1002/ asi.23135.
- [185] Xiao Y, Lu LYY, Liu JS, Zhou Z. Knowledge diffusion path analysis of data quality literature: A main path analysis. J Informetr 2014. https://doi.org/10.1016/j.joi.2014.05.001.
- [186] ISO. TECHNICAL REPORT ISO / TR 14049 Environmental management Life cycle assessment — Examples of application of ISO 14041 to goal and scope definition and 2000;2000.

- [187] ISO. ISO/TR 14049:2012. Environmental management Life cycle assessment Illustrative examples on how to apply ISO 14044 to goal and scope definition and inventory analysis. 2012.
- [188] ISO. Iso 14041. Environ Manag Life Cycle Assess Goal Scope Defin Invent Anal 1998;1998.
- [189] ISO. MS ISO/TS 14072:2014 2014.
- [190] Moro A, Joanny G, Moretti C. Emerging technologies in the renewable energy sector: A comparison of expert review with a text mining software. Futures 2020;117:102511. https://doi.org/10.1016/j.futures.2020.102511.
- [191] Batagelj V, Mrvar A. Pajek Analysis and Visualization of Large Networks, 2011. https:// doi.org/10.1007/978-3-642-18638-7_4.
- [192] Ciano MP, Strozzi F, Minelli E, Pozzi R, Rossi T. The link between lean and human resource management or organizational behaviour: a bibliometric review. XXIV Summer Sch. "Francesco Turco" – Ind. Syst. Eng., 2019.
- [193] Strozzi F, Colicchia C, Creazza A, Noè C. Literature review on the 'smart factory' concept using bibliometric tools. Int J Prod Res 2017;55:1–20. https://doi.org/10.1080/00207543.20 17.1326643.
- [194] Liu JS, Lu LYY. An integrated approach for main path analysis: Development of the Hirsch index as an example. J Am Soc Inf Sci Technol 2012. https://doi.org/10.1002/ asi.21692.
- [195] Peñaloza D, Erlandsson M, Falk A. Exploring the climate impact effects of increased use of bio-based materials in buildings. Constr Build Mater 2016;125:219–26. https://doi. org/10.1016/j.conbuildmat.2016.08.041.
- [196] Chen C, Habert G, Bouzidi Y, Jullien A, Ventura A. LCA allocation procedure used as an incitative method for waste recycling: An application to mineral additions in concrete. Resour Conserv Recycl 2010;54:1231–40. https://doi.org/10.1016/j.resconrec.2010.04.001.
- [197] Bailis R, Kavlak G. Environmental implications of Jatropha biofuel from a silvi-pastoral production system in central-west Brazil. Environ Sci Technol 2013;47:8042–50. https:// doi.org/10.1021/es303954g.
- [198] Schau EM, Fet AM. LCA studies of food products as background for environmental product declarations. Int J Life Cycle Assess 2008;13:255–64. https://doi.org/10.1065/ lca2007.12.372.
- [199] Heijungs R, Guinée JB. Allocation and "what-if" scenarios in life cycle assessment of waste management systems. Waste Manag 2007. https://doi.org/10.1016/j. wasman.2007.02.013.
- [200] Heijungs R. Ten easy lessons for good communication of LCA. Int J Life Cycle Assess 2014. https://doi.org/10.1007/s11367-013-0662-5.
- [201] Marvuglia A, Cellura M, Heijungs R. Toward a solution of allocation in life cycle inventories: The use of least-squares techniques. Int J Life Cycle Assess 2010;15:1020–40. https://doi.org/10.1007/s11367-010-0214-1.
- [202] Steubing B, Wernet G, Reinhard J, Bauer C, Moreno-Ruiz E. The ecoinvent database version 3 (part II): analyzing LCA results and comparison to version 2. Int J Life Cycle Assess 2016;21:1269–81. https://doi.org/10.1007/s11367-016-1109-6.
- [203] Brander M, Wylie C. The use of substitution in attributional life cycle assessment. Greenh Gas Meas Manag 2011;1:161–6. https://doi.org/10.1080/20430779.2011.637670.

- [204] Nhu TT, Dewulf J, Serruys P, Huysveld S, Nguyen C V., Sorgeloos P, et al. Resource usage of integrated Pig-Biogas-Fish system: Partitioning and substitution within attributional life cycle assessment. Resour Conserv Recycl 2015;102:27–38. https://doi.org/10.1016/j. resconrec.2015.06.011.
- [205] Weidema B. ISO system expansion = substitution. 20 Consult 2014:1.
- [206] Forman GS, Hahn TE, Jensen SD. Greenhouse gas emission evaluation of the GTL pathway. Environ Sci Technol 2011;45:9084–92. https://doi.org/10.1021/es202101b.
- [207] Brockmann D, Pradinaud C, Champenois J, Benoit M, Hélias A. Environmental assessment of bioethanol from onshore grown green seaweed. Biofuels, Bioprod Biorefining 2015;9:696–708. https://doi.org/10.1002/bbb.1577.
- [208] Cherubini E, Franco D, Zanghelini GM, Soares SR. Uncertainty in LCA case study due to allocation approaches and life cycle impact assessment methods. Int J Life Cycle Assess 2018;23:2055–70. https://doi.org/10.1007/s11367-017-1432-6.
- [209] Herrmann IT, Jørgensen A, Bruun S, Hauschild MZ. Potential for optimized production and use of rapeseed biodiesel. Based on a comprehensive real-time LCA case study in Denmark with multiple pathways. Int J Life Cycle Assess 2013;18:418–30. https://doi. org/10.1007/s11367-012-0486-8.
- [210] Forman GS, Hauser AB, Adda SM. Life cycle analysis of gas to liquids (GTL) derived linear alkyl benzene. J Clean Prod 2014;80:30–7. https://doi.org/10.1016/j.jclepro.2014.05.058.
- [211] Weidema BP, Frees N, Nielsen AM. Marginal production technologies for life cycle inventories. Int J Life Cycle Assess 1999. https://doi.org/10.1007/BF02979395.
- [212] Pelletier N, Allacker K, Pant R, Manfredi S. The European Commission Organisation Environmental Footprint method: Comparison with other methods, and rationales for key requirements. Int J Life Cycle Assess 2014;19:387–404. https://doi.org/10.1007/s11367-013-0609-x.
- [213] Cherubini F, Strømman AH, Ulgiati S. Influence of allocation methods on the environmental performance of biorefinery products - A case study. Resour Conserv Recycl 2011. https://doi.org/10.1016/j.resconrec.2011.06.001.
- [214] Ahlgren S, Björklund A, Ekman A, Karlsson H, Berlin J, Börjesson P, et al. Review of methodological choices in LCA of biorefinery systems - key issues and recommendations. Biofuels, Bioprod Biorefining 2015. https://doi.org/10.1002/bbb.1563.
- [215] Mackenzie SG, Leinonen I, Kyriazakis I. The need for co-product allocation in the life cycle assessment of agricultural systems—is "biophysical" allocation progress? Int J Life Cycle Assess 2017. https://doi.org/10.1007/s11367-016-1161-2.
- [216] Azapagic A, Clift R. Allocation of environmental burdens in co-product systems: Process and product-related burdens (part 2). Int J Life Cycle Assess 2000;5:31–6. https://doi. org/10.1007/BF02978557.
- [217] Ekvall T, Finnveden G. Allocation in ISO 14041 a critical review. J Clean Prod 2001;9:197– 208. https://doi.org/10.1016/S0959-6526(00)00052-4.
- [218] González-García S, Moreira MT, Feijoo G. Environmental performance of lignocellulosic bioethanol production from alfalfa stems. Biofuels, Bioprod Biorefining 2010;4:118–31. https://doi.org/10.1002/bbb.204.
- [219] Finnveden G, Albertsson AC, Berendson J, Eriksson E, Höglund LO, Karlsson S, et al. Solid waste treatment within the framework of life-cycle assessment. J Clean Prod 1995;3:189–99. https://doi.org/10.1016/0959-6526(95)00081-X.

- [220] Jungmeier G, Werner F, Jarnehammar A, Hohenthal C, Richter K. Allocation in LCA of wood-based products - Experiences of cost action E9: Part II. Examples. Int J Life Cycle Assess 2002.
- [221] Tehrani Nejad M. A, Saint-Antonin V. Factors driving refinery CO2 intensity, with allocation into products: Comment. Int J Life Cycle Assess 2014;19:24–8. https://doi. org/10.1007/s11367-013-0634-9.
- [222] Moretti C, Moro A, Edwards R, Rocco MV, Colombo E. Analysis of standard and innovative methods for allocating upstream and refinery GHG emissions to oil products. Appl Energy 2017;206:372–81. https://doi.org/10.1016/j.apenergy.2017.08.183.
- [223] International Dairy Federation. A common carbon footprint approach for dairy: The IDF guide to standard lifecycle assessment methodology for the dairy sector. Bull Int Dairy Fed 2015.
- [224] van der Harst E, Potting J, Kroeze C. Comparison of different methods to include recycling in LCAs of aluminium cans and disposable polystyrene cups. Waste Manag 2016;48:565–83. https://doi.org/10.1016/j.wasman.2015.09.027.
- [225] Silva DAL, Lahr FAR, Pavan ALR, Saavedra YMB, Mendes NC, Sousa SR, et al. Do woodbased panels made with agro-industrial residues provide environmentally benign alternatives? An LCA case study of sugarcane bagasse addition to particle board manufacturing. Int J Life Cycle Assess 2014;19:1767–78. https://doi.org/10.1007/s11367-014-0776-4.
- [226] Palmieri N, Forleo MB, Giannoccaro G, Suardi A. Environmental impact of cereal straw management: An on-farm assessment. J Clean Prod 2017;142:2950–64. https://doi. org/10.1016/j.jclepro.2016.10.173.
- [227] Vergé X, Maxime D, Desjardins RL, Vanderzaag AC. Allocation factors and issues in agricultural carbon footprint: A case study of the Canadian pork industry. J Clean Prod 2016;113:587–95. https://doi.org/10.1016/j.jclepro.2015.11.046.
- [228] Tufvesson LM, Tufvesson P, Woodley JM, Börjesson P. Life cycle assessment in green chemistry: Overview of key parameters and methodological concerns. Int J Life Cycle Assess 2013;18:431–44. https://doi.org/10.1007/s11367-012-0500-1.
- [229] Vidal R, Martínez P, Garraín D. Life cycle assessment of composite materials made of recycled thermoplastics combined with rice husks and cotton linters. Int J Life Cycle Assess 2009;14:73–82. https://doi.org/10.1007/s11367-008-0043-7.
- [230] Clift R. Chairman's report of session 3: Causality and allocation procedures., 1994, p. In: (HuPPES and SCHNEIDF.R 1994) pp. 3-4.
- [231] Clift R. Report from SETAC-Europe Working Group on Life Cycle Inventory Analysis. Annu. Meet. Taormina, May 1996. SETAC-Europe, Brussels, 1996, p. 17, 1996, p. In: Abstract Book, 6th SETAC-Europe.
- [232] Azapagic A, Clift R. Life cycle assessment and linear programming environmental optimisation of product system-. Comput Chem Eng 1995;19:229–34. https://doi. org/10.1016/0098-1354(95)87041-5.
- [233] ISO. ISO (1996): ISO/TC207/SC5AXIG2: CD 14 041.2. N99, DIN 1996.
- [234] Azapagic A, Clift R. Allocation of Environmental Burdens in Co-product Systems : Product-related Burdens (Part 1). Int J Life Cycle Assess 1999;4:357–69.
- [235] Ekvall T, Weidema BP. System boundaries and input data in consequential life cycle inventory analysis. Int J Life Cycle Assess 2004;9:161–71. https://doi.org/10.1007/ BF02994190.

- [236] Thrane M. LCA of Danish fish products: New methods and insights. Int J Life Cycle Assess 2006;11:66–74. https://doi.org/10.1065/lca2006.01.232.
- [237] Schmidt JH, Weidema BP. Shift in the marginal supply of vegetable oil. Int J Life Cycle Assess 2008;13:235–9. https://doi.org/10.1065/Ica2007.07.351.
- [238] Dalgaard R, Schmidt J, Halberg N, Christensen P, Thrane M, Pengue WA. LCA for soybean meal. LCA Food Prod 2008;10:240–54. https://doi.org/10.1065/lca2007.06.342.
- [239] Bier JM, Verbeek CJR, Lay MC. An eco-profile of thermoplastic protein derived from blood meal Part 1: Allocation issues. Int J Life Cycle Assess 2012;17:208–19. https://doi. org/10.1007/s11367-011-0349-8.
- [240] Wardenaar T, Van Ruijven T, Beltran AM, Vad K, Guinée J, Heijungs R. Differences between LCA for analysis and LCA for policy: A case study on the consequences of allocation choices in bio-energy policies. Int J Life Cycle Assess 2012;17:1059–67. https:// doi.org/10.1007/s11367-012-0431-x.
- [241] van der Werf HMG, Nguyen TTH. Construction cost of plant compounds provides a physical relationship for co-product allocation in life cycle assessment. Int J Life Cycle Assess 2015;20:777–84. https://doi.org/10.1007/s11367-015-0872-0.
- [242] Pradel M, Aissani L, Canler JP, Roux JC, Villot J, Baudez JC, et al. Constructing an allocation factor based on product- and process-related parameters to assess environmental burdens of producing value-added sludge-based products. J Clean Prod 2018. https://doi.org/10.1016/j.jclepro.2017.10.112.
- [243] Pradel M, Aissani L. Environmental impacts of phosphorus recovery from a "product" Life Cycle Assessment perspective: Allocating burdens of wastewater treatment in the production of sludge-based phosphate fertilizers. Sci Total Environ 2019;656:55–69. https://doi.org/10.1016/j.scitotenv.2018.11.356.
- [244] Curran MA, Mann M, Norris G. The international workshop on electricity data for life cycle inventories. J Clean Prod 2005;13:853–62. https://doi.org/10.1016/j.jclepro.2002.03.001.
- [245] Sollazzo G, Longo S, Cellura M, Celauro C. Impact analysis using life cycle assessment of asphalt production from primary data. Sustain 2020;12:1–21. https://doi.org/10.3390/ su122410171.
- [246] Sustain Euro Road. About Sustain Euro Road 2021. https://sustainableroads.eu/aboutthe-project/ (accessed February 8, 2021).
- [247] Bizarro DEG, Steinmann Z, Nieuwenhuijse I, Keijzer E, Hauck M. Potential carbon footprint reduction for reclaimed asphalt pavement innovations: Lca methodology, best available technology, and near-future reduction potential. Sustain 2021;13:1–20. https://doi.org/10.3390/su13031382.
- [248] Pantini S, Borghi G, Rigamonti L. Towards resource-efficient management of asphalt waste in Lombardy region (Italy): Identification of effective strategies based on the LCA methodology. Waste Manag 2018;80:423–34. https://doi.org/10.1016/j. wasman.2018.09.035.
- [249] Giani MI, Dotelli G, Brandini N, Zampori L. Comparative life cycle assessment of asphalt pavements using reclaimed asphalt, warm mix technology and cold in-place recycling. Resour Conserv Recycl 2015;104:224–38. https://doi.org/10.1016/j.resconrec.2015.08.006.
- [250] Ecochain. Environmental Cost Indicator (MKI) Overview 2019. https://ecochain.com/nl/ knowledge/milieukosten-indicator-mki/ (accessed February 8, 2021).

- [251] Moretti L, Mandrone V, D'Andrea A, Caro S. Comparative "from cradle to gate" life cycle assessments of Hot Mix Asphalt (HMA) materials. Sustain 2017;9. https://doi.org/10.3390/ su9030400.
- [252] Schwarz A, Overmars L, Godoi Bizarro D, Keijzer E, Kuling L, van Horssen A. LCA Achtergrondrapport voor brancherepresentatieve Nederlandse asfaltmengsels 2020. TNO rapport 10987 2020.
- [253] He M, Tu C, Cao DW, Chen YJ. Comparative analysis of bio-binder properties derived from different sources. Int J Pavement Eng 2019;20:792–800. https://doi.org/10.1080/102 98436.2017.1347434.
- [254] Moretti C, Corona B, Hoefnagels R, Vural-Gürsel I, Gosselink R, Junginger M. Review of life cycle assessments of lignin and derived products: Lessons learned. Sci Total Environ 2021;770:144656. https://doi.org/10.1016/j.scitotenv.2020.144656.
- [255] Biobaseddelta. Chain works together to green road construction on a large scale. Program Chaplin has started 2020. https://biobaseddelta.com/news/program-chaplinhas-started/.
- [256] CEN ECFS. EN 15804:2013 Standards Publication Sustainability of construction works — Environmental product declarations — Core rules for the product category of construction products. Int Stand 2013.
- [257] Keijzer E, Kootstra L, Schwarz A, Bizarro D, Kuling L, Van Horssen A, et al. Product Category Rules voorbitumineuze materialen in verkeersdragers en waterwerken in Nederland (PCR Asfalt) 2020:1–71.
- [258] Bouwkwaliteit S. Determination Method Environmental performance Buildings and civil engineering works 2019;31:1–83.
- [259] Martinez-Arguelles G, Giustozzi F, Crispino M, Flintsch GW. Laboratory investigation on mechanical performance of cold foamed bitumen mixes: Bitumen source, foaming additive, fiber-reinforcement and cement effect. Constr Build Mater 2015;93:241–8. https://doi.org/10.1016/j.conbuildmat.2015.05.116.
- [260] Giustozzi F, Crispino M, Toraldo E, Mariani E. Mix design of polymer-modified and fiberreinforced warm-mix asphalts with high amount of reclaimed asphalt pavement: Achieving sustainable and high-performing pavements. Transp Res Rec 2015;2523:3–10. https://doi.org/10.3141/2523-01.
- [261] Borghi A, Jiménez del Barco Carrión A, Lo Presti D, Giustozzi F. Effects of Laboratory Aging on Properties of Biorejuvenated Asphalt Binders. J Mater Civ Eng 2017;29:04017149. https://doi.org/10.1061/(asce)mt.1943-5533.0001995.
- [262] Brovelli C, Crispino M, Pais JC, Pereira PAA. Assessment of Fatigue Resistance of Additivated Asphalt Concrete Incorporating Fibers and Polymers. J Mater Civ Eng 2014;26:554–8. https://doi.org/10.1061/(asce)mt.1943-5533.0000837.
- [263] Durão V, Silvestre JD, Mateus R, de Brito J. Assessment and communication of the environmental performance of construction products in Europe: Comparison between PEF and EN 15804 compliant EPD schemes. Resour Conserv Recycl 2020;156. https:// doi.org/10.1016/j.resconrec.2020.104703.
- [264] Milieudatabase. Nationale MilieuDATABASE. MAAKT CIRCULAIR BOUWEN MEETBAR. 2020. https://milieudatabase.nl/milieudata/ (accessed March 10, 2021).
- [265] ecoinvent. Allocation cut-off by classification 2021. https://www.ecoinvent.org/ database/system-models-in-ecoinvent-3/cut-off-system-model/allocation-cut-off-byclassification.html (accessed February 15, 2021).

- [266] Tomani P. The lignoboost process. Cellul Chem Technol 2010;44:53-8.
- [267] EPD. Product Category Rules According To Iso 14025. Arable crops. VERSION 2.0. 2016.
- [268] EPD. PRODUCT CATEGORY RULES (PCR). CONSTRUCTION PRODUCTS. PCR 2019:14 VERSION 1.0. 2019.
- [269] ISO. ISO 21930:2017 Sustainability in buildings and civil engineering works Core rules for environmental product declarations of construction products and services. 2017.
- [270] Indexmundi. Wood Pulp Monthly Price Euro per Metric Ton 2020. https://www. indexmundi.com/commodities/?commodity=wood-pulp&months=120¤cy=eur (accessed March 10, 2021).
- [271] EUROSTAT. Electricity prices for non-household consumers bi-annual data (from 2007 onwards) 2020. https://appsso.eurostat.ec.europa.eu/nui/submitViewTableAction. do.
- [272] Carpos P, De Vita A, Tasios N, Siskos P, Kannavou M, Preopoulos A, et al. EU Reference Scenario 2016 - Energy, transport and GHG emissions - Trends to 2050. 2016. https://doi. org/10.2833/9127.
- [273] EAPA. Asphalt in figures 2017. 2017.
- [274] Peterson CL, Hustrulid T. Carbon cycle for rapeseed oil biodiesel fuels. Biomass and Bioenergy 1998;14:91–101. https://doi.org/10.1016/S0961-9534(97)10028-9.
- [275] EREF. Carbon Sequestration in Landfills: Documentation from Field Samples 2014. https://erefdn.org/carbon-sequestration-in-landfills-documentation-from-fieldsamples/#:~:text=Landfills have significant value as,in a landfill will biodegrade (accessed March 10, 2021).
- [276] Khan MU, Ahring BK. Lignin degradation under anaerobic digestion: Influence of lignin modifications -A review. Biomass and Bioenergy 2019;128. https://doi.org/10.1016/j. biombioe.2019.105325.
- [277] Dessbesell L, Yuan Z, Leitch M, Paleologou M, Pulkki R, Xu CC. Capacity Design of a Kraft Lignin Biorefinery for Production of Biophenol via a Proprietary Low-Temperature/ Low-Pressure Lignin Depolymerization Process. ACS Sustain Chem Eng 2018;6:9293– 303. https://doi.org/10.1021/acssuschemeng.8b01582.
- [278] Abbati De Assis C, Greca LG, Ago M, Balakshin MY, Jameel H, Gonzalez R, et al. Techno-Economic Assessment, Scalability, and Applications of Aerosol Lignin Microand Nanoparticles. ACS Sustain Chem Eng 2018;6:11853–68. https://doi.org/10.1021/ acssuschemeng.8b02151.
- [279] Elgowainy A, Han J, Cai H, Wang M, Forman GS, Divita VB. Energy efficiency and greenhouse gas emission intensity of petroleum products at U.S. Refineries. Environ Sci Technol 2014;48:7612–24. https://doi.org/10.1021/es5010347.
- [280] Vos-Effting S de, Keijzer E, Jansen B, Zwamborn A, Mos J, Beentjes T, et al. LCA-Achtergrondrapport voor Nederlandse Asfaltmengsels. Rapport voor opname van brancherepresentatieve asfaltmengsels in de Nationale Milieudatabase. Versie 2.1. TNO 2017 R11029 (Versie 21) 2018.
- [281] Han J, Forman GS, Elgowainy A, Cai H, Wang M, Divita VB. A comparative assessment of resource efficiency in petroleum refining. Fuel 2015;157:292–8. https://doi.org/10.1016/j. fuel.2015.03.038.
- [282] Eurobitume. The Eurobitume Life-Cycle Inventory for Bitumen 2019:48.

- [283] Hischier R, Reichart I. Multifunctional electronic media-traditional media: The Problem of an Adequate Functional Unit. Int J Life Cycle Assess 2003;8:201–8. https://doi. org/10.1007/BF02978472.
- [284] Yuan X, Tang Y, Li Y, Wang Q, Zuo J, Song Z. Environmental and economic impacts assessment of concrete pavement brick and permeable brick production process - A case study in China. J Clean Prod 2018;171:198–208. https://doi.org/10.1016/j. jclepro.2017.10.037.
- [285] Ayer NW, Tyedmers PH, Pelletier NL, Sonesson U, Scholz A. Co-product allocation in life cycle assessments of seafood production systems: Review of problems and strategies. Int J Life Cycle Assess 2007;12:480–7. https://doi.org/10.1065/lca2006.11.284.
- [286] Majeau-Bettez G, Dandres T, Pauliuk S, Wood R, Hertwich E, Strømman AH. Choice of Allocations and Constructs for Attributional or Consequential Life Cycle Assessment and Input-Output Analysis 2017:1–15. https://doi.org/10.1111/jiec.12604.
- [287] Azapagic A, Clift R. Life cycle assessment and linear programming environmental optimisation of product system-. Comput Chem Eng 1995;19:229–34. https://doi. org/10.1016/0098-1354(95)87041-5.
- [288] Havukainen J, Nguyen MT, Väisänen S, Horttanainen M. Life cycle assessment of smallscale combined heat and power plant: Environmental impacts of different forest biofuels and replacing district heat produced from natural gas. J Clean Prod 2018. https://doi.org/10.1016/j.jclepro.2017.10.241.
- [289] Boschiero M, Cherubini F, Nati C, Zerbe S. Life cycle assessment of bioenergy production from orchards woody residues in Northern Italy. J Clean Prod 2016. https:// doi.org/10.1016/j.jclepro.2015.09.094.
- [290] Bloess A, Schill WP, Zerrahn A. Power-to-heat for renewable energy integration: A review of technologies, modeling approaches, and flexibility potentials. Appl Energy 2018;212:1611–26. https://doi.org/10.1016/j.apenergy.2017.12.073.
- [291] Lombardi F, Rocco MV, Colombo E. A multi-layer energy modelling methodology to assess the impact of heat-electricity integration strategies: The case of the residential cooking sector in Italy. Energy 2019:1249–60. https://doi.org/10.1016/j.energy.2019.01.004.
- [292] Chiaroni D, Chiesa M, Chiesa V, Franzò S, Frattini F, Toletti G. Introducing a new perspective for the economic evaluation of industrial energy efficiency technologies: An empirical analysis in Italy. Sustain Energy Technol Assessments 2016;15:1–10. https:// doi.org/10.1016/j.seta.2016.02.004.
- [293] Paletto A, Bernardi S, Pieratti E, Teston F, Romagnoli M. Assessment of environmental impact of biomass power plants to increase the social acceptance of renewable energy technologies. Heliyon 2019;5. https://doi.org/10.1016/j.heliyon.2019.e02070.
- [294] González-García S, Bacenetti J. Exploring the production of bio-energy from wood biomass. Italian case study. Sci Total Environ 2019;647:158–68. https://doi.org/10.1016/j. scitotenv.2018.07.295.
- [295] Bacenetti J, Fusi A, Azapagic A. Environmental sustainability of integrating the organic Rankin cycle with anaerobic digestion and combined heat and power generation. Sci Total Environ 2019. https://doi.org/10.1016/j.scitotenv.2018.12.190.
- [296] Götz T, Saurat M, Kaselofsky J, Obernberger I, Brunner T, Weiss G, et al. First Stage Environmental Impact Assessment of a New Highly Efficient and Fuel Flexible Mediumscale CHP Technology Based on Fixed-bed Updraft Biomass Gasification and a SOFC. 27th Eur. Biomass Conf. Exhib., 2019, p. 1586–94. https://doi.org/10.5071/27thEUBCE2019-4DO.2.4.

- [297] Brunner T, Biedermann F, Obernberger I, Hirscher S, Schöch M, Milito C, et al. Development of a new highly efficient and fuel flexible medium.scale CHP technology based on fixed-bed updraft biomass gasification and a SOFC 2018;72:249–67.
- [298] European Commission. Technology readiness levels (TRL). Horiz 2020 Work Program 2014-2015 Gen Annex Extr from Part 19 - Comm Decis C 2015:4995.
- [299] Chianese S, Fail S, Binder M, Rauch R, Hofbauer H, Molino A, et al. Experimental investigations of hydrogen production from CO catalytic conversion of tar rich syngas by biomass gasification. Catal Today 2016:182–91. https://doi.org/10.1016/j. cattod.2016.04.005.
- [300] Fail S, Diaz N, Benedikt F, Kraussler M, Hinteregger J, Bosch K, et al. Wood gas processing to generate pure hydrogen suitable for PEM fuel cells. ACS Sustain Chem Eng 2014;2:2690–8. https://doi.org/10.1021/sc500436m.
- [301] Brunner T, Weiss T, Mandle C, Obernberger I, Ramerstorfer C. D8 . 4 : Preliminary techno-economic performance analysis of the new technologies. 2016.
- [302] Evangelisti S, Lettieri P, Clift R, Borello D. Distributed generation by energy from waste technology: A life cycle perspective. Process Saf Environ Prot 2015;93:161–72. https://doi. org/10.1016/j.psep.2014.03.008.
- [303] Lee YD, Ahn KY, Morosuk T, Tsatsaronis G. Environmental impact assessment of a solid-oxide fuel-cell-based combined-heat-and-power-generation system. Energy 2015;79:455–66. https://doi.org/10.1016/j.energy.2014.11.035.
- [304] Rillo E, Gandiglio M, Lanzini A, Bobba S, Santarelli M, Blengini G. Life Cycle Assessment (LCA) of biogas-fed Solid Oxide Fuel Cell (SOFC) plant. Energy 2017;126:585–602. https:// doi.org/10.1016/j.energy.2017.03.041.
- [305] Adams PWR, Mcmanus & Small-scale biomass gasification CHP utilisation in industry:Energy and environmental evaluation. Sustainable Energy Technologies and Assessments, 6. pp Small-scale biomass gasification CHP utilisation in industry: Energy and environmental evaluation 2014;6:129–40. https://doi.org/10.1016/j.seta.2014.02.002.
- [306] Mehmeti A, McPhail SJ, Pumiglia D, Carlini M. Life cycle sustainability of solid oxide fuel cells: From methodological aspects to system implications. J Power Sources 2016;325:772–85. https://doi.org/10.1016/j.jpowsour.2016.06.078.
- [307] Corona B, Shen L, Junginger M. Preliminary market study for Europe : Detailed market assessment of 4 EU member state markets. 2018.
- [308] Höök M, Tang X. Depletion of fossil fuels and anthropogenic climate change-A review. Energy Policy 2013. https://doi.org/10.1016/j.enpol.2012.10.046.
- [309] Conibear L, Butt EW, Knote C, Arnold SR, Spracklen D V. Residential energy use emissions dominate health impacts from exposure to ambient particulate matter in India. Nat Commun 2018. https://doi.org/10.1038/s41467-018-02986-7.
- [310] Hartmann DL, Tank a. MGK, Rusticucci M. IPCC Fifth Assessment Report, Climatie Change 2013: The Physical Science Basis. IPCC AR5 2013. https://doi.org/10.1017/ CBO9781107415324.
- [311] Rabl A, Spadaro J V., Holland M. Description of the RiskPoll software. How Much Is Clean Air Worth?, 2014. https://doi.org/10.1017/CBO9781107337831.020.
- [312] van Zelm R, Huijbregts MAJ, den Hollander HA, van Jaarsveld HA, Sauter FJ, Struijs J, et al. European characterization factors for human health damage of PMI0 and ozone in life cycle impact assessment. Atmos Environ 2008. https://doi.org/10.1016/j. atmosenv.2007.09.072.

- [313] Posch M, Seppälä J, Hettelingh JP, Johansson M, Margni M, Jolliet O. The role of atmospheric dispersion models and ecosystem sensitivity in the determination of characterisation factors for acidifying and eutrophying emissions in LCIA. Int J Life Cycle Assess 2008. https://doi.org/10.1007/s11367-008-0025-9.
- [314] Frischknecht R, Steiner R, Arthur B, Norbert E, Gabi H. Swiss Ecological Scarcity Method : The New Version 2006. 2006.
- [315] van Oers L, de Koning A, Guinée J, Huppes G. Abiotic resource depletion in LCA. 2002.
- [316] Primas A. Life Cycle Inventories of new CHP systems. Ecoinvent. 2007;0.
- [317] EUROSTAT. Energy data 2019.
- [318] Perić M, Komatina M, Antonijević D, Bugarski B, Dželetović Ž. Life Cycle Impact Assessment of Miscanthus Crop for Sustainable Household Heating in Serbia. Forests 2018. https://doi.org/10.3390/f9100654.
- [319] Caslin B, Finnan J, Easson L. Miscanthus best practice guidelines. 2010.
- [320] Staffell I, Ingram A, Kendall K. Energy and carbon payback times for solid oxide fuel cell based domestic CHP. Int J Hydrogen Energy 2012;37:2509–23. https://doi.org/10.1016/j. ijhydene.2011.10.060.
- [321] KG BTG& C. Pyrotex KE ® More than just hot gas de-dusting. 2017.
- [322] Biganzoli L, Rigamonti L, Grosso M. LCA evaluation of packaging re-use: the steel drums case study. J Mater Cycles Waste Manag 2019;21:67–78. https://doi.org/10.1007/ s10163-018-00817-x.
- [323] Rigamonti L, Grosso M, Sunseri MC. Influence of assumptions about selection and recycling efficiencies on the LCA of integrated waste management systems. Int J Life Cycle Assess 2009. https://doi.org/10.1007/s11367-009-0095-3.
- [324] Valente A, Iribarren D, Dufour J. End of life of fuel cells and hydrogen products: From technologies to strategies. Int J Hydrogen Energy 2019. https://doi.org/10.1016/j. ijhydene.2019.01.110.
- [325] Milà i Canals L, Bauer C, Depestele J, Dubreuil A, Freiermuth Knuchel R, Gaillard G, et al. Key Elements in a Framework for Land Use Impact Assessment Within LCA (11 pp). Int J Life Cycle Assess 2007;12:5–15. https://doi.org/10.1065/lca2006.05.250.
- [326] Strazza C, Del Borghi A, Costamagna P, Gallo M, Brignole E, Girdinio P. Life Cycle Assessment and Life Cycle Costing of a SOFC system for distributed power generation. Energy Convers Manag 2015;100:64–77. https://doi.org/10.1016/j.enconman.2015.04.068.
- [327] Sadhukhan J. Distributed and micro-generation from biogas and agricultural application of sewage sludge: Comparative environmental performance analysis using life cycle approaches. Appl Energy 2014;122:196–206. https://doi.org/10.1016/j. apenergy.2014.01.051.
- [328] Osman A, Ries R. Life cycle assessment of electrical and thermal energy systems for commercial buildings. Int J Life Cycle Assess 2007;12:308–16. https://doi.org/10.1065/ lca2007.02.310.
- [329] Anne Renouf M, Pagan RJ, Wegener MK. Life cycle assessment of Australian sugarcane products with a focus on cane processing. Int J Life Cycle Assess 2011;16:125–37. https:// doi.org/10.1007/s11367-010-0233-y.
- [330] IEA. World Energy Outlook 2018: Electricity 2018.
- [331] Crippa M, Oreggioni G, D G, Muntean M, Schaaf E, Lo Vullo E, et al. Fossil CO2 and GHG emissions of all world countries. 2019. https://doi.org/10.2760/687800.

- [332] IATA. International Air Transport Assciation Annual Review 2020. 76th Annu Gen Meet 2020.
- [333] Doliente SS, Narayan A, Tapia JFD, Samsatli NJ, Zhao Y, Samsatli S. Bio-aviation Fuel: A Comprehensive Review and Analysis of the Supply Chain Components. Front Energy Res 2020;8. https://doi.org/10.3389/fenrg.2020.00110.
- [334] Wei H, Liu W, Chen X, Yang Q, Li J, Chen H. Renewable bio-jet fuel production for aviation: A review. Fuel 2019;254. https://doi.org/10.1016/j.fuel.2019.06.007.
- [335] Arat HT, Sürer MG. Experimental investigation of fuel cell usage on an air Vehicle's hybrid propulsion system. Int J Hydrogen Energy 2020;45:26370–8. https://doi.org/10.1016/j. ijhydene.2019.09.242.
- [336] EASA. Sustainable Aviation Fuel ' Facilitation Initiative ' 2019; EASA_REP_R.
- [337] de Jong S, Hoefnagels R, Faaij A, Slade R, Mawhood R, Junginger M. The feasibility of short-term production strategies for renewable jet fuels - a comprehensive technoeconomic comparison. Biofuels, Bioprod Biorefining 2015. https://doi.org/10.1002/ bbb.1613.
- [338] IRENA. Biofuels for Aviation Technology Brief. 2017. https://doi.org/10.1016/b978-0-12-809806-6.00012-2.
- [339] O'Connell A, Kousoulidou M, Lonza L, Weindorf W. Considerations on GHG emissions and energy balances of promising aviation biofuel pathways. Renew Sustain Energy Rev 2019. https://doi.org/10.1016/j.rser.2018.11.033.
- [340] Pavlenko AN, Searle S, Christensen A. The cost of supporting alternative jet fuels in the European Union 2019:20.
- [341] Tsoutsos TD, Tournaki S, Paraíba O, Kaminaris SD. The Used Cooking Oil-to-biodiesel chain in Europe assessment of best practices and environmental performance. Renew Sustain Energy Rev 2016;54:74–83. https://doi.org/10.1016/j.rser.2015.09.039.
- [342] Talens Peiró L, Lombardi L, Villalba Méndez G, Gabarrell i Durany X. Life cycle assessment (LCA) and exergetic life cycle assessment (ELCA) of the production of biodiesel from used cooking oil (UCO). Energy 2010;35:889–93. https://doi.org/10.1016/j. energy.2009.07.013.
- [343] ASTM. Standard Specification for Aviation Turbine Fuel Containing Synthesized Hydrocarbons ASTM D7566-20b. 2020.
- [344] Achinas S, Li Y, Achinas V, Euverink GJW. Biogas potential from the anaerobic digestion of potato peels: Process performance and kinetics evaluation. Energies 2019;12. https:// doi.org/10.3390/en12122311.
- [345] Mars AE, Veuskens T, Budde MAW, Van Doeveren PFNM, Lips SJ, Bakker RR, et al. Biohydrogen production from untreated and hydrolyzed potato steam peels by the extreme thermophiles Caldicellulosiruptor saccharolyticus and Thermotoga neapolitana. Int J Hydrogen Energy 2010;35:7730–7. https://doi.org/10.1016/j. ijhydene.2010.05.063.
- [346] Broeren MLM, Kuling L, Worrell E, Shen L. Environmental impact assessment of six starch plastics focusing on wastewater-derived starch and additives. Resour Conserv Recycl 2017;127:246–55. https://doi.org/10.1016/j.resconrec.2017.09.001.
- [347] Moretti C, López-Contreras A, de Vrije T, Kraft A, Junginger M, Shen L. From agricultural (by-)products to jet fuels: Carbon footprint and economic performance. Sci Total Environ 2021;775:145848. https://doi.org/https://doi.org/10.1016/j.scitotenv.2021.145848.

- [348] Capaz R, Posada J, Seabra J, Osseweijer P. Life cycle assessment of renewable jet fuel from ethanol: An analysis from consequential and attributional approaches. Eur. Biomass Conf. Exhib. Proc., vol. 2018, 2018, p. 1336–43.
- [349] van Zanten HHE, Bikker P, Meerburg BG, de Boer IJM. Attributional versus consequential life cycle assessment and feed optimization: alternative protein sources in pig diets. Int J Life Cycle Assess 2018;23. https://doi.org/10.1007/s11367-017-1299-6.
- [350] JRC, CONCAWE, EUCAR (JEC). Well-To-Wheels Report Version 4.a: Well-to Wheels analysis of future automotive fuels and powertrains in the European context. 2014.
- [351] Zampori L, Pant R. Suggestions for updating the Product Environmental Footprint (PEF) method. JRC Tech Reports 2019:248.
- [352] Feednavigator. https://www.feednavigator.com/Article/2013/04/12/UK-starchbyproducts-project-underway 2013. https://www.feednavigator.com/Article/2013/04/12/ UK-starch-byproducts-project-underway (accessed July 17, 2021).
- [353] Stokstad E. Nitrogen crisis from jam-packed livestock operations has 'paralyzed' Dutch economy. Science (80-) 2019. https://doi.org/10.1126/science.aba4504.
- [354] Vural Gursel I, Moretti C, Hamelin L, Jakobsen LG, Steingrimsdottir MM, Junginger M, et al. Comparative cradle-to-grave life cycle assessment of bio-based and petrochemical PET bottles. Sci Total Environ 2021;793:148642. https://doi.org/10.1016/j. scitotenv.2021.148642.
- [355] Goedkoop M, Heijungs R, Huijbregts M, Schryver A De, Struijs J, Zelm R Van. ReCiPe 2008. 2009. https://doi.org/10.029/2003JD004283.
- [356] Ponsioen T, Blonk H. Case studies for more insight into the methodology and composition of carbon footprints of table potatoes and chips 2011.
- [357] Van Hecke W, Joossen-Meyvis E, Beckers H, De Wever H. Prospects & potential of biobutanol production integrated with organophilic pervaporation – A technoeconomic assessment. Appl Energy 2018;228:437–49. https://doi.org/10.1016/j. apenergy.2018.06.113.
- [358] Van Hecke W, De Wever H. High-flux POMS organophilic pervaporation for ABE recovery applied in fed-batch and continuous set-ups. J Memb Sci 2017. https://doi. org/10.1016/j.memsci.2017.06.058.
- [359] Dunn JB, Mueller S, Wang M, Han J. Energy consumption and greenhouse gas emissions from enzyme and yeast manufacture for corn and cellulosic ethanol production. Biotechnol Lett 2012. https://doi.org/10.1007/s10529-012-1057-6.
- [360] Breitkreuz K, Menne A, Kraft A. New process for sustainable fuels and chemicals from bio-based alcohols and acetone. Biofuels, Bioprod Biorefining 2014. https://doi. org/10.1002/bbb.1484.
- [361] Lodi G, De Guido G, Pellegrini LA. Simulation and energy analysis of the ABE fermentation integrated with gas stripping. Biomass and Bioenergy 2018. https://doi. org/10.1016/j.biombioe.2018.06.012.
- [362] Cloete S, Giuffrida A, Romano MC, Zaabout A. Economic assessment of the swing adsorption reactor cluster for CO2 capture from cement production. J Clean Prod 2020;275. https://doi.org/10.1016/j.jclepro.2020.123024.
- [363] Gevo. Sustainable aviation fuel. 2019.
- [364] Elgowainy A, Han J, Wang M, Carter N, Stratton R, Hileman J, et al. GREET_Life_Cycle_ Analysis_of_Aviation_Fuels. 2012. https://doi.org/10.2172/1255237.

- [365] Di Marcoberardino G, Liao X, Dauriat A, Binotti M, Manzolini G. Life Cycle Assessment and Economic Analysis of an Innovative Biogas Membrane Reformer for Hydrogen Production. Processes 2019. https://doi.org/10.3390/pr7020086.
- [366] Topham S, Bazzanella A, Schiebahn S, Luhr S, Zhao L, Otto L, et al. Carbon dioxide. Ullmann's Encycl Ind Chem 2014. https://doi.org/10.1002/14356007.a05_165.pub2.
- [367] Althaus H, Chudacoff M, Hischier R, Jungbluth N, Osses M, Primas A, et al. Life cycle inventories of chemicals. ecoinvent report No.8, v2.0. 2007.
- [368] Young B, Krynock M, Carlson D, Hawkins TR, Marriott J, Morelli B, et al. Comparative environmental life cycle assessment of carbon capture for petroleum refining, ammonia production, and thermoelectric power generation in the United States. Int J Greenh Gas Control 2019;91. https://doi.org/10.1016/j.ijggc.2019.102821.
- [369] Duynie. Potato Peel. Product description 2021. https://www.duynie.co.uk/applications/ duynie/default/assets/pdf/product/1871/Potato_Peel.pdf?groep=.
- [370] Ncobela CN, Kanengoni AT, Hlatini VA, Thomas RS, Chimonyo M. A review of the utility of potato by-products as a feed resource for smallholder pig production. Anim Feed Sci Technol 2017;227:107–17. https://doi.org/10.1016/j.anifeedsci.2017.02.008.
- [371] Feedipedia. Potato peels, fresh 2019. https://www.feedipedia.org/node/12471.
- [372] Nelson ML. Utilization and application of wet potato processing coproducts for finishing cattle. J Anim Sci 2010;88. https://doi.org/10.2527/jas.2009-2502.
- [373] Heuvelmans K, Vogel S. The European Feed Mix. Successful Ingredients for the World's Second-Largest Feed Market. 2017.
- [374] CBS. Decline in pig farming 2020. https://www.cbs.nl/en-gb/news/2020/34/decline-inpig-farming (accessed April 28, 2021).
- [375] van der Hilst F, Hoefnagels R, Junginger M, Londo M, Shen L, Wicke B. Biomass Provision and Use, Sustainability Aspects BT - Encyclopedia of Sustainability Science and Technology. In: Meyers RA, editor., New York, NY: Springer New York; 2018, p. 1–30. https://doi.org/10.1007/978-1-4939-2493-6_1048-1.
- [376] Zhao X, Taheripour F, Malina R, Staples MD, Tyner WE. Estimating induced land use change emissions for sustainable aviation biofuel pathways. Sci Total Environ 2021:146238.
- [377] Ahlgren S, Di Lucia L. Indirect land use changes of biofuel production-a review of modelling efforts and policy developments in the European Union. Biotechnol Biofuels 2014;7:1–10.
- [378] IPCC. 2006 IPCC guidelines for national greenhouse gas inventories. 2006.
- [379] Donke ACG, Novaes RML, Pazianotto RAA, Moreno-Ruiz E, Reinhard J, Picoli JF, et al. Integrating regionalized Brazilian land use change datasets into the ecoinvent database: new data, premises and uncertainties have large effects in the results. Int J Life Cycle Assess 2020;25:1027–42. https://doi.org/10.1007/s11367-020-01763-3.
- [380] van Zeist W-J. White paper; Direct Land Use Change Tool (version-2016.1). White Pap Version 20161 2016:1–4.
- [381] PlasticsEurope. Reformer hydrogen. Eco-profiles of the European Plastics Industry. 2005.
- [382] IEA. Report extract. A new era for CCUS. 2021. https://www.iea.org/reports/ccus-inclean-energy-transitions/a-new-era-for-ccus (accessed July 8, 2021).

- [383] Dou Z, Toth JD, Westendorf ML. Food waste for livestock feeding: Feasibility, safety, and sustainability implications. Glob Food Sec 2018;17:154–61. https://doi.org/10.1016/j. gfs.2017.12.003.
- [384] Rust NA, Ridding L, Ward C, Clark B, Kehoe L, Dora M, et al. How to transition to reducedmeat diets that benefit people and the planet. Sci Total Environ 2020;718. https://doi. org/10.1016/j.scitotenv.2020.137208.
- [385] Plevin RJ, Beckman J, Golub AA, Witcover J, O'Hare M. Carbon accounting and economic model uncertainty of emissions from biofuels-induced land use change. Environ Sci Technol 2015;49:2656–64.
- [386] Song XP, Hansen MC, Potapov P, Adusei B, Pickering J, Adami M, et al. Massive soybean expansion in South America since 2000 and implications for conservation. Nat Sustain 2021. https://doi.org/10.1038/s41893-021-00729-z.
- [387] Fritsche UR, Sims REH, Monti A. Direct and indirect land-use competition issues for energy crops and their sustainable production-an overview. Biofuels, Bioprod Biorefining 2010;4:692–704.
- [388] van der Hilst F. Location, location, location. Nat Energy 2018;3:164–5. https://doi. org/10.1038/s41560-018-0094-3.
- [389] Vera I, Wicke B, Hilst F van der. Spatial Variation in Environmental Impacts of Sugarcane Expansion in Brazil. Land 2020;9:397.
- [390] Wicke B, Verweij P, Meijl H, Vuuren DP, Faaij AP. Indirect land use change: review of existing models and strategies for mitigation. Biofuels 2012;3. https://doi.org/10.4155/ bfs.11.154.
- [391] Weidema BP, Simas MS, Schmidt J, Pizzol M, Løkke S, Brancoli PL. Relevance of attributional and consequential information for environmental product labelling. Int J Life Cycle Assess 2020;25:900–4. https://doi.org/10.1007/s11367-019-01628-4.
- [392] Weidema BP, Pizzol M, Schmidt J, Thoma G. Social responsibility is always consequential — Rebuttal to Brander, Burritt and Christ (2019): Coupling attributional and consequential life cycle assessment: A matter of social responsibility. J Clean Prod 2019;223:12–3. https://doi.org/10.1016/j.jclepro.2019.03.136.
- [393] Adams PWR, Mezzullo WG, McManus MC. Biomass sustainability criteria: Greenhouse gas accounting issues for biogas and biomethane facilities. Energy Policy 2015. https:// doi.org/10.1016/j.enpol.2015.08.031.
- [394] Brander M. Attributional and consequential methods are both necessary for managing responsibility – Reply to Weidema et al. (2019). J Clean Prod 2019;228:8–9. https://doi. org/10.1016/j.jclepro.2019.04.307.
- [395] Djomo SN, Humbert S, Dagnija Blumberga. Life cycle assessment of hydrogen produced from potato steam peels. Int J Hydrogen Energy 2008;33:3067–72. https:// doi.org/10.1016/j.ijhydene.2008.02.006.
- [396] Hijazi O, Mettenleiter S, Samer M, Abdelsalam E, Wiecha JG, Ziegler KL, et al. Life cycle assessment of biogas production in small-scale in Columbia. 2019 ASABE Annu Int Meet 2019. https://doi.org/10.13031/aim.201900099.
- [397] Ochs D, Wukovits W, Ahrer W. Life cycle inventory analysis of biological hydrogen production by thermophilic and photo fermentation of potato steam peels (PSP). J Clean Prod 2010;18. https://doi.org/10.1016/j.jclepro.2010.05.018.

- [398] UNEP/SETAC. Global guidance principles for life cycle assessment databases—a basis for greener processes and products. UNEP/ SETAC Life Cycle Initiative, United Nations Environment Programme, Paris. 2011.
- [399] Sala S, Amadei A, Beylot A, Ardente F. The evolution of life cycle assessment in European policies over three decades. Int J Life Cycle Assess 2021. https://doi.org/10.1007/s11367-021-01893-2.
- [400] Raos G. Molecular geometry and molecular graphics: Natta's polypropylene and beyond. Adv. Intell. Syst. Comput., 2019. https://doi.org/10.1007/978-3-319-95588-9_4.
- [401] Natta G. Nouvelles recherches dans le domaine de la chimie des olefines. World Pet. Congr. Proc., 4th World Petroleum Congress, WPC 1955; Rome; Italy; 6 June 1955 through 15 June 1955; Code 141555: 1955, p. Volume 1955-June, 1955, Pages 271-290.
- [402] PlasticsEurope. Plastics the Facts 2017 2017.
- [403] Reports and Data. Polypropylene Market By Type, By Grade, By Molding Techniques, By Application, By End-Users (Automotive, Packaging, Construction, Electrical & Electronics, Consumer Goods, And Others), And Segment Forecasts. 2019.
- [404] PlasticsEurope. Polypropylene (PP). 2014.
- [405] Shen L, Worrell E, Patel M. Present and future development in plastics from biomass. Biofuels, Bioprod Biorefining 2010;4:25–40. https://doi.org/10.1002/bbb.189.
- [406] Suwanmanee U, Varabuntoonvit V, Chaiwutthinan P, Tajan M, Mungcharoen T, Leejarkpai T. Life cycle assessment of single use thermoform boxes made from polystyrene (PS), polylactic acid, (PLA), and PLA/starch: Cradle to consumer gate. Int J Life Cycle Assess 2013;18:401–17. https://doi.org/10.1007/s11367-012-0479-7.
- [407] Weiss M, Haufe J, Carus M, Brandão M, Bringezu S, Hermann B, et al. A Review of the Environmental Impacts of Biobased Materials. J Ind Ecol 2012. https://doi.org/10.1111/ j.1530-9290.2012.00468.x.
- [408] Choi B, Yoo S, Park S II. Carbon footprint of packaging films made from LDPE, PLA, and PLA/PBAT blends in South Korea. Sustain 2018;10. https://doi.org/10.3390/su10072369.
- [409] Ingrao C, Gigli M, Siracusa V. An attributional Life Cycle Assessment application experience to highlight environmental hotspots in the production of foamy polylactic acid trays for fresh-food packaging usage. J Clean Prod 2017. https://doi.org/10.1016/j. jclepro.2017.03.007.
- [410] van der Harst E, Potting J, Kroeze C. Multiple data sets and modelling choices in a comparative LCA of disposable beverage cups. Sci Total Environ 2014;494–495:129–43. https://doi.org/10.1016/j.scitotenv.2014.06.084.
- [411] European Bioplastics. Bioplastics market data 2017 Global production capacities of bioplastics 2017-2022 2017:4.
- [412] European Bioplastics. Bioplastics market data. 2019.
- [413] Machado PG, Walter A, Cunha M. Bio-based propylene production in a sugarcane biorefinery: A techno-economic evaluation for Brazilian conditions. Biofuels, Bioprod Biorefining 2016;10:623–33. https://doi.org/10.1002/bbb.1674.
- [414] Niaounakis M. Biopolymers: Applications and Trends, William Andrew Publishing, Oxford, p. 185-232. 2015.
- [415] Kikuchi Y, Oshita Y, Mayumi K, Hirao M. Greenhouse gas emissions and socioeconomic effects of biomass-derived products based on structural path and life cycle analyses: A case study of polyethylene and polypropylene in Japan. J Clean Prod 2017;167:289–305. https://doi.org/10.1016/j.jclepro.2017.08.179.

- [416] Gay M, Pope B, Wharton J. Propylene From Biomass. 2011.
- [417] Neste. https://www.neste.com/ikea-and-neste-take-significant-step-towards-fossilfree-future 2018.
- [418] Mayumi K, Kikuchi Y, Hirao M. Life Cycle Assessment of biomass derived resin for sustainable chemical industry. Chem Eng Trans 2010;19:19–25. https://doi.org/10.3303/ CET1019004.
- [419] EUBIA. Used cooking oil 2020. https://www.eubia.org/cms/wiki-biomass/biomassresources/challenges-related-to-biomass/used-cooking-oil-recycling/ (accessed January 28, 2020).
- [420] Patel R, Khan A, Mujtaba IM, Butterfield R, Mercuri E, Manca D. Bio fuel: Process for synthesis of biodiesel from used cooking oil: Feasibility and experimental studies. 2017. https://doi.org/10.4324/9781315153209.
- [421] Neste. https://www.neste.com/releases-and-news/neste-and-lyondellbasell-announcecommercial-scale-production-bio-based-plastic-renewable-materials 2019.
- [422] Ramanathan V, Feng Y. Air pollution, greenhouse gases and climate change: Global and regional perspectives. Atmos Environ 2009;43:37–50. https://doi.org/10.1016/j. atmosenv.2008.09.063.
- [423] Rosenbaum RK, Bachmann TM, Gold LS, Huijbregts MAJ, Jolliet O, Juraske R, et al. USEtox - The UNEP-SETAC toxicity model: Recommended characterisation factors for human toxicity and freshwater ecotoxicity in life cycle impact assessment. Int J Life Cycle Assess 2008. https://doi.org/10.1007/s11367-008-0038-4.
- [424] Frischknecht R, Braunschweig A, Hofstetter P, Suter P. Human health damages due to ionising radiation in life cycle impact assessment. Environ Impact Assess Rev 2000. https://doi.org/10.1016/S0195-9255(99)00042-6.
- [425] Boulay A-M, Bare J, De Camillis C, Döll P, Gassert F, Gerten D, et al. Consensus building on the development of a stress-based indicator for LCA-based impact assessment of water consumption: outcome of the expert workshops. Int J Life Cycle Assess 2015;20:577–83. https://doi.org/10.1007/s11367-015-0869-8.
- [426] Nikander S. Greenhouse gas and energy intensity of product chain: case transport biofuel. Helsinki University of technology, 2008.
- [427] Althaus H-J, Chudacoff M, Hischier R, Jungbluth N, Osses M, Primas A. ecoinvent report n°8: Life cycle inventories of chemicals. 2007.
- [428] Fröhlich et al. Fröhlich J., Liebich, A., Trojek, S., Volz, S., Lauwigi, C., Giegrich, J.Eco-profiles and Environmental Product Declarations of the European Plastics Manufacturers: Chlorine (The chlor-alkali process) 2013.
- [429] US EPA. Liquified Petroleum Gas Combustion. AP 42, Compil Air Pollut Emiss Factors, Vol 1 Station Point Area Sources 2008:8–11.
- [430] Büsser S. Life Cycle Inventory of E85 , LPG supply in Switzerland and Biogasmix 2008. ESU-Services Ltd 2010.
- [431] Karimzadeh R, Godini HR, Ghashghaee M. Flowsheeting of steam cracking furnaces. Chem Eng Res Des 2009;87:36–46. https://doi.org/10.1016/j.cherd.2008.07.009.
- [432] Joosten LAJ. Process Data Descriptions for the production of synthetic organic materials Input data for the MATTER study. 1998.
- [433] PlasticsEurope. Eco-profiles and Environmental Product Declarations of the European Plastics Manufacturers: Polypropylene (PP) 2014:1–4. https://doi.org/10.1016/j. susc.2010.09.014.

- [434] European Commision. RED (2009). Renewable Energy Directive 2009/28/EC 2009.
- [435] De Mora EF, Torres C, Valero A. Thermoeconomic analysis of biodiesel production from used cooking oils. Sustain 2015;7:6321–35. https://doi.org/10.3390/su7056321.
- [436] Nederlandse Emissieautoriteit. Rapportage Energie voor Vervoer in Nederland 2017 2017:1–55.
- [437] Toop G, Alberici S, Spoettle M, Steen H van. Trends in the UCO market. Ecofys 2013.
- [438] Bhandari R, Trudewind CA, Zap P. Life Cycle Assessment of Hydrogen Production Methods – A Review 2012.
- [439] Fernández-Dacosta C, Shen L, Schakel W, Ramirez A, Kramer GJ. Potential and challenges of low-carbon energy options: Comparative assessment of alternative fuels for the transport sector. Appl Energy 2019;236:590–606. https://doi.org/10.1016/j. apenergy.2018.11.055.
- [440] Saleh HE. Effect of variation in LPG composition on emissions and performance in a dual fuel diesel engine. Fuel 2008;87:3031–9. https://doi.org/10.1016/j.fuel.2008.04.007.
- [441] Thamsiriroj T, Murphy JD. A critical review of the applicability of biodiesel and grass biomethane as biofuels to satisfy both biofuel targets and sustainability criteria. Appl Energy 2011;88:1008–19. https://doi.org/10.1016/j.apenergy.2010.10.026.
- [442] Svanes E, Vold M, Hanssen OJ. Effect of different allocation methods on LCA results of products from wild-caught fish and on the use of such results. Int J Life Cycle Assess 2011. https://doi.org/10.1007/s11367-011-0288-4.
- [443] European Commission. Organisation Environmental Footprint Guide. Eur Comm Res Centre-Institute Environ Sustain 2013. https://doi.org/doi:10.3000/19770677.L_2013.124. eng.
- [444] Schrijvers DL, Loubet P, Sonnemann G. An axiomatic method for goal-dependent allocation in life cycle assessment. Int J Life Cycle Assess 2021. https://doi.org/10.1007/ s11367-021-01932-y.
- [445] FAO. FAOSTAT 2020. https://doi.org/http://www.fao.org/faostat/en/#data/QC/.
- [446] Pathak PD, Mandavgane SA, Puranik NM, Jambhulkar SJ, Kulkarni BD. Valorization of potato peel: a biorefinery approach. Crit Rev Biotechnol 2018;38:218–30. https://doi.org/ 10.1080/07388551.2017.1331337.
- [447] Neste. Neste delivers first batch of 100% renewable propane to European market 2018.
- [448] van der Hulst MK, Huijbregts MAJ, van Loon N, Theelen M, Kootstra L, Bergesen JD, et al. A systematic approach to assess the environmental impact of emerging technologies: A case study for the GHG footprint of CIGS solar photovoltaic laminate. J Ind Ecol 2020;24:1234–49. https://doi.org/10.1111/jiec.13027.
- [449] PlasticsEurope. Benzene, Toluene, and Xylenes (Aromatics, BTX).Eco-profiles and Environmental Product Declarations of the European Plastics Manufacturers. 2013.

ACKNOWLEDGMENTS

I would like first to thank the co-authors of each chapter of my PhD work. In particular, I am grateful to all three of my promoters for their supervision, advice, patience, and trust.

In particular, I would like to express my deepest gratitude to Martin and Li for offering me the opportunity to make this PhD journey. I have greatly enjoyed doing my PhD in Utrecht and if I could go back in time, I would make the same choice again.

Martin, you and the other professors of our E&R group have shaped such a nice working environment made by lots of wonderful people. You have always been very supportive and kind during all these four years and made me work in the best conditions. It has been an honor and pleasure working for you.

Li, I think that you are the best and most challenging reviewer that I will ever find. Each of your comments ended in improving the quality of my work significantly to the point that we left the journals' reviewers almost with no comments. Thank you for your teachings and patience during these four years and for having pushed me to think always once more.

Blanca, I have been your first PhD student. I can say it out loud: you have been a great co-promoter! Thank you for our countless methodological discussions, patience and your friendly support in the most stressful moments of my PhD journey.

Ric, I really thank you for your guidance, trust and support.

A special thanks to the colleagues that have been my dear friends during my time in Utrecht: Ivan, Ody, Jing, Anand, Steven, Jesus, Lukas, Tarek, Hui, Anna, Lotte, Paul and Juraj.

It is time to thank my wonderful wife and parents, but unfortunately, I do not have an entire chapter available. Martina, you are my power and happiness and I could have never managed to finalize this thesis without your constant support and love. Papà e Mamma siete genitori eccezionali e raggiungere questo traguardo sarebbe stato impossibile senza il vostro amore, sacrifici e insegnamenti.

A special thanks to the wonderful city of Utrecht, which has always made me feel at home. I will miss you a lot!

CURRICULUM VITAE

Christian Moretti was born on December 31, 1992 in Varese, Italy. He joined the Copernicus Institute of Sustainable Development of Utrecht University as a junior researcher/PhD candidate in August 2017. His PhD research focused on life cycle environmental impacts of bio-based and petroleum chemicals, asphalts, bioenergy, and biofuels. His previous research activity was carried out at the Joint Research Centre (JRC) of the European Commission in the sustainable transport unit. He holds a master of science in Energy engineering from Politecnico di Milano. Before entering the academic sector, he was shortly employed in metallic (Varroc Group) and fashion (Missoni SPA) industries. As of February 2022, he joined the Department of Environmental Systems Science of the Swiss Federal Institute of Technology (ETH).

Publications in scientific journals:

- Analysis of standard and innovative methods for allocating upstream and refinery GHG emissions to oil products. Moretti C., Moro A., Edwards R., Rocco M. & Colombo E. - Applied Energy, 2017 doi:10.1016/j.apenergy.2017.08.183.
- A carbon footprint assessment of multi-output biorefineries with international biomass supply: a case study for the Netherlands. Vera Concha, I.C., Hoefnagels, E.T.A., van der Kooij, Aldert, Moretti, C. & Junginger, H.M. Biofuels, Bioproducts and Biorefining, 2020 doi: 10.1002/bbb.2052
- 3. Emerging technologies in the renewable energy sector: A comparison of expert review with a text mining software. Moro, A., Joanny, G., Moretti, C., Futures, 2020, doi: 10.1016/j.futures.2020.102511.
- Environmental Life Cycle Assessment of polypropylene made from used cooking oil. Moretti C., Junginger M., Shen. L., Resources, Conservation & Recycling, 2020, doi: 10.1016/j.resconrec.2020.104750.
- Combining Biomass Gasification and Solid Oxid Fuel Cell for Heat and Power Generation - An Early-Stage Life Cycle Assessment. Moretti, C., Corona B., Ruhlin V., Götz T., Junginger M., Brunner T., Obernberger I. & Shen L., Energies, 2020. doi: 10.3390/en13112773.
- Reviewing ISO Compliant Multifunctionality Practices in Environmental Life Cycle Modeling. Moretti, C., Corona B., Edwards R, Junginger M., Moro A., Rocco M & Shen L., Energies, 2020, doi: 10.3390/en13143579.

- Review of Life Cycle Assessments of lignin and derived products: lessons learned. Moretti C., Corona B., Hoefnagels R., Vural-Gürsel I., Gosselink R. & Junginger M. Science of the total Environment, 2021. 10.1016/j.scitotenv.2020.144656.
- From agricultural (by-)products to jet fuels: carbon footprint and economic performance. Moretti C., López-Contreras A., de Vrije T., Kraft A., Junginger M. & Shen L. Science of the total environment, 2021. 10.1016/j.scitotenv.2021.145848.
- Cradle-to-grave life cycle assessment of single-use cups made from PLA, PP and PET. Moretti C., Hamelin L., Jakobsen L., Junginger M., Steingrimsdottir M., Høibye L. & Shen L. Resources, Conservation & Recycling, 2021. 10.1016/j. resconrec.2021.105508.
- Comparative cradle-to-grave life cycle assessment of bio-based and petrochemical PET bottles. Vural-Gürsel I., Moretti C., Hamelin L., Jakobsen L., Junginger M., Steingrimsdottir M., Høibye L. & Shen L. Science of the Total Environment, 2021. 10.1016/j.scitotenv.2021.148642.
- Kraft lignin as a bio-based ingredient for Dutch asphalts: an attributional LCA. Moretti C., Corona B., Hoefnagels R., van Veen M., Vural-Gursel I., Strating T., Gosselink R. & Junginger M. Science of the Total Environment, 2021. 10.1016/j. scitotenv.2021.150316.
- Attributional and consequential LCAs of a novel bio-jet fuel from Dutch potato by-products. Moretti C., Vera I., Junginger M., López-Contreras A. & Shen L. Science of the Total Environment, 2022. 10.1016/j.scitotenv.2021.152505.
- Definition of product system and solving multifunctionality in ISO 14040-14044: issues and proposed amendments – Towards a more open & general LCA framework. Schaubroeck T., Schrijvers D., Schaubroeck S., Moretti C., Zamagni A., Pelletier N., Huppes G. & and Brandão M. Frontiers in Sustainability, 2022. 10.3389/frsus.2022.778100.
- Using lignin from local biorefineries for asphalts: LCA case study for the Netherlands. Moretti C., Hoefnagels R., van Veen M., Corona B., Obydenkova S., Russel S., Jongerius A., Vural-Gürsel I. & Junginger M. Journal of Cleaner production, 2022. 10.1016/j.jclepro.2022.131063.